## University of Wollongong

## **Research Online**

# University of Wollongong Thesis Collection 2017+

University of Wollongong Thesis Collections

2018

## Unraveling the biogeochemical behaviour of arsenic and antimony in soils and bioavailability to agricultural plants

Lien Kim Ngo University of Wollongong

Follow this and additional works at: https://ro.uow.edu.au/theses1

## University of Wollongong Copyright Warning

You may print or download ONE copy of this document for the purpose of your own research or study. The University does not authorise you to copy, communicate or otherwise make available electronically to any other person any copyright material contained on this site.

You are reminded of the following: This work is copyright. Apart from any use permitted under the Copyright Act 1968, no part of this work may be reproduced by any process, nor may any other exclusive right be exercised,

without the permission of the author. Copyright owners are entitled to take legal action against persons who infringe their copyright. A reproduction of material that is protected by copyright may be a copyright infringement. A court may impose penalties and award damages in relation to offences and infringements relating to copyright material. Higher penalties may apply, and higher damages may be awarded, for offences and infringements involving the conversion of material into digital or electronic form.

Unless otherwise indicated, the views expressed in this thesis are those of the author and do not necessarily represent the views of the University of Wollongong.

## **Recommended Citation**

Ngo, Lien Kim, Unraveling the biogeochemical behaviour of arsenic and antimony in soils and bioavailability to agricultural plants, Doctor of Philosophy thesis, School of Chemistry, University of Wollongong, 2018. https://ro.uow.edu.au/theses1/416

Research Online is the open access institutional repository for the University of Wollongong. For further information contact the UOW Library: research-pubs@uow.edu.au



# Unraveling the biogeochemical behaviour of arsenic and antimony in soils and bioavailability to agricultural plants

Lien Kim Ngo

This thesis is presented as part of the requirements for the conferral of the degree:

Doctor of Philosophy (PhD, Environmental Chemistry)

> Supervisors: Prof. Dianne Jolley Prof. Peter Teasdale Dr. William Bennett

The University of Wollongong School of Chemistry

November, 2018

## DECLARATION

I, Lien Kim Ngo, declare that this thesis is submitted in partial fulfillment of the requirements for the conferral of the degree of Doctor of Philosophy in the School of Chemistry, the University of Wollongong, is wholly my own work unless otherwise referenced or acknowledged. This document has not been submitted for qualifications at any other academic institution.

Lien Kim Ngo

August 2017

## ABSTRACT

The enrichment of soil arsenic (As) and antimony (Sb) is putting increasing pressure on the environment and human health. The biogeochemical behaviour of Sb and its uptake mechanisms by plants are poorly understood and generally assumed to be similar to that of As. Accurate assessment of the bioavailability of these toxic elements to agricultural crops grown in contaminated soils under different scenarios is important for the management of contaminated soils and minimising the risk of human exposure.

In this study, the diffusive gradients in thin films (DGT) technique and sequential extraction procedure (SEP) were applied to assess the partitioning and lability of As and Sb under agricultural conditions in historically contaminated soils with various physicochemical properties. The performance of DGT in predicting As and Sb uptake by two cultivars of radish (*Raphanus sativus*), water spinach (*Ipomoea aquatica*) grown in these soils were investigated, compared to soil solution analysis and sequential extraction. The competitive interactions of As and Sb in single (As-only and Sb-only) and multi-contaminant (As+Sb) soils and the interactive effects of As and Sb on accumulation by *I. aquatica* were also examined. Furthermore, the effect of changing redox conditions on the behaviour and speciation of As and Sb in soils using multiple *in situ* DGT samplers and their accumulation and speciation in *I. aquatica* cultivated in such changing environments were investigated.

The results showed that As and Sb exhibited contrasting biogeochemical behaviours. Irrespective of the method, all labile fractions showed that both As and Sb were firmly bound to the solid phases in historically contaminated soils, and that Sb was less mobile than As, even though total soil Sb concentrations were higher than total soil As. When the contaminated soils were subjected to anaerobic conditions (flooding), DGT-labile As increased with As(III) being the dominant species, but DGT-labile Sb decreased with Sb(V) dominating. The mobilisation of As was coupled with the reductive dissolution of Fe(III) oxyhydroxides, but this did not occur with Sb. Under aerobic condition, the lability of As in As+Sb soils was lower than that in As-only soils. In contrast, during the flooding period, labile As in As+Sb soils decreased and was higher than that in As-only soils, which was likely due to the presence of Sb.

The bioassays demonstrated very low bioaccumulation of As and Sb in the two cultivars of radish (R.sativus). The white icicle radish exhibited the same uptake trend as the cherry belle radish, but was less able to translocate As and Sb from roots to shoots than

the cherry belle cultivar. In contrast, much greater bioaccumulation of As and Sb in water spinach (*Ipomoea aquatica*) were observed in which As concentration in edible shoots of *I. aquatica* far exceeded the permissible limit of As in food set by the Food Standard Agency Australia and New Zealand and FAO/WHO. The bioaccumulation factor for As in *I. aquatica* was >1, being much higher than that of Sb, while the translocation factor of Sb was higher than that of As, indicating differences in uptake patterns. Compared with single-element soils, the average bioaccumulation factor of As in *I. aquatica* was grown in flooded and non-flooded conditions. Flooded *I. aquatica* grew healthily and had greater biomass than non-flooded *I. aquatica*. Concentrations of As and Sb in *I. aquatica* tissues of flooded plants were over 2 times higher than those in non-flooded plants. Inorganic As species were found in plant tissues with both As(III) and As(V) present in non-flooded *I. aquatica* and mainly As(V) present in flooded *I. aquatica*. The reason for this difference is unknown.

Arsenic and Sb in tissues of *R. sativus* and *I. aquatica* were strongly correlated with their labile concentrations measured by DGT, soil solution, and SEP. These techniques are useful measures for predicting bioavailable As and Sb in the historically contaminated soil to the test plants. The coupling of SEP and DGT was useful in understanding the biogeochemical behaviour of As and Sb in aerobic soils. The combined mercapto-silica-DGT, Metsorb-DGT, and diffusive equilibration in thin films (DET) for Fe successfully examined the speciation and lability of As and Sb in flooded soils and differentiated the mechanisms of the mobilisation of As and Sb in soils under changes in redox conditions. DGT measurements in planted soils captured changes in As mobility induced by *I. aquatica*, leading to the better reflection of As uptake by tissues of *I. aquatica*. Importantly, the relationships between As(III)/As(V) in soils measured by DGTs and As(III)/As(V) in plants allowed predicting the preferential As species uptake by flooded *I. aquatica* and explaining the proposed As speciation in flooded *I. aquatica*.

Overall, the biogeochemical behaviour of Sb was different to As. The *in situ* sampling capability, which can be readily utilized in different environmental conditions, and the selective measurement capability for As(III) and Sb(III) made DGT an effective technique for this study. Therefore, DGT is a very promising tool for measuring the bioavailability and speciation of As and Sb in soils, identifying differences in their biogeochemical behaviour, and predicting their uptake by plants under various conditions.

## ACKNOWLEDGEMENTS

I would like to deeply thank the Australian Awards – Vietnam for offering me a PhD scholarship, which brought me to Australia for my PhD research at University of Wollongong. This was a challenging journey because things happened to me that I had not thought of and out of my control. This journey can be considered as a film of my life which is meaningful and unforgettable to me.

I would like to thank all in the UOW School of Chemistry, especially the technical staff for their assistance with equipment access and operations. Special thanks to Helen Price for training DGT gel synthesis, probe assembly and handing, and ICP-MS analysis. Thanks to Prof. Alan Chivas, A/Prof. Brian Jones, Lili Yu, David Wheeler, and Jose Abrantes (UOW School of Earth and Environmental Sciences) for allowing equipment access and pushing resolving the series of ICP-MS problems. Thanks also to the technical staff at the UOW School of Civil, Mining, and Environmental Engineering, especially Alan Grant and Ritchie McLean for the access of soil grinders and to the technical staff and friends at the UOW Ecological Research Centre for their assistance in the growth chamber operations and space access.

I would like to thank Ed Burton and Scott Johnson (Southern Cross University), John Coffey (NSW EPA), and Clayton Colmer (NSW Soil Conservation Service) for their assistance and facilitating site access of contaminated soil and thank Mark Dee, Graham Hunt, and Dick Westera (staff at UOW Landscape Department) for their help in collecting clean soils in Wollongong.

I would also like to thank my "adopted supervisor" A/Prof. Glennys O'Brien, Senior Prof. Stephen Pyne, Dr. Christopher Hyland, Dr. Wilford Lie, and Ellen Manning for friendly chats and caring of me during my hard time at UOW. Special thanks also to Benjamin Pinch for working with me and sharing challenges during my first year of research. Gabriella Macoustra, Lakmini Egodawatta, Kaitlyn Lieschke, Thang Dau, and Nghi Phan are thanked for your assistance with preparing soils and plants. Thanks to Elvio, Cassandra, Gabriella, and Darren (Jolley group) for being great lab mates, office mates, social chats, and sharing emotion. Thanks also to Tim, Francesca, Aleica, Megan, Brett, Jessica, Rachel, Sarah, Nicholas, Gabriel, and Marc (Jolley group) for friendly chats. Thanks to Josh King (CSIRO) for his advice on microwave operation and metal analysis. Thanks to friends in Pyne group, Keller group, and Kelso group for kindly sharing the use of sonicators, a few chemicals, and access to liquid nitrogen. Thanks to some Vietnamese friends (Nguyet, Diep, Hien, Quan, Ngo, Doan, Son, Han, Nghi) for their friendly support and sharing emotion. Thanks also to scholarship officers in the UOW Australian Awards Unit, especially Nhan and Wendy for their great support with documents and procedures.

I would like to thank Prof. William Maher and Mr Frank Krikowa (University of Canberra) for their expertise and collaboration for the analysis of As and Sb species in plants. I learnt skills and experience and got directions from my time working with Frank. Thanks to Dr. Dang Nguyen and Prof. Hung Ngo (Can Tho University) for their advice on growing plants and fertilising.

I would like to deeply thank Phuong Nguyen (my special Vietnamese friend, support network, "sample photographer") and Kel Magrath (my special Australian friend, "field supervisor", "adopted grandfather") for all they did for me. They were always behind me and tried to help me. Deep thanks to Phuong for being there to share my up-and-down period, encouraging me, and taking care of me when things did not go my way, especially when I suffered the accident. Deep thanks to Kel for being my "driver" taking me out to buy items for experiments, supervising me collect soil in the field, transporting soil and lab equipment from places to places, going to the hospital with me, and so on. What Phuong and Kel did for me have meant so much to me and made my time at UOW less difficult. I really owe you both a debt of gratitude.

Thanks to my siblings, especially my oldest brother for taking care of my mother in the toughest time for my role so that I could do the scholarship application at the last minute and have concentration on studies. My scholarship application and PhD journey could not occur without your support. Thank you very much, brother!

Last and most importantly, I would like to appreciate my superlative supervisors, Professor Dianne Jolley, Professor Peter Teasdale, and Dr William Bennett. Special thanks for all your precious time and effort, outstanding support, and sharp comments during my PhD, which has greatly helped my development in terms of writing and science. Particularly deep thanks to Dianne, I am very fortunate to have you my supervisor. I have insufficient words to adequately express my gratitude to you, Dianne. You has given me full of assistance, encouragement, and shared my happiness, sadness, emotion, and tragedy. I am eternally grateful for all your time and hard work, being beyond a normal supervisor. I could not have been at this point without you. I have learned not only academic lessons but also life skills and cultural lessons from you. What you have done for me has meant so much to me and made my PhD journey become an unforgettable part of my life. I would like to express my undying gratitude to you. I really owe you a debt of gratitude. You are not only my supervisor but also "my mother", my idol as an inspiring female scientist and a family-oriented woman.

## THESIS PUBLICATION

## Published

 <u>Ngo, L.K.</u>, Pinch, B.M., Bennett, W.W., Teasdale, P.R., Jolley, D.F., 2016. Assessing the uptake of arsenic and antimony from contaminated soil by radish (*Raphanus sativus*) using DGT and selective extractions. *Environmental Pollution*. 216, 104-114.

## THESIS PRESENTATIONS

## **Seminar presentations (abstracts)**

- <u>Ngo, L.K.</u>, Bennett, W.W., Teasdale, P.R., Jolley, D.F., 2016. Comparison of the distribution and availability of arsenic and antimony in contaminated soils. RACI Analytical and Environmental Division National Symposium, 18 – 20<sup>th</sup> July, Adelaide, South Australia, Australia.
- <u>Ngo, L.K.</u>, Jolley, D.F., Pinch, B.M., Bennett, W.W., Teasdale, P.R., 2015. An evaluation of arsenic and antimony bioavailability in historically contaminated soil. DGT Conference 2015 "From DGT Research to Environmental Assessment", 28<sup>th</sup> September 1<sup>st</sup> October, Donostia-San Sebastián, Spain.
- <u>Ngo, L.K.</u>, Pinch, B.M., Bennett, W.W., Teasdale, P.R., Jolley, D.F., 2014. Distribution and bioavailability of arsenic and antimony in contamianted soils and edible plants. 22<sup>nd</sup> Annual RACI Research and Development Conference, 13 – 15<sup>th</sup> December, Adelaide, South Australia, Australia.

## Seminar posters (abstracts)

 <u>Ngo, L.K.</u>, Bennett, W.W., Teasdale, P.R., Jolley, D.F., 2016. Availability of arsenic and antimony in contaminated soils assessed by DGT, sequential extraction, and bioassay with water spinach (*Ipomoea aquatica*). 7<sup>th</sup> SETAC (Society of Environmental Toxicology and Chemistry) World Congress, 6 – 10<sup>th</sup> November, Orlando, Florida, USA.

Chapter 1. General introduction
1.1. Statement of context
1.2. Chemistry
1.3. As and Sb contamination in soils
1.3.1. Sources of As and Sb
1.3.2. As and Sb concentrations in contaminated soils
1.4. Chemical behaviour of As and Sb in soils 4
1.4.1. Speciation, solubility, and mobility 4
1.4.2. As and Sb retention in soils
1.4.3. Associations of As and Sb in soils 12
1.5. Mechanism of metal(loid) uptake from soils by plants
1.5.1. Concentration and speciation of metal(loid)s in soil solution
1.5.2. Metal(loid) movements in soils to roots
1.5.3. Rhizosphere environment
1.5.4. Transfer of metal(loid)s from the root surface into roots
1.5.5. Accumulation of metal(loid)s into root and transport to shoots
1.6. Bioaccumulation of As and Sb in edible vegetables
1.7. Factors controlling bioavailability of As and Sb in soils and their uptake by plants 26
1.7.1. Arsenic and Sb speciation
1.7.2. Redox conditions and Fe, Al, and Mn oxide/hydroxides
1.7.3. pH
1.7.4. Organic matter
1.7.5. Soil texture
1.7.6. Phosphate
1.7.7. Plant factors

1.8. Conventional methods for estimating bioavailability of As and Sb in soils	32
1.8.1. Soil solution	33
1.8.2. Chemical extraction	33
1.9. Diffusive Gradients in Thin Films (DGT) technique	35
1.9.1. Principles of DGT in waters	36
1.9.2. Principles of DGT in soils and sediments	39
1.9.3. DGT-determined effective concentration ( $C_E$ ) as a predictor of metal bioavailability to plants	44
1.9.4. Experimental evidence of DGT applications in measuring As and Sb bioavailability in soils to plants	46
1.10. Aim and objectives	47
Chapter 2. General methods	49
2.1. General washing methods	49
2.2. Test soils	49
2.2.1. Historically contaminated soils	49
2.2.2. Recently contaminated soils	49
2.3. Test plants	50
2.4. Pot experiments	51
2.4.1. Bioassays with radishes (Raphanus sativus L.)	51
2.4.2. Bioassays with water spinach (Ipomoea aquatica L.)	51
2.5. Soil characteristic measurements	52
2.5.1. Maximum water holding capacity (MWHC)	52
2.5.2. Moisture content	52
2.5.3. Soil particle size analysis	53
2.5.4. Soil pH	53
2.5.5. Organic matter (OM)	53
2.5.6. Total Kjeldahl nitrogen	54

2.5.7. Extractable phosphorous	54
2.5.8. Amorphous Fe and Al	54
2.5.9. Total free Fe and Al	54
2.6. Soil As and Sb concentrations	55
2.6.1. Sequential extraction procedure	55
2.6.2. Dissolved As and Sb concentrations in soil solution	55
2.6.3. Total soil As and Sb concentrations	55
2.7. DGT measurements	56
2.7.1. DGT devices and assembly	56
2.7.2. Gel solution	57
2.7.3. Diffusive gel	57
2.7.4. Binding gels	58
2.7.5. Procedure for deploying DGT in soils	58
2.7.6. Calculation of DGT measured concentrations	59
2.8. Plant analysis	60
2.8.1. Pre-digestion	60
2.8.2. Plant digestion	60
2.9. Analytical method	60
2.10. Data analysis	61
Chapter 3. Assessing the uptake of arsenic and antimony from contaminated soil by	У
radish ( <i>Raphanus sativus</i> ) using DGT and selective extractions	62
3.1. Introduction	63
3.2. Methods	65
3.2.1. General experimental	65
3.2.2. Experimental design	65
3.2.3. Soil characteristics	66

3.2.4. Soil and plant analyses	66
3.2.5. Sequential extraction procedure	66
3.2.6. DGT and soil solution measurements	66
3.2.7. Data analysis	66
3.3. Results and discussion	67
3.3.1. Soil characteristics	67
3.3.2. Labile As and Sb in soils assessed by sequential extraction	69
3.3.3. Labile As and Sb in soils assessed by DGT technique	73
3.3.4. Uptake of As and Sb by radish (Raphanus sativus)	75
3.3.5. The relationship between As and Sb bioaccumulation and soil e	exposure
concentrations	
3.4. Conclusions	
Chapter 4. DGT and selective extractions reveal differences in arsenic and a	ntimony
uptake by the white icicle radish (Raphanus sativus)	
4.1. Introduction	
4.2. Materials and methods	85
4.2.1. General methods	85
4.2.2. Soil preparation and plant bioassays	85
4.2.3. Sequential extraction procedure	86
4.2.4. DGT and soil solution measurements	86
4.2.5. Data analysis	86
4.3. Results and discussion	86
4.3.1. Soil properties	86
4.3.2. Fractionation and mobility of As and Sb in soils	87
4.3.3. Labile As and Sb in soils assessed by DGT	
4.3.4. Uptake of As and Sb by radish (long white icicle, Raphanus sativus).	

4.3.5. Evaluating the effectiveness of geochemical measurements as predictors of
biological uptake
4.4. Conclusions
Chapter 5. Assessment of arsenic and antimony lability in historically co-contaminated
soils and their uptake by water spinach (Ipomoea aquatica)102
5.1. Introduction
5.2. Methods103
5.2.1. General methods103
5.2.2. Experimental design
5.2.3. Soil characterisation104
5.2.4. Soil and plant analyses104
5.2.5. Sequential extraction procedure (SEP)105
5.2.6. DGT and soil solution measurements105
5.2.7. Data analysis
5.3. Results and discussion106
5.3.1. Soil characteristics106
5.3.2. Sequential extraction109
5.3.3. DGT measurements111
5.3.4. Uptake of As and Sb by water spinach (Ipomoe aquatica)113
5.3.5. The relationship between As and Sb bioaccumulation and soil concentrations 117
5.3.6. Risk assessment of As and Sb in <i>I. aquatica</i> to human health
5.4. Conclusions
Chapter 6. Biogeochemical behaviour of As and Sb in soils and their uptake by water
spinach (Ipomoea aquatica): Implications for lability, distribution, and competition127
6.1. Introduction
6.2. Methods129
6.2.1. General methods

6.2.2. Experimental design	129
6.2.3. Soil characterisation	130
6.2.4. Soil and plant analyses	130
6.2.5. Sequential extraction procedure	130
6.2.6. DGT and soil solution measurements	131
6.2.7. Data analysis	131
6.3. Results and discussion	131
6.3.1. Soil characteristics	131
6.3.2. As and Sb fractionations in soils	131
6.3.3. Lability of As and Sb in soils	135
6.3.4. Uptake of As and Sb by water spinach ( <i>Ipomoe aquatica</i> )	138
6.3.5. The relationship between As and Sb bioaccumulation and soil concentration	ions 145
6.3.6. Risk assessment of As and Sb in <i>I. aquatica</i> to human health	147
6.4. Conclusions	151
Chapter 7. Mobilisation of As and Sb in soils induced by redox changes identified DGT and their uptake by water spinach ( <i>Ipomoea aquatica</i> )	d by 153
7.1. Introduction	153
7.2. Methods	156
7.2.1. General method	156
7.2.2. Experimental design	156
7.2.3. Soil and plant analyses	158
7.2.4. DGT procedures	159
7.2.5. Data analysis	160
7.3. Results and discussion	160
7.3.1. DGT measurements in flooded historically contaminated soils	160
7.3.2. DGT measurements in flooded recently contaminated soils	166

7.3	B.3. Accumulation of As and Sb in water spinach ( <i>Ipomoea aquatica</i> )	175
7.3	3.4. Arsenic and antimony speciation in water spinach (Ipomoea aquatica)	179
7.3	8.5. Evaluating the performance of DGT as a predictor of bioaccumulation and	
spe	eciation in <i>I. aquatica</i>	185
7.4. 0	Conclusions	191
Chapter	r 8. Concluding remarks and future directions	192
8.1.7	The geochemical behaviour of As and Sb in contaminated soils	193
8.1 seq	.1. The partitioning and lability of As and Sb in various soils determined by quential extraction procedure (SEP)	193
8.1 filr	.2. Labile As and Sb in various soils measured by the diffusive gradients in thin ms (DGT) technique	194
8.1 ind	.3. The dynamics of lability and speciation of As and Sb in contaminated soils luced by redox changes identified by DGT	194
8.2. 0	Competitive interactions of As and Sb in soils under different redox conditions	195
8.3. U differ	Uptake of As and Sb by a variety of vegetables cultivated in various soils under rent redox conditions	197
8.3	3.1. Uptake trend	197
8.3	3.2. Speciation transformation of As in <i>I. aquatica</i> 1	.98
8.3	3.3. Risk assessment of As and Sb in <i>I. aquatica</i> to human health	198
8.4. ] veget	Evaluation of the performance of DGT in predicting As and Sb uptake by tables compared with other soil extraction methods	200
8.5.	Evaluation of the combination of multiple DGT and DET to investigate	202
benav		202
8.6. F	-uture directions	203
8.6	5.1. Geochemical behaviour of As and Sb in soils	203
8.6	5.2. Uptake of As and Sb by plants	204

## LIST OF FIGURES

**Figure 1.2** Movement of elements to the root surface (taken from Marschner and Rengel 2012); (1): root interception, (2): mass flow (convection), and (3): diffusion. .. 17

**Figure 4.3** Bioaccumulation factors (BAF) and translocation factors (TF) for As and Sb of two cultivars of radish cultivated in various soils. Cherry bell radish – Mining ( $\blacksquare$ ) (chapter 3), White icicle radish – Mining ( $\Box$ ), White icicle radish – Cattle dip ( $\boxtimes$ )......96

**Figure 5.1** The percentage of As and Sb associated with different binding fractions for bioassay soils (mean of all concentration for each soil, mean  $\pm$  SE, n = 15)......109

**Figure 6.1** Biomass of *I. aquatica* cultivated in soils amended with different concentrations of As and Sb. Each value is mean  $\pm$  SE (n=3). The asterisks in the figure indicate significant differences (p<0.05) compared with the control plant tissues. ..... 139

## LIST OF TABLES

<b>Table 1.1</b> As and Sb concentrations in contaminated soils (dry weight)
<b>Table 2.1</b> Sequential extraction procedure applied for determination of the partitioning of As and Sb in soils.       56
<b>Table 3.1</b> Total and labile concentrations measured by SEP, DGT, soil solution in aconcentration gradient obtained from historically contaminated soils, and R value of Asand Sb (mean $\pm$ SE, n = 3)
<b>Table 3.2</b> Bioaccumulation factor and translocation factor for As and Sb in <i>R. sativus</i> $(mean \pm SE, n = 4)$
Table 4.1 Physical and chemical properties of bioassay soils (dry mass, mean ± SE,n≥3)
<b>Table 4.2</b> Total and labile concentrations measured by SEP, DGT, soil solution inbioassay mining soils, including R-value ( $R=C_{DGT}/C_{sol}$ ) for As and Sb (mean ± SE, n =3)
<b>Table 4.3</b> Total and labile concentrations measured by SEP, DGT, soil solution in bioassay cattle dip soils, including R-value (R= $C_{DGT}/C_{sol}$ ) for As (mean ± SE, n = 3)88
<b>Table 5.1</b> Physical and chemical properties of bioassay soils (dry mass, mean $\pm$ SE, $n \ge 3$ )
<b>Table 5.2</b> Total concentrations in a concentration gradient obtained from historically contaminated soils, R values, $K_d$ (distribution coefficients, L/kg), and kinetic parameters including $T_c$ (the response time of the desorption process, s), and $k_{-1}$ (the desorption rate constant, s <sup>-1</sup> ) obtained from DIFS of As and Sb (mean ± SE, n = 3) 108
<b>Table 5.3</b> Tissue mass, As and Sb concentrations, bioaccumulation factor (BAF) andtranslocation factor (TF) for As and Sb in <i>I. aquatica</i> exposed to contaminated soils for35 days (mean $\pm$ SE, n = 3).114
<b>Table 5.4</b> Values of EDI (mg/kg/day), HQ, and CR of As for the ingestion of <i>I.aquatica</i> shoots grown in studied soils123
Table 5.5 Values of EDI (mg/kg/day) and HQ of Sb for the ingestion of <i>I. aquatica</i> shoots grown in studied soils

**Table 6.1** As concentrations in each fraction of soils amended with (i) As only and (ii)with a mixture of As and Sb (mean  $\pm$  SE, n = 3). Numbers in brackets are % of total soilconcentrations.134

**Table 6.2** Sb concentrations in each fraction of soils amended with (i) Sb only and (ii)with a mixture of As and Sb (mean  $\pm$  SE, n = 3). Numbers in brackets are % of total soilconcentrations.134

**Table 6.5** Correlation coefficients ( $R^2$ ) between As and Sb concentration in tissues of *I*. *aquatica* cultivated in different soils and  $C_{SEP}$ ,  $C_{DGT}$ , and  $C_{sol}$  of As and Sb in soils. 146

**Table 6.7** Values of shoot Sb (mg/kg, WW), EDI (mg/kg/day), non-carcinogenic risk(HQ), and cancer risk (CR) for the ingestion of water spinach (*I. aquatica*) shoots grownin Sb-only and As+Sb soils150

**Table 7.2** Total As and As species concentrations in roots and shoots of *I. aquatica*cultivated in non-flooded and flooded soils. Data are presented as mean  $\pm$  SE (n = 3).Values in parentheses are the percentage of As species. There was no detectable MMAand DMA.181

## **Chapter 1. General introduction**

### **1.1. Statement of context**

Accumulation of As and Sb in soils is putting increasing pressure on the environment and becoming a matter of great concern. This is because they are non-biodegradable, persistent, and toxic substances, and consequently pose a severe threat to human and the ecosystem health via various pathways (direct intake or the food web, interactions of the contaminated environment-plant-animal-human). Currently there is very little known about the biogeochemical behaviour of Sb and its uptake mechanisms by plants, and it is generally assumed to be the same as that of As. In order to assess the risk to human health associated with contamination of As and Sb in soils, there is a global initiative to develop suitable techniques for assessing labile As and Sb, the distribution of various chemical forms of As and Sb in soils, the dynamics of As and Sb in soils under different conditions, and their uptake by agricultural crops. Current approaches typically utilise conventional chemical methods, including total metal digests and sequential extraction techniques, however these have not been reliable measures of metal(loid) bioavailability in soils and nor have they been successful predictors of metal(loid) bioaccumulation in many cases. Despite limitations of sequential extraction such as lack of selectivity, and potential reprecipitation and readsorption during extraction, it is still useful for providing information on the partitioning of metal(loids) in various soil binding sites. The diffusive gradients in thin films (DGT) technique is a relatively novel approach that relies on time-integrated fluxes from soils to the device, which reflects similar metal exposure scenarios between soils and plant roots. DGT has been successfully applied for measuring metal lability in soils and metal uptake by plants, however its applications for assessing labile As and Sb in soils and predicting As and Sb accumulation by edible crops have been limited. This study will investigate the biogeochemical behaviour of Sb compared with As and evaluate the effectiveness of DGT in predicting As and Sb accumulation by a variety of edible plants in different agronomic practices to better understand the relationship between DGT measurements and plant uptake. These results will also be correlated with conventional approaches to identify the most appropriate techniques in measuring bioavailable As and Sb in soils and predicting their uptake by edible plants on the long-term basic.

## 1.2. Chemistry

Arsenic (As) and antimony (Sb) are metalloids belonging to Group 15 of the periodic table (Kabata-Pendias & Mukherjee 2007). As and Sb display the same range of oxidation states in environmental systems (-3 to +5) because they have identical  $s^2p^3$  outer orbital electron configuration. They commonly exist as oxyanions in +5 states (arsenate and antimonate) in the relatively aerobic environments or in +3 states (arsenite and antimonite) in anaerobic environments. As and Sb in the environment occur in both inorganic and organic forms, with the two common inorganic forms being arsenite and arsenate, antimonite and antimonite, respectively.

While these similarities in the chemistry of As and Sb exist, there are differences in the structure of their pentavalent oxyanions. Arsenate  $(AsO_4^{3-})$  is tetrahedral, a similar structure to phosphate  $(PO_4^{3-})$ , thus arsenate and phosphate compete for the binding sites in soil and the accumulation of arsenate by plants occurs via phosphate transporters. In contrast, antimonate  $(Sb(OH)_6^{-})$  is octahedral. However, antimonite  $(Sb(OH)_3)$  is structurally similar to arsenite  $(As(OH)_3)$  (Wilson et al. 2010).

## 1.3. As and Sb contamination in soils

#### 1.3.1. Sources of As and Sb

Both As and Sb naturally occur as trace elements in soil mainly from the weathering of soil minerals. The majority of natural sources of As and Sb are rocks, volcanic ore deposits, and hydrothermal metal sulphide deposits either associated or co-precipitated with metal ores, or separate mineral ores. Common As minerals are arsenopyrite (FeAsS), orpiment (As<sub>2</sub>S<sub>3</sub>), enargite (Cu<sub>3</sub>AsS<sub>4</sub>), realgar (AsS), and As metal oxides (Arai 2010; Smith & Huyck 1999). Higher As concentrations are found in sedimentary rocks than igneous rocks, thus soils and parent materials formed from sedimentary rocks often have elevated As concentrations (Smith 1998). The concentrations of As in the earth's crust and uncontaminated soils are 1.8 - 2.5 and 6.8 mg/kg, respectively. In terrestrial environments, inorganic forms (arsenite and arsenate) are generally dominant and highly affected by redox conditions and pH. Inorganic arsenic species under oxidising conditions methylated microorganisms, can be by producing monomethylarsonic acid (MMA), dimethylarsinic acid (DMA) and trimethylarsine oxide (TMAsO) (Mandal & Suzuki 2002). Ultimately organic As species are converted into inorganic As and  $CO_2$  by oxidative degradation, or into volatile As compounds, arsine (e.g. AsH<sub>3</sub>) gas by reduction (Arai 2010).

Naturally occurring Sb is commonly found to combine with S, As, and Pb to form over 100 minerals. Sb is usually found in stibnite  $(Sb_2S_3)$ , kermisite  $(Sb_2S_2O)$ , guettardite  $(Pb(Sb,As)_2S_4)$ , cervantite  $(Sb(III)Sb(V)O_4)$ , docrasite  $(Sb(III)_2(Sb(III),As)_2)$ , and various oxides (valentinite  $Sb_2O_3$ ) of stibnite oxidation. Of these, stibnite is the only mineral commercially mined, being a source of metallic antimony. Sb is generally present in trace amounts in Ag, Cu, and Pb ores. Similar to As, Sb can be methylated and volatilised as SbH<sub>3</sub> in the environment (Arai 2010). The concentrations of Sb in the earth's crust and uncontaminated soils are 0.2 - 0.9 and 0.2 - 2.3 mg/kg, respectively (Kabata-Pendias 2011).

Substantial amounts of arsenic enter the environment through anthropogenic inputs including industrial processes, agricultural use, smelting, and mining activities. Industrial processes include smelting of As-containing ores and by-products of fossil fuel combustion (e.g. fly coal ash). Approximately 60% of As emissions are related to Cu-smelting and coal combustion (Matschullat 2000). Industrial uses include timber treatment, leather tanning, photoelectric devices, electronic, paints, Pb-acid batteries, Cu-based alloys, glassware, cosmetics, and fireworks (Kabata-Pendias 2011; Mahimairaja et al. 2005; Matschullat 2000; Smith 1998). Agricultural applications of As include pesticides, herbicides, algaecides, antibacterial agents for livestock, cattle dips, seed treatment, fertilizers, and As-enriched irrigation (Adriano 2001; Mahimairaja et al. 2005; Smith 1998).

Most Sb pollution originates from mining and smelting activities, coal and fuel combustion, industrial and municipal waste, outfall of sewage, shooting ranges, fertilizers, and emission from vehicles where Sb is used as a fire-retardant in brake linings (Adriano 2001; Arai 2010; Bhattacharya et al. 2007; He et al. 2012; Tschan et al. 2009b). Industrial uses include Pb-acid batteries, cable sheathings and ammunition, semiconductors, flame-retardants such as textiles, papers, plastics, and adhesives, a paint pigment, ceramic opacifier, catalyst, mordant, and glass decolouriser. The wide uses of Sb have led to substantial inputs of Sb into the environment under dust emissions and incineration of industrial and municipal wastes.

In general, Sb is a co-contaminant in soil. In mine tailings Sb is associated with the primary resource element (usually As), in orchards Sb is found with As, in shooting ranges with Pb, in dust and incineration of waste with other elements such as As, Cd, Cr, Cu, Co, Hg, Mn, Ni, Pb, V, and Zn.

## 1.3.2. As and Sb concentrations in contaminated soils

The concentrations of As and Sb in uncontaminated soils are generally less than 10 mg/kg (Kabata-Pendias 2011). However, their concentrations in contaminated soils vary greatly and can reach 310000 mg As/kg and 80200 mg Sb/kg, depending upon the nature and contamination sources (Mahimairaja et al. 2005; Wilson et al. 2010). Arsenic and Sb concentrations in soils contaminated by anthropogenic inputs are presented in Table 1.1, a review of recent publications. For all soils in Table 1.1, the total soil As concentrations exceeded the maximum tolerable concentration of As in agricultural soils set by EU (20 mg/kg) (Álvarez-Ayuso et al. 2016; Bhattacharya et al. 2010b) and the total soil Sb concentrations were several orders of magnitude larger than concentrations in unpolluted soils in China (< 3 mg Sb/kg) (He 2007) and Europe and North America (< 10 mg Sb/kg) (Filella et al. 2002a; Okkenhaug et al. 2011).

## 1.4. Chemical behaviour of As and Sb in soils

## 1.4.1. Speciation, solubility, and mobility

pH and redox potential ( $E_h$ ) greatly influence metalloid oxidation state and consequently their solubility and mobility in soils. In geochemical environments, inorganic As is primarily present in two oxidation states (+3 and +5). In the oxic environment, arsenate, As(V), is the main species. As(V) exists as H<sub>3</sub>AsO<sub>4</sub> (pK<sub>1</sub> = 2.20, pK<sub>2</sub> = 6.97, pK<sub>3</sub> = 12.13) at pH below 2, in the pH range (2 – 11) As(V) species are charged due to the dissociation of H<sub>3</sub>AsO<sub>4</sub> to H<sub>2</sub>AsO<sub>4</sub><sup>-</sup> and HAsO<sub>4</sub><sup>2-</sup>. At low E<sub>h</sub> value (-200 to +300 mV), As is found in the +3 oxidation state as H<sub>3</sub>AsO<sub>3</sub> (pK<sub>1</sub> = 9.22, pK<sub>2</sub> = 12.1, pK<sub>3</sub> = 13.4) which does not dissociate until pH 9. At higher pH values H<sub>3</sub>AsO<sub>3</sub> exists as H<sub>2</sub>AsO<sub>3</sub><sup>-</sup> and HAsO<sub>3</sub><sup>2-</sup>. At E<sub>h</sub> below -250 mV the formation of insoluble compounds such as AsS, As<sub>2</sub>S<sub>3</sub> occurs in the presence of sulphide. Elemental arsenic and arsine can be formed under very strong anaerobic conditions (Arai 2010; Bissen & Frimmel 2003). In the normal range of soil pH (4 – 9), As(III) species are uncharged, while As(V) species are usually negatively charged. In addition to the pH, the mobility of As in soils is influenced by the available sorbent types (e.g. Fe, Al, Mn hydroxides). The surfaces of Fe, Al, and Mn oxides/hydroxides and clay minerals are only positively charged below soil pH 8, 5, 3, and 4, respectively (Lombi & Holm 2010). The lack of charge on As(III) species compared to successive deprotonated forms of As(V) species leads to less potential for association with solid phases such as oxides/hydroxides and clay minerals in soils (Wilson et al. 2010). Thus, As(III) species are generally more mobile than As(V) species in environmental systems, hence more toxic (Van Herreweghe et al. 2003; Wilson et al. 2010).

In the soil/water environment As can form solubility products with aluminum, iron, calcium, and sulphur. Arsenate ions precipitate with Fe(III) and Al(III) to form amorphous scorodite and aluminum arsenate at acidic conditions, while calcium arsenate precipitates at alkaline conditions (Fergusson 1990). The solubility constants for iron and aluminum arsenates are low  $(10^{-11})$  being smaller than that for calcium arsenate (10<sup>-5</sup>), illustrating that Fe and Al play a role in controlling As availability in soils (Arai 2010; Smith 1998; Smith et al. 1999). In oxic systems, arsenate is the predominant species and readily binds to mineral compositions. In reducing conditions, adsorbed As can be released via the reductive dissolution of adsorbents, for instance, the reduction of ferric ion into ferrous ion. In addition, a direct conversion of As(V) to As(III) increases mobile As concentration because of the weak sorption of As(III) on soil constituents. However, the decrease in mobile As concentration in long-term flooded soils has been observed (Onken & Hossner 1995). This can be explained by the resorption of As on solids (Onken & Adriano 1997) and co-precipitation of Mn<sub>3</sub>(AsO<sub>4</sub>)<sub>2</sub> (Masscheleyn et al. 1991) under long-term and moderately anoxic conditions (0 - 100)mV).

Soil concentration range (mg/kg)		Contamination source	Country	Reference
As	Sb	_		
826 - 1606	2735 - 4517	Mine sites	Australia	Wilson et al. 2013
479	2412	Historic gold-antimony mine	Australia	Doherty et al. 2017
4400	11152	Former Sb processing plant	Australia	Doherty et al. 2017
191 - 38600	2.5 - 237	Former mining activities	Australia	Ashley & Lottermoser 199
1.8 - 40	0.1 - 39.4	Antimony mine 300 km upstream	Australia	Tighe et al. 2005a
59 – 176	180 - 554	Antimony mine 2 km upstream	Australia	Telford et al. 2009
42 - 4530	60 - 230	Abandoned mining areas	Spain	Casado et al. 2007
246 - 758	14.1 - 324	Former mining activities	Spain	Álvarez-Ayuso et al. 2012
7.4 – 3115	0.2 - 35.7	Mining-affected soils	Spain	Pérez-Sirvent et al. 2012
146 - 540	525 - 4463	Old mining sites	Slovakia	Vaculík et al. 2013
9.16 – 447	4.86 - 2058	Forest soils near the smelter	Czech Republic	Ettler et al. 2010
4.33 – 154	3.12 - 131	Agricultural soils near the smelter	Czech Republic	Ettler et al. 2010
7.4 – 596	74.2 - 16389	Mine and its surrounding sites	China	Li et al. 2014
40.02 - 400.2	610 - 54221	Old antimony mine	China	Wei et al. 2011
50 - 17428	10 - 1187	Former mining areas	Scotland	Gál et al. 2006
16 - 691	1.63 - 11.44	Former mining areas	Italv	Gál et al. 2006
11.1 – 651.1	30.5 - 5986.4	Mining-impacted soils	Portugal	Pratas et al. 2005
2220	80200	Smelter site	New Zealand	Wilson et al. 2004
4.4 - 65.1	2.5 - 175	Urban soils impacted by Pb-Zn smelter	France	Douay et al. 2008
37 – 13040	7.4 - 13610	Abandoned mining area	Slovakia	Rapant et al. 2006
	35 - 17500	Shooting range	Switzerland	Johnson et al. 2005
	26 - 1150	Mining contaminated area	France	Denys et al. 2009
	27.7 - 15112.9	Previous mining area	Italy	Baroni et al. 2000
	11.89 - 709.84	Mining and smelting areas	England	Flynn et al. 2003
	100.6 - 5045	Areas affected by mining activities	China	He 2007
	527 - 11798	Areas affected by mining activities	China	Okkenhaug et al. 2011
	14 - 15100	Areas affected by mining activities	Spain	Murciego et al. 2007
	585 - 3184	Areas affected by mining activities	Spain	Álvarez-Ayuso et al. 2013
	1300 - 17500	Shooting range	Switzerland	Scheinost et al. 2006
	629 - 8230	Shooting range	Switzerland	Robinson et al. 2008
200 - 310000		Abandoned gold mining sites	Canada	Meunier et al. 2010
16000		Former mining areas	Thailand	Visoottiviseth et al. 2002
5.3 - 2035.3		Former mining areas	Italv	Baroni et al. 2004
129 – 4779		Cattle dip sites	Australia	Rahman et al. 2017
21 – 1406		Cattle dip sites	Australia	Niazi et al. 2011

**Table 1.1** As and Sb concentrations in contaminated soils (dry weight).

The speciation of Sb is also influenced by pH and redox potential and greatly affects its solubility and mobility in the geochemical environments. This means that the speciation influences Sb retention in soils and soil processes. In the typical soil environment, pH ranging from 4 - 9 and  $E_h$  ranging from - 300 to + 900 mV (Husson 2013), Sb(III) and Sb(V) are the dominant species (Filella et al. 2009). Over the relevant environmentally pH range, the negatively charged ion (Sb(OH)<sub>6</sub><sup>-</sup>) is the dominant Sb(V) species, while in reduced conditions the uncharged Sb(OH)<sub>3</sub>, antimonous acid with pK<sub>a</sub> = 11.9 is the most abundant Sb(III) species (Filella et al. 2002a; Wilson et al. 2010). Thus, like As, Sb(III) species may be more mobile and toxic than that of Sb(V) (Filella et al. 2002b; Wilson et al. 2010). Under reducing conditions, the solubility of Sb is limited by the solubility of Sb(III) sulphides (e.g. stibnite) and oxides (e.g. Sb(OH)<sub>3</sub>, Sb<sub>2</sub>O<sub>3</sub>, Sb<sub>2</sub>O<sub>4</sub>) (Arai 2010). In addition, Sb(III) can readily complex with Cl<sup>-</sup> to form SbCl<sup>2+</sup>, SbCl<sub>2</sub><sup>+</sup>, SbCl<sub>3</sub>, and SbCl<sub>4</sub><sup>-</sup> in acidic aqueous solutions (Oelkers et al. 1998).

Under oxic soil environments, the oxidation of Sb(III) to Sb(V) is likely to occur and Sb(V) should be the dominant sorbed Sb species. In a batch experiment using a water extraction of 12 Japanese soils, Nakamaru et al. (2006) reported that anionic Sb(V) was the major Sb species in soil solution. Oorts et al. (2008) also stated that the majority of Sb(V) (70%) was found within 2 days of the addition of Sb as  $Sb_2O_3$  into soil. Using XAFS analysis, Mitsunobu et al. (2006) showed that Sb in mining-impacted soil was present as Sb(V) over a wide range of redox (- 140 to + 360 mV, pH 8) and suggested that Sb(V) was very stable in the soil. Under anoxic conditions, the reductive dissolution of Fe, Al, and Mn oxides could lead to the release of adsorbed Sb.

#### 1.4.2. As and Sb retention in soils

Both As and Sb are strongly retained in soils (Ettler et al. 2010; Flynn et al. 2003; McLaren et al. 1998; Wilson et al. 2010). The degree of retention determines their labile and bioavailable fractions, which in turn determine their persistence, reaction, transformation, and toxicity. The retention is influenced by many factors such as metalloid species present and soil properties. Understanding the As and Sb retention processes in soils is a foundation for understanding their biogeochemical behaviour and for accurate and adequate risk

assessment in various systems. Adsorption is the most important mechanism in controlling the metalloid bioavailability (Wilson et al. 2010).

## 1.4.2.1. Adsorption

The adsorption of As and Sb in soils involves two different mechanisms, namely nonspecific and specific adsorption. Non-specific adsorption refers to the electrostatic attraction between positive charges of adsorbents and negative charges of As and Sb anions (Manful et al. 1989; Sadiq 1997). Conversely, specific adsorption refers to the predominant sorb via ligand exchange, the incorporation of As and Sb species as ligands in the coordinated shell of mineral compounds (Manful et al. 1989; Sadiq 1997). The adsorption of As and Sb in soils is controlled by soil characteristics and environmental factors such as type of oxides/hydroxides and clay minerals and their amount present, organic matter content, pH, As and Sb species, and competing ions (Frost & Griffin 1977; Grafe et al. 2001; Manning & Goldberg 1996a; Wilson et al. 2010).

#### Adsorption on oxides and hydroxides

According to Arai (2010), anion adsorption on soil components is a function of the net surface charge density of the sorbent and the chemical speciation of sorbate, which in turn depends upon the pH. Generally, the adsorption of anions on minerals increases with decreasing pH due to the negatively charged chemical species and the positively charged mineral surfaces when pH < PZC (point of zero charge). Thus, As and Sb are expected to strongly sorb onto surfaces of metal oxides through electrostatic interaction when pH – PZC is <0 and to dominantly sorb through ligand exchange when pH – PZC is >0.

Of all the mineral components, oxides/hydroxides of Fe, Mn, and Al are actively involved in adsorption processes (Deschamps et al. 2003; Smith et al. 1999). Soils with high amounts of metal oxide minerals can retain higher amounts of As (Smith et al. 1999). The surface charge of various minerals is pH dependent (Figure 1.1). Under acidic systems, these minerals have positive surface charges, thus strongly retaining oxyanions of As and maybe Sb (Sadiq 1997); however, the positive surface charges vary with metal oxide/hydroxide type. The surfaces of Mn, Al, and Fe oxide/hydroxide limit the retention of As and potentially Sb when soil pH is greater than 3, 5, and 8, respectively (Figure 1.1, Sadiq 1997). In the pH range of 7 - 9, carbonate minerals are positively charged, thus play an important role in As retention (Mahimairaja et al. 2005; Sadiq 1997). The adsorption of As onto minerals also depends on the As species (As(III) or As(V)) present. At acidic and neutral conditions, As(V) exists as negatively charged  $H_2AsO_4^-$  and  $HAsO_4^{2^-}$ , and the negative charge increases as medium pH increases (Smith 1998; Zhang & Selim 2008). Conversely, As(III) is present as neutral species,  $H_3AsO_3$ , at pH <9.2.

A rise in pH decreases As(V) adsorption but increases As(III) adsorption in soils (Smith et al. 1999). This decline of As(V) adsorption may be related to two interaction factors: (i) increased negative surface potential on adsorbents and (ii) increased amount of As(V) anions in soil solution due to deprotonation of As(V) compounds (Manful et al. 1989; Smith et al. 1999). Both arsenate and arsenite display a maximum adsorption at a certain pH. This pH value is variable and adsorbent-dependent. For instance, the adsorption As(V) maxima was achieved at pH 5 and for As(III) was pH 7 on alumina (Xu et al. 1988, 1991). As(III) sorption maxima was observed on amorphous Fe(OH)<sub>3</sub> at about pH 7, As(V) at about pH 4 (Pierce & Moore 1982).



**Figure 1.1** Generalised overview of charges distribution on soil colloids, showing the dependence of soil pH on charges formation of different soil compositions and minerals (adopted from Sadiq 1997).

Oxide/hydroxides are also known to be a sorbent for Sb in soil (Manaka 2006; Mitsunobu et al. 2006; Wilson et al. 2010). Sb(III) strongly sorbs to oxide/hydroxides and the sorption follows the order MnOOH > Al(OH)<sub>3</sub> > FeOOH, with a gradual decrease in the amount sorbed as pH rises above 6. Below pH 6, more than 80% of Sb(III) added is retained by these sorbents (Thanabalasingam & Pickering 1990). Adsorption of Sb(III) on goethite is stronger than As(III) given the same initial concentrations of anion and goethite (Leuz et al. 2006). In addition, goethite has a greater affinity for Sb(III) than Sb(V) (Xi & He 2013). Sb(V) adsorption to Fe hydroxide is more pH dependent than that of Sb(III) with an adsorption maxima below pH 7 (Leuz et al. 2006). According to Tighe and Lockwood (2007), 95% of Sb(V) was sorbed by crystalline Fe hydroxide across pH 2.5 - 7 with a maximum sorption at about pH 4. Sb(V) sorption on hematite is also strong and maximum at similar pH 4 (Pierce & Moore 1982). However, Sb(V) is not as strongly sorbed to goethite as As(V) over a wide pH range (Leuz et al. 2006), influencing on their mobility and bioavailability in specific systems. The primary sorption mechanism is inner-sphere surface complexes and binuclear bridging complexes (Fendorf et al. 1997; Leuz et al. 2006; Scheinost et al. 2006).

#### Adsorption on silicate clay minerals

Silicate clay minerals are also known to be important in As and Sb retention in soils (Frost & Griffin 1977; Gál et al. 2006; Manning & Goldberg 1996a). According to Wilson et al. (2010), adsorption of anions in clay minerals is associated with broken clay particle edges, commonly via surface ligand exchange mechanisms (e.g.  $-M-OH + H_2AsO_4^- = -M-H_2AsO_4 + OH^-$ ). More As is absorbed by illite and montmorillonite clay minerals than by kaolinite, possibly due to higher surface area of the former clay minerals (Frost & Griffin 1977; Manning & Goldberg 1997). The adsorption of two inorganic As species on silicate minerals also differs. Generally, clay minerals adsorb greater amounts of As(V) than As(III) (Lin & Puls 2000). As(V) adsorption on kaolinite and montmorillonite clay minerals increases at low pH, peaking at about pH 5, and thereafter decreasing (Frost & Griffin 1977; Goldberg & Glaubig 1988). Conversely, As(III) adsorption on kaolinite continuously increases between pH 4 – 9, but peaks near pH 7 on montmorillonite (Frost & Griffin 1977). Sb adsorption on clay minerals is less studied, yet the Sb adsorption on Al-

silicate minerals appears to be important, depending upon Sb origin (Gál et al. 2006). Like As(III), the adsorption of Sb(III) on kaolinite is strongly dependent upon pH and decreases with increasing pH (Xi et al. 2016). The presence of competitive anions have no obvious effect on the adsorption of Sb(III) on kaolinite.

## Adsorption on organic matter

The role of organic matter in the adsorption of As is complex. Some studies have reported that organic molecules decrease As retention by competing with As for sorption sites on mineral surfaces (Grafe et al. 2001; Redman et al. 2002). The degree of adsorption is dependent on pH, As speciation, and type of competing organic molecules. According to Grafe et al. (2001), peat fulvic and humic acids diminished As(V) adsorption on goethite, whereas no effect was found with citric acid. The maximum reduction of As(V) adsorption on fulvic acid occurred at pH 3 – 8 and at pH 6 – 9 for humic acid. The decrease of As(III) adsorption on these organic acids followed the order of citric acid > fulvic acid ~ humic acid at a pH range of 3 – 8. Redman et al. (2002) reported that the adsorption of As(V) and As(III) on hematite was reduced by organic compounds. In contrast, the role of organic matter as an As adsorbent has been shown (Cao & Ma 2004; Cao et al. 2003; Saada et al. 2003), with the amine groups of humic acid being considered responsible for As retention on organic matter (Saada et al. 2003).

The association of Sb with soil organic matter has also been reported (Ceriotti & Amarasiriwardena 2009; Clemente et al. 2008). Up to 30% of total Sb(III) species readily bound to humic acid at environmentally relevant conditions (Buschmann & Sigg 2004; Filella et al. 2002b). The binding strength of Sb (III) is comparatively stronger than As(III). Sb(III) was found to constitute up to 34% of total Sb in soil organic acids (Ettler et al. 2007) and the interaction with humic acid significantly influenced mobility. In contaminated shooting range soils humic acid had a high capacity to complex and tightly bind the predominant Sb species, Sb(V), trapping most of it in the organic layer (Steely et al. 2007). Significant amounts of added Sb(V) (56%) sorbed on humic acid at pH 4 was also reported by Tighe et al. (2005b). Ligand exchange with Sb and formation of negatively charged Sb complexes with phenolic, carboxylic, and hydroxyl-carboxylic groups may be a mechanism of Sb association with organic matter. The association might be due to the
formation of bridging products between metalloid anions, organic matter, and Ca (Buschmann & Sigg 2004; Steely et al. 2007).

In general, the information of soil components and properties (e.g. % clay, % silt, % sand; Fe and Al oxide content; organic matter content, pH, redox potential) allows inferences of the formation of binding phases in soils and the availability of As and Sb in soils studied.

### 1.4.2.2. Competition

Phosphate ions are analogues of  $A_{S}(V)$  ions and can compete with As for adsorption sites on mineral compounds in soil, suppressing the adsorption of As (Manful et al. 1989; Roy et al. 1986). Phosphate is able to compete with arsenate for available adsorption sites, both non-specific and specific (Qafoku et al. 1999). According to Liu et al. (2001), phosphate and arsenate primarily compete for surface sites on goethite, but certain soil sites are uniquely specific for phosphate and arsenate adsorption. Other anions such as nitrate, chloride, sulphate, and molybdate also influence the adsorption of arsenate, but their effect on As adsorption is not profound (Livesey & Huang 1981; Manful et al. 1989; Manning & Goldberg 1996b). Competitive effects on Sb adsorption are relatively unknown. According to a recent study by Xi et al. (2016), the presence of competitive anions such as nitrate, sulphate, and phosphate have no obvious effect on the adsorption of Sb(III) on kaolinite. Addition of P as superphosphate enhanced the release of As and Sb in contaminated firing range soils (Kilgour et al. 2008). The addition of phosphate decreased the distribution coefficients of Sb in Japanese agricultural soils, with 20 - 40% of sorbed Sb (Nakamaru et al. 2006) and 0.2 - 1.3% of total soil Sb (Nakamaru & Sekine 2008) in phosphate exchangeable phase. It seems that the mobilisation of Sb by phosphate addition is dependent on the Sb species present and the extent of ligand exchangeable Sb.

# 1.4.3. Associations of As and Sb in soils

The mobility and bioavailability of As and Sb in soils are strongly influenced by their association with soil solid phases. The binding strength depends on the binding phase to which As and Sb are bound. Arsenic and Sb partitioning in soils can be operationally-defined by geochemical fractions including dissolved, soluble, labile, Fe and Al oxides (reducible), organic matter (oxidisable), and residual (Larios et al. 2013).

#### 1.4.3.1. As and Sb in dissolved, mobile or labile fractions

The dissolved As and Sb in soils can be measured by extracting soils with deionised water at a certain ratio of soil and water. This phase can be isolated by centrifugation and ultrafiltration (Ettler et al. 2007). Casado et al. (2007) reported that the mobile As and Sb extracted from abandoned mining-impacted soils were 0.02 - 0.7% of total soil As and 0.02-0.27% of total soil Sb, reflecting that the plant-available fractions were extremely low compared to their concentrations in soils. Generally, the concentrations of dissolved or soluble As and Sb only account for a small proportion of their total soil concentrations (<2%) and greater than 99% of As and Sb in soil solution were arsenate and antimonate (Wilson et al. 2010). The readily available fraction, the non-specifically sorbed As and Sb, which are outer sphere complexes or freely exchangeable fraction, are obtained from the first extraction step with  $(NH_4)_2SO_4$  of sequential extraction procedure (SEP) developed by Wenzel et al. (2001). This fraction contains the labile species comprised of free ions and soluble inorganic and organic complexes. It consists of the most labile and available chemical species for plant uptake (Anawar et al. 2008; Casado et al. 2007; Larios et al. 2012a). The freely exchangeable As is significantly positively correlated with As concentration in plant shoots (Niazi et al. 2011).

The adsorped fraction includes As and Sb that are specifically-sorbed to soil surfaces as inner-sphere complexes. Arsenic and Sb in this phase are released by anion exchange processes using ammonium phosphate, NH<sub>4</sub>H<sub>2</sub>PO<sub>4</sub> (Wenzel et al. 2001). Ammonium phosphate has been previously validated in SEPs for As in oxidized soils (Keon et al. 2001; Wenzel et al. 2001) and mine wastes (Drahota et al. 2014). It has been stated in section 1.4.2 that both As and Sb are strongly adsorbed onto mineral surfaces. Many studies have demonstrated that phosphate solutions are efficient in extracting adsorbed As in soils and mine wastes (Drahota et al. 2001; Larios et al. 2012b; Larios et al. 2013; Wenzel et al. 2001). These authors suggest that phosphate extracts the freely exchangeable As (outer-sphere complexes) as well as strongly adsorbed As (inner-sphere complexes), depending on the ionic strength and duration of extraction. The quantity of anions in these fractions is used to measure trace elements which are mobilised and readily released back to the environment. Arsenic and Sb in the non-specifically sorbed and specifically sorbed

phases are defined as the labile and bioavailable fractions (Wenzel et al. 2001). Thus, ammonium phosphate has been used as a single extraction to determine the labile fraction of As and Sb in soils (Ettler et al. 2007; Hamon et al. 2004). Labile As and Sb in long-term contaminated soils account for a small proportion of their total soil concentrations, less than 15% (Ettler et al. 2010; Kim et al. 2014; Ngo et al. 2016; Niazi et al. 2011; Wilson et al. 2010), and are highly correlated with As and Sb concentrations in plants (Gonzaga et al. 2008; Ngo et al. 2016; Niazi et al. 2011). In addition, in long-term contaminated soils labile Sb is much lower than labile As.

## 1.4.3.2. As and Sb incorporated into oxide/hydroxides

Mineral oxide/hydroxides are strong scavengers of As and Sb in soils (Drahota et al. 2014; Tighe et al. 2005a). Fe and Al oxides are typically the fraction binding the greatest proportion of As in the environment (Lock et al. 2016). The selected reagents used for extracting As and Sb associated with amorphous and crystalline Fe and Al are oxalate buffer and oxalate buffer plus ascorbic acid at 96 °C (Wenzel et al. 2001). The fractionation of As in 20 soils conducted by Wenzel et al. (2001) showed 12 - 73% of total As associated with amorphous Fe and Al oxides, but 13 - 39% associated with crystalline forms. Many later studies using or adapting the improved procedure of Wenzel et al. (2001) have reported the same trend (Álvarez-Ayuso et al. 2016; Gál et al. 2006; Kim et al. 2014; Müller et al. 2007; Niazi et al. 2011).

Similar to As, the proportion of Sb extracted from metal oxides is much greater than from the labile fraction. In a study on chemical fractionation of Sb in acid soil by Tighe and Lockwood (2007), 30 - 47% of total Sb was associated with non-crystalline Fe and Al hydroxides. In another study on speciation and availability of Sb in active mining impacted-soil by Okkenhaug et al. (2011), 22 - 66% of total Sb was reported to be associated with amorphous and crystalline Fe and Al oxides. The proportion of Sb associated with oxide fractions is dependent on Sb origin (He 2007).

# 1.4.3.3. As and Sb associated with organic matter and sulphide

As and Sb might be incorporated into different forms of organic and sulphide material known as the oxidisable fraction. The binding between As and Sb and organic matter is due to adsorption and complexation (as discussed in section 1.4.2.1). Oxidising reagents such as

hydrogen peroxide  $(H_2O_2)$  in acid medium (Jones et al. 1997; Lintschinger et al. 1998),  $Na_4P_2O_4$  (Larios et al. 2013), and a mixture of KClO<sub>3</sub>, HCl, and HNO<sub>3</sub> (Drahota et al. 2014) are usually applied to leach As and Sb bound to organic matter and sulphides. Applying a mixture of hydrogen peroxide and ammonium acetate to extract As associated with organic fraction, Jones et al. (1997) found about 10% of total As in this fraction but Lintschinger et al. (1998) found no significant amount of Sb in this fraction (<4.1%). In a comparative study on the partitioning of As and Sb in forest soils, (Ettler et al. 2010) used the same reagents and found that up to 32% of the total Sb bound in the organic and sulphide fraction, while 6 - 22% of the total As was in this fraction. In the same study with agricultural soils, Ettler et al. (2010) reported that Sb bound to organic matter and sulphide fraction was 2 - 12%, whereas As bound in the oxidisable fraction was 2 - 7%. The proportion of Sb in the organic and sulphide fraction appears to be higher than As. It can be generalised that the proportion of As and Sb in the oxidisable fraction depends on the contaminant, soil sources, and minerals. This fraction is negligible in soil surface horizons, but it may be important or even predominant in soil rich in sulphide or organic matter contents such as forest soils.

# 1.4.3.4. As and Sb in residual fraction

The residual fraction contains mainly silicates and other resistant minerals holding As and Sb in the crystalline lattice structure. These As and Sb are not soluble under natural environmental conditions, thus are not biologically available, and only become available as mineral weathering occurs over a long period of time. The destruction of the highly recalcitrant As and Sb fraction is obtained by strong acid digestion with a mixture of the following strong acids such as HF, HClO<sub>4</sub>, HCl, and HNO<sub>3</sub> (He 2007; Tighe & Lockwood 2007). The residual fraction depends not only upon the contaminant source, but also factors controlling As and Sb mobility such as dissolution of primary phases, As and Sb adsorption by soil components, and interactions with other variables (e.g. pH, redox potential, competing ions) (Wilson et al. 2010). In general, the residual As and Sb is high in which the proportion of Sb in residual phase is higher than As.

#### 1.5. Mechanism of metal(loid) uptake from soils by plants

This section will describe the processes by which (metal)loids are taken up by plant roots and how these (metal)loids are subsequently distributed within the plant tissues. This is relevant because it provides a foundation for the interpretation of soil processes around the root zone and As and Sb uptake by plants in this research.

#### 1.5.1. Concentration and speciation of metal(loid)s in soil solution

The general paradigm in metal(loid) uptake from soils by plants is that plant roots absorb element ions via soil solution and the concentration and species of dissolved ions affects the uptake rate (McLaughlin et al. 2011). In soil solution, dissolved metal(loid)s exist as various forms categorised as free ions, inorganic complexes, and organic complexes (Hooda 2010; McLaughlin 2001). The total quantity of metal(loid) ions in soil solution affects the plant uptake of (metal)loid ions from the soil solution (Alloway 1995; Barber 1995). The metal(loid)s commonly enter the plant roots as ions via ion channels having capacity to concentrate elements from soil solution. According to Campbell (1995) and McLaughlin et al. (2011), if the metal(loid) uptake by plants is well predicted by free ion activity, the plant acquisition must be slow and the uptake does not induce the depletion of metal(loid) concentration in the rhizosphere.

The uptake rate generally increases with increasing ion concentrations in soil solution, which is called a concentration-dependent pattern or plant tissue concentrations increasing as soil concentrations increase. Such patterns are important in risk assessment in perspective of identifying soil concentrations at which plant tissue concentrations are below targeting values. This pattern is the basis of the application of bioaccumulation factor concept in risk assessment (McLaughlin et al. 2011).

## 1.5.2. Metal(loid) movements in soils to roots

The uptake of metal(loid)s from soil by plants depends on ion transport from the bulk soil to the root surface. There are three principal processes for metal(loid) transport in soils: 1) root interception, 2) mass flow (convection), 3) diffusion (Figure 1.2).



**Figure 1.2** Movement of elements to the root surface (taken from Marschner and Rengel 2012); (1): root interception, (2): mass flow (convection), and (3): diffusion.

Root interception describes the movement of roots through the soil when they develop. As roots extend they replace spaces formerly occupied by soil, putting them in contact with elements sorbed to various solid phases. The proportion of elements intercepted by roots is based on a) the amount of available elements in the soil volume occupied by the roots, b) root volume as a percentage of total soil volume, and c) the percentage of the total soil volume occupied by pores (Marschner & Rengel 2012). In general, only a small proportion of available elements can be supplied by root interception.

Mass flow is the movement of dissolved ions through the soil to the root in the convective flow of water caused by plant water absorption. Transpiration from the plant shoots governs the convective transport; thus, the contribution of mass flow to the element supply depends on the portion of water transpired and the content of elements in the soil solution (Marschner & Rengel 2012). Comparison of the rate of supply with the rate of influx provides the significance of mass flow. Alternatively, mass flow can be calculated by multiplying water use per plant by the concentration in the soil solution. Differences in water use attributed to crop, climate, and moisture conditions result in variations in the significance of mass flow. Mass flow can have importance in the supply of water-soluble micro and macro-elements into plants (Barber 1995).

When a particular nutrient (micro- and macro- nutrient) is not sufficiently supplied to the root by mass flow and root interception, the concentration of available nutrients in the soil at the root surface is reduced because of continued uptake. This leads to a concentration gradient perpendicular to the root surface and subsequently nutrients diffuse along the gradient toward the root surface (Barber 1995). Equilibrium is not reached since roots absorb nutrients, causing the continuous diffusion to the root along the concentration gradient. In general, the diffusional supply of nutrients is the most important path for plant uptake. The distance in which the concentration gradient extends is dependent on the rate of diffusion. The distance for diffusive movement of nutrients through the soil to the root is usually in the range of 0.1 to 15 mm (Barber 1995; Marschner 1995). Thus, only the soil nutrients in this zone can contribute to the supply of diffusive nutrients to the root. Practically, if the depletion of ions occurs, ions are replenished from liquid or solid/complexed forms. This means that a fraction of complexed ions is also in the bioavailable form in soils, provided that the environmental condition favours the sufficient rapid desorption and dissociation. For this reason, bioavailability functionally includes both the activity in soil solution and the labile bound forms (McLaughlin et al. 2011). It is also reasonable that diffusive fluxes correlate well with the plant uptake, provided that the metal(loid) is depleted in the rhizosphere (Nolan et al. 2005).

Field-based data for some plants and soils show that the mass flow exceeded the root uptake of As, consequently As may have accumulated around the roots (Chen et al. 2009). This may be related to the As root absorption power and its mechanism regulating the uptake by plants. These will be discussed in sections 1.5.4 and 1.5.5.

## 1.5.3. Rhizosphere environment

The rhizoshpere is defined as a few millimetres (1 - 2 mm) of soil surrounding the plant roots and affected by plant activities (Alloway 1995). The rhizosphere conditions effectively control the supply of contaminants to plant roots. Factors affecting the rhizosphere include biological parameters (e.g. the root system including exudates, nutrient status, microorganisms), and physicochemical conditions (e.g. pH, redox potential, soluble organic matter, depletion and accumulation of ions) (Fitz & Wenzel 2002; Gobran et al. 2001). Plant roots have the ability to modify the solubility and availability of elements in the soils, influencing the biogeochemical conditions in the root area (Hinsinger 2000; Marschner 1995; Mengel & Kirkby 2001). For instance, organic exudates are capable of mobilising nutrients, making anions (e.g. phosphates) and cations (Cu, Zn) more available. Moreover, plants are able to alter the pH of the rhizosphere thanks to the release of organic acids serving as soil solution buffer (Marschner 1995). In addition, the biogeochemical cycle in the rhizosphere is influenced by microorganisms. If bacterial activities in the rhizosphere are high, methylation and reduction is favoured (Renella et al. 2007).

Since phosphate and arsenate are chemically analogous, all processes for P mobilisation are likely to mobilise As. For example, organic acids having a low molecular weight (citric and malic acids) are capable of displacing arsenate from a position of retention in soils (Redman et al. 2002; Wenzel 2009). In addition, plant strategies include the attachment of Fe oxide/hydroxides to alter the mineral surfaces retaining the As, which can potentially solubilise As (Fitz & Wenzel 2002). Moreover, arsenate adsorbed by corn roots can be significantly reduced by corn mycorrhizae, association between fungi and plant roots (Yu et al. 2009).

### **1.5.4.** Transfer of metal(loid)s from the root surface into roots

In general, there are two key mechanisms controlling the uptake of metal(loid)s by plant roots, passive and active (Alloway 1995; McLaughlin 2001). Passive (non-metabolic) uptake involves the diffusion of metal(loid) ions from soil solution into root endodermis without demanding energy and uptake of ions in response to the established concentration gradient. Active uptake takes place if ions have a net negative charge and acts against a concentration gradient. This process requires metabolic energy and thus can be inhibited by toxins. The barrier to plant uptake mechanisms which require energy is the plasma membrane inside the cell wall (Barber 1995). Two main ways for solutes to move from the soil solution to the xylem vessels in roots are apoplasmic or symplasmic flow (Figure 1.3). Apoplasmic pathways involve the movement of solutes through intercellular spaces with pores in cell walls of the cortex up to the Casparian strip. The Casparian strip around the stele. Symplasmic pathways involve the direct movement of solutes through the symplasm of cells in the cortex up to the Casparian strip (from cell to cell, selective transport crossing membranes). In the Casparian strip, the solute ions in these two flows have to cross a plasma membrane in the cytoplasm to enter the xylem vessels. Then they can be transported to the shoots.

The adsorption of As by roots occurs by diffusion from soil solution in the root apoplast, followed by the symplast, penetrating to the interior of the plant cells (Mengel & Kirkby 2001). Ions pass through the plasma membrane via transport proteins specific for one or more elements with similar chemical characteristics. Both phosphate and aquaporins (water channel) are involved in the transmembrane transport and adsorption of As. Many studies have illustrated that arsenate uses a phosphate transporter (Esteban et al. 2003; Meharg & Macnair 1992; Zhao et al. 2009) and arsenite enters the membrane via aquaporin channel (Isayenkov & Maathuis 2008; Ma et al. 2008) since they share the same chemistry in terms of ion size and charge density.



**Figure 1.3** Simplified scheme of the root, displaying pathways of metal transport across the root to the stele (adopted from Luo 2008).

The mechanisms of Sb uptake by plants are relatively unknown. In the preliminary study by Tschan et al. (2008), antimonate uptake by plants was not affected by phosphate, suggesting that the same pathway as phosphate does not occur for antimonate. This difference is because the structure of antimonate is octahedral, while that of arsenate and phosphate is tetrahedral. Additionally, antimonate has a larger size and lower charge density, setting antimonate apart from the other oxyanions (Tschan et al. 2008). According to the review of Tschan et al. (2009b), antimonite may cross plasma membranes passively via aquaporins since  $As(OH)_3$  and  $Sb(OH)_3$  are similar in their structural conformation and charge distribution with glycerol. Such transport would agree with the observed proportionality between water-soluble Sb concentrations and Sb in plants. However, aquaporins are not open for antimonate. The uptake of antimonate by plants would require regulation by transporters.

# 1.5.5. Accumulation of metal(loid)s into root and transport to shoots

Once inside the cell, arsenate is rapidly reduced to arsenite, using glutathione as a reductant, with the catalysis of arsenate reductase (AsO<sub>4</sub><sup>3-</sup> + 2GSH  $\rightarrow$  AsO<sub>3</sub><sup>3-</sup> + GSSG) (Moreno-Jiménez et al. 2012; Verbruggen et al. 2009). Thus, it has been shown that the majority of As existing in plant tissues is arsenite (Tripathi et al. 2007), irrespective of which As species was in the plant growing medium. Arsenite is known to have a high affinity for -SH groups and to be complexed and stored in vacuoles, despite being able to be transported to shoots via the xylem. The complexation of As with glutathione (GSH) and phytochelatins (PCs), has been identified in various plants such as Holcus lanatus, P.cretica, Brassica juncea, Rauvolfia serpentine, and Helianthus annuus (Montes-Bayón et al. 2004; Pickering et al. 2000; Raab et al. 2004; Raab et al. 2005; Schmoger et al. 2000). The movement of ions via the xylem is regulated by both the flow of transpiration stream, and membrane transport proteins (Moreno-Jiménez et al. 2012). Two transporters (Lsi1 and Lsi2) have been recently described between the epidermis and endodermis and mediate arsenite entrance into xylem or its effux to the external medium (Moreno-Jiménez et al. 2012; Zhao et al. 2009). Accumulation of arsenite in vacuoles might explain the decrease in As transport into the xylem (Zhao et al. 2009). The transport of As in most plants is commonly not very effective, thus As tends to be accumulated in roots, with an exception for some plant species unusually effective in accumulating As in shoots. The reduction process in plant roots may contribute to the physiological mechanism plants use to block the transport of As via the xylem and limit the flow of As into shoots, hence protecting them from the effect of As.

For the long-distance transport of As in plants, arsenate is considered to act similarly to phosphate, while no plant transporter has been identified for mediating transport of arsenite from root to shoot; furthermore, the role of chelators for this transport has not been described (Tripathi et al. 2007). Pickering et al. (2000) and Raab et al. (2005) identified unbound As(III) and As(V) in xylem sap of *Brassica juncea* and *Helianthus annuus*, in addition to As-PC complex. Furthermore, the majority of As was present as complexes in roots (Vazquez Reina et al. 2005), and the amount of As complexed by –SH groups was negatively correlated with the amount of As translocated to shoots (Huang et al. 2008). With these findings, many authors postulated that complexed As is not transported through the xylem and As(III) and As(V) are the main species transported from roots to shoots where similar reduction and sequestration occur (Farooq et al. 2016; Mendoza-Cózatl et al. 2011; Raab et al. 2005; Tripathi et al. 2007).

Generally, once the ions have been absorbed by the roots, the transport of solutes in plants may occur in the vascular system of xylem and phloem. Translocation is an important process in determining trace element concentrations in plant tissues. The translocation of solutes to the shoots depends on the element concerned, plant species and variety, plant age, external concentration of the element, and solution composition (Alloway 1995; Barber 1995). In the leaves, ions may be incorporated into proteins or translocated around the plant in the phloem with substances made by photosynthesis. Plants are grouped into different categories based on how much metal(loid) is accumulated in shoots (Baker 1981). A hyperaccumulator plant shows a high transfer factor, leading to extreme amounts of metal(loid)s in leaves (Baker 1981). It is reported that As hyperaccumulators can rapidly translocate a large proportion of metal(loid)s from roots to shoots, while non-accumulators are likely to sequester metal(loid)s in the roots (Baker 1981). The As concentration in non-accumulator shoots rarely exceeds 2 mg As/kg (Horswell & Speir 2006).

In contrast to As, the Sb accumulation in roots, the mechanisms of transport from roots to shoots, and Sb storage in leaves remains unknown (Feng et al. 2013). Based on the higher Sb(V) concentrations in rice shoots than in roots and pore water of growing media, Okkenhaug et al. (2012) postulated that rice may preferentially accumulate Sb(V) and oxidise Sb(III) to Sb(V) in shoots. Whether Sb is involved in redox reactions in plants is unclear. Additionally, whether Sb(V) is accumulated by roots and reduced to Sb(III), which is in turn complexed with phytochelatins is still unclear (Feng et al. 2013). In a recent review on cellular and molecular mechanisms of Sb transport, toxicity, and resistance there was evidence of intracellular reduction, however the mechanism of Sb(V) reduction to Sb(III) outside the cell is not understood (Tamás 2016). In the cytoplasm, Sb(V) may also be reduced to Sb(III) enzymatically or non-enzymatically and then sequestrated into vacuoles of yeast cells. However, the involvement of PCs in Sb(III) resistance may be limited due to trace amounts of Sb(glutathione)<sub>3</sub> detected. There is little information on the production and responses of PCs to Sb exposure in plants, further investigation is needed.

### 1.6. Bioaccumulation of As and Sb in edible vegetables

Metal(loid)s are accumulated by plants when they are exposed to soils or water contaminated with metalloids. Metal(loid) uptake by plants, especially edible crops, presents the risk to human health through the food chain (Wilson et al. 2014). The scenario of metal(loid)-enriched edible plants has led to many hydroponic and soil studies via greenhouse and field based experiments, conducted to understand their uptake potential by various vegetable species and subsequent risks that may occur to humans. The maximum permissible As concentration (MPC) of 0.1 mg/kg (wet weight, WW) recommended by FAO/WHO and Australian MPC of 1 mg/kg (WW) for foodstuffs, specific to cereals have usually been referred to when evaluating the risks to humans from the consumption of contaminated vegetables (Burló et al. 1999; Carbonell-Barrachina et al. 1999a; Rahman & Naidu 2009; Uddh-Söderberg et al. 2015; Wilson et al. 2014). Numerous studies reveal that As concentrations in edible tissues of crops grown in historically contaminated soils are lower than MPCs (Kabata-Pendias 2011; Kabata-Pendias & Mukherjee 2007; Warren et al. 2003). However, some vegetable species such as silverbeet, spinach, tomato, and turnip can accumulate As in their edible parts to concentrations exceeding the critical values (Burló et al. 2003).

al. 1999; Carbonell-Barrachina et al. 1999a; Carbonell-Barrachina et al. 1999b; Rahman & Naidu 2009).

Ample greenhouse and field based studies on As have been conducted to understand its mechanism of uptake by various plants. Generally, As accumulation in various parts of edible crops followed the order roots >> stems > leaves > fruits (Cobb et al. 2000; Kabata-Pendias 2011; Smith et al. 2009). Rice and wheat accumulated As from contaminated soils and waters in the order root >> stem > leaf > grain> husk (Kabata-Pendias 2011). The contaminated soil particles or As adsorbed on the iron layer on the root surfaces of edible root crops may contribute to high As bioaccumulation in roots. Where inadequate washing is exercised, this may result in direct intake of As by human, in spite of low As concentrations in plant tissues (Carbonell-Barrachina et al. 1999a; Carbonell-Barrachina et al. 1999b).

The soil-plant transfer model is commonly used to predict metal(loid) uptake by plants and estimate the potential risk present to humans via digestion of vegetables produced from contaminated soils. The simple and commonly-used parameters are bioaccumulation factor (BAF) determined by the ratio of metal(loid) concentration in plant to metal(loid) concentration in soil and translocation factor (TF) defined as the ratio of metal(loid) concentration in shoots to metal(loid) concentration in roots (Álvarez-Ayuso et al. 2012; Antunes et al. 2006; Murray et al. 2009; Sharifi et al. 2014; Vamerali et al. 2010). A higher BAF reflects the high availability of metal(loid)s in soil and thus greater plant efficiency in transferring metal(loid) from soil to the plant. A low BAF indicates the low availability of metal(loid)s or strong adsorption of metal(loid)s to soil constitutes (Douay et al. 2013). TF is an index to evaluate the ability of plants to translocate metal(loid)s from roots to shoots, used for classification of plants into different categories. Tolerant plants tend to limit soilroot and root-shoot transfers, thus have low bioaccumulation in tissues, while hyperaccumulator plants actively take up metal(loid)s from soil to roots and translocate metal(loid)s from roots to shoot tissues. Plants with high BAF (normally >1) are often classified as hyperaccumulators and can be used for phytoextraction. Plants with high BAF (>1) and low TF (<1) are suitable for phytostabilisation (Vamerali et al. 2010; Vithanage et al. 2012). The existing literature on the As uptake by plants reveals that the typical bioaccumulation factor of As is about 0.2 (Wilson et al. 2013).

In terms of Sb research, a range of studies has been recently reported on Sb phytoavailability (He 2007; Wanat et al. 2014), toxicity to plants (Oorts et al. 2008; Sanderson et al. 2014; Shtangeeva et al. 2011; Tschan et al. 2009a; Vaculík et al. 2015), accumulation in plants (Álvarez-Ayuso et al. 2013; Conesa et al. 2011; Hajiani et al. 2017; Qi et al. 2011; Shtangeeva et al. 2014; Tschan et al. 2009a), and uptake mechanisms (Corrales et al. 2014; Mathews et al. 2011; Tisarum et al. 2015; Tisarum et al. 2014; Tschan et al. 2010; Tschan et al. 2008). In addition, a number of studies have focused on Sb uptake by rice with respect to the effect of the iron plaque and speciation on its uptake due to concerns with As uptake by rice (Cai et al. 2016; Cui et al. 2015; Huang et al. 2012; Okkenhaug et al. 2012; Ren et al. 2014). Most of these preliminary studies are field-based and hydroponic experiments. Only a few studies have investigated Sb accumulation in edible vegetables and food crops (Hammel et al. 2000; Ngo et al. 2016; Wilson et al. 2014). A recent review on Sb bioavailability with research perspectives for sustainable agricultures also reported Sb concentrations in a range of edible plants (Pierart et al. 2015). However, almost all edible plants in the cited studies were collected in the fields. This would lead to some limitations in terms of evaluating Sb bioaccumulation and understanding the mechanism of Sb uptake by edible vegetables. Some field studies showed that Sb concentrations in plant shoots were unusually high (exceeding 100 mg/kg) due to foliar uptake of Sb (Baroni et al. 2000; He 2007). According to McLaughlin et al. (2011), in air non-gaseous metal(loid) associated with small particulates can be deposited on plants.

Antimony accumulation varies widely among plant species (Pierart et al. 2015; Tschan et al. 2009b). In contrast to high Sb concentrations in plants collected in the fields, low Sb accumulation in plants grown in heavily contaminated soils was also reported (Tschan et al. 2009b). The typical bioaccumulation factor for Sb is about 0.02 in the reviewed literature (Wilson et al. 2013). Based on published data Tschan et al. (2009b) performed regression analysis and found that there was a significant correlation between extractable Sb concentrations in soils and Sb concentrations in plants. Interestingly, the slope of the

regression line was almost exactly to 1, indicating proportionality. Such proportionality is found to extend over five orders of magnitude of Sb concentrations. It was hypothesised that Sb uptake by plants would follow passive transport by convection with the stream of water transport into and through plants. Sb accumulated in edible plants may present a health risk to human via the food chain; however, there has been no maximum permissible concentration of Sb in foodstuffs, especially edible vegetables. The danger is when plants can accumulate high concentrations of Sb while still looking healthy.

# 1.7. Factors controlling bioavailability of As and Sb in soils and their uptake by plants

# 1.7.1. Arsenic and Sb speciation

Among As species, As(III) is more toxic than As(V) and inorganic species are more toxic than organic species with the order: As(III) > As(V) > MMA > DMA (Baig et al. 2010). Inorganic As species are dominant in plants (Abedin et al. 2002; Smith et al. 2008a) and can be transformed into organic species by methylation with microbial activities, especially in paddy soils (Takamatsu et al. 1982). Various As species have distinct solubilities and hence display different bioavailabilities to plants. The uptake of these four As species by common vegetable species showed a variable trend. For example, the As accumulated in tomatoes followed the trend: As(III) ~ As(V) >> MMA > DMA (Burló et al. 1999). Marin et al. (1992) reported that the As availability to rice followed the trend As(III) > MMA > As(V) > DMA. Similarly, Meharg and Hartley-Whitaker (2002) observed As(III) and MMA are more available to rice plants.

Inorganic species of Sb predominate over its organic species in environmental systems (Wilson et al. 2010). Sb(III) is generally more toxic than Sb(V) (Gebel 1997; He & Yang 1999). Sb speciation greatly influences Sb uptake by plants Feng et al. (2013). For example, rice was more efficient in accumulating Sb(III) than Sb(V), and Sb(V) was the main species in rice tissues (Okkenhaug et al. 2012; Ren et al. 2014). In addition, Shtangeeva et al. (2012) reported that wheat (*Triticum aestivum* L.) exposed to SbCl<sub>3</sub> accumulated more Sb in seeds and tissues than that exposed to SbCl<sub>5</sub>. The opposite was true for rye (*Secale cereal* L.) under the same growing conditions. These reports indicate that the knowledge of As and Sb speciation in soils is important in assessing the As and Sb bioavailability and toxicity to plants.

#### 1.7.2. Redox conditions and Fe, Al, and Mn oxide/hydroxides

Fe, Al, Mn oxides have been largely applied for immobilisation of As and Sb in contaminated soils (Bagherifam et al. 2014; Doherty et al. 2017; Okkenhaug et al. 2016; Warren et al. 2003), in which iron-based amendments were the most effective with up to 90% decrease of As and Sb in soil pore water (Doherty et al. 2017). In addition, Bagherifam et al. (2014) stated that natural metal oxides of Mn, Al, and Fe reduced As bioavailability (82%) and Sb (60%) and significantly decreased plant uptake and bioaccessibility of As and Sb in soils. In the field trials to assess As uptake by vegetables from contaminated soils and soils treated with iron oxides, Warren et al. (2003) reported that application of 0.5% of Fe oxides reduced As bioavailability and its uptake by beetroot, cauliflower, lettuce, potato, radish, and spinach by a mean of 32%. The pot experiment conducted on As uptake by lettuce cultivated in contaminated soils applied with ferric sulfate also observed reduction of As concentration in lettuce by 11% (Warren & Alloway 2003). For paddy soils, As and Sb mainly found in rice roots due to their adsorption by the iron plaque formed on the root surfaces, reducing As and Sb in rice plants (Cai et al. 2016; Okkenhaug et al. 2012). Many studies on paddy soils showed the significant correlations between As fractions and iron fractions, suggesting that iron redox cycling may directly influence As fractionation in soils, which may thereafter indirectly influence As bioavailability and uptake by rice plants (Fu et al. 2011; Liu et al. 2015). Additionally, significantly negative correlations between As in rice grain and As bound to amorphous Fe oxide were found, indicating that amorphous Fe oxide acts as a sink for As.

However, in reducing environments such as flooded paddy soils and waterlogged soil scenarios, arsenite and antimonite were predominant and the solubility of As and Sb increased sharply in soil solution (Hockmann et al. 2014a; Okkenhaug et al. 2012; Takahashi et al. 2004; Xu et al. 2008). The increase in As and Sb availability is due to the reductive dissolution of Fe (hydr)oxides and the reduction of As (V) and Sb(V) to As(III) and Sb(III), respectively, through microbial processes (Gorny et al. 2015; Hockmann et al. 2014a; Hockmann et al. 2014b; Islam et al. 2004; Masscheleyn et al. 1991; Nakamaru & Altansuvd 2014), resulting in more As and Sb available for plant uptake (Wan et al. 2013a; Xu et al. 2008). Flooding contaminated soils greatly enhanced Fe and Mn concentrations in

soil pore water, followed by an increase in As concentration in rice and reduction of rice growth (Hartley et al. 2010). According to Williams et al. (2007), the redox condition is driving force of the high bioaccumulation of As in paddy rice compared to other grain crops.

# 1.7.3. pH

Soil pH is one of the important parameters controlling the mobility and phytoavailability of As and Sb since it controls As and Sb speciation and retention in soils via adsorption. The specificity of As and Sb adsorption on the surfaces of soil components is influenced by pH, depending on the soil texture and types of mineral constitutes (Rosas-Castor et al. 2014). The difference in mobility of As and Sb is attributed to the effect of pH on surface charge of mineral constitutes and protonation or deprotonation of As and Sb species (as discussed in section 1.4.2.1). A few studies have described the effect of soil pH on As mobility in the presence of plants (Kim Anh et al. 2013; Tu & Ma 2003) who reported the enhanced As uptake by As-hyperaccumulating plants in an acidic environment. In a recent study, Ahmed et al. (2011) found a positive correlation between As in grain and soil pH. The As uptake can be enhanced at higher pH since the rise in soil pH increases negative surface charges on mineral components (as discussed in section 1.4.2.1), facilitating desorption of As from Fe oxides (Marin et al. 1992; Sahoo & Kim 2013), leading to the mobilisation of labile As in the rhizosphere, enhancing As accumulation in plants (Fitz & Wenzel 2002). The effect of soil pH on As and Sb mobility in the absence of plants (laboratory batch studies) could be different from studies where plant species are included since plants are capable of altering the rhizosphere pH by producing organic exudates (as discussed in section 1.5.3). There is very limited literature about the effect of pH on the uptake of metal(loid)s, especially Sb by vegetables.

# 1.7.4. Organic matter

The effect of organic matter (OM) on As and Sb mobility and their uptake by plants is inconsistent (Sahoo & Kim 2013). OM can control As and Sb mobility in soils since it forms soluble and insoluble complexes with As and Sb (Das et al. 2008; Wang & Mulligan 2006; Williams et al. 2011). An increase in OM can promote the desorption of As from solid phase (Buschmann et al. 2006; Redman et al. 2002; Sahoo & Kim 2013; Turpeinen et

al. 1999), increasing As mobility in soils. This is because it can enhance activities of soil microbes and decrease soil redox (Turpeinen et al. 1999), favouring the reductive dissolution of Fe oxyhydroxides (Harvey et al. 2002). The release of As through the reductive dissolution of oxyhydroxides linked to OM was observed (Selim Reza et al. 2010). In addition, dissolved organic matter (DOM) can increase As mobility since its negative charges have high potential to compete with As for adsorption sites in soils (Lin et al. 2008; Sahoo & Kim 2013). Arsenic solubility in soils was found to be positively associated with OM (Harvey et al. 2002; Huang et al. 2006; Rowland et al. 2006; Turpeinen et al. 1999). This may be the result of the positive correlation between As concentration in rice and soil OM (Bhattacharya et al. 2010a). An increase in DOM can enhance As(III) and As(V) mobility in soils (Dobran & Zagury 2006).

In contrast, OM has a high affinity for As because of the formation of insoluble organo-As complexes and their adsorption onto organic colloids (Cao & Ma 2004; Cao et al. 2003; Das et al. 2008; Sahoo & Kim 2013). Thus, soils with high contents of OM can significantly reduce As mobility (Lund & Fobian 1991), resulting in less As available to plants and accumulation in plants. Rahaman et al. (2011) investigated the effect of OM on the transport of As in rice and observed that the addition of organic amendments considerably reduce As concentration in rice grains, which was also supported by Fu et al. (2011) reporting a negative relationship between rice grain As and soil OM. However, it has been recently claimed that organic amendments may have insignificant effects on the As accumulation in plant tissues compared to unamended soils (McBride et al. 2013). The first study on the effect of biochar amendments on the mobility and bioavailability of As and Sb simultaneously present in contaminated soils conducted by Lomaglio et al. (2016) showed that the addition of biochar increased the solubility of As and Sb together with the increase in soil pore water pH. In addition, the indicator plants, dwarf beans (*Phaseolus* vulgaris L.), accumulated greater As and Sb concentrations in various tissues. This may be explained by an increase in soil pH which induced the mobilisation of humic acids, displacing As and Sb from inorganic and organic binding sites and made them available. Additionally, the alkaline conditions generated by the biochar addition resulted in the desorption of As and Sb from soil particles.

#### 1.7.5. Soil texture

Soil texture also affects metal(loid) bioavailability to crops (Adriano 2001). In general, finer textured soils (clay and silt) have more surface active area and more Fe oxides present than coarse (sand) soils, thus clay soils have a greater metal(loid) adsorption potential compared with sandy soils due to greater sorption sites in clay (Fitz & Wenzel 2002; Heikens et al. 2007; Mahimairaja et al. 2005; McLaughlin et al. 2011). As a result, clay soils have less bioavailable metal(loid)s and less phytotoxicity of metal(loid)s than sandy or loamy soils (McLaughlin et al. 2011; Warren et al. 2003). O'Neill (1995) reported that As concentration in plants is lower in silty clay loam than loamy sand. A field study on potential health risks of As and Sb by Sharifi et al. (2014) showed that higher As and Sb uptake by plants at the site linked to coarse-grained texture (dominant by sand) compared to sites with finer texture (dominant by clay) because of higher bioavailable As and Sb in sandy soils.

# 1.7.6. Phosphate

Phosphate is also an important factor influencing As solubility in soils and its uptake by plants (Fitz & Wenzel 2002) as previously discussed (section 1.4.2.2). Since phosphate and arsenate are chemical analogues, they compete for the same adsorption sites in soils via ligand exchange mechanisms, resulting in the desorption of As and the increase in As solubility and bioavailability in soils and rhizosphere (Bogdan & Schenk 2009; Smith et al. 2002). Many studies reported that increasing phosphate concentration enhances As mobility in soils (Cao et al. 2003; Signes-Pastor et al. 2007; Smith et al. 2002). Hence, application of phosphate into soils can promote As accumulation in plants, supported by numerous studies using greenhouse experiments (Bogdan & Schenk 2009; Creger & Peryea 1994; Geng et al. 2005; Hartley et al. 2009; Hossain et al. 2009). Nevertheless, there is a contrasting effect in some cases since phosphate has an inhibitory effect competing with arsenate for the same transport channel when crossing the plasma membrane (Meng et al. 2011; Tripathi et al. 2007; Zhao et al. 2009). Some studies have reported that the addition of phosphate in solution decreases As accumulation in plants (Pigna et al. 2010; Rahman & Hasegawa 2011; Tu & Ma 2003; Wang et al. 2002; Wang & Duan 2009). The overall effect of

phosphate on arsenate accumulated into root cells may depend upon the balance between these two mechanisms.

The effect of phosphate on arsenite uptake by plants is negligible (Wang et al. 2002). This insignificant effect may be due to differences in their uptake mechanisms in which arsenite is transported through aquaporins, whereas phosphate and arsenate share the same transporter (Abedin et al. 2002; Meharg & Jardine 2003; Meharg & Macnair 1990). For Sb, the effect of phosphate on Sb mobility in soils and its uptake by plants is relatively unknown. In a study on Sb mobility in Japanese soils, Nakamaru et al. (2006) reported that the distribution coefficients of Sb decreases with increasing phosphate concentration used as an extractant, suggesting that Sb sorption and desorption was influenced by phosphate. However, the specific adsorption mechanism could not be explained since only 20 - 40% of Sb sorbed on soils was extracted by phosphate. In terms of plant uptake, Tschan et al. (2008) reported that the addition of phosphate did not affect the accumulation of antimonate in maize (*Zea mays*) and sunflower (*Helianthus annuus*) and deducted that plants did not use the phosphate uptake route to accumulate antimonate. Thus, further studies are needed to understand these mechanisms.

Apparently, these factors (e.g. pH, redox conditions, Fe and Al oxides, organic matter, phosphate) affecting the adsorption/desorption, precipitation/dissolution or complexation/decomplexation of As and Sb in soils determine the solubility, biovailability, and speciation of As and Sb in soils, which in turn influences their uptake by plants. This provides the background for the explanation of As and Sb biogeochemical behaviour in various soils and plants in this research.

# 1.7.7. Plant factors

# 1.7.7.1. Crop species and cultivars

The ability of plants to accumulate As in their tissues varies with crop species and varieties. For example, in a hydroponic study with similar growth conditions, tomatoes accumulated higher As concentrations than bean plants (Carbonell-Barrachina et al. 1997). Moreover, among various rice cultivars, IR 50, Red Minikit, and White Minikit accumulated higher As concentrations in grain (0.24 - 0.31 mg/kg) than Jaya, Nayanmani, Ratna, and Lal Sanna (0.14 - 0.20 mg/kg) (Bhattacharya et al. 2010c). A similar scenario was found for

Sb, demonstrated by a study of Cai et al. (2016) reporting that the japonica rice (cultivars N and Z) exhibited a higher accumulation and stronger translocation tendency of Sb from the root to stem than indica hybrid rice (cultivars F and G). The variation is likely due to genetic and morphological differences (Zhang & Duan 2008), although it may be influenced by other factors such as differences in As concentrations in soils and environmental conditions (Sahoo & Kim 2013). The genotype differences in As accumulation is probably due to differences in root structure, controlling root aeration and the formation of iron plaque on root surfaces (Chaturvedi 2006; Liu et al. 2004) or dissimilarities in As tolerance genes (Dasgupta et al. 2004). The variation of arsenic accumulation by rice amongst genotypes is affected by root structure (Dwivedi et al. 2010; Wu et al. 2011a). Thus, the selection of appropriate cultivars is crucial to limit As concentrations in edible parts of plants for food safety.

# 1.7.7.2. Duration of crop cultivation

The As accumulation by plants also depends upon the course of crop cultivation. Jedynak et al. (2010) reported that 6-week old mustard plants accumulated about 3 times higher As concentrations in tissues than 3-week old plants. This tendency indicates that As uptake by plants occurs continuously during their lifetime.

The accumulation of As and Sb by various edible vegetable species and different cultivars of the same edible vegetable cultivated in various contaminated soils will be performed in this research to evaluate their behaviour with crop species and plant cultivars. This is important for the appropriate selection of vegetable species and varieties in terms of food safety.

#### 1.8. Conventional methods for estimating bioavailability of As and Sb in soils

As and Sb exist in various soil binding sites in which some forms are highly soluble, labile, and available for plant uptake, affecting the food chain, while others are inert and rarely affect the amount of metal(loid) in the soil solution. Thus, determination of As and Sb bioavailability in soil plays an essential role in the prediction of plant uptake and the assessment of environmetal toxicity problems (Wilson et al. 2010). A number of methods have been applied for assessing of bioavailability of soil As and Sb.

#### **1.8.1. Soil solution**

Plants take up nutrients and contaminants mainly from the soil solution (Barber 1995). The measurement of total dissolved elements in soils is considered as a mean of assessing the bioavailability and toxicity of trace elements in soils. Trace elements in the soil solution exist in a variety of chemical forms such as free hydrated ions, complexed with a wide range of naturally occuring organic and inorganic ligands and bound to colloidal material (Hooda 2010). The simple procedures to obtain soil solution are centrifugation (Thibault & Sheppard 1992) and a Rhizon soil solution sampler (Menzie & Guppy 2000; Tiensing et al. 2001). Centrifugation tends to extract a higher amount of solutes in the soil solution when compared to Rhizon samplers (Lorenz et al. 1994; Tiensing et al. 2001). The solution extracted by the Rhizon sampler is from larger pores and the physical structure of soil sample remains intact and is not disturbed. Hence, the Rhizon sampler is considered to better represent the soil pore water extracted by plants than centrifugation (Nolan et al. 2003; Tiensing et al. 2001). The use of soil solutions to predict bioavailability of metals to plants has given mixed results.

### **1.8.2.** Chemical extraction

A variety of chemical extractions have been used as surrogate measurements for predicting bioavailable As and Sb in soils. The underpinning idea is that a chemical extraction procedure is capable of isolating species or forms of elements that represent their mobile pool in soil defined as exchangeable, mobile and plant-available fraction. Ideally, a test should extract only the labile fraction and should not alter the partitioning between the phases. Many attempts have been made to find a water-soluble extractant able to isolate the labile elements in soil to relate this faction to phytoavailability.

Single extraction procedures have been widely used and are still applied due to their simplicity from laboratory-operational aspects. The common extractants can be divided into three categories: organic chelators, mildly acidic solutions, and neutral salts. Chemical extractions have returned varying results in measuring bioavailable As and Sb in soils and predicting their uptake by plants. For Sb, there have been varying degrees of effectiveness in extracting Sb from soil found in decreasing order  $NH_4NO_3 > EDTA > CH_3COOH > CH_3COONH_4$  (He 2007). For As, chemical solutions of NaHCO<sub>3</sub>, NH<sub>4</sub>Cl, NH<sub>4</sub>F, NaOH,

HCl, and  $H_2SO_4$  poorly predicted bioavailability of As to edible rape. The efficiency of various single extractants in predicting As bioavailability to plants is less successful when a range of soil types is included (Song et al. 2006). Neither ammonium sulphate nor ammonium phosphate extraction could explain phytotoxicity of arsenate added to 16 soils with various properties (Song et al. 2006). However, in other studies, phosphate solution has been proven to be a good predictor of As and Sb bioavailability in soils (Ettler et al. 2007; Huang et al. 2006).

Sequential extraction procedures (SEP) involve processing a sample of soils with a series of extractants in order to determine the concentrations of trace elements in different fractions Wenzel et al. (2001). Sequential extraction can provide information about geochemical partitioning of trace elements in soils, which can supply the distribution of trace elements in various phases and evaluate their lability, bioavailability, and thus toxicity. There are a number of sequential extraction procedures proposed by many authors, however the methods proposed by Wenzel et al. (2001) have been widely applied for As and Sb fractionation in soils. The associations of As and Sb with various binding phases in soils were described in section 1.4.3.

Chemical extraction approaches are commonly used in environmental research; however, most extractants are less specific than desired. The problems of these methods are non-selectivity of extractants and redistribution of elements during extractions (Young et al. 2005). However, the information about the partitioning of metal(loid)s in various soil binding phases can provide useful information on the potential mobility and availability of metal(loid)s under specific environmental conditions (Wilson et al. 2010). In addition, fractionation by sequential extraction can be useful in understanding the chemical reactions of metal(loid)s or their transformation in soils. Information obtained from validated sequential extraction procedures can be useful in making long-term risk assessment and providing supplementary information for remediation decisions (Hooda 2010).

In this research, SEP can provide valuable information on the partitioning of As and Sb in several solid pools, their potential mobility and labile pools, compared to the performance of the *in situ* technique, diffusive gradients in thin films (DGT), in measuring labile and bioavailable As and Sb in soils.

#### **1.9.** Diffusive Gradients in Thin Films (DGT) technique

DGT was initially developed for measuring trace metal species in aqueous media (Davison & Zhang 1994; Zhang & Davison 1995), then extended to soil and sediment deployments (Davison et al. 2000; Zhang et al. 1995). The DGT technique is capable of measuring various ionic species (e.g. cationic metals, anionic metalloids, phosphorus, sulphides) depending on the binding agent used. Some typical examples include Chelex 100 resin for trace metals (Cusnir et al. 2014; Dahlqvist et al. 2002; Garmo et al. 2003; Zhang & Davison 1995); Ferrihydrite (FeOOH), Metsorb (TiO<sub>2</sub>), and ZrO<sub>2</sub> for As, Sb, Se, V, Mo, P, W (Bennett et al. 2010; Ding et al. 2010; Guan et al. 2015; Luo et al. 2010; Österlund et al. 2010; Panther et al. 2013; Panther et al. 2008; Panther et al. 2010; Price et al. 2013; Sun et al. 2014; Zhang et al. 1998a); silver iodide for sulphides (Teasdale et al. 1999); 3mercaptopropyl-functionalised silica gel for As(III), Sb(III), and methyl-Hg (Bennett et al. 2016b; Bennett et al. 2011; Clarisse & Hintelmann 2006; Gao et al. 2014); and mixed binding agents (e.g. Chelex-Metsorb, Chelex-Ferrihydrite, Chelex-ZrO<sub>2</sub>) for both cations and anions (Huynh et al. 2012; Mason et al. 2005; Panther et al. 2014; Sun et al. 2015; Wang et al. 2017). The DGT technique has been tested in various sample matrices including waters, soils, and sediments. For metal(loid)s, examples of some important applications of DGT are investigation of metal speciation in water (Davison & Zhang 1994; Zhang & Davison 2000), sediment biogeochemical processes (Bennett et al. 2012a; Gao et al. 2009; Stockdale et al. 2009; Wu et al. 2011b), bioavailability in sediments (Amato et al. 2014; Simpson et al. 2012), and bioavailability in soils (Huynh et al. 2010; Nolan et al. 2005; Wang et al. 2014; Zhang et al. 2001).

The DGT assembly consists of a plastic base in the form of a piston and a cylindrical cap that accommodates a resin layer overlaid with a layer of diffusive gel which is covered by a filter membrane to protect it from particles. The cap has a window of known surface area allowing the target elements to contact the filter membrane (Figure 1.4). A layer of hydrogel impregnated with binding resins or powders for target analytes is often used as the binding gel layer and a polyacrylamide gel is usually used as the diffusive gel layer (Zhang & Davison 1995). The principles of DGT application for measurements of solutes in solutions, soils, and sediments are described below.



**Figure 1.4** Cross section of DGT assembly: water and soil sampling piston (a) (taken from Bennett et al. 2010), sediment probe (b) (taken from Harper et al. 1998).

# 1.9.1. Principles of DGT in waters

The DGT technique is based on a simple device that is capable of accumulating labile chemical species on a binding agent after diffusion across a filter membrane and a hydrogel which serves as a diffusion layer (Figure 1.5). A binding agent, selective to the elements of interest or species of element in solution, is embedded in a thin layer of hydrogel (binding-gel). It is separated from the bulk solution by an ion-permeable gel layer (diffusive gel) and a filter membrane with thickness of  $\Delta g$  (Davison & Zhang 1994; Zhang & Davison 1995).

The diffusive gel layer is separated from the bulk solution by a diffusive boundary layer (DBL) with thickness of  $\delta$ , where transport of target ions is limited to molecular diffusion. After the device is immersed in solution, a steady state linear concentration gradient is established between the bulk solution and the resin gel. Based on employing this simple steady state condition the DGT technique can be used to determine concentrations of chemical species in situ.



**Figure 1.5** Schematic representation of a steady state concentration gradient of an analyte through a DGT assembly, where C: the concentration of analyte; DBL: diffusive boundary layer;  $\Delta g$ : thickness of diffusive gel layer;  $\delta$ : thickness of diffusive boundary layer (taken from Davison et al. 2000).

Ion diffusion through the gel layer is driven by Fick's first law of diffusion, as shown in Equation 1.1.

$$J = \frac{D(C - C)}{\Delta g + \delta}$$
 Equation 1.1

where J (mol cm<sup>-2</sup> s<sup>-1</sup>) is the flux of metal(loid) ions diffusing through the diffusive gel layer to the resin gel; D (cm<sup>-2</sup> s<sup>-1</sup>) is the diffusion coefficient of metal(loid) ions in the gel; C is the concentration of a metal(loid) ion in the bulk solution; C' is the concentration of a metal(loid) ion at the interface between the binding gel and the diffusive gel. When the diffused metal(loid) ions are efficiently bound by resin gel, C' is effectively zero, providing that the binding resin is not saturated. In well-stirred solutions, Davison and Zhang (2012) have shown that the boundary layer thickness ( $\delta$ ) can be neglected. Therefore, Equation (1.1) can be simplified to Equation 1.2.

$$J = \frac{DC}{\Delta g}$$
 Equation 1.2

Following deployment, the resin layer is retrieved and the mass of the accumulated ions in this layer is measured. The mass can be measured directly in the binding-gel layer with techniques that are capable of analysing solids such as X-ray Fluorescence (XRF) or Proton-induced X-ray Emission (PIXE). More commonly, metal(loid) ions in the resin layer can be eluted with a known volume of extraction solution (e.g. 1 M HNO<sub>3</sub>, 1 M NaOH, 1 M NaOH/1 M H<sub>2</sub>O<sub>2</sub>) (Bennett et al. 2016a). The extract can then be analysed instrumentally upon appropriate dilution, such as Atomic Absorption Spectrophotometry (AAS), Inductively Coupled Plasma-Atomic Emission Spectrometry (ICP-AES), and Inductively Coupled Plasma-Mass Spectrometry (ICP-MS) to determine the mass of accumulated metal(loid)s using Equation 1.3.

$$M = \frac{C_e(V_g + V_e)}{f_e}$$
 Equation 1.3

where  $C_e$  is measured concentration in the eluent;  $V_g$  is the volume of gel in the resin gel layer;  $V_e$  is the volume of added extraction solution; and  $f_e$  is extraction efficiency of extraction solution for the metal(loid). In practice,  $V_g$  is often neglected (Zhang & Davison 1995). The measured mass M can be used to calculate the flux through the known area of the exposed diffusive layer, A (cm<sup>2</sup>) after a given deployment time.

$$J = \frac{M}{At}$$
 Equation 1.4

where A is a known exposure area; t is deployment time. Combining and rearranging Equation 1.2 and 1.4 give Equation 1.5.

$$C_{DGT} = \frac{M\Delta g}{DtA}$$
 Equation 1.5

The bulk concentration of metal(loid)s in the original sample ( $C_{DGT}$ ) can be quantified from the measured mass of metal(loid)s in the resin layer, the thickness of the diffusive layer, ion diffusion coefficient, the time of immersion and exposure area. This feature of DGT makes it ideal for determination of labile chemical species in solution in situ.

#### **1.9.2.** Principles of DGT in soils and sediments

The principles for application of DGT in sediments and soils are different from those in solution. The interpretation of DGT measurements is not as easy in sediments and soils as in solutions. The DGT measurements of metal ions in solutions can be interpreted as concentrations because of the well-mixed conditions in solutions, while pore waters of sediments and soils are not well mixed, and so the concentrations of metal ions in the sediments and soil in the vicinity of the DGT assembly may be diminished (Figure 1.6).

Solute ions can be replenished from the solid phase to the solution in the layers of soils near to the device (Zhang et al. 1998b; Zhang et al. 1995; Zhang et al. 2001). If the mechanism of transport for solutes in pore water of sediments and soils is diffusion only, the depletion of metal ions in the pore water adjacent to the device becomes continuously larger with time. Therefore, the flux of solute ions to the DGT device, which is induced by the concentration gradients through the gel layer, continuously decreases with time. Harper et al. (1998) reported that in the case of no resupply of solutes to the pore water, after the 24-hour deployment the interfacial pore water concentration between the soils or sediments and DGT device ( $C_a$ ) is about 0.06  $C_{pw}$ , where  $C_{pw}$  is the labile pore water concentration in the bulk solution. In practice,  $C_a$  is often much greater than this value (Zhang et al. 1998b), meaning that there is a significant resupply of solutes. This is because the depletion of solute concentration in pore waters can mobilise solutes from the solid phase.

If the resupply from the solid phase is rapid and unchanged during the deployment time, the principles developed for solutions can be applied for deployments in sediments and soils, using  $C_a$ , the interfacial pore water concentration between the DGT device and sediments or soils, instead of the bulk solute concentration. Once the steady state is achieved throughout the deployment,  $C_a$  can be calculated from the mass obtained from DGT. Therefore, the concentration measured by DGT,  $C_{DGT}$ , is corresponding to the pore water concentration adjacent to the DGT device,  $C_a$ , as shown in Equation 1.6.

$$C_{DGT} = C_a = \frac{M\Delta g}{DtA}$$
 Equation 1.6



**Figure 1.6** Schematic representation of a cross section through a DGT device in contact with sediments or soils. Pseudo steady-state concentration gradients are demonstrated for three cases: (a) unsustained, (b) sustained and (c) partially sustained case. Ca is the interfacial pore water concentration between the soils or sediments and DGT device (taken from Harper et al., 1999).

However, in practice the steady state is not constant during the deployment.  $C_a$  will change with the deployment time, and  $C_{DGT}$  is a time-averaged value of  $C_a$ . If the flux of solute ions to the resin layer at any given deployment time is governed by  $C_a$ ,  $C_{DGT}$  can be rewritten:

$$C_{DGT} = \frac{1}{t} \int_{0}^{t} C_{a}(t_{i}) dt$$
 Equation 1.7

where  $C_a(t_i)$  is  $C_a$  defined as a function of deployment time and t is the deployment time. In practice,  $C_{DGT}$  will rise rapidly at the beginning time of deployment thanks to the establishment of the steady state and then decrease gradually after several hours due to the depletion of resupply of solutes from the solid phase. The initial increase in  $C_{DGT}$  was observed to be negligible after 1-hour deployment (Zhang et al. 1995).

### Solid phase resupply

Generally, the DGT depletes pore water concentrations adjacent to the device. Therefore, the mean interfacial concentration will be less than the concentration of labile species in the bulk pore water,  $C_{pw}$ . The extent of this underestimation can be interpreted as the ratio, R,

of DGT determined concentration to the pore water concentration measured by other techniques (Harper et al. 1999; Harper et al. 1998).

$$R = \frac{C_{DGT}}{C_{pw}}$$
 Equation 1.8

R varies between 0 and 1. R could reach the highest value if there was an infinite and rapid resupply of solutes from the solid phase. In practice, the rate of resupply is limited by the rate of the release of solutes from the solid phase, leading to an R value < 1. In sediments and soils, the higher the value of R, the faster the rate of the release of solutes, and the larger the capacity for remobilisation of solutes from the solid phase (Harper et al. 1999). Harper et al. (1999; 2000) stated that the response of DGT to deployment in sediments and soils will belong to one of three scenarios described below and demonstrated in Figure 1.6.

(a) Unsustained case ( $R = R_{diff}$ ), where  $R_{diff}$  is the ratio obtained when solutes reach the resin layer through diffusion only. There are no resupply of labile species to the pore water and no steady state. This means that there are no metal ions in the exchangeable fraction of the sediments and soils, or the rate of desorption of metal ions from the solid phase is slow. Therefore, the DGT device is replenished only by the diffusion of dissolved species in the pore waters, resulting in continuous depletion. Characteristics of sediments and soils, the design of the DGT device, and the deployment time decide the value of  $R_{diff}$ . For a typical sampler deployed in high porosity sediments for 24 hours,  $R_{diff} = 0.1$  ( $C_{DGT} = 0.1C$ ) (Harper et al. 1998).

(b) Sustained case (0.95 < R < 1). Labile species removed from the solution by the DGT sink are rapidly resupplied from the solid phase. This is due to the fast rates of solute mixing in the sediments and soils, or the large capacity of the release of solutes from the solid phase, and that the rate of resupply from the solid phase is faster than the rate of removal by the DGT assembly. Well-mixed pore water only happens in surficial estuarine and sediments. The concentration measured by DGT can be considered as the concentration of labile metal species in the pore waters.

(c) Partially sustained case ( $R_{diff} < R < 0.95$ ). There is a certain level of resupply of the solutes from the solid phase to the solution because of  $R > R_{diff}$ . However, it is insufficient

to maintain the initial bulk concentrations of pore water and to prevent some depletion of concentrations of pore water between DGT device and sediments or soils.

# Kinetic of metal exchange and release

The response of DGT devices and thus the ratio, R, depends on a variety of factors including sampler design, diffusion coefficients, and the kinetics and capacity of solid phase resupply. If methods and sampler designs are standardised in use, R can be considered as a measurement of the capacity of the solid phase to resupply solutes to the pore water and the kinetics of this process.

Some studies were conducted to interpret the R value. Zhang et al. (1998b) described

$$R = K_d K_1$$
 Equation 1.9

where  $K_d$  is a distribution coefficient characterising the partitioning of labile metal species between solid phase and pore water and  $K_1$  is the rate constant of the resupply of solutes from solid phase to pore water. Based on the Equation 1.9, the concentration of labile species in solution is fully sustained when the dissociation rate constant ( $K_1$ ) and  $K_d$  are large (proportion of them in solution being high).

Harper et al. (1998; 2000) developed the DIFS (DGT Induced Fluxes in Sediments or soils) model (Figure 1.7), based on a model of the exchange of solutes between the solid phase and pore water to interpret the value of R. The exchange between a dissolved phase (pore water) and a solid phase (sorbed one) is assumed to be a first order reversible process, described by Equation 1.10.

$$C_d \xrightarrow{k_1} C_s$$
 Equation 1.10

where  $C_d$  and  $C_s$  are the concentrations of solutes in pore water and in sorbed phase, respectively;  $k_1$  and  $k_{-1}$  are the rate constants for sorption and desorption, respectively. At equilibrium, the ratio of sorbed to dissolved concentration is defined as the distribution coefficient,  $K_d$ . The relationship between  $k_1$ ,  $k_{-1}$ , and  $K_d$  is shown by Equation 1.11, where  $P_c$  is the particle concentration.

$$K_d = \frac{C_s}{C_d} = \frac{1}{P_c} \frac{k_1}{k_{-1}}$$
 Equation 1.11



**Figure 1.7** Basic concepts of solute transport and reaction in DIFS model (modified from Harper et al. 2000).

When DGT devices are deployed into soils and sediments, pore water concentrations adjacent to the devices become gradually depleted due to the accumulation of solutes onto the binding resin. In response to this perturbation, solutes desorb from the solid phase to replenish the pore water concentrations to re-establish the equilibrium. The response time  $T_c$  required to resupply solutes pore water concentrations from the solid phase is referred to the reciprocal of the sum of  $k_1$  and  $k_{-1}$  as shown in Equation 1.12.

$$T_c = \frac{1}{k_1 + k_{-1}}$$
 Equation 1.12

The ability of the solid phase to resupply solutes to solution depends on the kinetics of sorption and desorption processes and the available reservoir of solutes sorbed to the solid phase. The DIFS model predicts a relationship between R,  $K_d$ , and the response time  $T_c$ . The R value is quantitatively related to  $K_d$  and the response time  $T_c$  of the (de)sorption process. If R and  $K_d$  are known,  $T_c$  can be obtained from equation 1.13, which is derived from plotting R against  $T_c$  for different values of  $K_d$  (c and d are parameters of the fitted sigmoidal curve).

$$T_c = c \left(\frac{1-R}{R-d}\right)^2$$
 Equation 1.13

After the 24-hour deployment the one dimensional model (the domain comprising the horizontal transport) yields the ratio of DGT estimated to initial pore water concentration of 0.06 while the two dimensional approach (the domain comprising the vertical and horizontal diffusion) obtains a ratio of 0.1. If there is a rapid resupply from the solid phase and the depletion of pore water concentrations within 2 mm of the filter, the results from two models are not different (Harper et al. 1998).

# **1.9.3.** DGT-determined effective concentration $(C_E)$ as a predictor of metal bioavailability to plants

When DGT is deployed into a soil or sediment, it removes metals from solution phase by accumulating them on the binding resin. In response to the removal, the concentration of metals in soil solution immediately adjacent to the DGT device is lowered. Local depletion of metals in soil solution induces the release and resupply of labile metals from complexes in soil solution and from the exchangeable metal fraction of the solid phase. Like DGT, plants accumulate metals by removing them from soil, which locally lowers metal concentration in the soil solution, inducing the resupply of metals from labile species in soil solution and the labile metal pool in the solid phase (Zhang & Davison 2006; Zhang et al. 2001). The similarities between metal uptake by DGT and plant uptake under diffusioncontrolled conditions have been discussed and modelled (Degryse et al. 2009; Lehto et al. 2006a, 2006b). The soil perturbation is similar if the rate of metal removal by DGT and the plant is similar, enhancing the ability of DGT to mimic plant uptake of metals. Models of metal uptake by DGT and plants showed that fluxes to DGT and plants are generally similar for typical values of diffusive layer thickness used in DGT devices (Lehto et al. 2006a, 2006b). Several studies have shown good correlations between C<sub>DGT</sub> and concentrations of metals in plant tissues (Degryse et al. 2009; Zhang & Davison 2015). For DGT and plant uptake limited by diffusion, there is diffusional supply of metals from soil solution and additional resupply of labile metals released from the solid phase. This is because C<sub>DGT</sub>, the time-averaged concentration at the soil and DGT interface, is less than

 $C_{sol}$ , dissolved metal concentration in soil solution, which can be confusing as the  $C_{DGT}$  has the additional resupply of metals from the solid phase (Lehto 2016). To overcome this conceptual difficulty and to allow further interpretation on DGT measurements, the DGTmeasured concentration can be converted to an effective concentration ( $C_E$ ) which was first introduced by Zhang et al. (2001) to predict the bioavailability of metals to plants.  $C_E$  is defined as the ratio of  $C_{DGT}$  to  $R_{diff}$ , as shown in Equation 1.14.

$$C_E = \frac{C_{DGT}}{R_{diff}}$$
 Equation 1.14

 $C_E$  represents the metal concentration which is available and exchangeable from both soil solution and solid phase.  $R_{diff}$  is the ratio of the mean interfacial concentration for the diffusion-only case (no resupply from the solid phase) to the concentration in soil solution.  $R_{diff}$  may be calculated using the numerical DIFS model (Harper et al. 2000; Harper et al. 1998; Sochaczewski et al. 2007). DIFS is used to simulate the deployment in a diffusion only scenario by accurately specifying the diffusion characteristics of soil, but assuming unsustained case of soil (e.g. very high response time  $T_c$  and very low available labile reservoir  $K_d$ ). Therefore, the effective concentration of  $C_E$  can be obtained directly from the DGT measurements.

Input parameters of particle concentration ( $P_c$ , g/ml), soil porosity ( $\phi$ ) and diffusion coefficient in soil ( $D_s$ ) were calculated with Equations 1.15, 1.16, 1.17 (Boudreau 1996).

$$P_c = m/V$$
 Equation 1.15

$$\varphi = d_p/(P_c+d_p)$$
 Equation 1.16

$$D_s = D_o/(1 - \ln(\phi^2))$$
 Equation 1.17

Where m is the total mass of soil particles, V is the soil solution volume,  $d_p$  is the density of soil particles, and  $D_o$  is the diffusion coefficient of each metals in water.

 $C_E$  is described as hypothetical constant dissolved concentration at the DGT interface needed to maintain the uptake by DGT device during the deployment, without the supply from the solid phase (Zhang & Davison 2015). The ratio between  $C_{DGT}$  and  $C_E$  ( $R_{diff}$ ) depends on soil density, porosity, diffusion coefficient, and deployment time, and is usually about 0.1 for a 24-hour deployment.  $C_E$  is usually around 10 times higher than  $C_{DGT}$ . Hence,  $C_{DGT}$  is generally very strongly correlated with  $C_E$ . In some datasets covering soils with various porosities, there may be less strong correlation between  $C_{DGT}$  and  $C_E$  due to variation in  $R_{diff}$  values. Some studies assessing the correlation between plant response with  $C_{DGT}$  and  $C_E$  did not observe any advantage in using  $C_E$  (Luo et al. 2014; Mason et al. 2010). Since the interpretation of  $C_{DGT}$  is easier and does not require additional calculation of  $R_{diff}$  by DIFS model simulation,  $C_{DGT}$  has still been used in predicting metal uptake by plants (Degryse & Smolders 2016).

# **1.9.4.** Experimental evidence of DGT applications in measuring As and Sb bioavailability in soils to plants

With regard to assessment of the lability and bioavailability of elements, DGT is useful in estimating the labile fraction of elements in soil, and has been successfully employed to measure bioavailable metals to plants (Degryse et al. 2009). The dynamical DGT approach integrates a variety of soil properties influencing metal adsorption/release (Williams et al. 2011; Zhang et al. 2001), and can mimic metal uptake by plants where depletion of metal concentrations adjacent to plant roots induces the release and resupply of metals from the solid phase (Zhang & Davison 2006; Zhang et al. 2001). There is extensive literature relating soil and sediment DGT measurements to metal bioavailability (Zhang & Davison 2015). Most studies have focused on the bioavailability of divalent cationic metals (e.g. Cu, Zn, Cd, Pb, Ni) in soils to plants using standard Chelex-DGT devices, but there are a few studies that have compared DGT measurements with biouptake of As and none was for Sb. The first DGT use for sampling As in soil was conducted by Fitz et al. (2003) who successfully applied Ferrihydrite-DGT to investigate the resupply of labile As pools after phytoextraction and to monitor the efficiency of As phytoremoval. Cattani et al. (2009) assessed the availability of As to roots of As hyperaccumulators and illustrated the sensitivity of Ferrihydrite-DGT in monitoring root-induced changes of As in soils. The DGT technique (Ferrihydrite being the binding resin) was also considered as a sensitive tool for evaluating As mobility changes in rhizophere of two plants with different abilities in accumulating As (Senila et al. 2013) and changes in As availability during aging (Liang et al. 2014). Mojsilovic et al. (2011) applied Ferrihydrite-DGT measurements of As and P

to model wheat (*T. aestivum*) arsenic toxicity using a dataset of soils contaminated with As and reported that the DGT-derived As/P ratio was demonstrated as a promising predictor of As phytotoxicity. DGT was also used to predict As uptake by rice (*Oryza sativa*) collected in the fields (Williams et al. 2011) and edible rape grown in 43 soils (Wang et al. 2014). However, DGT has not previously been used to measure labile Sb in soils and predicting As and Sb uptake by edible vegetables.

Another advanced application of DGT in soil is evaluation of the distribution and dynamics of labile metal(loids) in the rhizophere of plants by high-resolution chemical imaging using localised DGT sampling combined with spatially resolved analysis by laser ablation - inductively coupled plasma - mass spectrometry (LA-ICP-MS), which is extensively reviewed by Santner et al. (2015). Santner et al. (2012) applied for the first time DGT coupled with LA-ICP-MS for chemical imaging of labile phosphorus in the rhizophere of two *Brassica napus* cultivars, advancing the understanding of P dynamics around the plant roots and plant P acquisition. In a recent study applying DGT-planar optode sandwich sensor, Williams et al. (2014) observed that elevated levels of As, Fe(II), and Pb exist at the boundary of the aerobic rhizophere region during the formation of iron plaque on the flooded rice roots. The release of As and Fe(II) was expected to co-occur since As was adsorbed on negatively charged surfaces through Fe(II)-bridges. pH decreases and Fe(II) mobilisation was co-localised, which can be because of slowed Fe(II) oxidation and/or Fe(II) desorption from freshly from iron oxide. These findings provide a new perspective in the mobilisation and accumulation of As in submerged rice.

# 1.10. Aim and objectives

The aim of this study is to investigate the biogeochemical behaviour of Sb compared to that of As in contaminated soils in relation to their uptake by importantly agricultural plants identified by the diffusive gradients in thin films (DGT) technique in comparison to selective extractions. The specific objectives of this research are:

• To investigate the partitioning of As and Sb in various soil binding phases and the performance of DGT in measuring labile As and Sb in different types of contaminated soils. DGT performance will be compared with conventional extraction techniques including soil solution and sequential extraction.
- To evaluate the performance of the DGT technique in measuring labile As and Sb in historically co-contaminated soils to root vegetables, two cultivars of radish (*Raphanus sativus*) in pot experiments and comparisons between predictions of As and Sb accumulation by radishes based on DGT measurements and current techniques (soil solution, sequential extraction) will be provided.
- To assess the performance of the DGT technique in measuring the lability of As and Sb in historically co-contaminated soils with varying physico-chemical soil properties and recently amended soils and predicting their accumulation by leafy vegetables, water spinach (*Ipomoea aquatica*) in pot experiments, in comparison to selective extractions. Differences in lability and bioaccumulation of As and Sb among soils will be evaluated in relation to the incorporation of bioavailable measurements into soil risk assessments.
- To investigate the fractionation and lability of As and Sb in single and multicontaminant amended soils and their competitive interactions when they co-exist in soils in relation to their uptake by water spinach (*Ipomoea aquatica*) in pot experiments, identified by sequential extraction and DGT.
- To investigate the mobilisation and speciation of As and Sb in soils induced by redox changes using multiple binding resins in DGT. Changes in As and Sb biouptake by water spinach (*Ipomoea aquatica*) cultivated in contaminated soils subjected to different redox environmental conditions will be evaluated and linked to changes in As and Sb mobility in soils predicted by DGTs. The speciation transformation of As in *I. aquatica* determined by high pressure liquid chromatography inductively coupled plasma mass spectrometry (HPLC-ICP-MS) will be investigated and linked to As speciation in soils measured by DGTs.

# **Chapter 2. General methods**

#### 2.1. General washing methods

Deionised water (18.2 M $\Omega$ /cm; Milli-Q Academic Water System; Millipore) was used to prepare all solutions. All chemicals were analytical reagent grade or higher. All plasticware and glassware were washed by soaking in 10% (v/v) HNO<sub>3</sub> (BDH, AnalaR) for at least 24 h, followed by multiple rinses with deionised water.

#### 2.2. Test soils

#### 2.2.1. Historically contaminated soils

A soil highly contaminated with As and Sb was collected from a decommissioned antimony processing facility in Urunga, New South Wales (NSW), Australia. A second soil contaminated with only As was collected adjacent to a cattle dip site in Queensland (QLD), Australia. Uncontaminated soils were collected in Wollongong, NSW. All soils were air-dried and then crushed to pass through a 2-mm sieve. To establish an As and Sb concentration gradient of test soils, varying ratios of contaminated and uncontaminated soils were thoroughly mixed with a cement mixer to ensure homogeneity.

#### 2.2.2. Recently contaminated soils

Clean soils were spiked to replicate recently contaminated soils. Each portion of a clean soil was amended with solutions of various concentrations of Na<sub>2</sub>HAsO<sub>4</sub>.7H<sub>2</sub>O, KSb(OH)<sub>6</sub>, and their mixture to achieve three series of soils contaminated with As only, Sb only, and both As and Sb, respectively. Following the addition of As/Sb solutions into soils, the spiked soils were homogeneously mixed and turned over vigorously several times. Moisture content of soils was retained at 60% of maximum water holding capacity (MWHC) after spiking. The artificially contaminated soils were left to equilibrate for eight weeks, during which time it was regularly stirred. After the equilibration period, all soils were air-dried, crushed using a mortar and pestle, and mixed homogeneously for the bioassays. A small proportion of soil samples was further crushed to pass a 2-mm sieve and stored in clean plastic bags for further analyses.

#### 2.3. Test plants

This research investigated a range of plant species to evaluate the response of a variety of edible plant species and cultivars exposed to As and Sb in various contaminated soils under different conditions and the performance of different techniques in predicting As and Sb uptake by plants with different root structures and physiology, which is essential for appropriate soil risk assessments and ensures food safety.

Radish (*Raphanus sativus* L.) is widely grown in Europe, Asia, Africa, South America, and the United States. Radish is the easiest, quickest, and most commercial crop to grow. It grows best at the temperature at  $\sim 15 - 20$  °C. Radish is cultivated on all types of soils from mud to sandy. Light-textured, fertile sandy loams produce the highest yield and best quality of roots. There are many radish cultivars, in which the two most popular ones with different root structures such as cherry belle and icicle were chosen for our studies. The cheery belle is a globe-shaped and bright red-coloured variety, while the icicle is a long, cylindrical-shaped and clear white-coloured variety. Both of them have short shoots and are mature in  $\sim 4 - 5$  weeks after emergence (Hartmann et al. 1981).

Water spinach (*Ipomoea aquatica* L.) is a vascular plant and widely grown and distributed in Asia. It is tropical and subtropical plant and grow well at the temperature of >24 °C. *I. aquatica* can be cultivated in land or in semiaquatic systems. *I. aquatica* is easily cultivated with little care and develops fast. It has fibrous roots, hollow stems, and ~15 cm long and 2 cm wide arrowhead shaped leaves. The soft stems and leafy portions are edible and rich in important vitamins (A, C), iron, and nutrients, which is popular for daily food (Göthberg et al. 2002).

The chosen plants have important nutritional values; thus, the knowledge of the plant responses to As and Sb in contaminated soils and useful techniques for predicting As and Sb accumulation by these plants is important for the risk management action to ensure human health and the development of sustainable agriculture.

#### **2.4.** Pot experiments

#### 2.4.1. Bioassays with radishes (*Raphanus sativus* L.)

Appropriate amounts of deionized water were added into test soils to have moisture content at ~60% of the water holding capacity (WHC). Nutrients were also added at the rate of 0.1 g N (as urea), 0.2 g P (as KH<sub>2</sub>PO<sub>4</sub>), and 0.3 g K (as K<sub>2</sub>SO<sub>4</sub>) kg<sup>-1</sup> soil to ensure that plant growth was not affected by nutrient deficiencies. Soils were allowed to settle and equilibrate in a greenhouse for two weeks prior to the commencement of bioassays.

Following soil equilibration, prepared soils were added into plastic pots which were lined with high density polyethylene (HDPE) bags. 1.5 and 2.0 kg of soils were added into plastic pots ( $10 \times 15$  cm) and poly(vinyl chloride) (PVC) cylindrical pots ( $25 \times 10$  cm) for the cultivation of cherry belle radish (chapter 3) and white icicle radish (chapter 4), respectively. Four replicates were performed for each treatment. Seeds were directly sown into soil pots (Day 0), and after emergence, the seedlings were thinned to three per pot. The pots were randomly arranged in the growth chambers set at 14:10 h day:night cycle, 20:16 °C day:night temperatures, 75% relative humidity, and a minimum photon flux of 500 µmol m<sup>-2</sup> s<sup>-1</sup>. During the bioassay, pots were weighed daily and deionised water was used to maintain the soil moisture at ~60% WHC. After 35 days, plants were harvested, rinsed thoroughly in deionised water to remove adhering soil. Plants were sectioned into shoots, peel (including tap root), and flesh using a stainless steel blade and vegetable peeler. Wet tissue mass (g) was recoreded.

## 2.4.2. Bioassays with water spinach (Ipomoea aquatica L.)

Appropriate amounts of deionized water were added into test soils (historically contaminated and freshly contaminated soils) to have moisture content at ~60% of the water holding capacity (WHC). Nutrients were also added at the rate of 0.15 g N (as urea), 0.1 g P (as  $NH_4H_2PO_4$ ), and 0.04 g K (as  $K_2SO_4$ ) kg<sup>-1</sup> soil to ensure that plant growth was not affected by nutrient deficiencies. Soils were allowed to settle and equilibrate in a greenhouse for two weeks prior to the commencement of bioassays.

Following soil equilibration, 1.5 kg of prepared soils was added into plastic pots ( $10 \times 15$  cm) which were lined with high density polyethylene (HDPE) bags. Three replicates were

performed for each treatment. Seeds were directly sown into soil pots (Day 0), and after emergence, the seedlings were thinned to six per pot. The pots were randomly arranged in the growth chambers set at 12:12 h day:night cycle, 30:25 °C day:night temperatures, 75% relative humidity, and a minimum photon flux of 500  $\mu$ mol m<sup>-2</sup> s<sup>-1</sup>. During the bioassay, pots were weighed daily and deionised water was used to maintain the soil moisture at ~60% WHC. After 35 days, plants were harvested, rinsed thoroughly in deionised water to remove adhering soil, and sectioned into roots and shoots using a stainless steel blade. Wet tissue mass (g) was recorded.

#### 2.5. Soil characteristic measurements

#### 2.5.1. Maximum water holding capacity (MWHC)

Maximum water holding capacity was determined by placing 20 - 30 g of air-dried 2-mmsieved soil in a funnel having a Whatman filter paper. The funnel was held in a container of water. Water was added to saturate the soil and then allowed to stand for 2 h. After that, the funnel was removed from the container and allowed to drain overnight. A blank measurement with only the filter paper was also performed simultaneously using the same procedure to get the weight of the wet filter paper. The maximum water holding capacity of soils was calculated using Equation 2.1.

MWHC (%) = (
$$W_{wet soil + wet filter paper} - W_{wet filter paper} - W_{dry soil}) \times 100\%/W_{dry soil}$$

Equation 2.1

where  $W_{wet soil}$  is the mass of the wet soil and filter paper after drainage,  $W_{wet filter paper}$  is the mass of the wet filter paper, and  $W_{dry soil}$  is the mass of dry soil.

#### 2.5.2. Moisture content

The measurement of the moisture content of soil was made by the gravimetric method based on quantification of the mass of water lost after drying at 105 °C (Buurman et al. 1996). The method involves weighing a proportion of homogenous sample of wet soil into a known weight crucible, placing the crucible in a drying oven at 105 °C overnight, then allowing it to cool to room temperature in a desiccator and reweighing. The difference between soil wet weight (WW) and soil dry weight (DW) was used to calculate the

percentage moisture content ([WW-DW]  $\times$  100/WW) which was then applied to transform values of chemical analyses on wet soil into dry weight.

#### 2.5.3. Soil particle size analysis

Soil particle size distribution was determined by using a laser diffraction particle size analyser (Mastersizer 2000; Malvern Instruments Ltd, Worcestershire, United Kingdom). Soil was added into water and diluted to give an interference of 10 - 20% of signal detection. The percentage of particles was calculated by scattering of 4 mW, 632.8 nm red laser; and 0.3 mW, 470 nm blue laser in a convergent beam setting.

#### 2.5.4. Soil pH

Measurement of soil pH involves weighing 5 g of air-dried 2-mm-sieved soil into a 50 ml centrifuge tube, adding 25 ml of deionised water, shaking for 2 h, and then allowing the tube stand for 1 h. When the suspension settled on the bottom, a pH meter was used to measure pH of the soil in the supernatant (Buurman et al. 1996). Caution was taken to avoid the pH electrode coming in to contact with the settled particles, leading to errors of pH readings. The pH electrode was calibrated using standard buffers of pH 4 and pH 7 before measuring. The electrode was rinsed well with deionised water between measurements and stored in the preservation solution, 3 M KCl, when it was not used.

#### 2.5.5. Organic matter (OM)

Organic matter of soils was determined by 'loss on ignition', the weight loss between 105 °C for drying and 550 °C for ashing samples (Heiri et al. 2001). The procedure involves weighing a proportion of solid sample (approximately 1 g) into an ashing vessel, placing the vessel containing sample in the drying oven at 105 °C overnight. The vessel was then removed from the drying oven, allowed to cool at room temperature, and weighed accurately. The vessel was progressively ashed at 550 °C for 24 h in the muffle furnace, followed by cooling in the dry atmosphere and recording the accurate weight. The percentage of organic matter in soil was calculated by the formula:

$$%OM = [(W_{105} - W_{550}) \times 100]/W_{105}$$
 Equation 2.2

where  $W_{105}$  is mass of soil at 105 °C and  $W_{550}$  is mass of soil at 550 °C. The sensitivity of this method for determining organic matter content approximates 0.2-0.5% OM.

#### 2.5.6. Total Kjeldahl nitrogen

Total Kjeldahl nitrogen was determined by the digestion of ~1 g dried soil in 12 mL of concentrated  $H_2SO_4$  with a catalyst mixture of CuSO<sub>4</sub>, Na<sub>2</sub>SO<sub>4</sub>, and Se, followed by the distillation of ammonia by addition of an excess of NaOH solution. The ammonia was trapped in a boric acid solution. The formed ammonium was then titrated with sulphuric acid standardised with sodium carbonate solution (Rayment & Lyons 2011).

#### 2.5.7. Extractable phosphorous

Extractable phosphorous (P-Olsen) was determined by the extraction of ~1 g dried soil with 20 mL of 0.5 M sodium bicarbonate solution at a pH of 8.5 for 30 min. The supernatant solution was filtered through a Whatman filter paper. The extracted phosphate was then analysed colourimetrically with ammonium molybdate (Rayment & Lyons 2011).

#### 2.5.8. Amorphous Fe and Al

Amorphous Fe and Al in soils were determined by the extraction of 1 g soil with 50 mL of 0.2 M ammonium oxalate-oxalic acid solution at pH of  $3.00 \pm 0.05$  for 4-h shaking in the dark at room temperature. The supernatant was centrifuged at  $2200 \times$  g for 15 minutes, filtered, and acidified to a final concentration of 2% HNO<sub>3</sub> for analyses of Fe and Al using inductively coupled plasma-optical emission spectroscopy (ICP-OES) (Buurman et al. 1996; Rayment & Lyons 2011).

#### 2.5.9. Total free Fe and Al

Total free Fe and Al in soils were determined by the extraction of 1 g soil with 1 g sodium dithionite and 50 mL of sodium citrate solution (22%) for 16-h shaking. The extract was then diluted 2 times with deionised water and 5 drops of superfloc solution (0.2%) was added and shaken vigorously. The supernatant was centrifuged at  $2200 \times g$  for 15 minutes, filtered, and acidified to a final concentration of 2% HNO<sub>3</sub> for analyses of Fe and Al using inductively coupled plasma-optical emission spectroscopy (ICP-OES) (Buurman et al. 1996; Rayment & Lyons 2011).

#### 2.6. Soil As and Sb concentrations

#### 2.6.1. Sequential extraction procedure

A five-step sequential extraction procedure developed by Wenzel et al. (2001) was used for determination of As and Sb in various soil binding phases (Table 2.1). After each extraction step, the samples were centrifuged at  $2200 \times g$  for 15 minutes, filtered, and acidified to a final concentration of 2% HNO<sub>3</sub> for analyses of As and Sb in different fractions using ICP-MS (section 2.9).

#### 2.6.2. Dissolved As and Sb concentrations in soil solution

Soil solution was extracted from a soil paste after DGT deployment by centrifuging at 2200  $\times$  g for 15 min at room temperature. Details of soil to water were as per section 2.7.5. The resulting soil solutions were taken up by syringe, filtered through a 13-mm diameter, 0.45- $\mu$ m polysulphone filter and then acidified to achieve a final concentration of 2% HNO<sub>3</sub>. The concentrations of As and Sb in the filtered soil solution were measured by ICP-MS (section 2.9).

#### 2.6.3. Total soil As and Sb concentrations

Total As and Sb in soil samples were analysed using a microwave assisted aqua regia extraction procedure, as per US EPA method 3051A. Finely ground dried soil (<0.5 g) was combined with 12 mL of a mixture of HNO<sub>3</sub>:HCl of 3:1 (v/v) (65% Suprapur®, Merck) in a TFM closed digestion vessel, left overnight, vented to allow gas to escape, then microwave heated (MARS Xpress, CEM). After cooling to room temperature, digests were filtered (<0.45  $\mu$ m) and diluted to 2% acid with deionised water for analyses by ICP-MS (section 2.9).

Fraction	Target phase	Extractant	Extraction conditions	Soil solution ratio	Wash step
1	Non- specifically sorbed	0.05 M (NH <sub>4</sub> ) <sub>2</sub> SO <sub>4</sub>	4 h shaking	1:25	
2	Specifically sorbed	0.05 M (NH <sub>4</sub> )H <sub>2</sub> PO <sub>4</sub>	16 h shaking	1:25	
3	Bound to amorphous Fe oxides	0.2 M (NH <sub>4</sub> ) <sub>2</sub> C <sub>2</sub> O <sub>4</sub> /H <sub>2</sub> C <sub>2</sub> O <sub>4</sub> , pH 3.25	4 h shaking in the dark	1:25	0.2 M NH <sub>4</sub> - oxalate buffer; pH 3.25, (1:12.5), 10 min shaking in the dark
4	Bound to crystalline Fe oxides	0.2 M (NH <sub>4</sub> ) <sub>2</sub> C <sub>2</sub> O <sub>4</sub> /H <sub>2</sub> C <sub>2</sub> O <sub>4</sub> + 0.1 M C <sub>6</sub> H <sub>8</sub> O <sub>6</sub> , pH 3.25	30 minutes in water basin at 96 $\pm$ 3 °C in the light	1:25	0.2 M NH <sub>4</sub> - oxalate buffer; pH 3.25, (1:12.5), 10 min shaking in the dark
5	Residual	Differences in sum of four fractions and the total concentrations			

**Table 2.1** Sequential extraction procedure applied for determination of the partitioning of As and Sb in soils.

# 2.7. DGT measurements

#### 2.7.1. DGT devices and assembly

DGT piston devices (an exposure window of  $3.14 \text{ cm}^2$ ), sediment probes ( $24 \text{ cm} \times 4 \text{ cm} \times 0.5 \text{ cm}$ , an open window of  $1.8 \text{ cm} \times 15 \text{ cm}$ ), and agarose-based cross-linker were purchased from DGT Research Ltd (UK) (<u>http://www.dgtresearch.com</u>). Ammonium persulphate and tetramethylethylenediamine (TEMED) was purchased from Sigma-Aldrich. All equipment for DGT gel synthesis and handling (e.g. glass plates, plastic spacers, plastic containers, tweezers, chopping board, DGT pistons) was detergent-washed, rinsed with deionised water, and soaked in 10% HNO<sub>3</sub> for 24 h. An additional acid-wash in a second 10% HNO<sub>3</sub> for 24 h was applied to ensure the complete cleaning procedure. For washing of the filter membrane, 5% HNO<sub>3</sub> soaking for 8 h was applied. Gel synthesis and handling, DGT device manipulations were performed in an AURA SD4 Laminar Flow Cabinet

following standard procedures of DGT Research. For assembling DGT pistons (3.14 cm<sup>2</sup> sampling window), a titanium dioxide binding gel disc (0.4 mm thickness) was placed on base, followed by a polyacrylamide diffusive gel disc (0.8 mm thickness), and topped by a 0.45  $\mu$ m cellulose nitrate filter membrane (~0.1 mm thickness), then clamped with a window (Zhang & Davison 1995; Zhang et al. 1995). For assembling DGT sediment probes, a titanium dioxide/mercapto-silica binding gel trip was placed onto the base plate followed by a diffusive gel trip and polysulphone membrane (0.45  $\mu$ m pore-size), then covered by the window plate. Before deployment, DGT pistons/probes were conditioned in 0.01 M NaCl and degassed by purging nitrogen gas overnight. A sheet of titanium dioxide/mercapto-silica gel was added to the conditioning solution in order to minimise contamination.

#### 2.7.2. Gel solution

Standard gel stock solution used in the synthesis of binding and diffusive gels was prepared as described by Zhang & Davison (1995), Zhang et al. (1998a) by combining 47.5 mL of deionised with 37.5 mL of 40% acrylamide solution (BDH, Poole). The solution was mixed thoroughly before adding 15 g of an agarose derived cross-linker (2%, DGT Research Ltd, Lancaster, UK). The gel solution was further mixed and stored at 4 °C in the refrigerator for 3 months.

#### 2.7.3. Diffusive gel

To make diffusive gels, 10 ml of gel stock solution was transferred to an acid washed tube. A further 70  $\mu$ L of fresh ammonium persulphate solution (10%) was added and well mixed, followed by an addition of 25  $\mu$ L of TEMED (*N*,*N*,*N'N'*-Tetramethylethylenediamine, 99%). The solution was vigorously stirred for 3 – 4 min. The solution was cast between two acid-washed glass plates separated by a plastic spacer round three edges having an approximate thickness (0.5 mm) and placed in an oven at 42 – 45 °C for 1 h. After the removal of gels from the plates, the gels were hydrated in deionised water and changed water several times for 24 h to achieve a stable thickness. After hydration, the pH of the solution containing the gels should be around 6.5 – 7. Otherwise, more deionised water change was applied to flush out any impurities remained and reduce the pH. The thickness of the gels was ~0.8 mm when the gels were fully hydrated. The diffusive gels were stored

in deionised water until used (Davison et al. 2000; Davison & Zhang 1994; Zhang & Davison 1995).

#### 2.7.4. Binding gels

Titanium dioxide powder (Metsorb) was used as binding agent for As and Sb measurements. 1 g of dry titanium dioxide was added to 10 mL of gel stock solution. The solution was then sonicated for 5 min, and mixed well for 1 min. A further 60  $\mu$ L of fresh ammonium persulphate solution (10%) and 20  $\mu$ L of TEMED (99%) was added to this solution and stirred well. Thereafter the gel solution was immediately cast between two acid-washed glass plates separated by a plastic spacer of appropriate thickness (0.25 mm). The gels were set at 45 °C in an oven for 30 min, then removed from the glass plates, hydrated in deionised water with 3-4 times changing water during 24 h. The hydrated gels were then stored in deionised water at 4 °C in the refrigerator until used (Bennett et al. 2010).

3-mercaptopropyl-functionalised silica gel was used as binding agent for As(III) and Sb(III) measurements. To prepare mercapto-silica binding layers, bisacrylamide-cross-linked polyacrylamide was replaced the standard agarose-cross-linked polyacrylamide since it leaded to homogeneous distribution of mercapto-silica. In brief, 1 g of dry mercapto-silica was added to 10 mL of bisacrylamide-cross-linked polyacrylamide gel stock solution. 200  $\mu$ L of ammonium persulphate solution (10%) and 8  $\mu$ L of TEMED (99%) was added. The gel mixture was well stirred and cast between two acid-washed Perspex plates using the 0.5 mm thickness spacer. The binding gels have the thickness of 0.4 mm because the gels shrink slightly after hydration (Bennett et al. 2011).

#### 2.7.5. Procedure for deploying DGT in soils

Blank DGT pistons/probes were treated in the same way as the sample devices.

#### **Preparation of soil samples**

Subsamples of air-dried and 2-mm sieved soils (~80 g) were wetted to 100% maximum water holding capacity by adding the appropriate amount of deionised water. The soil and water were well mixed to have a smooth paste using an acid-washed plastic spatula. The paste was incubated in HDPE pots for 24 h at  $20 \pm 1$  °C. During the equilibration and

deployment, lids were loosely placed on pots to avoid evaporation and to allow air-flow into pots.

#### **Deployment of DGT pistons in soils**

DGT devices were carefully placed on the soil paste with gentle pressure to ensure complete contact between the filter membrane of the device and the soil surface. Each soil sample was performed in three replicates. DGT assemblies were deployed for 24 h at  $20 \pm 1$  °C. On retrieval, DGT devices were washed with deionised water to remove soil particles and then disassembled. The resin gel discs were removed from the DGT devices and placed in the clean vials. For DGT sediment probes, binding gel strips were sliced to the desired resolution using Teflon<sup>®</sup>-coated razor blades.

To elute elements, titanium dioxide and mercapto-silica gel disks/slices were immersed in 1 mL of solution with a final concentration of 1 M NaOH : 1 M  $H_2O_2$  for 24 h (Bennett et al. 2016b; Panther et al. 2013). The concentration of elements in the eluent was measured by ICP-MS (section 2.9).

#### 2.7.6. Calculation of DGT measured concentrations

The mass of elements in the resin gel (M) can be calculated using Equation 2.3.

$$M = C(V_{gel} + V_{eluent})/f_e$$
 Equation 2.3

where C is the concentration of elements in eluent, measured by ICP-MS, in ( $\mu$ g/L), V<sub>eluent</sub> is the volume of eluents used for elution of elements accumulated on gels, V<sub>gel</sub> is the volume of resin gel, and f<sub>e</sub> is the elution factor for each analyte. The time averaged concentration of elements at the interface of the soil and the DGT device can be obtained using Equation 2.4.

$$C_{DGT} = M\Delta g / (DAt)$$
 Equation 2.4

where  $\Delta g$  is the thickness of the diffusive gel (0.8 mm) plus the thickness of filter membrane (0.1 mm), D is the diffusion coefficient of elements in the diffusive gel (4.97 ×  $10^{-6}$  cm<sup>2</sup>s<sup>-1</sup> for As and  $5.42 \times 10^{-6}$  cm<sup>2</sup>s<sup>-1</sup> for Sb at 20 °C), t is the deployment time (s), and A is the exposed area of the gel disc (A = 3.14 cm<sup>2</sup>).

#### 2.8. Plant analysis

#### 2.8.1. Pre-digestion

After harvesting, plants were rinsed with tap water and then rinsed with deionised water thoroughly to remove soil particles, and excess of water was eliminated with tissue paper. Plant characteristics such as shoot length, root length, fresh weights were recorded. Plants were divided into shoots and roots with a stainless steel blade and dried in an oven at 70 °C for 48 h (Agbenin & Welp 2012; Álvarez-Ayuso et al. 2012; Niazi et al. 2011; Soriano-Disla et al. 2010; Wang et al. 2014). Weights of dried shoots and roots were recorded. The oven-dried plant tissues were ground with a mortar and pestle and stored in clean plastic zip-bags for analyses of total As and Sb concentrations in plants. For analysis of concentrations of As and Sb species in plants, the washed plants were freeze dried for 48 h at -50 °C and subsequently ground to fine powder under liquid nitrogen.

#### 2.8.2. Plant digestion

Finely ground plant sample (<0.5 g) was combined with 12 mL of a mixture of HNO<sub>3</sub>:H<sub>2</sub>O<sub>2</sub> 1:2 (v/v) (65 and 30%, respectively, Suprapur®, Merck) in a TFM closed digestion vessel, left overnight, vented to allow gas to escape, and then microwave heated (MARS Xpress, CEM). After cooling to room temperature, digests were filtered (<0.45  $\mu$ m) and diluted to 2% acid with deionised water for analyses by ICP-MS (section 2.9).

#### 2.9. Analytical method

Total As and Sb analyses were performed using an Octopole Reaction Cell Inductively Coupled Plasma-Mass Spectrometer (ORC ICP-MS; Agilent 7500ce) by measuring the signal at m/z 75 for As and m/z 121 for Sb. Rh and In (final concentrations of 10  $\mu$ g/L) were used as internal standards to account for instrument drift. Quality control standards prepared at 25  $\mu$ g/L were analysed every 15 samples.

For quality control purposes, reagent blanks, duplicate analyses, certified reference materials, and matrix spike recovery were used. A soil certified reference material (Montana II soil, SRM 2711a; NIST) returned recoveries of 80-118% for the studied analytes. A plant certified reference material (Tomato leaves, SRM 1573a; NIST) had typical recoveries of 80-120% of the studied analytes.

### 2.10. Data analysis

The statistical testing and analysis and the correlations were carried out using SPSS 21 (IBM Corp.). All statistical tests were performed at the  $\alpha = 0.05$  level, unless otherwise stated. Statistical significant differences in As and Sb labile fractions, the kinetic resupply of As and Sb in soils, As and Sb concentrations in plant tissues between treatments were identified using analysis of variance (ANOVA) with post hoc Tukey HSD comparisons where needed. Independent-sample T-test was used to evaluate any significant differences between As/Sb concentrations in roots and shoots, As/Sb bioaccumulation in two plant cultivars, As/Sb availability in individual and co-contaminated soils. Correlation analysis was performed to investigate the relationships between As and Sb labile fractions measured by different techniques and their concentrations in different plant tissues.

# Chapter 3. Assessing the uptake of arsenic and antimony from contaminated soil by radish (*Raphanus sativus*) using DGT and selective extractions

This chapter has been published.

;

Ngo, L.K., Pinch, B.M., Bennett, W.W., Teasdale, P.R., Jolley, D.F., 2016. Assessing the uptake of arsenic and antimony from contaminated soil by radish (*Raphanus sativus*) using DGT and selective extractions. *Environmental Pollution*. 216, 104-114. DOI: 10.1016/j.envpol.2016.05.027.

#### **3.1. Introduction**

Arsenic and antimony are highly toxic elements, producing mutagenic and carcinogenic effects in humans (Mandal & Suzuki 2002; Winship 1987), and are considered priority pollutants by the United States Environmental Protection Agency (USEPA 1979). A major route of exposure is consumption of crops grown in contaminated soils. While background concentrations of arsenic and antimony in soils are low (<10 mg/kg) (Wilson et al. 2010), contamination can result from activities such as mining, waste disposal, incineration, fuel combustion, smelting, and shooting activities (He et al. 2012; Hockmann et al. 2015; Telford et al. 2009; Wilson et al. 2010), with mining operations making the greatest contribution (Takahashi et al. 2010). These activities can increase environmental concentrations by orders of magnitude (Wilson et al. 2010), with mining-affected soils previously reported to contain up to 17,400 mg As/kg and 16,389 mg Sb/kg (Gál et al. 2007; Li et al. 2014). A major concern is the disposal of waste material from mine sites, which has resulted in contamination of agricultural soils (Hammel et al. 2000), especially when the environmental conditions favour mobilisation of these contaminants. The biogeochemical behaviour of As and Sb is strongly dependent on chemical speciation and parameters like soil pH and redox potential, which can affect important retention mechanisms such as adsorption and precipitation (Wilson et al. 2010; Wilson et al. 2014).

The biogeochemical behaviour of As has been investigated widely, whilst the understanding of Sb biogeochemistry and bioavailability is comparatively limited (Filella et al. 2009; Wilson et al. 2014). Because they are both Group 15 metalloids and exist as oxyanion species with predominant oxidation states of +3 and +5, researchers have anticipated that As and Sb will behave in a similar manner (Wilson et al. 2010). Various speciation studies on As have identified its soil mobility and bioavailability, utilising procedures that are classic (soil water extracts/soil solution), operational (sequential extractions) and functional (plant bioassay) (Bacon & Davidson 2008; Meharg & Hartley-Whitaker 2002; Vithanage et al. 2012; Wenzel et al. 2001). A sequential extraction procedure developed by Wenzel et al. (2001) has been particularly useful in identifying the mobile fraction of As in soils and predicting As phytoavailability in long-term contaminated soils (Niazi et al. 2011), but few such studies have been done for Sb

63

(Okkenhaug et al. 2011). Recent studies by Hockmann et al. (2014a; 2015) have focused on the mobility of Sb under differing redox conditions in the laboratory and field, however, there is a general lack of information on the assessment of Sb soil pools and phytoavailability in soils. For this reason the As fractionation scheme was applied for Sb (Ettler et al. 2010; Okkenhaug et al. 2011). In terms of Sb uptake by plants, most studies to date have determined Sb accumulation for phytostabilisation and phytoremediation in field collected plants (Álvarez-Ayuso et al. 2013; Casado et al. 2007; Murciego et al. 2007) and preliminarily assessed Sb uptake by edible plants grown in field or greenhouse trials (He 2007; Tschan et al. 2009a; Wilson et al. 2014).

The diffusive gradients in thin films (DGT) technique is an *in situ*, passive sampling method that has been successfully used in soils to measure phytoavailable metals (Degryse et al. 2009). The DGT approach incorporates a variety of soil properties influencing metal adsorption/desorption (Williams et al. 2011), and can mimic metal uptake by plant roots where lower metal concentrations in the vicinity of plant root zones leads to the release and resupply of metals from the solid phase (Zhang et al. 2001). The first DGT application in soil for As sampling was conducted by Fitz et al. (2003) who successfully used DGT to monitor the efficiency of As phytoremoval and the resupply of labile As pools after phytoextraction. The suitability of DGT to monitor root-induced changes in soils was successful for available soil-As to roots in hyperaccumulator plants (Cattani et al. 2009); changes in the root-soil interface of two plants with differing As accumulation abilities (Senila et al. 2013); changes in As bioavailability during aging (Liang et al. 2014); and to predict As uptake by edible rape grown in 43 soils (Wang et al. 2014). However, DGT has not previously been used to measure labile Sb in soil, perhaps as the method validation has only recently been described (Panther et al. 2013).

The aims of this study were to investigate the lability of As and Sb in a contaminated soil serially diluted to produce a contamination gradient. Lability was measured using both chemical extractions and DGT techniques, and these were compared with As and Sb bioaccumulation in radish (*Raphanus sativus*), a common edible vegetable species worldwide. Importantly, this study assessed the ability of chemical extractions and DGT measurements to predict the bioavailability of As and Sb to *R. sativus*, which is critical for

future research on the risk of As and Sb phytoaccumulation by plants grown in contaminated soils.

#### **3.2. Methods**

#### 3.2.1. General experimental

General washing methods were done as per section 2.1.

## 3.2.2. Experimental design

#### **3.2.2.1. Experimental soils**

Soil highly contaminated with As (4800 mg/kg) and Sb (7900 mg/kg) was collected from a decommissioned antimony processing facility in Urunga NSW, Australia. This was a sandy soil and had low organic matter (0.95%) and pH (3.2). The soil also contained a range of other metals including Cd, Cu, Zn, Cr, Co, Ni, and V at 0.003, 18, 28, 28.4, 2.2, 3.2, and 29.1 mg/kg, respectively, which are all below the reported global background soil concentrations (Kabata-Pendias 2011). This indicates that these metals are not main contaminants and their effects on As and Sb behaviour were negligible. The uncontaminated diluent sandy soil with higher organic matter (8%) and pH (6.6) was collected in Wollongong, NSW. All soils were air-dried and then crushed to pass through a 2-mm sieve. The historically contaminated soil was thoroughly mixed with the uncontaminated soil at various ratios from 1:200 to 1:10 to establish an As and Sb concentrations were estimated to be in the concentration ranges of previous studies on As and Sb accumulation by plants in pot experiments (Niazi et al. 2011; Tschan et al. 2009a).

#### **3.2.2.2. Pot experiment**

Pot experiments were conducted with the cherry belle radish (*Raphanus sativus*) in controlled growth chambers as per section 2.4.1. Harvested plant tissues were oven-dried at 70 °C for 48 h, and dry mass recorded (g). Dried tissues were ground with a mortar and pestle and stored in clean plastic bags prior to As and Sb analyses.

# **3.2.3.** Soil characteristics

Soil particle size, pH, total Kjeldahl nitrogen, and available phosphorous was determined as per section 2.5.

# 3.2.4. Soil and plant analyses

Finely ground soil (~0.4 g) was digested with aqua regia in a closed microwave digestion system (MARS Xpress, CEM) at 175 °C (ramp time 15 min, 5 min holding time). Finely ground plant (~0.3 g) was microwave digested with a mixture of HNO<sub>3</sub>:H<sub>2</sub>O<sub>2</sub> (1:2, v/v) (65% and 30%, respectively) at 180 °C (ramp and holding time 15 min each). After cooling to room temperature, digests were filtered (<0.45  $\mu$ m) and diluted to 2% acid with deionised water for analyses. Total As and Sb was analysed by inductively coupled plasma-mass spectrometry (ICP-MS; Agilent 7500ce) by measuring the signal at m/z 75 for As and m/z 121 for Sb as per section 2.9.

For each analytical batch, reagent blanks, standard reference materials, spike recovery, and duplicate analyses were used as quality control measures. Soil standard reference material (Montana II soil, SRM 2711a; NIST) returned recoveries of As and Sb within the ranges of 101-105% and 80-90%, respectively. Plant standard reference material (Tomato leaves, SRM 1573a; NIST) were within 110-120% and 80-86% of certified values for As and Sb, respectively.

# 3.2.5. Sequential extraction procedure

The chemical fractionation (soil binding phases) of both As and Sb in the test soils was determined as per Wenzel et al. (2001), due to the expected geochemical similarities of As and Sb (Tighe & Lockwood 2007). A soil:solution ratio of 1:25 was used for all extractions and prepared as per section 2.6.1.

# 3.2.6. DGT and soil solution measurements

DGT and soil solution measurements were conducted as per sections 2.6.2 and 2.7.

# 3.2.7. Data analysis

The basic statistical analysis and the correlations were carried out using SPSS Ver. 21 (IBM Corp., Armonk, New York, USA). Statistically significant differences between As

and Sb in tissue dry masses were analysed using analysis of variance (ANOVA) at a significance level of p<0.05. Correlation analysis investigated the relationship between As and Sb labile fractions measured by various techniques and their concentrations in different tissue compartments. All results are presented as mean  $\pm$  standard error (SE) (dry mass), unless otherwise stated. To evaluate the bioavailability of As and Sb for *R. sativus*, the bioaccumulation factor (BAF) was calculated by the ratio of the total As/Sb concentrations in roots as mg/kg dry weight (DW) and total As/Sb in soils (mg/kg DW) (Sharifi et al. 2014; Vamerali et al. 2010). To compare the distribution of As and Sb in *R. sativus* roots and shoots, the translocation factor (TF) was calculated by the ratio of the total As/Sb in shoots (mg/kg DW) and total As/Sb in roots (mg/kg DW) (Vamerali et al. 2010).

#### **3.3. Results and discussion**

#### **3.3.1.** Soil characteristics

Soil properties affect the solid phase speciation and mobility of As and Sb (Wilson et al. 2010). In this study 12 test soils were prepared by mixing a soil contaminated with both As and Sb with an uncontaminated soil having similar texture, and the physical and chemical properties of soils were analysed prior to use in pot experiments. The test soils properties were very similar for each of the 12 dilutions, as demonstrated by the low standard deviation in the following measurements: silty sand (73.0  $\pm$  0.4% sand, 25.0  $\pm$  0.3% silt and  $2.00 \pm 0.04\%$  clay); slightly acidic (pH 6.50  $\pm 0.03$ ); 7.11  $\pm 0.08\%$  organic matter content; and mean extractable phosphorous and total nitrogen of 74  $\pm$  8 and 102  $\pm$  3 mg/kg, respectively. Thus, it is likely that the differences in As and Sb behavior were due to the differences in their concentrations. The total As and Sb concentrations in 12 bioassay soils ranged from 13.3 - 400 mg As/kg and 11.8 - 720 mg Sb/kg (the actual concentrations of each test soil presented in Table 3.1). These values were above the average background total As and Sb soil concentrations (<10 mg/kg, Wilson et al. 2010) and in the wide range of total soil As and Sb found in different areas around the world: 7.4 - 17,400 mg As/kgand 3.12 - 16,389 mg Sb/kg in mining, smelting, and industrial regions as well as soils surrounding these areas of Spain, China, Scotland, and Czech Republic (Álvarez-Ayuso et al. 2013; Casado et al. 2007; Ettler et al. 2010; Gál et al. 2007; Li et al. 2014; Okkenhaug et al. 2011).

As					Sb					
Soil	Total <sub>As</sub> (mg/kg)	C <sub>SEP-labile As</sub> (mg/kg)	C <sub>DGT-As</sub> (µg/L)	C <sub>sol-As</sub> (µg/L)	Ratio $R_{As}$ ( $R = C_{DGT}/C_{sol}$ )	Total <sub>Sb</sub> (mg/kg)	C <sub>SEP-labile Sb</sub> (mg/kg)	C <sub>DGT-Sb</sub> (µg/L)	C <sub>sol-Sb</sub> (µg/L)	Ratio $R_{Sb}$ ( $R = C_{DGT}/C_{sol}$ )
<b>S</b> 1	$13.3\pm0.3$	$0.92\pm0.01$	$1.9\pm0.4$	$6.6\pm0.7$	$0.29\pm0.07$	$11.8\pm0.6$	$0.140\pm0.002$	$1.9\pm0.1$	$10.3\pm0.2$	$0.19\pm0.01$
<b>S</b> 2	$25\pm3$	$1.51\pm0.04$	$2.6\pm0.6$	$6.8\pm0.5$	$0.4 \pm 0.1$	$26 \pm 2$	$0.295\pm0.005$	$2.5\pm0.3$	$19\pm1$	$0.14\pm0.02$
<b>S</b> 3	$30 \pm 1$	$1.86\pm0.09$	$2.7\pm0.4$	$7.7\pm0.2$	$0.35\pm0.05$	$39 \pm 1$	$0.41\pm0.02$	$4.0\pm0.3$	$28\pm3$	$0.14\pm0.02$
<b>S</b> 4	$40 \pm 2$	$2.51\pm0.06$	$3.9\pm0.2$	$8.4\pm0.6$	$0.47\pm0.04$	$56\pm5$	$0.591 \pm 0.008$	$5.1\pm0.2$	$35\pm3$	$0.15\pm0.01$
S5	$49\pm1$	$2.96\pm0.06$	$4.4\pm0.9$	$9.8\pm0.4$	$0.4 \pm 0.1$	$70 \pm 2$	$0.727\pm0.009$	$6.4\pm0.3$	$43\pm3$	$0.15\pm0.01$
<b>S</b> 6	$51 \pm 1$	$3.31\pm0.06$	$4.1 \pm 0.3$	$10.9\pm0.6$	$0.38\pm0.04$	$80 \pm 4$	$0.83\pm0.01$	$6.9\pm0.1$	$52\pm5$	$0.13\pm0.01$
<b>S</b> 7	$77 \pm 3$	$3.97\pm0.01$	$5.7\pm0.8$	$11.5\pm0.6$	$0.49\pm0.08$	$110\pm10$	$0.853\pm0.008$	$8.5\pm0.4$	$57\pm5$	$0.15\pm0.01$
<b>S</b> 8	$79\pm3$	$4.23\pm0.07$	$4.5\pm0.5$	$14.1\pm0.5$	$0.32\pm0.03$	$127\pm 6$	$0.96\pm0.03$	$8.7\pm0.2$	$65\pm4$	$0.13\pm0.01$
<b>S</b> 9	$90\pm2$	$5.21\pm0.09$	$4.40\pm0.05$	$15.0\pm0.6$	$0.29\pm0.01$	$149\pm3$	$1.34\pm0.01$	$10.7\pm0.1$	$84\pm5$	$0.13\pm0.01$
<b>S</b> 10	$180\pm10$	$8.07\pm0.07$	$11 \pm 2$	$22 \pm 1$	$0.52\pm0.07$	$280\pm10$	$2.14\pm0.04$	$17.7\pm0.6$	$128\pm3$	$0.138\pm0.006$
S11	$210\pm20$	$9.8\pm0.1$	$13.8\pm0.7$	$28.1\pm0.7$	$0.49\pm0.03$	$370\pm30$	$2.60\pm0.05$	$22.5\pm0.4$	$167 \pm 7$	$0.135\pm0.006$
S12	$400\pm20$	$18.5\pm0.4$	$30 \pm 4$	$67.1\pm0.7$	$0.45\pm0.06$	$720\pm30$	$4.8 \pm 0.1$	$46.7\pm0.9$	$350 \pm 1$	$0.133\pm0.003$

**Table 3.1** Total and labile concentrations measured by SEP, DGT, soil solution in a concentration gradient obtained from historically contaminated soils, and R value of As and Sb (mean  $\pm$  SE, n = 3).

#### **3.3.2.** Labile As and Sb in soils assessed by sequential extraction

The As and Sb associated with the different soil binding phases are shown in Figures 3.1. The binding strength of As and Sb to soil increased with the strength of sequential extraction solutions, and for the test soils the extractable As and Sb in soils increased with the increasing total soil As and Sb concentrations (Table A.1). Although As and Sb have been assumed to have similarities in geochemical behaviour (Ettler et al. 2010), clear differences in association with soil binding phases were observed.

For As, the non-specifically sorbed (SO<sub>4</sub><sup>2-</sup> extractable) As in the bioassay soils increased from 0.048 ± 0.004 (S1) to 1.49 ± 0.05 (S12) mg/kg, and only constituted 0.31 ± 0.01% of the total soil As. The specifically sorbed (PO<sub>4</sub><sup>3-</sup>-extractable) As in soils varied from 0.87 ± 0.01 (S1) to 17.0 ± 0.4 (S12) mg/kg, which was  $5.4 \pm 0.2\%$  of the total As. Arsenic was primarily associated with amorphous, crystalline Fe and Al oxides, and residual fractions, each represented 31-33% of the total soil As (Figure 3.1a).

For Sb, the non-specifically sorbed fraction in the bioassay soils varied between 0.056  $\pm$  0.001 (S1) and 1.75  $\pm$  0.01 (S12) mg/kg, representing 0.33  $\pm$  0.02% of the total soil Sb. The specifically sorbed Sb ranged from 0.084  $\pm$  0.001 (S1) to 3.1  $\pm$  0.1 (S12) mg/kg, constituting 0.60  $\pm$  0.03% of the total Sb. The Sb associated with amorphous and crystalline Fe and Al oxides was 11.3  $\pm$  0.8 and 15  $\pm$  1%, respectively. Antimony was mainly present in the residual soil fractions accounting for 73  $\pm$  2% of the total Sb (Figure 3.1b). This fraction is strongly associated with the soil minerals and is generally geochemically immobile (Filella et al. 2009; Okkenhaug et al. 2011). Antimony in this fraction is likely to exist as the antimony sulphide mineral, stibnite (Álvarez-Ayuso et al. 2013).



**Figure 3.1** (a) As associated with different binding fractions for bioassay soils (mg/kg, dry mass, mean  $\pm$  SE, n = 3). Soils are plotted from left to right as S1 to S12 for each binding phase. Inset graphs depict the As and Sb associated with each fraction as a percentage of the total soil concentration (mean  $\pm$  SE, n = 36).



**Figure 3.1** (b) Sb associated with different binding fractions for bioassay soils (mg/kg, dry mass, mean  $\pm$  SE, n = 3). Soils are plotted from left to right as S1 to S12 for each binding phase. Inset graphs depict the As and Sb associated with each fraction as a percentage of the total soil concentration (mean  $\pm$  SE, n = 36).

The soil binding phases that represent the bioavailable fractions are defined as the combination of the non-specifically sorbed and specifically sorbed fractions (Wenzel et al. 2001). Based on this, Sb was much less bioavailable (at  $0.93 \pm 0.04\%$ ) than As (5.7  $\pm$  0.2%). The limited research comparing the As and Sb mobility in contaminated soils has also shown generally higher extractable/available As compared to Sb. For example, Ettler et al. (2007) showed that the extractability of Sb was systematically lower than As (0.1 M Na<sub>2</sub>HPO<sub>4</sub> yielded 9% and up to 34% of Sb and As, respectively), and Müller et al. (2007) reported 10% and up to 15% of Sb and As, respectively, in the mobile fraction (0.05 M Na<sub>2</sub>HPO<sub>4</sub>). However, in soils from an historical mining site in Scotland, Gál et al. (2007) found higher Sb mobility (0.01-8.8% of the total Sb) than As (0.01-0.6% of the total As) in the exchangeable binding phase (with 1 M NH<sub>4</sub>NO<sub>3</sub>).

Binding to Fe and Al oxides was significantly lower for Sb (11-15%) than for As (31-33%), a trend which is supported by other studies (Ettler et al. 2010; Müller et al. 2007). This may be because arsenic has also been shown to outcompete Sb for soil sorption sites since As has a higher affinity for the Fe oxides than Sb (Casiot et al. 2007). The literature about the effect of redox state on the interaction of As and Sb with iron oxides also support the result. Under oxic conditions both As and Sb are present as As(V) and Sb(V) (Wilson et al. 2010), and As(V) interacts more strongly with Fe and Mn oxides than As(III), while the opposite is true for Sb (Fendorf et al. 2007; Leuz et al. 2006). In addition, As(V) has a higher affinity for iron oxides than Sb(V) (Wilson et al. 2010). These lead to higher proportions of As found in oxide phases compared with Sb. A study of As fractionation in 20 contaminated soils by Wenzel et al. (2001) also showed that As was predominantly associated with the amorphous Fe and Al oxides fraction, irrespective of the sources of arsenic contamination, making the Fe and Al oxides the most dominant sink for As in soils.

In the current study most of the Sb was strongly bound to the residual fraction of mining soils  $(73 \pm 2 \%)$ , which was more than double the percentage of residual-As. Interestingly, this was unaffected by the diluent control soils or the interaction of the plants. Previous research has similarly found that Sb is predominantly partitioned in the residual fraction (Álvarez-Ayuso et al. 2013; Ettler et al. 2010). A higher proportion of Sb in the residual phase is most likely to be due to Sb present as stibnite, which is the mineral typically mined

(Ålvarez-Ayuso et al. 2013) and known to be associated with the Urunga site. The residual fraction may also include Sb associated with organic matter; while As and Sb can both adsorb to humic acids, antimonite tends to accumulate more strongly in organic horizons of soils (Ceriotti & Amarasiriwardena 2009). Dousova et al. (2015) confirmed these trends by demonstrating that arsenate was predominantly bound to mineral soils, while antimonate was strongly adsorbed in organic soils. This suggests that Sb-lability may increase if concentrations of soil organic matter decrease (e.g. from microbial degradation). The amount of Sb or As associated with the organic fraction of soil is not identified in the SEP used here (Wenzel et al. 2001), and it is likely to be in measured as part of the residual fraction.

The differences in As and Sb chemical fractionation and lability may be related to the differences in their mineral solubilities, redox properties, and their pentavalent oxyanions structures, where arsenate is tetrahedral antimonate is octahedral (Okkenhaug et al. 2012; Wilson et al. 2010).

Based on the results, it is likely that As and Sb were both not very labile under the oxic conditions, and Sb was more strongly retained in the solid phase than As. However, these contaminants may be remobilised into soil pore water under anoxic conditions (Hockmann et al. 2015; Xu et al. 2008), and As is at more risk than Sb due to the slow rate of the release of Sb from mineral phases like stibuite. This warrants further investigation.

#### 3.3.3. Labile As and Sb in soils assessed by DGT technique

DGT-labile concentrations,  $C_{DGT}$ , and total As and Sb concentrations in soil solution,  $C_{sol}$ , are reported in Table 3.1. The capacity of the solid phase to resupply As and Sb in soil solutions was calculated from these values as R (R =  $C_{DGT}/C_{sol}$ , 0 < R < 1). An R value of 1 indicates that desorption/dissolution processes within the soils can sustain analyte soil solution concentrations in response to the flux into the DGT sampler (Harper et al. 1999; Harper et al. 1998), whereas a very low R indicates very limited or no resupply of labile species from the solid phase and only diffusional resupply.

In this study the soil solution concentrations ( $C_{sol}$ ) of As and Sb and the DGT-labile ( $C_{DGT}$ ) As and Sb generally increased with total concentrations. The values of  $C_{sol}$  and  $C_{DGT}$  of As and Sb were low ranging from 0.005 to 0.05% (1.9 – 67 µg As L<sup>-1</sup>) and from 0.006 to

0.09% (1.9 - 350 µg Sb L<sup>-1</sup>) of total soil As and Sb, respectively. The low lability could be due to the extent of aging with the contaminated soil and the stability of As associations (Liang et al. 2014). The average  $K_d$  value (sorbed/dissolved concentrations, mL/g) for As was 290 ± 20, which was much higher than that for Sb (15.4 ± 0.3), indicating more Sb in soil water.  $C_{DGT}$  of As and Sb were lower than  $C_{sol}$  in all soils, however the  $C_{DGT}$  increased linearly with soil solution concentrations (Figure 3.2). The average R, regression slope, for As was 0.41 ± 0.03 indicating partially sustained resupply from the solid phase and suggests that solid phase As was likely to be mobile for the plant bioassays. On the other hand, the average R of 0.14 ± 0.07 for Sb suggests that there was minimal resupply from the solid phase. The higher regression slope for As (0.41) compared to Sb (0.14) could be explained by the fact that DGT accumulated As more quickly than it was resupplied or more of the Sb being in the soil solution was in an inert form, perhaps as dissolved organic complexes, which were not DGT-labile.

It is also noticeable that despite total Sb concentrations in soils being nearly double those of As, the DGT-derived concentrations of Sb were only 26 to 34% higher than the DGT-As. The R ratios of As and Sb remained relatively constant throughout the soil concentration series, indicating that the behaviour of As and Sb in the bioassay soils were unaffected by the soil dilution. Thus the control soil (diluent) had no marked effect on the resupply rates of As and Sb from the solid to the dissolved phase.



**Figure 3.2** The relationship between DGT-labile ( $C_{DGT}$ ) As and Sb after 24-h deployment and concentrations in extracted soil solutions (mean  $\pm$  SE, n = 3). Dashed line represents a 1:1 gradient, regression set through zero. Note that total Sb soil concentrations were much higher than total As concentrations.

#### **3.3.4.** Uptake of As and Sb by radish (*Raphanus sativus*)

#### 3.3.4.1. Plant yield

*R. sativus* had an average wet tissue mass of  $23.4 \pm 0.6$  g (n = 48) which was not affected (p>0.05) by the contaminated soils following the 35-day exposure bioassay, as there was also no significant difference in tissue moisture (87.2 ± 0.3%, n = 48) between plants exposed to different concentrations. *R. sativus* responses in this study were not consistent with the study by Smith et al. (2008b), who observed the reduction of wet weight mass of *R. sativus* grown hydroponically (0.8 – 50 mg As/L) and in mine waste soil treatments (660 and 1100 mg As/kg). However, the As-lability in our test soils reached 0.067 mg As/L (C<sub>sol-As</sub>) and had a total soil As of 400 mg/kg, which was much lower than that in Smith et al. (2008b).

#### 3.3.4.2. Accumulation of As and Sb in *R. sativus* tissues

The As and Sb concentrations in different *R. sativus* tissues increased with increasing soil exposure concentrations (Figure 3.3). Tissue concentrations were low in comparison to the total soil concentrations, which is consistent with the low lability of As and Sb in soils as determined by both SEP, C<sub>sol</sub> and DGT measurements. Both As and Sb concentrations in the R. sativus tissues followed the trend shoots > peel > root flesh. The flesh concentrations remained particularly low irrespective of soil As and Sb concentrations, and it accumulated <3 times the As and <9 times the Sb found in the shoots. For all total soil concentrations <70 mg As/kg (equivalent to DGT-labile As of  $<4.5 \mu$ g/L), the *R. sativus* peel and flesh concentrations were not significantly different (p>0.05), however, above this concentration the peel accumulated more As than the flesh. Antimony followed a similar trend, however the difference in Sb accumulation between the peel and flesh was less than for As. Overall As was accumulated to a greater concentration than Sb in all plant compartments even though total soil and soil water Sb concentrations were higher. The uptake pattern in this study is supported by Bhatti et al. (2013) who reported that the highest amount of As in R. sativus shoots, followed by peel and peeled roots when the plants were irrigated with arsenic-contaminated water every 10 days until harvest. This indicates that the distribution of As in various tissues of this R. sativus species is not affected by contamination exposure mode.



**Figure 3.3** The relationship between total soil concentrations and the accumulation of As (a) and Sb (b) in *R. sativus* tissues (mg/kg dry mass, mean  $\pm$  SE, n = 4) following cultivation in contaminated soils for 35 days. Shoots ( $\blacktriangle$ ), Peel ( $\Box$ ), and Flesh ( $\diamondsuit$ ). All data have error bars; where they are not seen they are smaller than the symbols.

The actual As and Sb concentrations accumulated in different compartments ranged from 0.1 - 2.5 and 0.02 - 0.56 mg/kg in roots (flesh plus peel) and 0.1 - 2.2 and 0.03 - 1.31 mg/kg in shoots, respectively (Figure A.1). The distribution of As between *R. sativus* tissues was affected by the As soil concentrations. Below total soil concentrations of 90 mg As/kg slightly higher amounts of As were found in shoots, but As tended to be accumulated more in roots than in shoots when *R. sativus* were exposed to higher soil As concentrations. In comparison, there was significantly (p<0.05) higher Sb in shoots than roots in almost all studied soils, except for the lowest treatment (S1). This is consistent with previous studies on arsenic toxicity and accumulation in *R. sativus* exposed to different forms of As in media (Carbonell-Barrachina et al. 1999a; Tlustoš et al. 1998).

The soil-plant transfer model is one of approaches to predict contaminant uptake by plants, in which the translocation factor (TF) and the bioaccumulation factor (BAF) are computed to describe the scenario at the screening level of risk assessments (McLaughlin et al. 2011; Murray et al. 2009). Translocation factors evaluated the ability of plant to translocate As and Sb from roots to shoots and is based on the ratio of total concentrations in shoots compared to roots (mg/kg DW). In our study the TF were different for As (0.8 - 2) and Sb (3-4.1) (Table 3.2), with the translocation rate of Sb being approximately 2.5 times higher than that of As which means that the R. sativus had a significantly (p<0.05) stronger capacity to transport Sb from the roots to the shoots than As. The translocation factors of As and Sb tended to decrease with increasing soil As and Sb concentrations. The lower translocation of solutes could be from the reduction of xylem sap that occurs when plants were exposed to high concentrations of metals. A study of the impact of As on the metabolite profile and production of xylem sap in cucumbers (Cucumis sativus L.) grown hydroponically reported that 96% of xylem sap was reduced when cucumber was exposed to 1 mg As(V)/L (Uroic et al. 2012), possibly because the reduction of sap is a mechanism of plant to avoid As translocation to shoots to protect photosynthesis in shoots.

The bioaccumulation factor (BAF) (also known as the transfer factor, transfer coefficient, or uptake factor) is determined from the ratio of the total concentration in plant roots and the total soil concentration (mg/kg DW). The BAF is an index used to evaluate the transfer of a metal from soils to plants. A higher BAF reflects a greater efficiency of plants to accumulate a metal and a relatively more labile metal fraction (Douay et al. 2013). Plants

with a low BAF (<1) tend to have limited transfer between soil and the roots, so this plant is considered to be "tolerant". Plants with high BAF and TF (>1) have potential to be used for phytoextraction remediation purposes (Vamerali et al. 2010). However, if contaminants are highly accumulated in edible parts of plants or if the plants applied for phytotechnologies are accidentally consumed by animals, they pose a risk if consumed. Thus, the bioaccumulation factor is commonly determined to ensure food safety.

The BAF of As and Sb in *R. sativus* roots varied between 0.0049 - 0.008 and 0.00028 - 0.003, respectively (Table 3.2), showing that the uptake rate of As in roots was much higher (2.5 – 21 times) than that of Sb, however, the transfer of As from soils to *R. sativus* roots was more efficient than Sb. The BAF values for As and Sb were much lower than the typical BAF reported for As (about 0.2) and for Sb (about 0.01) reviewed from literature (Wilson et al. 2013), suggesting that the phytoavailable As and Sb in test soils were low and the plant acquisition of these metalloids from the historically contaminated oxic soils were small.

Soil	Bioaccumulation factor (BAF)		Translocation factor (TF)		
	As	Sb	As	Sb	
<b>S</b> 1	$0.008 \pm 0.002$	$0.003\pm0.001$	$1.4 \pm 0.4$	$1.3 \pm 0.5$	
S2	$0.0049 \pm 0.0004$	$0.0015 \pm 0.0002$	$2.0 \pm 0.3$	$3 \pm 1$	
<b>S</b> 3	$0.0059 \pm 0.0006$	$0.0005 \pm 0.0001$	$1.5 \pm 0.1$	$3.6 \pm 0.5$	
S4	$0.0058 \pm 0.0004$	$0.00028 \pm 0.00001$	$1.2 \pm 0.1$	$3.9 \pm 0.4$	
S5	$0.0061 \pm 0.0003$	$0.0008 \pm 0.0001$	$1.15\pm0.07$	$2.0 \pm 0.2$	
<b>S</b> 6	$0.0078 \pm 0.0007$	$0.0006 \pm 0.0001$	$1.09\pm0.05$	$4.1 \pm 0.6$	
<b>S</b> 7	$0.0057 \pm 0.0003$	$0.0004 \pm 0.0001$	$1.1 \pm 0.1$	$2.7\pm0.1$	
<b>S</b> 8	$0.0056 \pm 0.0002$	$0.00042 \pm 0.00002$	$0.95\pm0.07$	$3.1 \pm 0.5$	
S9	$0.0071 \pm 0.0003$	$0.0008 \pm 0.0001$	$0.89\pm0.06$	$2.0 \pm 0.1$	
S10	$0.0050 \pm 0.0003$	$0.0006 \pm 0.0001$	$0.83\pm0.06$	$2.4 \pm 0.3$	
S11	$0.0058 \pm 0.0004$	$0.00066 \pm 0.00004$	$0.81\pm0.09$	$2.5 \pm 0.3$	
S12	$0.0061 \pm 0.0004$	$0.00079 \pm 0.00003$	$0.9 \pm 0.1$	$2.3 \pm 0.3$	

**Table 3.2** Bioaccumulation factor and translocation factor for As and Sb in *R. sativus* (mean  $\pm$  SE, n = 4).

This is consistent with the low percentage of bioavailable As and Sb (sum of the non-specifically sorbed and specifically sorbed fractions obtained from the SEP), which was between 0.31 - 5.4% and 0.33 - 0.6% of the total As and Sb, respectively (Table 3.1). The low C<sub>DGT</sub> of As and Sb also support this, with an average of 0.008% of total As and 0.009% of the total Sb. Our results support the well-established hypothesis that As phytoavailability is more closely related to available As than to total soil As (Cattani et al. 2009). These data suggest that the low lability of As and Sb in test soils resulted in low bioaccumulation of As and Sb in the *R. sativus*.

Our study was performed at much lower labile concentrations than previous publications, thus the As accumulated within our *R. sativus* tissues (Figure 3.3) was considerably lower than that reported by other studies:  $36 \pm 8$  and  $10 \pm 3$  mg As/kg (DW) in roots and shoots, respectively, of *R. sativus* grown hydroponically in 2 mg As/L nutrient solution (Smith et al. 2009), and 5 mg As/kg (DW) for both roots and shoots of *R. sativus* grown in soils with 1 N HCl-extractable As concentrations of 25 mg/kg (Gutierrez et al. 2010). In addition to our lower exposure concentrations, Australian soils are typically very high in Fe. Thus the most probable parameter limiting the As and Sb accumulation in our bioassay is adsorption by iron oxides and minerals within the soil particles, demonstrated by high percentages of As (64%) and Sb (27%) found in Fe and Al oxide phases and 31% of total As and 73% of total Sb found in residual phases (Figure 3.1). Arsenic is most prominently found in soils bound to Fe and Al-oxides (Kim et al. 2014; Niazi et al. 2011) and these minerals are dominant components of Australian soils (Smith et al. 2003).

Other plants have similarly accumulated more As in the roots than shoots (Cobb et al. 2000; Smith et al. 2009). The mechanism of metal and metalloid tolerance in plants may relate to plants limiting the upward movement of toxic elements by concentrating them in the plant roots, including via complexation in phytochelatins and sequestration in vacuoles (Schmoger et al. 2000; Zhao et al. 2010).

Greater As accumulation in comparison to Sb has also been reported in the literature (Wei et al. 2011; Wilson et al. 2014). The differences in bioaccumulation most likely arise from differences in analyte behaviour, due to their structural conformation of arsenate and antimonate, as the most common Sb species in soils is the octahedral oxyanion  $Sb(OH)_6^-$ , which differs from the tetrahedral oxyanions of As(V). The mechanisms of As uptake by

plants have been studied in depth, as arsenate and phosphate are chemical analogs, enabling arsenate to be taken up by plant roots through the phosphate pathway (Zhao et al. 2010). However, the uptake mechanisms of antimonate are still unclear (Feng et al. 2013), with research showing that antimonate was not accumulated via phosphate transporters by maize and sunflower in hydroponic experiments (Tschan et al. 2008). Further research on mechanisms of Sb uptake by plants is required to fully understand Sb bioavailability.

It is important to note that plants harvested from soils often contain a visible layer of iron plaque (e.g. precipitated iron oxyhydroxides) on the roots (Tripathi et al. 2014). If this layer is not removed prior to analysis, the tissue concentration of As and Sb could be erroneously high due to their sorption to this layer of iron plaque and thus the inclusion of a fraction that was not actually accumulated by the plant. The removal of root skin prior to cooking and eating is also a usual practice. In fact, as there was much higher As concentrations in the peel, it would be important to reduce contaminant exposure by peeling root vegetables before consumption.

# **3.3.5.** The relationship between As and Sb bioaccumulation and soil exposure concentrations

The relationships between As and Sb concentrations in *R. sativus* tissues and the 'labile' soil exposure concentrations measured by SEP, DGT (as  $C_{DGT}$ ), and soil solution ( $C_{sol}$ ) are presented in Figure 3.4. The concentrations of As and Sb in plant tissues showed a highly significant positive correlation with all of the labile As and Sb fractions ( $R^2 = 0.96 - 0.99$ ), indicating that As and Sb in these phases are readily available to plant uptake.

The 'bioavailable' fractions of SEP, and labile concentrations measured by DGT, and soil solution were better predictors for As and Sb in roots (greater correlation coefficients) than in plant shoots, with the exception of soil solution-As. Arsenic in shoots seems to be limited by that in roots, demonstrated by the low translocation factor. On the other hand, the Sb concentration in shoots is much higher than that in roots, consistent with the higher translocation factor. The better correlations between labile As and Sb in soils and their uptake by roots can be explained by the fact that roots are in direct contact with soils and may be better indicators for soil bioavailable elements than shoots, where bioaccumulation of elements is influenced by the translocation processes, normally governed by physiological characteristics of plants (Chaignon et al. 2003). This also confirmed the

findings that the bioaccumulation factor was one of the parameters reflecting the bioavailability of As and Sb in the test soils and their transfer from soils to roots, while the different translocation of As and Sb from roots to shoots solely depends upon the nature of As and Sb characteristic within this *R. sativus*. McLaughlin et al. (2011) also reported that the translocation factor is specific to the plant species, genotype, and a given element.

According to Degryse et al. (2009), metal uptake by plants can be governed by both plant and soil processes. If the plant acquisition of metals from soil is slower than the supply of metals by diffusion, the depletion of metals at the vicinity of root area is negligible. In this case, the metal accumulated by plants can be assumed to be controlled by the free ion in solution (Campbell 1995). If metal concentrations in soil solutions are proportional to free ion activities, good relationships between metals in plants and in soil solutions may be observed. If the metal accumulation by plants is rapid compared to the diffusional supply, depletion of free ion concentrations occurs within the rhizosphere, leading to the resupply of metals from the soil solution complexes and from the solid phase (Zhang et al. 2001). If the concentration of metals is depleted at the root zone, many processes affecting the metal supply to plants occur including the diffusional transport of metals to the roots and the kinetics of release of metals from solid phases to soil solutions. In these cases, free ion activities and soil solution measures are not successful in predicting metal uptake by plants (Lehto et al. 2006b). According to previous studies (Ernstberger et al. 2005; Luo et al. 2014), metal concentrations in soil solution are derived from labile and non-labile metal. The DGT measurement comprises the free ion in soil solution and the resupply of labile metal from complexes in solution and solid phases, but excludes inert metal bound to solid phases and solution complexes. In this study, the soil solution and DGT are equally effective in predicting As and Sb accumulation by R. sativus, suggesting that the plant accumulated the free ions and the resupply was not a governing factor, consistent with uptake being limited by the plant physiology.



**Figure 3.4** The relationship between tissue As and Sb (mg/kg, dry mass) and bioavailable fractions of As and Sb in soils, as measured by  $C_{SEP \ labile}$ ,  $C_{DGT}$ , and  $C_{sol}$ . Shoot ( $\blacktriangle$ ), Root (o). All data have error bars; where they are not seen they are smaller than the symbols.

A previous study on the assessment of As bioavailability in 43 contaminated soils using DGT and conventional methods also demonstrated strong correlations between As in edible rape and As concentrations obtained from DGT and soil solution, while As contents in the plant were scattered with respect to As obtained from different single extraction (Wang et al. 2014). From these data, the authors concluded that DGT proved to be an effective tool for the assessment and prediction of As bioavailability in soils. However, the research presented in this chapter is the first study to report the application of DGT in predicting Sb accumulation by plants, and comparing this with other soil extraction procedures.

#### **3.4.** Conclusions

The soil historically contaminated with both As and Sb have different soil binding phases for these contaminants. As was chemically more labile and mainly bound to the Fe and Al oxides, yet Sb was much less labile and most likely predominantly associated with organic matter, silicates, or other resistant minerals like sulphides. The coupling of SEP, soil solution and DGT provided essential information on the soil As and Sb speciation, which may be useful in understanding the lability and phytoavailability of As and Sb in soils. The SEP procedure revealed that a high proportion of As was associated with Fe and Al oxides which is not mobile and not DGT-labile. Thus, it is not likely to be the bioavailable fractions in soils under oxic conditions.

Both As and Sb concentrations in *R. sativus* tissues displayed the trend of shoots > peel > flesh. Since there were higher amounts of As and Sb in the peel, peeling off radish skins before consumption would be important to reduce contaminant exposure. The *R. sativus* bioassay showed low bioaccumulation of As and Sb in comparison with total soil concentrations. Arsenic was accumulated more than Sb, however, Sb had a higher ability to translocate from root to shoot than As, to the extent that much higher Sb concentrations were observed in shoots, suggesting a quite different form of transport within the plant.

The DGT, soil solution, and SEP-labile fractions performed comparably well in predicting the accumulation of As and Sb in *R. sativus*. The ability of plants to accumulate As and Sb in their tissues is known to vary with varieties, thus, further studies are needed to identify if these observations are established with the other cultivar of radish and other soil types.
Chapter 4. DGT and selective extractions reveal differences in arsenic and antimony uptake by the white icicle radish (*Raphanus sativus*)

# 4.1. Introduction

In chapter 3, the DGT technique was used for simultaneous measurements of labile As and Sb in contaminated soils ranging from 13.3 to 400 mg As/kg and 11.8 to 720 mg Sb/kg and to predict uptake by the cherry belle radish (*Raphanus sativus*); the results indicated that these two elements behaved quite differently (chapter 3). In terms of their geochemical behaviour, Sb was mainly associated with residual phases and resistant minerals such as silicates or sulphides, while As was primarily bound to Fe and Al oxides. DGT measurements indicated that As was more labile and partially resupplied from the soil solid phase, while Sb was minimally resupplied and therefore most likely present in inert forms. With respect to their uptake by the cherry bell radish (*R. sativus*), arsenic accumulated in roots was 2.5 - 21 times higher that of Sb. However, Sb was translocated from roots to shoots at much greater extent than As by a factor of 2.5, illustrating different uptake trends between these two metalloids.

The ability of plants to accumulate As in their tissues varies with crop species and varieties. For example, the tomato variety Marmande accumulated much more As in roots than tomato cultivar Muchamiel, whereas the latter took up higher As concentrations in shoots and fruits than the former (Burló et al. 1999). Moreover, among various rice cultivars, the BR11 variety accumulated higher As concentrations in grain (0.104 - 1.835 mg/kg) than the others (0.058 - 0.117 mg/kg) (Meharg & Rahman 2003). Similar results were found for Sb, demonstrated by Cai et al. (2016) reporting that the Japonica rice (cultivars N and Z) exhibited a higher accumulation and stronger translocation tendency of Sb from root to stem than Indica hybrid rice (cultivars F and G). The selection of non-accumulating cultivars is an important risk management strategy to minimise human exposure to contaminants via vegetable consumption (McLaughlin et al. 2011). Thus, investigations at

the variety scale are also needed to determine if different cultivars of the same species behave similarly with respect to simultaneous uptake of As and Sb in a range of soil types.

In this study the DGT technique was combined with a sequential extraction approach to assess the lability of As and Sb in relation to their uptake by the *R. sativus* cultivar, the long white icicle radish. This cultivar has a very long taproot, which is different from the round shape of the root of the cherry belle radish examined in our previous study (chapter 3). The differences in uptake by these cultivars have not been reported previously. The aims of this study were to investigate (1) the biogeochemical behaviour of As and Sb in soils prepared from two historically contaminated soils; (2) the bioaccumulation of As and Sb in the white icicle radish grown in these test soils; and (3) the ability of sequential extraction and DGT measurements to predict As and Sb uptake by the plant. The results of this study on As and Sb bioaccumulation by the white icicle cultivar will be compared with recent data obtained for the cherry belle cultivars of the same species.

#### 4.2. Materials and methods

#### 4.2.1. General methods

General washing methods and soil characterisation were done as per sections 2.1 and 2.5. Soil and plant digestion, and analysis of As and Sb in digested samples by ICP-MS were conducted as per sections 3.2.4 and 2.9. For quality assurance, reagent blanks, certified reference materials, spikes, and duplicates were analysed every 15 samples. A soil standard reference material (Montana II soil, SRM 2711a; NIST) returned recoveries of 101-105% and 80-90% for As and Sb, respectively. A plant standard reference material (Tomato leaves, SRM 1573a; NIST) yielded recoveries of 110-120% and 80-86% for As and Sb, respectively.

# 4.2.2. Soil preparation and plant bioassays

Soil highly contaminated with (4800 mg/kg) and Sb (7900 mg/kg) was collected from the area around a decommissioned antimony processing facility in Urunga, New South Wales (NSW), Australia (Warnken et al. 2017) (mining soil, "S"). Details about soil compositions and physical properties were described in section 3.2.2. A second contaminated soil was collected adjacent to a cattle dip site in Queensland (QLD), Australia (cattle dip soil, "Q").

This was a silty-sand soil and contained 1360 mg As/kg and negligible concentrations of other contaminants. An uncontaminated soil was collected in Wollongong, NSW. All soils were air-dried and crushed to pass through a 2-mm sieve. In order to establish test soils, the uncontaminated and the contaminated soils were mixed in a cement mixer at various ratios from 1:200 to 1:10 to give a range of As and Sb concentrations for the mining soil (S1 – S12), and at ratios from 1:100 to 1:3 to give a series of As concentrations for the cattle dip soil (Q1 – Q12). Plant bioassays were conducted with the long white icicle radish (*Raphanus sativus*), an agriculturally important edible vegetable, as per section 2.4.1. Harvested plant tissues from the same pot were combined, weighed, and oven-dried at 70 °C for 48 h to a constant weight with the dry mass (g) recorded. Dried tissues were ground with a mortar and pestle and stored in clean plastic bags prior to As and Sb analyses.

# 4.2.3. Sequential extraction procedure

Sequential extraction was conducted as per section 2.6.1.

# 4.2.4. DGT and soil solution measurements

DGT and soil solution measurements were conducted as per section 2.6.2 and 2.7.

#### 4.2.5. Data analysis

Statistical analysis and calculations of bioaccumulation factor (BAF) and translocation factor (TF) were performed as per section 3.2.7.

# 4.3. Results and discussion

# 4.3.1. Soil properties

The average physical and chemical properties of each series of test soils are summarised in Table 4.1. Soil characteristics are important parameters in controlling the mobility and distribution of As and Sb in solid phases (Wilson et al. 2010). The soils prepared from the cattle dip soil were contaminated with As only, while the mining soils were contaminated with both As and Sb. The physico-chemical properties of all test soils were characterised and analysed prior to performing pot experiments. The soils were a silty-sand (66 - 69% sand, 28 - 31% silt, and 3% clay), slightly acidic to slightly alkaline (pH 6.46 - 7.09), and had relatively similar amounts of nitrogen (83 - 90 mg/kg). The soils diluted from the cattle dip site had slightly higher amounts of organic matter and extractable phosphorus than

those prepared from the mining soil. The total As and Sb concentrations in bioassay soils varied between 13 - 456 and 11 - 680 mg/kg, respectively (Tables 4.2 and 4.3). These values are typical of the total As and Sb in various soils around the world.

#### 4.3.2. Fractionation and mobility of As and Sb in soils

The differences in the percentage of As and Sb associated with various binding phases for the two contaminated soils are presented in Figure 4.1. Generally, the amounts of soil As and Sb extracted in each fraction increased with the increasing total soil As and Sb concentrations (Table B.1, B.2). In both soil types, relatively low amounts of As were released in the SO<sub>4</sub><sup>2-</sup> extraction step targeting non-specifically sorbed (readily available) As and also in the  $PO_4^{3-}$  extraction step targeting specifically sorbed As. The non-specifically sorbed As in diluted cattle dip and mining soils only accounted for  $1.70 \pm 0.08\%$  and  $0.26 \pm$ 0.01% of the total As, respectively. The specifically sorbed As in diluted cattle dip and mining soils only constituted  $11.1 \pm 0.2\%$  and  $5.1 \pm 0.2\%$  of the total As, respectively. The sum of the non-specifically sorbed and specifically sorbed As, known as the labile As (SEP-labile), in diluted cattle dip soils ( $\sim$ 13%) was more than double that in diluted mining soils (~5%). Arsenic was predominantly found in the oxides and residual phases for both soils, relatively equally associated with amorphous and crystalline oxide fractions (33-35%) in the mine site soils, but was primarily bound to amorphous oxides ( $66 \pm 1\%$ ) at the cattle dip site with a lower proportion bound to crystalline Fe and Al oxides  $(14 \pm 1\%)$ . This is likely due to the higher concentrations of amorphous Fe in diluted cattle dip soils (Table 4.1).

Characteristics	Diluted cattle dip soils (Q1 – Q12)	Diluted mining soils (S1 – S12)
sand (%)	$69 \pm 1$	$66 \pm 1$
silt (%)	$28 \pm 1$	$31 \pm 1$
clay (%)	$3.0\pm0.3$	$3.0 \pm 0.1$
pH	$7.09\pm0.06$	$6.46\pm0.03$
organic matter (%)	$8.0\pm0.1$	$6.9\pm0.1$
extractable Phosphorous (mg/kg)	$99\pm7$	$70 \pm 10$
total Kjeldahl Nitrogen (mg/kg)	$83\pm7$	$90\pm7$
amorphous Fe (mg/kg)	$8.3 \pm 0.4$	$5.2 \pm 0.1$
crystalline Fe (mg/kg)	$6.4 \pm 0.5$	$9.4\pm0.2$

**Table 4.1** Physical and chemical properties of bioassay soils (dry mass, mean  $\pm$  SE, n $\geq$ 3).

	As-Mining						Sb-Mining				
Soil	Total	$C_{\text{SEP-labile}}$	$C_{DGT}$	$C_{sol}$	Ratio $R_{As}$	Total	$C_{\text{SEP-labile}}$	$C_{DGT}$	$C_{sol}$	Ratio $R_{Sb}$	
	(IIIg/Kg)	iig/kg) (iiig/kg)		(µg/L)	$(\mathbf{K} - \mathbf{C}_{\text{DGT}}/\mathbf{C}_{\text{sol}})$	(IIIg/Kg)	(IIIg/Kg)	(µg/L)	(µg/L)	$(\mathbf{K} - \mathbf{C}_{\text{DGT}}/\mathbf{C}_{\text{sol}})$	
<b>S</b> 1	$13.1 \pm 0.3$	$0.86 \pm 0.01$	$2.12 \pm 0.02$	$13.2 \pm 0.6$	$0.162 \pm 0.007$	$11.0 \pm 0.6$	$0.125 \pm 0.002$	$1.3 \pm 0.3$	$12.3 \pm 0.5$	$0.11 \pm 0.03$	
S2	$18.3\pm0.8$	$1.06\pm0.07$	$2.83\pm0.04$	$16.4\pm0.5$	$0.173\pm0.005$	$23.2\pm0.6$	$0.255\pm0.006$	$2.9\pm0.2$	$22 \pm 1$	$0.132\pm0.006$	
<b>S</b> 3	$38\pm3$	$2.22\pm0.03$	$4.3\pm0.2$	$20.6\pm0.2$	$0.21\pm0.01$	$42 \pm 7$	$0.420\pm0.06$	$4.1 \pm 0.2$	$33 \pm 3$	$0.13\pm0.02$	
<b>S</b> 4	$44 \pm 0.4$	$2.65\pm0.06$	$5.93\pm0.04$	$27.3\pm0.4$	$0.218\pm0.005$	$55 \pm 7$	$0.59\pm0.01$	$5.7 \pm 0.1$	$45 \pm 4$	$0.129\pm0.008$	
S5	$52\pm 6$	$2.93\pm0.05$	$7.2\pm0.5$	$31.6\pm0.9$	$0.23\pm0.02$	$69 \pm 2$	$0.666\pm0.007$	$8.0 \pm 0.6$	$52 \pm 4$	$0.154\pm0.008$	
<b>S</b> 6	$60 \pm 10$	$3.70\pm0.09$	$7.6\pm0.2$	$29.2\pm0.7$	$0.259 \pm 0.004$	$76 \pm 2$	$0.738 \pm 0.008$	$8.0 \pm 0.4$	$55\pm 6$	$0.15\pm0.02$	
<b>S</b> 7	$80\pm9$	$3.85\pm0.04$	$7.1 \pm 0.9$	$35.8\pm0.5$	$0.20\pm0.02$	$120\pm20$	$0.88\pm0.01$	$11.6\pm0.5$	$78 \pm 2$	$0.15\pm0.01$	
<b>S</b> 8	$82 \pm 1$	$4.12\pm0.05$	$9.6 \pm 0.1$	$43 \pm 2$	$0.22\pm0.02$	$130\pm20$	$0.92\pm0.01$	$13.3\pm0.6$	$102 \pm 4$	$0.131\pm0.008$	
<b>S</b> 9	$94 \pm 4$	$5.08\pm0.04$	$10.4 \pm 0.4$	$47 \pm 2$	$0.220\pm0.004$	$190\pm20$	$1.57\pm0.04$	$14.9\pm0.5$	$110 \pm 4$	$0.136\pm0.008$	
S10	$142 \pm 1$	$5.71\pm0.07$	$12.2\pm0.3$	$62 \pm 2$	$0.199 \pm 0.002$	$260 \pm 4$	$1.80\pm0.06$	$20.7\pm0.6$	$167 \pm 6$	$0.124\pm0.007$	
S11	$220 \pm 30$	$9.81 \pm 0.1$	$16 \pm 2$	$73 \pm 3$	$0.22 \pm 0.01$	$400 \pm 20$	$2.59\pm0.04$	$28.1\pm0.4$	$219\pm3$	$0.128 \pm 0.002$	
S12	$420 \pm 20$	$18.1\pm0.2$	$34.9\pm0.6$	$136 \pm 4$	$0.256\pm0.005$	$680\pm50$	$4.1 \pm 0.1$	$57.9\pm0.9$	$484\pm8$	$0.120\pm0.003$	

**Table 4.2** Total and labile concentrations measured by SEP, DGT, soil solution in bioassay mining soils, including R-value ( $R=C_{DGT}/C_{sol}$ ) for As and Sb (mean  $\pm$  SE, n = 3).

**Table 4.3** Total and labile concentrations measured by SEP, DGT, soil solution in bioassay cattle dip soils, including R-value ( $R=C_{DGT}/C_{sol}$ ) for As (mean  $\pm$  SE, n = 3).

	As-Cattle dip								
Soil	Total	C <sub>SEP-labile</sub>	C <sub>DGT</sub>	$C_{sol}$	Ratio R <sub>As</sub>				
5011	(mg/kg)	(mg/kg)	(µg/L)	(µg/L)	$(R = C_{DGT}/C_{sol})$				
Q1	$16 \pm 1$	$1.98\pm0.03$	$3 \pm 1$	$14.5\pm0.3$	$0.23\pm0.07$				
Q2	$27 \pm 4$	$3.2\pm0.1$	$5.2 \pm 0.2$	$21.0\pm0.2$	$0.25\pm0.01$				
Q3	$30 \pm 1$	$4.9\pm0.2$	$10.3\pm0.5$	$39 \pm 1$	$0.26\pm0.01$				
Q4	$47 \pm 3$	$5.98 \pm 0.09$	$15.3\pm0.4$	$48.6\pm0.7$	$0.32\pm0.01$				
Q5	$53\pm 4$	$7.97\pm0.03$	$19 \pm 1$	$59.3\pm0.4$	$0.33\pm0.02$				
Q6	$62 \pm 3$	$9.3\pm0.2$	$23 \pm 1$	$72.1\pm0.3$	$0.31\pm0.02$				
Q7	$72\pm2$	$10.95\pm0.07$	$23.9\pm0.8$	$81 \pm 1$	$0.30\pm0.01$				
Q8	$86\pm4$	$11.8\pm0.07$	$27 \pm 1$	$86 \pm 1$	$0.31\pm0.01$				
Q9	$107\pm 6$	$15.0\pm0.6$	$34 \pm 3$	$104.3\pm0.4$	$0.32\pm0.03$				
Q10	$153 \pm 3$	$22.4\pm0.2$	$46 \pm 2$	$145 \pm 2$	$0.32\pm0.02$				
Q11	$176\pm8$	$30 \pm 1$	$55 \pm 2$	$169 \pm 3$	$0.33\pm0.01$				
Q12	$460 \pm 40$	$61.9 \pm 0.6$	$120 \pm 10$	$318 \pm 3$	$0.39\pm0.04$				

In general, the partitioning patterns of As in our test soils are consistent with those from other cattle dip (Niazi et al. 2011) and mining soils (Gonzaga et al. 2008), which also showed that the bioavailable As was only about 8% of the total As and that As was mostly (84.3%) associated with amorphous and crystalline Fe and Al oxides. The partitioning of As in these soil types indicates that Fe and Al oxides played a significant role in retention of As in contaminated soils, due to formation of inner-sphere, monodentate and mononuclear as well as bidentate binuclear complexes with soil Fe oxides (Wilson et al. 2010). The higher proportion of As in amorphous oxides and lower proportion of As in the residual fraction in the diluted cattle dip soils reflect the potentially mobile As in these soils, which could be released more easily than in the diluted mining soils, particularly under anoxic conditions. High fractions of As were found in mobile phases in the diluted mining soils.



**Figure 4.1** The percentage of As and Sb associated with different binding fractions for bioassay soils (mean of all concentrations for each soil, mean  $\pm$  SE, n = 36).

Sb distribution between soil binding phases differed to that of As, confirming the results of our previous study (Ngo et al., 2016). The sum of the non-specifically sorbed and specifically sorbed Sb, known as the labile Sb (SEP-labile), was extremely low (0.86% of total soil Sb compared with 5.3% for As in the same soil). Sb in Fe and Al oxide phases (13-18%) were less than half that of As because As has a greater tendency to bind to metal oxides in comparison to Sb (Wilson et al. 2010). In contrast, Sb was primarily bound to the residual phase (68%), which is consistent with previous studies (Álvarez-Ayuso et al. 2012; Serafimovska et al. 2013). The lower proportions of Sb in the metal oxide phases and the higher percentage of Sb in the residual phase compared with As could be because of association with organic matter (Dousova et al. 2015; Tighe et al. 2005) or sulfides (Gál et al. 2007; Okkenhaug et al. 2011). These organic and sulfide phases are not specifically targeted by the SEP of Wenzel et al. (2001), thus in our study, the Sb in these phases was likely to be contained in the residual fraction.

Generally, the partitioning of Sb in the test soils is supported by previous studies on Sb mobility and distribution in soils impacted by smelting and mining activities. For example, in studies on the distribution and availability of Sb, generally low mobility was also observed in mining and smelting site soils (Macgregor et al. 2015; Pérez-Sirvent et al. 2011). In studies on soils contaminated with both As and Sb, Ettler et al. (2010) found that in a smelter-polluted soil, the labile concentrations of Sb ranged from 0.6 - 4% and the residual fractions accounted for 65 - 94% of total Sb, while labile As was 1.2 - 22%. The distribution of As and Sb in different fractions of mining soils showed lower proportions of Sb in oxide phases compared with As and a very high fraction of Sb in the residual phase, which suggests that As could be potentially mobilised more easily than Sb and that Sb is strongly bound to the soil matrix and less geochemically mobile in the environment.

# 4.3.3. Labile As and Sb in soils assessed by DGT

The DGT-labile concentrations ( $C_{DGT}$ ) and total As and Sb concentrations in soil solutions ( $C_{sol}$ ) are presented in Tables 4.2 and 4.3. Generally, the DGT-labile and soil solution concentrations of As and Sb in soils increased with increasing total concentrations. The  $C_{DGT}$  and  $C_{sol}$  of As in the diluted cattle dip soils accounted for ~0.03% (3 – 120 µg/L) and ~0.1% (14.5 – 318 µg/L) of the total soil As, respectively, while those of As in the diluted

mining soils constituted only ~0.01% ( $2.12 - 34.9 \ \mu g/L$ ) and ~0.06% ( $13.2 - 136 \ \mu g/L$ ) of the total As. The C<sub>DGT</sub> and C<sub>sol</sub> of Sb in the diluted mining soils were only ~0.01% ( $1.3 - 57.9 \ \mu g/L$ ) and ~0.08% ( $12.3 - 484 \ \mu g/L$ ) of the total soil Sb. The low lability and ratio of dissolved to total concentration is consistent with the SEP results, which highlights a low relative proportion of both As and Sb present in the SEP labile fraction.

The R ratio of  $C_{DGT}$  and  $C_{sol}$  (0 < R < 1) was calculated to estimate the capacity of the soil solid phase to resupply As and Sb into soil pore water as described in section 3.3.3. The Rvalues (Tables 4.2, 4.3, and slopes from Figure B.1) for both As and Sb exhibited a relatively constant trend with increasing total soil As and Sb concentrations. This is consistent with the concentration range being obtained by mixing two soils – there seems to be little redistribution during this process. The average R-values for As in diluted cattle dip and mining soils were  $0.31 \pm 0.01$  and  $0.214 \pm 0.008$ , respectively, indicating partial resupply of As from the solid phases. A higher R-value for As from diluted cattle dip soils indicates a stronger resupply of As in this soil, consistent with the higher SEP-labile As in cattle dip soils compared with those in the mining soils. In contrast to As, the R-value of  $0.132 \pm 0.004$  for Sb suggests virtually no resupply from the soil solid phase. The lower Rvalue, regression slope, for Sb (0.12) compared to As (0.24) could be because of the stronger partitioning of Sb to the solid phase and more Sb in inert forms or organic complexes in the soil solution that were poorly labile and thus not measured by DGT. This indicates that As was resupplied more strongly than Sb from the solid phase, which was supported by the partitioning of As and Sb in various SEP-derived binding phases stating that As was more chemically labile than Sb, and that Sb was primarily bound to the soil matrix (residual phase).

It is interesting to note that the resupply of As in cattle dip soil was higher than that in the mining soil, despite no significant differences in total concentrations of As between the two soils. The greater lability and bioavailability of As in diluted cattle dip soils could be due to higher concentration of phosphates because phosphates can enhance the release of As from solid phases via competition for sorption sites (Cao & Ma 2004). It was also interesting that the resupply rate of As was higher than that of Sb within the mining soil, although the total soil As was less than total soil Sb. The lower lability of As and Sb in diluted mining soils

may also be due to the presence of sulphide minerals reported to be responsible for the immobilisation of As and Sb in mine waste sites (Álvarez-Ayuso et al. 2013; Kim et al. 2014).

# 4.3.4. Uptake of As and Sb by radish (long white icicle, *Raphanus sativus*)

# 4.3.4.1. Plant yield

The long white icicle radishes grown in cattle dip and mining soils had an average dry biomass of  $3.6 \pm 0.1$  g (n = 48), which was similar to controls and not affected (p>0.05) by the As and Sb exposure in the soils after the 35-day bioassay. These radish responses were consistent with the observation of cherry belle radishes (chapter 3).

# 4.3.4.2. Uptake trend

The accumulation of As and Sb in different white radish tissue compartments demonstrated a dose-dependent response and positive correlation with their total soil concentrations (Figure 4.2). Radish tissues (root flesh, peel, and shoot) accumulated As to greater concentrations than Sb even though total soil Sb concentrations were higher than total soil As, suggesting that As presents a greater risk than Sb in terms of food consumption/safety. However, bioaccumulation was low compared with the total As and Sb soil concentrations, which is in line with the low SEP and DGT labile As and Sb measurements. Tissue As displayed the trend of peel > shoot > flesh, while Sb in plant tissues followed a different trend of shoot > peel > flesh. The flesh As and Sb concentrations were consistently low, only accounting for 0.1 - 0.5% and 0.02 - 0.1% of their total soil concentrations, respectively, irrespective of soil concentrations and soil types. This trend was also observed by previous studies where the lowest As concentrations were in radish flesh (Bhatti et al. 2013) and peeled carrot (Codling et al. 2015; McBride 2013).



**Figure 4.2** The relationship between total soil concentrations and the accumulation of As and Sb in white radish tissues (mg/kg dry mass, mean  $\pm$  SE, n = 4) following cultivation in contaminated soils for 35 days. Shoots ( $\blacktriangle$ ), Peel ( $\Box$ ), and Flesh ( $\diamondsuit$ ). Figures A, B, and C represent As in cattle dip soils, As in mining soils, and Sb in mining soils, respectively.

In the taproot of radishes, the peel accumulated more As than the flesh, the differences between peel and flesh concentrations being more pronounced with increasing soil concentrations, a trend that has been previously observed in radishes (Bhatti et al. 2013; Warren et al. 2003) and carrots (Bhatti et al. 2013; Codling et al. 2015). The higher peel concentrations may be due to the direct contact between roots and contaminated soils, resulting in the possible adsorption of As on the unwashed iron layer on the outside of roots. Iron is known to be an important sorption phase in soils (Wilson et al. 2010) and can act as a barrier for metalloid uptake by plant roots (Tripathi et al. 2014). Another possibility is that the cells of the outer layer might accumulate As and Sb to a greater extent than other radish cells. Peeling appeared to significantly reduce As in roots, especially in highly As contaminated radishes, which is also supported by previous studies where peeling is recommended for carrots (McBride 2013) and potatoes (Codling et al. 2016). Sb also exhibited a higher concentration in the peel compared to the flesh, although not to the same degree as As. If iron plaque is the dominant factor responsible for increased concentrations of contaminants in the peel, Sb may not bind as strongly to this sorption phase compared to As, as supported by previous studies (Wilson et al. 2010).

For Sb accumulation by radishes cultivated in diluted mining soils, the total Sb in roots and shoots varied between 0.03 - 0.49 and 0.04 - 0.67 mg/kg, respectively, which were lower by a factor of 2.4 - 4.5 and 1.1 - 2.9 relative to As in roots and shoots for the same soil. The As concentrations in radishes grown in cattle dip soils were higher than those from mining soils. Arsenic concentrations accumulated in different compartments of radishes grown in diluted cattle dip and mining soils ranged from 0.13 - 4.27 and 0.13 - 1.40 mg/kg in roots (flesh plus peel); 0.11 - 3.54 and 0.10 - 0.56 mg/kg in shoots, respectively (Figure B.2). For radishes grown in the lowest concentration soils (S1 and Q1), where the control soil was dominant, As concentrations in roots and shoots of radish from both soils were the same. For radishes cultivated in soils of above 20 mg As/kg, As in roots and shoots from diluted mining soils, even though there were no significant differences in total As concentrations. This finding indicates that the bioavailable As concentrations in cattle dip soils were higher than those in mining soils, probably due to the greater phosphate concentration in cattle dip soils and the As being sources from pesticides (not measured) rather than mineralogy,

which is in agreement with the SEP and DGT-measured labile concentrations described in the previous sections.

Comparing As and Sb uptake by radishes grown in mining soils, both roots and shoots accumulated more As than Sb, despite total soil Sb concentrations being higher. Higher concentrations of As in radish compartments compared to Sb can be explained by the greater amounts of labile As than labile Sb in soils. These results confirm that total soil concentrations are not sufficient to evaluate the bioaccumulation in plants and the potential risk of contaminated soils, and that bioaccumulation is both plant and chemical species specific.

# 4.3.4.3. Bioaccumulation and translocation factors

The bioaccumulation factor (BAF) defined as the ratio of total concentration in plant roots and the total soil concentration (mg/kg DW) was calculated to evaluate the transfer of As and Sb from soil to radish (Table B.3). The BAF of As in radish roots from diluted cattle dip and mining soils ranged from 0.007 - 0.013 and 0.003 - 0.011, respectively, showing that the average uptake rate of As in the cattle dip soils was 1.6 times higher than the mining soils (Figure 4.3). The BAFs of As were much higher (3.3 - 5.8 times) than that of Sb in radish roots with values ranging from 0.0007 - 0.003. Lower Sb bioaccumulation compared with As has also been reported in the literature (Wilson et al. 2013; Wilson et al. 2014). A recent study on the biogeochemical behaviour of As and Sb in water, soil, and tailings in China also showed that the bioaccumulation factors of Sb were ~10 times less than those of As in plant/tailing and vegetable/soil systems (Fu et al. 2016).

The BAF values suggest that As was transferred from soils to roots more efficiently than Sb. The BAF for As and Sb in this study were lower than the BAF typically reported for As (~0.2) and for Sb (~0.01) reviewed from literature (Wilson et al. 2013). This means that the bioavailability of As and Sb to radish in these test soils was low and the radish studied may have had exclusion mechanisms for As and Sb. The low accumulation of As and Sb in the white icicle radish is consistent with the very low bioavailability of As and Sb (Tables 4.2 and 4.3).



**Figure 4.3** Bioaccumulation factors (BAF) and translocation factors (TF) for As and Sb of two cultivars of radish cultivated in various soils. Cherry bell radish – Mining ( $\blacksquare$ ) (chapter 3), White icicle radish – Mining ( $\Box$ ), White icicle radish – Cattle dip ( $\boxtimes$ ).

The translocation factor (TF) was obtained from the ratio of total concentrations of As and Sb in shoots compared to roots (mg/kg DW). The TF of As for radish from both diluted cattle dip and mining soils were < 1 with values varying between 0.37 - 0.9 and 0.35 - 0.77, respectively (Table B.3). The result illustrates that the translocation capacity is dependent on a given element and plant species, irrespective of soil types. The TF of Sb for radish grown in mining soils ranged from 0.85 - 1.69 being 1.3 - 3.3 times greater than for As. The limited translocation of As from roots to shoots for vegetables was also observed by Smith et al. (2009) and Bhatti et al. (2013). This may be a mechanism of As tolerance in plants, limiting the As movement to plant shoots by reducing root As(V) (the predominant As species in soils) to As(III) and sequestrating the reduced As in roots via phytochelatin (PC) complexation and stabilisation in vacuoles (Smith et al. 2008; Zhao et al. 2010).

The differences in As and Sb distribution in soil and radish compartments suggests different uptake mechanisms, possibly due to structural differences of the most common Sb and As species in soils, antimonate (Sb(V)) being octahedral and arsenate (As(V)) being tetrahedral (similar to phosphate) (Tschan et al., 2009). As a result, arsenate is taken up by roots via the phosphate transporter (Zhao et al., 2010). However, an analogous transport

path does not seem to occur for antimonate, with Tschan et al. (2008) showing that plant uptake was unaffected by phosphate. Nevertheless, the uptake mechanisms of Sb and the species accumulated by terrestrial plants and edible crops are still unknown (Feng et al., 2013; Tshan et al., 2009; Pierart et al., 2015). Thus, further studies on mechanisms of Sb uptake by a variety of plants are needed to fully understand Sb phytoavailability.

#### 4.3.4.4. Comparison of As and Sb uptake by two radish cultivars

With respect to As and Sb uptake by two cultivars of radish grown in soils prepared from the same parent contaminated soil, differences in uptake mechanisms were observed (Figure 4.3). On average, the bioaccumulation factor of As was much higher than that of Sb (7.5 times for cherry belle radish and 4.5 times for white icicle radish). Arsenic was somewhat restricted in the roots, but Sb was translocated from roots to shoots, illustrated by the higher translocation factor of Sb compared to As (2.5 times). The sequestration of As in roots was also observed in the white icicle radish cultivated in cattle dip soils.

In terms of cultivar behaviour, the bioaccumulation factors for As in both radish cultivars were not significantly different. The bioaccumulation factor for Sb in the white radish was 1.5 times higher than that in cherry belle radish, but not statistically significant (p>0.05). According to Barber (1995), the uptake of elements by plants is affected by the adsorption area of roots; the larger the adsorption area the higher the effective uptake. The surface area of the taproot in contact with the soils is larger for the white icicle radish. However, significantly higher bioaccumulation factors for As and Sb for the white icicle radish were not evident, indicating the possibility that uptake is through the minor roots. The uptake of the elements by radishes would occur at the apical region, root tips rather than over the entire root surfaces.

Regarding translocation characteristics, the cherry belle radish had a higher capacity to translocate As and Sb from roots to shoots than the white icicle radish by a factor of  $2.5 \pm 0.2$  and  $2.8 \pm 0.4$ , respectively. This is most likely due to the greater mean distance required to transport elements for the white icicle radish. The white icicle radish has a very long root, requiring more energy to move elements inside the xylem and plant organs compared to the small bulb of the cherry belle radish. Thus, the white icicle radish had lower efficiency to translocate As and Sb from roots to shoots.

These results confirm that the behaviour of As and Sb in relation to plant uptake was different and that the uptake and distribution trend are dependent on a given element and cultivar, irrespective of contamination sources. Taking the results of bioaccumulation and translocation of As into consideration, both the BAF and TF for As were low (<1) and therefore the white icicle radish can be considered to be an As tolerant plant (Vamerali et al. 2010).

# **4.3.5.** Evaluating the effectiveness of geochemical measurements as predictors of biological uptake

The relationships between As and Sb concentrations in roots and shoots of the white icicle radish and their labile fractions in soils obtained from SEP, DGT, soil solution measurements are presented in Figures 4.4 and 4.5. The As and Sb concentrations in plant tissues were highly positively correlated with all labile As and Sb fractions ( $R^2 = 0.87 - 0.97$  for As in cattle dip soils,  $R^2 = 0.93 - 0.99$  for As in mining soils, and  $R^2 = 0.92 - 0.98$  for Sb in mining soils), which indicates that tissue As and Sb were well predicted by these labile measures. The close relationships between SEP-labile As and Sb and tissue As and Sb indicates that the non-specifically sorbed and specifically sorbed As and Sb are the best predictors of the phytoavailability of As and Sb. The close relationships between plant As and SEP-labile As in this study are supported by the results of Gonzaga et al. (2008) stating that the total of the first two fractions from SEP was closely correlated with As uptake by the *P. vittata* (Chinese brake fern) and Niazi et al. (2011) reporting a positive correlation between exchangeable As from the first extraction and As in *B. juncea* (Indian mustard) shoots.

Arsenic and Sb uptake by the white icicle radish from the test soils had a stronger relationship with DGT-labile As and Sb than their soil solution concentrations, especially from cattle dip soils, which is demonstrated by the higher correlation coefficients for  $C_{DGT}$ . This suggests that there was some resupply of As and Sb from the solid phase or from the dissolved complexes of As and Sb in soil solution to sustain the depletion of As and Sb in the rhizophere. From the predictive perspective, DGT exhibited excellent performance in predicting the phytoavailability of As and Sb in the studied soils, compared to soil solution concentration method.



**Figure 4.4** The relationship between tissue As and Sb (mg/kg, dry mass) and bioavailable fractions of As and Sb in bioassay mining soils, as measured by SEP, DGT, and soil solution. Shoots ( $\blacktriangle$ ), Roots (o). All data have error bars; where they are not seen they are smaller than the symbols.



**Figure 4.5** The relationship between tissue As (mg/kg, dry mass) and bioavailable fractions of As in bioassay cattle dip soils, as measured by SEP, DGT, and soil solution. Shoots ( $\blacktriangle$ ), Roots (o). All data have error bars; where they are not seen they are smaller than the symbols.

The good predictors of DGT are also supported by previous studies showing significantly strong correlations between As concentrations derived from DGT and As concentrations in edible rape from 43 contaminated soils (Wang et al. 2014) and As in rice grain from 39 contaminated soils (Williams et al. 2011). DGT-measured concentrations of Ni from low and high Ni treated soils were also strongly correlated with Ni concentrations in radish (*R. sativus*) grown in those soils (Luo et al. 2014). Recently, Dočekalová et al. (2015) found high correlations between flux to DGT and flux to radish (*R. sativus*) for Cu and Cd.

Better relationships were found for the prediction of As and Sb in roots than in shoots by all bioavailable fractions obtained from SEP, DGT, and soil solution, except the  $C_{DGT-Sb}$  and

 $C_{sol-Sb}$ . This is expected because roots may be better indicators for bioavailable metals in soils due to the direct contact with soils, while the metal accumulation by shoots is determined by the translocation processes and governed by plant physiology. This pattern is supported by Nowack et al. (2004) who stated that shoots are not good indicators for bioavailable metals.

Considering As and Sb uptake by two cultivars of *R. sativus* grown in soils, labile measurements were better predictors for As and Sb uptake for cherry belle radish (chapter 3) (greater correlation coefficients) than white icicle radish, with the exception for root Sb predicted by SEP and root As predicted by DGT. This indicates that the capacity of these techniques to predict element uptake by plants is good, but depends on cultivars.

#### 4.4. Conclusions

This study showed that As and Sb exhibited contrasting behaviour in soils and white icicle radish (*R. sativus*). Sb was most likely dominantly associated with resistant minerals, silicates, and organic matter, while As was mainly associated with Fe oxides, irrespective of contamination sources. Sb and As bound to these phases were poorly mobile and not DGT-labile. Thus, the bioavailable fractions of As and Sb in historically contaminated soils were low in which Sb was geochemically much less labile than As. The DGT measurements also identified that regardless of soil types, As was partially sustained and more labile than Sb which had virtually no resupply from the solid phase to soil solution. For plant uptake, As was accumulated in the plant tissues more than Sb, but translocated less to shoots than Sb. The white icicle radish exhibited the same uptake trend as the cherry belle radish in our previous study, but was less able to translocate As and Sb from roots to shoots than the cherry belle cultivar. These results suggest that the *R. sativus* uptake was clearly element and cultivar-specific.

This study demonstrated that SEP, DGT, and soil solution were good approaches in measuring the bioavailable As and Sb in historically contaminated soils and predicting their uptake by root vegetables, in which DGT had better performance than soil solution concentration method. Further studies are needed to identify if these observations are consistent with leafy vegetables and other soil types.

# Chapter 5. Assessment of arsenic and antimony lability in historically co-contaminated soils and their uptake by water spinach (*Ipomoea aquatica*)

# **5.1. Introduction**

The lability and mobility of As and Sb is governed by oxides of Fe, Al, Mn, clay, and organic matter (OM) via adsorption which is a pH-dependent process. These adsorbents are efficient in the immobilisation of As and Sb in mining soils, especially soils with lower pH and higher redox values (García-Sanchez et al. 2002; Juhasz et al. 2008; Mitsunobu et al. 2006). The availability of As can be reduced when oxide concentrations of Fe, Al, Mn, and OM increase (De Oliveira et al. 2015). The retention of antimonate by OM has also been observed in recent years (Clemente et al. 2008; Dousova et al. 2015), with humic acids having a high capacity to complex and bind antimonate trapped in the soil organic layer in a Sb-contaminated shooting range (Steely et al. 2007).

The lability of metal(loid)s is closely related to their accumulation in many plants (McLaughlin et al. 2011). To our knowledge, most of the research on metal(loid) uptake by plants has been conducted with freshly contaminated or spiked soils. However, this approach has some limitations since the system is unlikely to be in equilibrium. The distribution of metal(loid)s in spiked soils changes from more labile to less labile forms with time and metal(loid)s are more labile in spiked soils than in field contaminated soils (Liang et al. 2014), leading to an overestimate of their toxicity in spiked soils. Thus, the assessment of risks posed by historic contamination is important. In addition, these investigations have focused on responses to a single contaminant, despite very few environments containing only one contaminant. Thus, research conducted with a historically contaminated soil containing a mixture of contaminants is realistic, ecologically and environmentally relevant, and important for reliable risk assessment.

To study the lability of As and Sb and to determine their phytoavailable fractions in contaminated soils, sequential extraction and diffusive gradients in thin films (DGT) have been applied. Recently, the combination of sequential extraction and DGT was successful

in assessing labile As and Sb from contaminated soils to two cultivars of radish (*Raphanus sativus*) (chapters 3 and 4). The performance of DGT in predicting As and Sb uptake by leafy vegetables has not been conducted.

Water spinach (*Ipomoea aquatica*) is used as a popular edible leafy vegetable and herbal plant in some regions of the world, especially Asia. It has fibrous roots, hollow stems, and wide arrowhead shaped leaves. The soft stems and leafy portions are edible and rich in important vitamins, minerals, nutrients, and fiber, which is popular for daily food (Göthberg et al. 2002). *I. aquatica* has been shown to be a Cd, Cu, Pb, and Cr accumulator plant (Bedabati Chanu & Gupta 2016; Göthberg et al. 2002; Weerasinghe et al. 2008). The uptake of As and Sb by *I. aquatica* and the risks associated with consumption of *I. aquatica* grown on historically contaminated sites have not been evaluated adequately.

In this study a combination of selective DGT and sequential extraction measurements was applied to assess the lability of As and Sb in a historically co-contaminated soil, and compared to their uptake by water spinach (*Ipomoea aquatica*), a popular edible leafy vegetable in some regions of the world, especially Asia. The aims were to investigate (1) the biogeochemical behaviour of As and Sb in a soil series prepared by mixing a soil historically contaminated with both As and Sb and three uncontaminated soils with low iron (L), medium iron (M), and high iron (H); (2) the bioaccumulation of As and Sb in *I. aquatica* grown in these test soils; (3) the ability of sequential extraction and DGT technique to predict As and Sb uptake by *I. aquatica*; and (4) the risks to human health due to consumption of As-enriched edible shoots of *I. aquatica*. This study will provide a better understanding of the behaviour of As and Sb in historically co-contaminated soils and determine the extent of As and Sb accumulation by water spinach (*Ipomoea aquatica*).

#### 5.2. Methods

# **5.2.1. General methods**

General washing methods were done as per section 2.1.

# 5.2.2. Experimental design

Soil contaminated with both As and Sb was collected from a decommissioned antimony processing facility in Urunga NSW, Australia. Details about soil compositions and physical

properties were described previously (Ngo et al. 2016). Uncontaminated soils were obtained from three sites in Wollongong, NSW, with soil-L having the lowest iron and highest pH values, soil-M containing medium iron content, and soil-H having high iron content. All soils were air-dried and crushed to pass through a 2 mm sieve. The contaminated soil was thoroughly mixed with each uncontaminated soil (control soil) at different ratios from 1:150 to 1:0.5 to establish three treatment series of test soils with increasing As and Sb concentrations. Soil moisture was raised to and maintained at ~60% of the water holding capacity (WHC) with deionised water. Soils were allowed to settle and equilibrate in a greenhouse for two weeks prior to the commencement of bioassays.

The 35-day bioassays with *I. aquatica* were conducted in controlled growth chambers as per section 2.4.2. Harvested plant tissues from the same pot were combined, weighed, and oven-dried at 70 °C for 48 h to a constant weight with the dry mass (g) recorded. Dried tissues were ground with a mortar and pestle and stored in clean plastic bags prior to As and Sb analyses.

# 5.2.3. Soil characterisation

Soil properties including particle size, pH, organic matter, total nitrogen, available phosphorus, and amorphous and free Fe and Al were determined as described in section 2.5.

# 5.2.4. Soil and plant analyses

Finely ground soil (~0.2 – 0.3 g, dry mass) was digested with aqua regia in a closed microwave digestion system at 120 °C (10 min ramp, 15 min hold) and then 190 °C (5 min ramp, 15 min hold). Finely ground plant (~0.2 – 0.3 g, dry mass) was microwave digested with a mixture of HNO<sub>3</sub>:H<sub>2</sub>O<sub>2</sub> (1:2, v/v) (65% and 30%, respectively) at 110 °C (10 min ramp, 15 min hold) and then 180 °C (5 min ramp, 15 min hold). After cooling to room temperature, digests were filtered (<0.45  $\mu$ m) and diluted to 2% acid with deionised water for analyses. Total As and Sb in soils and plants were analysed by ICP-MS as described in section 2.9.

For quality control, a combination of reagent blanks, duplicate analyses, certified reference materials, and matrix spike recovery were used. A soil certified reference material

(Montana II soil, SRM 2711a; NIST) returned recoveries of 90-115% for the studied analytes. A plant certified reference material (Tomato leaves, SRM 1573a; NIST) had typical recoveries of 95-110% for the studied analytes.

#### **5.2.5. Sequential extraction procedure (SEP)**

The procedure of Wenzel et al. (2001) was used to determine As and Sb associated with non-specifically sorbed, specifically sorbed, amorphous Fe and Al oxide, crystalline Fe and Al oxide, and residual fractions as described in section 2.6.1.

#### 5.2.6. DGT and soil solution measurements

DGT and soil solution measurements were conducted as described in sections 2.6.2 and 2.7. The DIFS (DGT induced fluxes in soils) model was applied to predict (de)sorption kinetic parameters for soils (Ernstberger et al. 2002; Ernstberger et al. 2005; Harper et al. 2000; Zhang et al. 2004). The reponse time ( $T_c$ ) of desorption process was obtained using the distribution coefficient ( $K_d$ ) and R as input parameters. The rate constant of resupply from the solid phase ( $k_{-1}$ ) was calculated from  $K_d$ ,  $P_c$  (soil particle concentration), and  $T_c$  using Equation 5.1.

$$k_{-1} = 1/((1 + K_d P_c)T_c)$$
 Equation 5.1

#### **5.2.7. Data analysis**

Statistical analysis and calculations of bioaccumulation factor (BAF) and translocation factor (TF) were performed as per section 3.2.7. To assess the risk associated with the As and Sb calculated exposure, the average exposure was compared with the toxicological reference dose. Both non-carcinogenic and carcinogenic risks to humans were evaluated based on the formula obtained from Environmental Health Australia - enHealth (2012) and USEPA (2011). The risk of non-carcinogenic effect was assessed by the hazard quotient (HQ) (Equation 5.2) being the ratio of estimated daily intake (EDI) to the reference dose (RfD).

$$HQ = EDI/RfD$$
 Equation 5.2

where EDI and RfD represent the estimated daily intake of As/Sb (mg/kg/day) and the reference dose for As/Sb, respectively. The RfD for As and Sb are 0.0003 mg/kg/day and

0.0004 mg/kg/day suggested by USEPA (1991, 1998) and applied previously (Khan et al. 2014; Li et al. 2014; Rehman et al. 2016). An HQ < 1 defines no potential risk to the exposed humans (Environmental Health Australia - enHealth 2012; USEPA 1998, 2011). The EDI was calculated according to Equation 5.3.

$$EDI = (C \times IR \times EF \times ED)/(BW \times LE)$$
 Equation 5.3

where C is the concentration of As in edible parts (shoots) of the plant (mg/kg wet weight, WW); IR is the ingestion rate of edible part (kg/person/day); EF is exposure frequency (days/year); ED is exposure duration (years, typically 70 years for an individual); BW is body weight (kg, 60 for adult and 24.5 for child); LE is life expectancy (days, 25550) (Rehman et al. 2016). The cancer risk (CR) was calculated by the multiplication of the estimated daily intake of As (mg/kg/day) to the cancer slope factor (CSF) according to Equation 5.4.

# $CR = EDI \times CSF$ Equation 5.4

The CSF (mg/kg/day)<sup>-1</sup> for As for adult and child have been set to 1.5 and 4.5, respectively (USEPA 1998, 2011). The CR is defined as the probability of an individual suffering cancer risk over a lifetime. For instance, a CR of  $1 \times 10^{-4}$  indicates a probability of 1 in 10,000 individuals developing cancer. A range of 1 in 10,000 to 1 in 1,000,000 is considered acceptable As cancer risk (Environmental Health Australia - enHealth 2012; Uddh-Söderberg et al. 2015; USEPA 1998, 2011).

#### 5.3. Results and discussion

#### 5.3.1. Soil characteristics

The average physicochemical characteristics of each series of test soils are presented in Table 5.1. The soils were all sandy-silts that were slightly acidic, with the pH of soil-L being 1 unit higher than the other two soils. In terms of nutrients (N and P), soil-L was higher in P and lower in N compared to soil-M and soil-H. The capacity of soil to retain As and Sb is highly correlated with oxides of Fe and Al, and organic matter (OM). The soils were characterised by their mineral contents, with soil-L being low in Fe and Al, soil-M having medium Fe and Al, and soil-H having the highest amounts of Fe and Al. There was a similar trend in OM and dissolved organic carbon (DOC) with soil-L lower in OM and

DOC than soil-M and soil-H. Based on their characteristics, soil-L is expected to have the highest available As and Sb because of low Fe, Al, OM and high pH, followed by soil-M and soil-H.

The prepared soils covered a wide range of total contaminant concentrations, from low to very high (42 - 4180 mg As/kg and 52 - 6610 mg Sb/kg, Table 5.2), which were all above average concentrations in non-contaminated soils (<10 mg/kg), but within the range of total As and Sb found in mining and smelting-impacted soils around the world (Wilson et al. 2010). This allowed the availability of As and Sb and their uptake by plants under various conditions of soil properties and total metalloid concentrations to be assessed. The high soil concentrations of As and Sb prepared were based on our pilot study showing that *I. aquatica* grown in highly co-contaminated soils (>1000 mg/kg of As and Sb) did not exhibit growth reduction and toxicity symptoms. Thus, the largest concentrations of each soil type were estimated to be high enough for the plant growth reduction and toxicity to be detectable.

Characteristic	Soil-L (S1 – S5)	Soil-M (S6 - S10)	Soil-H (S11 - S15)
Sand (%)	30 ± 3	31 ± 1	33 ± 3
Silt (%)	$63 \pm 2$	$61.0\pm0.5$	$60 \pm 3$
Clay (%)	$7.0\pm0.5$	$8.0 \pm 0.4$	$7.0 \pm 0.4$
pH	$6.5 \pm 0.2$	$5.1 \pm 0.2$	$5.3 \pm 0.2$
Organic matter (%)	$6.75\pm0.05$	$7.3 \pm 0.3$	$7.6 \pm 0.3$
Extractable Phosphorous (mg/kg)	$50 \pm 20$	$34 \pm 4$	$30 \pm 9$
Total Kjeldahl Nitrogen (mg/kg)	$120 \pm 30$	$280 \pm 60$	$240\pm20$
Amorphous Fe (g/kg)	$7.4 \pm 0.3$	$7.4 \pm 0.2$	$7.3 \pm 0.4$
Amorphous Al (g/kg)	$0.55\pm0.06$	$1.79\pm0.09$	$2.2 \pm 0.1$
Free Fe (g/kg)	$18.2\pm0.6$	$25.0\pm0.6$	$30 \pm 1$
Free Al (g/kg)	$1.01\pm0.08$	$2.0 \pm 0.1$	$3.1 \pm 0.2$

**Table 5.1** Physical and chemical properties of bioassay soils (dry mass, mean  $\pm$  SE, n $\geq$ 3).

**Table 5.2** Total concentrations in a concentration gradient obtained from historically contaminated soils, R values,  $K_d$  (distribution coefficients, L/kg), and kinetic parameters including  $T_c$  (the response time of the desorption process, s), and  $k_{-1}$  (the desorption rate constant, s<sup>-1</sup>) obtained from DIFS of As and Sb (mean ± SE, n = 3).

Soil	Treatment			As			Sb				
		Total <sub>As</sub> (mg/kg)	Ratio R <sub>As</sub> =C <sub>DGT</sub> /C <sub>sol</sub>	K <sub>d</sub> (L/kg)	T <sub>c</sub> (s)	$\frac{k_{-1}}{(10^{-6} s^{-1})}$	Total <sub>sb</sub> (mg/kg)	Ratio R <sub>Sb</sub> =C <sub>DGT</sub> /C <sub>sol</sub>	K <sub>d</sub> (L/kg)	T <sub>c</sub> (s)	$k_{-1}$ (10 <sup>-6</sup> s <sup>-1</sup> )
L	<b>S</b> 1	$42.1\pm0.2$	$0.65\pm0.01$	659	88	7.46	$52\pm 6$	$0.1173 \pm 0.0003$	17	11170	2.76
	S2	$160 \pm 30$	$0.57\pm0.06$	707	182	3.48	$200\pm30$	$0.20\pm0.02$	24	2473	7.49
	<b>S</b> 3	$270\pm9$	$0.60\pm0.02$	1010	149	2.87	$330\pm30$	$0.189 \pm 0.001$	28	3472	4.35
	S4	$670\pm100$	$0.682\pm0.001$	830	77	6.86	$870 \pm 170$	$0.21\pm0.01$	32	2138	6.37
	<b>S</b> 5	$1400\pm50$	$0.609\pm0.002$	419	109	9.46	$2277 \pm 130$	$0.214\pm0.001$	35	2130	5.69
М	S6	$65\pm3$	$0.36 \pm 0.04$	486	820	1.12	$90.7\pm0.6$	$0.137 \pm 0.006$	20	7877	2.78
	<b>S</b> 7	$176\pm2$	$0.46\pm0.03$	569	376	2.11	$220\pm30$	$0.166\pm0.006$	28	5125	3.10
	<b>S</b> 8	$330\pm20$	$0.32\pm0.01$	327	1112	1.23	$530\pm180$	$0.187\pm0.004$	33	3827	3.53
	S9	$950\pm50$	$0.40\pm0.03$	295	577	2.65	$1190\pm60$	$0.189 \pm 0.008$	40	3872	2.92
	S10	$2630\pm90$	-	-	-	-	$3790 \pm 180$	-	-	-	-
Н	S11	$101\pm 8$	$0.39\pm0.03$	639	595	1.14	140 ± 8	$0.143 \pm 0.008$	24	6975	2.55
	S12	$200\pm30$	$0.41\pm0.03$	498	512	1.64	$230\pm30$	$0.172\pm0.007$	26	4186	3.85
	S13	$510\pm30$	$0.38\pm0.01$	479	614	1.43	$670\pm70$	$0.183 \pm 0.005$	39	3817	2.79
	S14	$1280\pm40$	$0.48\pm0.01$	351	256	4.28	$1600\pm60$	$0.22\pm0.01$	41	1720	5.42
	S15	$4180\pm160$	-	-	-	-	$6610\pm460$	-	-	-	-

-: not measured due to analysis problems

#### **5.3.2. Sequential extraction**

The concentrations and proportions of As and Sb associated with different soil binding phases are presented in Table C.1 and Figure 5.1. Generally, the amounts of As and Sb extracted in each fraction increased with increasing soil concentrations. In all three soil types, As was mainly bound to the less labile fractions of the soil, amorphous Fe and Al oxides (38-45%), followed by crystalline Fe and Al oxides (33-40%). The residual As accounted for 10 - 24% of total As in all three soils. The available As (sum of non-specifically sorbed and specifically sorbed fractions) in soil-L (8.47% of total As) was significantly higher (p<0.05) than soil-M and soil-H (6.68 and 5.93%, respectively). Although the proportion of available As was quite low, it plays an important role in soil biogeochemical processes since they are the most mobile forms of As, especially in soils with high total As concentrations (Wenzel et al. 2001).



**Figure 5.1** The percentage of As and Sb associated with different binding fractions for bioassay soils (mean of all concentrations for each soil, mean  $\pm$  SE, n = 15).

Unlike As, Sb was primarily found in the residual fraction, representing 59-64% of total soil Sb, most likely as stibnite, which is the mineral typically mined (Álvarez-Ayuso et al. 2013). The Sb associated with crystalline (26-29% of total Sb) and amorphous (9.6-11.7%) Fe and Al oxides were the second and third most dominant phases. The high amount of Sb in the last three SEP fractions and the similar proportion within each fraction among all soil types (Figure 5.1) indicates that the partitioning of Sb in these fractions was not affected by soil Fe, Al, and organic matter from the different diluent control soils. The available Sb in soil-L (0.91%) was significantly higher (p<0.05) than soil-M (0.51%) and soil-H (0.41%). The very low proportion of Sb in available fraction (<1%) indicates that Sb was biogeochemically immobile and strongly bound to the soil matrix at all concentrations.

It is clear that both the available As and Sb in soil-L were significantly higher (p<0.05) than soil-M ~ soil-H, in which the As availability was much higher than that of Sb, even though total soil As was lower than total soil Sb. As expected, soil-L with the lowest concentrations of Fe, Al, and organic matter and the highest pH consequently had the highest available As and Sb. Soil-H with the highest amounts of Fe, Al, and organic matter and the lower pH had the lowest availability of As and Sb. This can be explained by the literature stating that As and Sb in soils were significantly sorbed by oxides of Fe and Al, organic matter under acidic conditions. This adsorption was reduced with increasing pH (Wilson et al. 2010). For example, in four Australian soils with pH ranging from 4.97 to 6.9, arsenate sorption was greater than arsenite and decreased as pH increased (Smith and Huyck 1999, cited in Wilson et al. 2010). The sorption of arsenate (Thanabalasingam & Pickering 1986) and antimonate (Tighe et al. 2005b) by humic acid was significant at around pH 5.5 and 4, respectively. Soils with a pH above 6 have less protonation of phenolic, carboxylic, and hydroxyl-carboxylic groups on particulate surfaces and a greater negative charge density on mineral surfaces, leading to the repulsion of As and Sb from binding sites and the release of As and Sb oxyanions (De Oliveira et al. 2015; Dousova et al. 2015). In addition, the greater phosphorous in soil-L may increase available As in soil solution due to competition of phosphate and arsenate binding on soil surfaces. The effect of phosphate on available Sb in soil is still unclear.

These observations are supported by previous studies on the bioavailability and fractionation of As in the whole soils. For example, Gonzaga et al. (2008) found that the amount of available As in a mining soil with 1626 mg Fe/kg was about six times lower than in a soil containing 509 mg Fe/kg. Juhasz et al. (2008) reported that soil with higher Fe and lower pH reduced As availability by 75% upon 12 months of aging. Wang et al. (2015) also stated that soil with pH 5.39 and total free Fe of 10.3 mg/kg had higher labile As fractions initially and one-year after spiking with 100 mgAs/kg than soil with pH 4.94 and total free Fe of 38.4 mg/kg.

The differences between As and Sb chemical fractionation and lability may be related to the differences in their pentavalent oxyanions structures, where arsenate is tetrahedral whereas antimonate is octahedral (Okkenhaug et al. 2012; Wilson et al. 2010). Arsenate has higher charge density and a smaller ionic radius (0.40), while antimonate has lower charge density and larger ionic radius (0.51), which results in a larger spatial structure for antimonate. As a result, As was more efficiently bound to active binding sites (e.g. metal oxides), explaining the higher proportions of As in Fe oxide fractions compared to Sb. The mineral solubility of As and Sb may also explain the differences in their lability. In mining soils, antimonate is primarily calcium antimonate ( $K_{sp}$  10<sup>-12.55</sup>), while arsenate are mainly iron and aluminium arsenate (K<sub>sp</sub> 10<sup>-11</sup>) and calcium arsenate (K<sub>sp</sub> 10<sup>-5</sup>). Antimonite is possibly present as stibuite (K<sub>sp</sub>  $1.6 \times 10^{-93}$ ), whereas arsenite is possibly present as orpiment, As<sub>2</sub>S<sub>3</sub>  $(K_{sp} 8.4 \times 10^{-16})$  (Arai 2010; Okkenhaug et al. 2011). The minerals of Sb have lower solubilities, which would contribute to the lower Sb availability. The fact that As and Sb existed together in the test soils may lead to an interaction and competition for binding sites which differed from previous studies based on single-contaminated soil. The mechanisms for As-Sb competition for soil binding sites are still unknown.

# 5.3.3. DGT measurements

The total soil concentrations, soil solution concentrations ( $C_{sol}$ ), DGT-labile concentrations ( $C_{DGT}$ ), and R values for As and Sb are presented in Tables C.2 and 5.2.  $C_{sol}$  and  $C_{DGT}$  of As and Sb increased with increasing total soil concentrations, but accounted for only a small proportion of total soil concentrations (0.004 - 0.02% for As, 0.002 - 0.04% for Sb). The soils had a low lability of As and Sb, possibly due to the aging of the parent

contaminated soil and the stability of complexes of As and Sb and solid phases of soils (Liang et al. 2014). As discussed in section 3.3.3, the R value ( $R=C_{DGT}/C_{sol}$ , 0 < R < 1) reflects the ability of soils to resupply a solute from the solid phase to soil solution; the higher the R value, the greater the resupply rate. The average R values for As for three soil types ranged from 0.38 – 0.62, indicating that As was partially resupplied from soils but was not sufficient to sustain constant pore water concentration. The average R value of soil-L ( $0.62 \pm 0.02$ ) was significantly higher (p<0.05) than soil-M ( $0.38 \pm 0.03$ ) and soil-H ( $0.41 \pm 0.02$ ) (Table 5.2). The highest resupply of As in soil-L probably resulted from its lowest Fe, DOC, and highest pH because the desorption of As may be favoured under these conditions as discussed in section 5.3.2. According to Williams et al. (2011), DOC and DOC-Fe complexes can bind As and reduce the dissolved-As concentration. Thus, higher Fe and DOC in soil-M and soil-L (Table 5.1) possibly led to more As in complex forms with DOC or DOC-Fe complexes, which would be unavailable for DGT. The resupply of As followed the trend of soil-L > soil-M ~ soil-H, which is consistent with the trend of C<sub>SEP-lable As</sub>.

For Sb, the average R values among soil types were relatively constant, ranging from 0.17-0.19. This means that the resupply of Sb from the solid phases to soil solution was lower than that of As, despite total soil Sb concentrations being higher than that of As. The relatively unchanged R values of Sb across soil types (varying soil properties) also support the conclusion that Sb was less labile and strongly bound to the solid phase.

DGT induced fluxes in soils (DIFS) was used to simulate the dynamic characteristics of As and Sb in soils using R and K<sub>d</sub> (the distribution coefficients). The values of T<sub>c</sub> (the response time of the resupply process) derived from DIFS model and K<sub>d</sub> were used to calculate desorption rate constant (k<sub>-1</sub>) as per Harper et al. (2000), Zhang et al. (2004), and Ernstberger et al. (2005). The calculated values of K<sub>d</sub>, T<sub>c</sub>, and k<sub>-1</sub> are presented in Table 5.2. On average, for As, soil-L had significantly higher values of K<sub>d</sub>, k<sub>-1</sub> and lower values of T<sub>c</sub> (p<0.05) than soil-M and soil-H. There were no significant differences in these average values between soil-M and soil-H. This indicates that soil-L had a higher labile pool of As in the solid phase, a higher rate constant of the As resupply process from the solid phase to soil solution, and shorter response time to the depletion of As than the other two soils. Consequently, the release potential and kinetics of available As from the solid phase was highest in soil-L, followed by soil-M and soil-H. This is in agreement with the highest values of  $R_{As}$  and greatest  $C_{SEP-labile As}$  in soil-M. For Sb, the average values of  $K_d$ , k. 1, and  $T_c$  in three soils were not significantly different (p>0.05), which is consistent with the trend of R values. This suggests that the kinetic release of Sb was not affected by soil types. Compared to As, Sb had significantly lower  $K_d$  and  $k_{-1}$  and higher  $T_c$  (p<0.05), suggesting the lower release and resupply of Sb from the solid phase to soil solution. This agrees with the lower R values for Sb and  $C_{SEP-labile Sb}$ .

The resupply of As and Sb were consistent with the previous studies on prediction of bioavailable of As and Sb in historically contaminated soils to plants (chapter 3 and 4) where As was partially resupplied, and the replenishment of Sb was minimal. Arsenic and Sb co-existed in test soils, thus competitive interactions for complex ligands in soil solutions and soil binding sites may occur, and this should be considered when comparing these results with single element studies.

# 5.3.4. Uptake of As and Sb by water spinach (*Ipomoea aquatica*)

#### 5.3.4.1. Plant yield

The *I. aquatica* dried tissue masses are presented in Table 5.3. The root and shoot dry mass varied widely with increasing soil concentrations, irrespective of soil types, ranging from 0.28 - 0.9 g roots/pot and 0.4 - 4.5 g shoots/pot. The most significant decrease (size and area of leaves, numbers of leaves, root length and shoot height) occurred at the highest treatment of each soil type, but toxicity symptoms such as chlorosis and wilting were not observed during plant growth. *I. aquatica* responses in this study are supported by Shaibur et al. (2009), who observed the dramatic reduction of root and shoot length, and leaf area of *I. aquatica* grown hydroponically at 50  $\mu$ M As treatment (equivalent to 3750  $\mu$ g As/L).

Soil	Treatment	As				Sb					Shoot
		Root As (mg/kg)	Shoot As (mg/kg)	BAF - As	TF - As	Root Sb (mg/kg)	Shoot Sb (mg/kg)	BAF - Sb	TF - Sb	Dry weight (g)	Dry weight (g)
L	<b>S</b> 1	$40\pm2$	$3.8\pm0.4$	$0.96\ \pm 0.04$	$0.094 \pm 0.007$	$0.84\pm0.09$	$0.66\pm0.03$	$0.016\pm0.002$	$0.8\pm0.1$	$0.6\pm0.1$	$4.4\pm0.9$
	S2	$107\pm8$	$11.7\pm0.6$	$0.66\ \pm 0.05$	$0.11\pm0.01$	$3.0 \pm 0.2$	$0.5\pm0.1$	$0.015\pm0.001$	$0.15\pm0.05$	$0.55\pm0.09$	$3.1\pm0.5$
	S3	$360\pm40$	$42\pm 4$	$1.4\ \pm 0.2$	$0.12\pm0.03$	$4.6\pm0.5$	$1.8\pm0.6$	$0.014\pm0.001$	$0.4 \pm 0.1$	$0.9\pm0.1$	$4.5\pm0.3$
	S4	$700\pm100$	$51 \pm 4$	$1.0\ \pm 0.2$	$0.09\pm0.02$	$15\pm3$	$1.32\pm0.04$	$0.018\pm0.003$	$0.09\pm0.01$	$0.6 \pm 0.1$	$2.0\pm0.3$
	S5	$2600\pm200$	$126 \pm 4$	$1.8\ \pm 0.1$	$0.049\pm0.003$	$80\pm20$	$10 \pm 2$	$0.035\pm0.008$	$0.14\pm0.03$	$0.28\pm0.08$	$1.41\pm0.05$
М	S6	41 ± 2	$4.4\pm0.2$	$0.63\ \pm 0.03$	$0.107 \pm 0.005$	$1.5\pm0.1$	$0.4 \pm 0.1$	$0.017 \pm 0.001$	$0.29\pm0.06$	$0.83\pm0.08$	$3.3 \pm 0.4$
	S7	$170 \pm 4$	$16\pm3$	$0.96\ \pm 0.02$	$0.09\pm0.02$	$4.0\pm0.2$	$0.35\pm0.02$	$0.018\pm0.001$	$0.09\pm0.01$	$0.9\pm0.2$	$3.9\pm0.8$
	S8	$610\pm10$	$36\pm 6$	$1.84\ \pm 0.03$	$0.06\pm0.01$	$11 \pm 1$	$1.5\pm0.2$	$0.021\pm0.002$	$0.13\pm0.01$	$0.57\pm0.05$	$2.8\pm0.2$
	S9	$1500\pm100$	$73\pm4$	$1.56\ \pm 0.08$	$0.050\pm0.004$	$41\pm 5$	$2.3\pm0.2$	$0.035\pm0.004$	$0.06\pm0.01$	$0.40\pm0.05$	$1.3 \pm 0.1$
	S10	-	-	-	-	-	-	-	-	-	-
Н	S11	$102 \pm 6$	$5.6\pm0.8$	$1.01 \ \pm 0.06$	$0.054\pm0.005$	$3.43\pm0.07$	$0.29\pm0.02$	$0.0238 \pm 0.0005$	$0.09 \pm 0.01$	$0.90\pm0.09$	$3.9\pm0.7$
	S12	$240\pm30$	$15\pm2$	$1.2\ \pm 0.2$	$0.063\pm0.005$	$5.7\pm0.4$	$0.54\pm0.04$	$0.025\pm0.002$	$0.10\pm0.01$	$0.621 \pm 0.001$	$3.0\pm0.5$
	S13	$580\pm100$	$47\pm2$	$1.1\ \pm 0.2$	$0.09\pm0.02$	$11.6\pm0.6$	$1.15\pm0.04$	$0.017\pm0.001$	$0.100\pm0.002$	$0.7\pm0.1$	$2.9\pm0.6$
	S14	$1860\pm200$	$89\pm 6$	$1.5\ \pm 0.1$	$0.048\pm0.002$	$51\pm4$	$1.9\pm0.3$	$0.032\pm0.002$	$0.04\pm0.01$	$0.56\pm0.08$	$1.4\pm0.2$
	S15	-	-	-	-	-	-	-	-	-	-

**Table 5.3** Tissue mass, As and Sb concentrations, bioaccumulation factor (BAF) and translocation factor (TF) for As and Sb in *I. aquatica* exposed to contaminated soils for 35 days (mean  $\pm$  SE, n = 3).

-: not measured for evaluating As/Sb uptake due to damaged roots and plant death

#### 5.3.4.2. Accumulation of As and Sb in *I. aquatica* tissues

As and Sb concentrations in *I. aquatica* tissues increased with increasing soil concentrations, regardless of soil types (Table 5.3), and were significantly higher in roots than shoots. The As accumulated in *I. aquatica* roots ranged from 40 - 2600 mg/kg, which was 14 times higher than the shoot concentrations (3.8 - 126 mg/kg). A similar uptake trend occurred for Sb in which roots (0.84 - 80 mg/kg) concentrated 10 times more Sb than shoots (0.54 - 10 mg/kg). It is possible that root bioaccumulation is a detoxification mechanism for *I. aquatica* to respond to metalloid exposure. Shaibur et al. (2009) also found 10-15 times greater As in roots (than stems and leaves) of I. aquatica cultivated hydroponically. There is no known data in the literature for Sb uptake by *I. aquatica*. Arsenic concentrations in edible parts of plants in this study were much higher than that of plants grown in non-contaminated soils (<2 mg As/kg) which is also the reference value for non-contaminated plants in the literature (Kabata-Pendias 2011). In the present study, when I. aquatica were simultaneously exposed to soil concentrations of  $\geq 160$  mg As/kg (equivalent to  $C_{sol-As} \ge 18.5 \ \mu g/L$  and  $C_{DGT-As} \ge 10.4 \ \mu g/L$ ) and  $\ge 200 \ mg \ Sb/kg$  (equivalent to  $C_{sol-Sb} \ge 71 \ \mu g/L$  and  $C_{DGT-Sb} \ge 14 \ \mu g/L$ ), As concentrations were  $\ge 11.7 \ mg/kg \ dry \ weight$ (DW) in shoots, corresponding to 1.17 mg/kg wet weight (WW) in edible shoots (based on the measured average moisture content of 90%). This exceeds the permissible value of As in foodstuffs, specific to cereals being 1 mg As/kg WW set by the Food Standard Agency of Australia and New Zealand (FSANZ 2013). Although I. aquatica accumulated high concentrations of As in edible shoots, no visible toxicity symptoms in those treatments were observed. This phenomenon is a concern which poses a risk to people consuming I. aquatica grown in As and Sb contaminated soils. This indicates that the highly Ascontaminated soil and underground water in some Asian regions pose significant risk and not suitable for cultivating water spinach (*I. aquatica*). There is no guideline value for Sb in foodstuffs.

The bioaccumulation factors (BAF) and translocation factors (TF) were also determined to evaluate As and Sb uptake by *I. aquatica*. The bioaccumulation factor (BAF) is based on the ratio of total root concentrations and total soil concentrations (mg/kg DW) and is an index used to assess the transfer of contaminants from soils to plant roots, and if >1 it is

defined as an accumulator (Vithanage et al. 2012). A higher BAF means a greater efficiency of plants to accumulate contaminants from soils. The BAF of As and Sb in roots of I. aquatica ranged from 0.63 - 1.84 and 0.014 - 0.035, respectively (Table 5.3). On average, the BAF of As was much higher (55.7 times) than Sb, indicating that As was more efficiently transferred from soils to *I. aquatica* roots than Sb. Greater As bioaccumulation in *I. aquatica* compared to Sb is consistent with the higher measured lability of As in soils. There was no significant difference (p>0.05) in the average BAF values for As (1.15, 1.27, and 1.19) and Sb (0.02, 0.023, 0.025) among soil-L, soil-M, and soil-H, respectively. I. aquatica with As BAFs > 1 can be defined as an As accumulator. The BAF values for As and Sb in *I. aquatica* roots were higher than the typical BAF for As (approximately 0.2) and Sb (approximately 0.01) reported in the literature (Kabata-Pendias 2011). This might suggest that I. aquatica was exposed to higher phytoavailable As and Sb or the I. aquatica acquisition of nutrients and metalloids was large compared to previous tested plants in the literature. Our study was performed at much higher As soil concentrations than other studies, thus the As accumulated in our *I. aquatica* tissues were considerably higher than those reported previously: 0.099 mg As/kg DW shoot of water spinach grown in soil of 33.1 mg As/kg (Yao et al. 2009); 0.195 and 0.362 mg As/kg shoot and root, respectively, for *I. aquatica* cultivated in soil amended with manure of 62.8 mg As/kg (Yao et al. 2010).

The translocation factor (TF) determined by the ratio of shoot to root concentrations (mg/kg DW) evaluates the ability of *I. aquatica* to translocate As and Sb from roots to shoots. Considering the TFs variation among soil types, the average TFs of As (0.09, 0.08, 0.06) and Sb (0.3, 0.14, 0.08) tended to decrease from soil-L, followed by soil-M and soil-H, but this was not statistically significant. On the whole, the translocation of both As and Sb were quite low, with all TF values <1, varying from 0.048 - 0.12 and 0.04 - 0.8, respectively (Table 5.3), in which the translocation factor of Sb was 2.2 times higher than As. This indicates that *I. aquatica* had a significantly (p<0.05) stronger capacity to translocate Sb from roots to shoots than As. This is supported by a previous study on As and Sb uptake by root vegetable (radish, *R. sativus*) showing that the translocation rate of Sb was 2.5 times higher than that of As (chapters 3 and 4). Thus, it appears that the movement of Sb from roots to shoots was more favoured than that of As, however the mechanism for this is still unknown. The limited translocation from roots to shoots (TFs <1) might be a mechanism of

plants to avoid toxicity, especially for As. Many researchers have shown that As detoxification occurs via reduction of arsenate to arsenite in roots, which is then complexed with thiols and sequestrated in root vacuoles to limit As transport from roots to shoots (Álvarez-Ayuso et al. 2016; Zhao et al. 2009). It is well known that the main transport pathway of arsenate uptake by plants is via the phosphate channel due to their chemical similarity, but this does not apply to Sb (Feng et al. 2013).

In general, As in roots was significantly more concentrated than Sb in *I. aquatica*, despite the total soil As being lower than Sb. The larger size of antimonate might prohibit its transport through the root cell wall compared to arsenate, thus restricting its accumulation. However, once accumulated in roots, Sb was translocated to shoots at the greater rate than As, illustrating the differences in uptake patterns of As and Sb by *I. aquatica*. Since As and Sb co-existed in the studied soils, the presence of As may affect Sb accumulation and translocation in *I. aquatica* and vice versa. The question of the occurrence of their competitive uptake by edible plants is still unknown. Greater accumulation of As compared to Sb has been observed for other plants grown in As and Sb co-contaminated soils (Wei et al. 2011; Wilson et al. 2014).

#### 5.3.5. The relationship between As and Sb bioaccumulation and soil concentrations

The relationship between As and Sb concentrations in *I. aquatica* tissues and their labile concentrations in soils measured by SEP ( $C_{SEP-labile}$ ), DGT ( $C_{DGT}$ ), and soil solution ( $C_{sol}$ ) are illustrated in Figure 5.2.  $C_{SEP-labile}$ ,  $C_{DGT}$ , and  $C_{sol}$  predicted tissue concentrations for As better than Sb, demonstrated by greater correlation coefficients for As. Root As was better predicted than shoot As, which can be explained by the fact that roots directly contact soils, whereas bioaccumulation in shoots depends on the translocation from roots to shoots, usually governed by plant physiology (Nowack et al. 2004). This is in agreement with results from the soil-plant transfer model in which the bioaccumulation factor was a reflection of the bioavailable fraction, while translocation factor was dependent on the element and plant characteristics (McLaughlin et al. 2011). These significant positive correlations between As concentrations in *I. aquatica* tissues and As labile fractions are also supported by previous studies on As uptake by plants (Ngo et al. 2016; Wang et al.

2014). In contrast, Sb concentrations in shoots of *I. aquatica* were better predicted than root Sb concentrations.

Sb concentrations in *I. aquatica* tissues were not as well correlated with  $C_{SEP-labile}$ ,  $C_{DGT}$ , and  $C_{sol}$  as As, illustrated by lower correlation coefficients (Figure 5.2). This indicates that As was well sustained and readily available to plant uptake than Sb, which is supported by the fact that the lability and release capacity of As from the solid phase and As bioaccumulation in all soils were much higher than those of Sb (sections 5.3.2, 5.3.3, and 5.3.4). This also suggests that the bioavailable fraction of Sb might not be the primary pool of Sb to plant uptake. When the soil Sb extracted from amorphous Fe and Al oxides were considered, a better correlation between the extractable Sb in soil and Sb in roots was observed (Figure 5.3), which is not the case for As. This indicates that Sb associated with amorphous Fe and Al oxides may contribute to the Sb uptake by *I. aquatica* or the plant induced the release and resupply of Sb from the solid phase, which was not captured by labile measures for Sb in soils, while As associated with amorphous Fe oxides may not be important for As uptake by the plant. In terms of the predictive ability, SEP performed well in predicting As and Sb concentrations in *I. aquatica* compared to DGT and soil solution analysis.

The release of As and Sb from amorphous Fe oxides occurred because Fe oxides bound As and Sb were less stable than those strongly bound to crystalline and residual fractions and could potentially be solubilised under favoured environmental conditions (Hockmann et al. 2015; Niazi et al. 2011). The release and solubilisation of As and Sb can be due to the effects of oxalic and citric acids excreted by *I. aquatica* roots and changes in surrounding conditions such as soil pH and redox potential. Another possible explanation is that the decrease in organic matter content due to microbial degradation over the test duration may result in the release of As and Sb associated with the organic fraction (Cattani et al. 2009; Lomaglio et al. 2016), providing more As and Sb for plant uptake. These processes were not included in the lability measurements. This is supported by Ptak and McBride (2015) reporting that organic acid exudates were able to readily mobilise Sb trapped by organic Fe complexes. According to Cattani et al. (2009), plant exudates such as oxalic and citric acids may complex elements such as Fe (and other metals complexed with As) or generate

environmental conditions for the development of soil microfauna and microorganisms. These processes would enhance the transformation and mobilisation of As in soils.

# 5.3.6. Risk assessment of As and Sb in I. aquatica to human health

The As concentrations in edible parts of water spinach (I. aquatica) were higher than the limit in food, specific to cereals (1 mg As/kg WW) set by the Food Standard Agency of Australia and New Zealand (FSANZ 2013); the permissible value for food (0.1 mg As/kg WW) recommended by joint Food and Agricultural Organisaion (FAO)/World Health Organisation (WHO) (JECFA 2011); and far exceeded the Chinese maximum permissible concentration of As in food (0.05 mg As/kg WW) (Bhatti et al. 2013). Thus, there is a potential risk of consuming this leafy vegetable cultivated in As-contaminated soils. The joint FAO/WHO Expert Committee on Food Additives (JECFA) has not reported guidelines relating to Sb safety. Nevertheless, the USEPA has recommended an oral reference dose (RfD) for Sb being 0.0004 mg/kg of body weight (BW)/day. The risk to individual health also depends on the vegetable intake rate, exposure time, lifetime, exposure duration, target population, and their body weight (Environmental Health Australia – enHealth 2012; USEPA 1998, 2011). The hazard quotient (HQ) and cancer risk (CR) calculations include these parameters and were used to further explore the potential risks to human health. Considering the dietary pattern of people in Asian countries where vegetables are consumed in three meals per day, the intake of 0.345 kg/day for adults and 0.232 kg/day for children (Khan et al. 2008; Khan et al. 2010; Wang et al. 2005), and the exposure frequency (EF) of 52 days in a year (Bhatti et al. 2013) were used for calculations. Both adults and children are considered in this assessment.


**Figure 5.2** The relationship between tissue As and Sb (mg/kg, dry mass) and bioavailable fractions of As and Sb in soils, as measured by  $C_{SEP-labile}$ ,  $C_{DGT}$ , and  $C_{sol}$ . Shoot (filled symbols), Root (unfilled symbols), Soil-L (o), Soil-M ( $\diamondsuit$ ), Soil-H ( $\square$ ). All data presented have error bars; many are smaller than symbols.



**Figure 5.3** The relationship between tissue As and Sb (mg/kg, dry mass) and bioavailable fractions of As and Sb and As and Sb extracted from amorphous Fe oxide phase in soils. Shoot (filled symbols), Root (unfilled symbols), Soil-L (o), Soil-M ( $\diamondsuit$ ), Soil-H ( $\Box$ ). All data have error bars; many are smaller than symbols.

The estimated daily intake (EDI), hazard quotient (HQ), and cancer risk (CR) of As and Sb via the consumption of *I. aquatica* shoots as a function of soil As and Sb are presented in Tables 5.4 and 5.5, respectively. Since there is no cancer slope factor for Sb, the CR value of Sb was not calculated. The EDI values of As for both adults and children were higher than the acceptable limit or the reference dose (RfD,  $3 \times 10^{-4}$  mg/kg/day) set by (USEPA 1998, 2011). The HQ values of As and Sb were calculated to assess the potential risk of non-cancer adverse health effect to humans through the As and Sb ingestion. The HQ values of As were > 1 for all cases, ranging from 1.04 - 34.5 for adults and 1.72 - 54.8 for children, indicating the substantial risk of non-carcinogenic effects for both adults and children consuming I. aquatica grown in contaminated soils with As  $\geq$ 42 mg/kg corresponding to  $C_{sol-As} \ge 5.27 \ \mu g/L$  or  $C_{DGT-As} \ge 3.42 \ \mu g/L$ . The HQ value of Sb was >1 only in S5. The HQ values of As were much higher than the HQ values of Sb, suggesting that the non-carcinogenic risks of As in *I. aquatica* shoots is much higher than that of Sb. Cumulatively, the HI (hazard index) values being the sum of HQ values of As and Sb were >1 for all cases. Higher HQ for children compared to adults inferred greater potential health risks for this group. Previous studies also reported that teenagers and children were at higher risk associated with the consumption of As-enriched vegetables (Jiang et al. 2015).

The carcinogenic risk was calculated to assess the probability of the exposed population developing cancer risk through the ingestion of As-contaminated vegetables during their lifetime. The CR values of As increased with the increasing soil As and far exceeded the probability of 1 in 10,000, indicating an increase in probability of the cancer development via consumption of *I. aquatica* shoots cultivated in soils contaminated with  $\geq$ 42 mg As/kg (C<sub>sol-As</sub>  $\geq$ 5.27 µg/L or C<sub>DGT-As</sub>  $\geq$ 3.42 µg/L). The CR values of As were higher for children than adults, indicating that children were at higher risk of cancer. Taking *I. aquatica* grown in the lowest treatment as an example, the cancer risk was 4.69 per 10,000 for adults, but was 23.2 per 10,000 for children. The average cancer risk development via consumption of *I. aquatica* shoots cultivated in the studied soils was 4.9 times higher for children than adults. Rehman et al. (2016) also reported that the development of cancer risk through ingestion of vegetables was lower in adults than children.

Soil	Treatment	Soil As	Shoot As	Estimated daily	v intake (EDI)	Hazard que	otient (HQ)	Cancer ri	isk (CR)
5011	Treatment	(mg/kg DW)	(mg/kg WW)	Adults	Children	Adults	Children	Adults	Children
L	<b>S</b> 1	42	0.38	$3.13 \times 10^{-4}$	$5.15  imes 10^{-4}$	1.04	1.72	$4.69  imes 10^{-4}$	$23.2 \times 10^{-4}$
	S2	160	1.17	$9.55\times10^{4}$	$15.7  imes 10^{-4}$	3.18	5.24	$14.3  imes 10^{-4}$	$70.7  imes 10^{-4}$
	<b>S</b> 3	270	4.20	$34.4  imes 10^{-4}$	$56.6  imes 10^{-4}$	11.5	18.9	$51.6\times10^{4}$	$255  imes 10^{-4}$
	S4	670	5.10	$41.8\times10^{4}$	$68.8  imes 10^{-4}$	13.9	22.9	$62.6\times10^{4}$	$309\times10^{4}$
	<b>S</b> 5	1400	12.63	$103  imes 10^{-4}$	$170  imes 10^{-4}$	34.5	54.8	$155  imes 10^{-4}$	$767\times10^{4}$
М	<b>S</b> 6	65	0.44	$3.60 \times 10^{-4}$	$5.93  imes 10^{-4}$	1.20	1.98	$5.40  imes 10^{-4}$	$26.7 \times 10^{-4}$
	<b>S</b> 7	176	1.59	$13.0\times10^{4}$	$21.5\times10^{4}$	4.35	7.16	$19.6\times10^{4}$	$96.6\times10^{4}$
	<b>S</b> 8	330	3.60	$29.5\times10^{4}$	$48.6\times10^{\text{-}4}$	9.83	16.2	$44.2\times10^{4}$	$218\times10^{4}$
	<b>S</b> 9	950	7.35	$60.2  imes 10^{-4}$	$99.1\times10^{\text{-}4}$	20.1	33.0	$90.3\times10^{4}$	$446\times10^{\text{-}4}$
	<b>S</b> 10	2630	-	-	-	-	-	-	-
Н	S11	101	0.56	$4.58  imes 10^{-4}$	$7.54 imes10^{-4}$	1.53	2.51	$6.87 imes10^{-4}$	$34.0  imes 10^{-4}$
	S12	200	1.47	$12.1  imes 10^{-4}$	$19.9\times10^{4}$	4.02	6.63	$18.1  imes 10^{-4}$	$89.5\times10^{\text{-}4}$
	S13	510	4.73	$38.8\times10^{4}$	$63.9  imes 10^{-4}$	12.9	21.3	$58.2  imes 10^{-4}$	$287  imes 10^{-4}$
	S14	1280	8.87	$72.7  imes 10^{-4}$	$120  imes 10^{-4}$	24.2	39.9	$109  imes 10^{-4}$	$539  imes 10^{-4}$
	S15	4180	-	-	-	-	-	-	-

Table 5.4 Values of EDI (mg/kg/day), HQ, and CR of As for the ingestion of *I. aquatica* shoots grown in studied soils.

-: not measured for evaluating As/Sb uptake due to damaged roots and plant death

Soil	Treatment	eatment Soil Sb	Shoot Sb	Estimated daily	y intake (EDI)	Hazard quotient (HQ)		
3011	Heatment	(mg/kg DW)	(mg/kg WW)	Adults	Children	Adults	Children	
L	<b>S</b> 1	52	0.066	$0.54  imes 10^{-4}$	$0.89  imes 10^{-4}$	0.14	0.22	
	S2	200	0.045	$0.37  imes 10^{-4}$	$0.60  imes 10^{-4}$	0.09	0.15	
	<b>S</b> 3	330	0.182	$1.49  imes 10^{-4}$	$2.45  imes 10^{-4}$	0.37	0.61	
	S4	870	0.132	$1.08  imes 10^{-4}$	$1.79  imes 10^{-4}$	0.27	0.45	
	S5	2277	0.990	$8.11  imes 10^{-4}$	$13.4 \times 10^{-4}$	2.03	3.34	
М	S6	91	0.044	$0.36  imes 10^{-4}$	$0.60  imes 10^{-4}$	0.09	0.15	
	<b>S</b> 7	220	0.035	$0.29\times 10^{4}$	$0.47  imes 10^{-4}$	0.07	0.12	
	<b>S</b> 8	530	0.145	$1.19\times10^{4}$	$1.96\times10^{\text{-}4}$	0.30	0.49	
	S9	1190	0.232	$1.90\times10^{4}$	$3.13\times10^{4}$	0.48	0.78	
	S10	3790	-	-	-	-	-	
Н	S11	140	0.029	$0.24  imes 10^{-4}$	$0.40  imes 10^{-4}$	0.06	0.10	
	S12	230	0.054	$0.44  imes 10^{-4}$	$0.73  imes 10^{-4}$	0.11	0.18	
	<b>S</b> 13	670	0.115	$0.94  imes 10^{-4}$	$1.55  imes 10^{-4}$	0.24	0.39	
	<b>S</b> 14	1600	0.188	$1.54  imes 10^{-4}$	$2.53  imes 10^{-4}$	0.39	0.63	
	S15	6610	-	-	-	-	-	

Table 5.5 Values of EDI (mg/kg/day) and HQ of Sb for the ingestion of *I. aquatica* shoots grown in studied soils.

-: not measured for evaluating As/Sb uptake due to damaged roots and plant death

It is noticeable that *I. aquatica* accumulated As at high levels exceeded the permissible limit for food consumption, but physical toxicity symptoms were not evident. The cancer risk assessment model also showed unacceptable non-carcinogenic (all HQ > 1) and carcinogenic risks (all CR >  $10^{-4}$ ) to both adults and children ingesting edible parts of *I. aquatica* cultivated in the studied soils, even in the lowest contaminated soils.

#### **5.4.** Conclusions

This study illustrated the contrasting behaviours of As and Sb in soils, their uptake by *I. aquatica*, and the importance of bioavailability in risk assessments. The SEP and DGT showed the lability of As in contaminated soils and its partial resupply from the solid phase varied with soil types, while for Sb, these were low and relatively constant across soil types. This confirms that irrespective of soil types, As was more geochemically labile than Sb, which was dominated by associations with recalcitrant minerals. Soil physicochemical properties especially Fe and Al oxides, DOC, and pH are important in controlling the lability and phytoavailability of As and Sb in contaminated soils. Thus, the incorporation of bioavailability measurements into soil risk assessments was essential for the management of As and Sb-contaminated soils.

*I. aquatica* had a high capacity to accumulate As and Sb with concentrations in roots being 10 and 14 times higher than shoots, respectively. A different transport pathway of As and Sb within *I. aquatica* was observed and *I. aquatica* was considered as an As accumulator plant. Arsenic accumulated in edible shoots at very high concentrations far exceeded the permissible limit of As in food set by the Food Standard Agency of China, Australia and New Zealand, and FAO/WHO, but no visible toxicity symptoms was observed in the plant, which is an alarming concern. Consumption of *I. aquatica* cultivated in studied soils, even at the lowest treatment (low contamination) posed the cancer risks to both adults and children. Thus, cultivation of *I. aquatica* in As-contaminated soils or water or irrigation of *I. aquatica* with As-enriched water should be restricted in terms of food safety and human health.

DGT, soil solution, and SEP-labile fractions were excellent in predicting metalloids uptake by *I. aquatica* cultivated in historically contaminated soils with the greater extent for As prediction. The analyses of relationships between As and Sb concentrations in *I. aquatica*  and their concentrations in various soil fractions support the conclusion that the activity of *I. aquatica* roots could induce the solubilisation and release of Sb from the solid phase. The combination of SEP and DGT was useful in determining the partitioning of As and Sb in soils and measuring their lability in different soil types. DGT was sensitive to differentiate the lability, capacity and kinetics of As and Sb release from the solid phase of various soils. The dynamic DGT technique integrated soil properties affecting soil processes and was less laborious than other measurements, thus it provides a promise for assessing bioavailable As and Sb in historically contaminated soils.

The question is does the presence of Sb influence the partitioning, lability and bioaccumulation of As by *I. aquatica* or vice versa. This is still unknown and warrants for further investigation. To do this a study on the fractionation and lability of As and Sb in its singly contaminated soil compared to that in co-contaminated soil and their uptake by *I. aquatica* grown in those soils needs to be performed.

# Chapter 6. Biogeochemical behaviour of As and Sb in soils and their uptake by water spinach (*Ipomoea aquatica*): Implications for lability, distribution, and competition

## **6.1. Introduction**

Arsenic is of concern due to its high concentrations in both anthropogenically contaminated sites and naturally in several areas such as India, West Bengal, and Bangladesh (Chowdhury et al. 2000; Fu et al. 2016). Intensive research has explored the biogeochemical behaviour and toxicity of As (Abernathy et al. 1999; Dixit & Hering 2003; Seyfferth et al. 2010), while studies on Sb have been limited. The behaviour and toxicity of Sb have been assumed to be comparable with that of As since they are both in group 15 of the periodic table. Due to insufficient data for Sb, the observed behaviour for As (uptake, detoxification mechanisms, methylation) has been commonly extended to Sb (Filella et al. 2007). However, Sb and As have different coordination geometry: arsenate is tetrahedrally coordinated to four oxygen atoms (e.g. similar to phosphate) whereas antimonate is octahedrally coordinated to six oxygen atoms and has a greater spatial structure than that of arsenate (radius ratios: Sb(V)/O = 0.51, As(V)/O = 0.4) (Qi & Pichler 2017). Recent evidence shows that these two metalloids behave differently under certain conditions, which is a challenge for interpreting Sb behaviour on the basis of As characteristics (Fu 2016). For example, previous chapters have shown that As was primarily associated with Fe oxides, while Sb was mainly found in residual phase, possibly sulphide or organic matter. Arsenic was present as As(III) and As(V), while Sb was dominantly present as Sb(V) in a soil-water system near an Sb mine (Mitsunobu et al. 2006), and bioaccumulation factors for Sb, determined by the ratio of plant and soil concentrations, were 10-fold less than those of As (Fu 2016; Ngo et al. 2016).

Arsenic and Sb are non-essential elements that are toxic to plants (Filella et al. 2002a; Gebel 1999); however, some plants can readily take up As and Sb from soils and accumulate them in edible tissues posing a risk to human health via the food chain. Ample research has been previously conducted to investigate the accumulation of As by a variety of plants in relation to its bioavailability in soils (Álvarez-Ayuso et al. 2016; Bhatti et al. 2013; Niazi et al. 2011; Rosas-Castor et al. 2014; Smith et al. 2009) and some research specific to Sb has been recently performed (Okkenhaug et al. 2011; Shtangeeva et al. 2014; Tschan et al. 2009a). There are a few studies focused on the simultaneous accumulation of As and Sb by edible crops cultivated in soils contaminated with both As and Sb (Ngo et al. 2016; Wilson et al. 2014), but their interactive effects remains unclear. Recent work on the effect of As on the uptake and bioaccumulation of Sb by As-hyperaccumulators, *P. cretica* (Feng et al. 2011) and *P.vittata* (Müller et al. 2013), reported that the presence of As enhanced Sb uptake by plants; however, they did not consider the effect of Sb on the As uptake as well as their competitive interactions in edible vegetables.

It is well established that the bioavailability of As and Sb in soils is controlled by many factors such as pH, redox potential (Mitsunobu et al. 2006; Okkenhaug et al. 2012), organic matter (Dousova et al. 2015; Williams et al. 2011), clay (García-Sanchez et al. 2002), mineral oxides (Dixit & Hering 2003; Kim et al. 2014; Leuz et al. 2006), and aging time (Juhasz et al. 2008; Li et al. 2014; Tang et al. 2007). However, these studies have only focused on single-element contaminated soils. Some studies investigated the fractionation and bioavailability of As and Sb that were co-existing in long-term contaminated soils (Ettler et al. 2010; Li et al. 2014; Müller et al. 2007; Ngo et al. 2016). Only one study by Liang et al. (2014) examined the bioavailability of As in As-contaminated soils in comparison to As+Pb contaminated soils. However, little attention has been given to the effect of As and Sb on their respective bioavailability in co-contaminated soils. To our knowledge, no study has been conducted to investigate whether the presence of As influences the partitioning, lability, and solubility of the co-existing Sb in soils and vice versa, which is important for the prediction of their bioavailability and potential environmental risk. In addition, the competitive uptake of As on the uptake and bioaccumulation of Sb by edible plants and vice versa in co-contaminated soils has not been explored.

Chapter 5 investigated the biogeochemical behaviour of As and Sb in historically cocontamianted soils with various soil properties. However, the competitive interactions between As and Sb in soils and competitive uptake by water spinach (*Ipomoea aquatica*) could not be interpreted due to the lack of soils having identical mineralogy and properties and contaminated with As only and Sb only, acting as references. To do this a study on the behaviour of As and Sb in soils spiked with As only and Sb only compared to those in soils spiked with As+Sb in the same manner is needed. In addition, *I. aquatica* is a very popular edible vegetable in Asia and was known to accumulate high amounts of As, posing cancer risks to human health (section 5.3.6, chapter 5). The question that the presence of Sb enhanced As accumulation by *I. aquatica* has been unknown and should be further explored.

The aims of this study are (1) to investigate the biogeochemical behaviour of As and Sb in single and multi-contaminant amended soils, (2) to investigate the interactive effects of As and Sb on accumulation by water spinach (*Ipomoea aquatica*), (3) to assess the performance of sequential extraction and the DGT technique for predicting As and Sb uptake by *I. aquatica*. This study will provide a better understanding of interactive effects of As and Sb in individual and co-occurring soils and in the studied plant.

# 6.2. Methods

### **6.2.1.** General methods

General washing methods were done as per section 2.1.

# 6.2.2. Experimental design

#### 6.2.2.1. Soil preparation

An uncontaminated soil was collected in Wollongong, NSW, then air-dried and crushed to pass through a 2-mm mesh sieve. Each portion of soil was amended with either single or mixed solutions of Na<sub>2</sub>HAsO<sub>4</sub>.7H<sub>2</sub>O and KSb(OH)<sub>6</sub> at various concentrations to create three series of soils contaminated with As only, Sb only, and both As and Sb. Eight treatments of each contaminated soil series were prepared, denoted as As1 – As8, Sb1 – Sb8, and AsSb1 – AsSb8. The amounts of As and Sb added into soils were targeted in the wide range of 20 – 600 mg As/kg and 20 – 6000 mg Sb/kg. The chosen concentrations of As and Sb were based on published studies on As bioavailability in As-spiked soils compared with that in soils spiked with a mixture of As+Pb (Liang et al. 2014) and As and

Sb uptake by plants cultivated in singly-spiked soils (Niazi et al. 2011; Tschan et al. 2010). In addition, the previous study (chapter 5) showed that *I. aquatica* grown in highly cocontaminated soils did not exhibit toxicity symptoms. Thus, the largest concentrations were estimated to be high enough for the plant toxicity to be detectable. Spiking soils was done as per section 2.2.2. These spiked soils are considered as recently contaminated soils.

# 6.2.2.2. Pot experiment

Spiked soil moisture was then raised to and maintained at ~60% of the water holding capacity (WHC) with deionised water. Soils were allowed to settle and equilibrate in a greenhouse for two more weeks prior to the commencement of bioassays. The 35-day bioassays with water spinach (*Ipomoea aquatica*) using pots were conducted in controlled growth chambers as per section 2.4.2. Harvested plant tissues from the same pot were combined, weighed, and oven-dried at 70 °C for 48 h to a constant weight with the dry mass (g) recorded. Dried tissues were ground with a mortar and pestle and stored in clean plastic bags prior to As and Sb analyses.

# 6.2.3. Soil characterisation

Soil physicochemical properties including particle size, pH, organic matter, total nitrogen, available phosphorus, and amorphous and free Fe and Al were determined as described in section 2.5.

# 6.2.4. Soil and plant analyses

Total As and Sb in soils and plants were digested and analysed as described in sections 5.2.4 and 2.9. For quality control, a combination of reagent blanks, duplicate analyses, certified reference materials, and matrix spike recovery were used. A soil certified reference material (Montana II soil, SRM 2711a; NIST) returned recoveries of 90-115% for the studied analytes. A plant certified reference material (Tomato leaves, SRM 1573a; NIST) had typical recoveries of 95-110% of the studied analytes.

#### **6.2.5. Sequential extraction procedure**

The improved sequential extraction scheme developed by Wenzel et al. (2001) was applied to partition both As and Sb in the studied soils as per section 2.6.1. The five extracted

fractions were As and Sb associated with non-specifically sorbed, specifically sorbed, amorphous Fe and Al oxide, crystalline Fe and Al oxide, and residual fractions.

#### 6.2.6. DGT and soil solution measurements

DGT and soil solution measurements were done as described in sections 2.6.2 and 2.7.

#### 6.2.7. Data analysis

Statistical analysis and calculations of bioaccumulation factor (BAF) and translocation factor (TF) were performed as per section 3.2.7. To assess the non-carcinogenic and carcinogenic risks associated with As and Sb exposure through diet, the hazard quotient (HQ) and cancer risk (CR) were calculated as described in section 5.2.7.

## 6.3. Results and discussion

#### **6.3.1.** Soil characteristics

The test soils were a sandy-silt (29.5% sand, 62.4% silt, 8.1% clay), slightly acidic (pH  $\sim$ 6.0), and had a total organic matter of 8.1%. The total Kjeldahl nitrogen and extractable phosphorous in soils were 92 and 30 mg/kg, respectively. The amorphous and crystalline Fe concentrations in soils were 7.9 g/kg and 27.1 g/kg, respectively. The measured concentrations of As and Sb in amended As-only, Sb-only, and As+Sb soils were presented in Tables 6.1 and 6.2.

# 6.3.2. As and Sb fractionations in soils

The partitioning of As and Sb in different soil fractions are presented in Tables 6.1 and 6.2. In general, reagents with increasing dissolution strength were used to sequentially extract As and Sb bound to various soil constituents, meaning that As and Sb extracted in the later fraction was less labile and bioavailable than those in the earlier fractions (Kim et al. 2014). These extractions are not perfectly selective to the target phase, so there can be some interactions between them. On average, As in the As-only soils was partitioned in the following order: associated with amorphous Fe and Al oxides ( $54 \pm 1\%$ ) > specifically sorbed ( $32 \pm 2\%$ ) > associated with crystalline Fe and Al oxides ( $17 \pm 1\%$ ) > non-specifically sorbed ( $3.2 \pm 0.7\%$ ) (Table 6.1). The pattern of Sb distribution in various soil binding phases differed from As. The partitioning of Sb in Sb-only soils followed the

sequence: associated with amorphous Fe and Al oxides > bound to crystalline Fe and Al oxides > specifically sorbed > non-specifically sorbed, except for high treatments ( $\geq$ 1560 mg Sb/kg) where the order was associated with amorphous Fe and Al oxides > non-specifically sorbed > bound to crystalline Fe and Al oxides > specifically sorbed (Table 6.2).

Considering changes of As and Sb in each fraction with their total loads in the soils, both As and Sb in non-specifically sorbed and specifically sorbed fractions increased in concentration and proportion. In contrast, As and Sb in amorphous Fe and Al oxides increased with each addition but proportions increased initially to ~60% (at total 84 mg As/kg) and ~70% (at total 78 mg Sb/kg), and then decreased to ~50% and ~55% at higher total concentrations of As and Sb, respectively. Arsenic in crystalline Fe and Al oxides increased but proportionally decreased from about 20 to 15% over the As-added series. This trend was also observed by Huang et al. (2016) and Tang et al. (2007).

When As and Sb were added together, the distribution pattern for As and Sb in As+Sb soils was similar to that in their singly amended soils but with some interesting differences. In As+Sb soils at low concentrations, the amounts of added As and Sb were comparable, but the Sb partitioning in various binding sites was different from that of As. The proportions of Sb in non-specifically sorbed, amorphous Fe and Al oxide, and crystalline Fe and Al oxide fractions were higher than that of As, while the proportions of Sb in specifically sorbed fraction was lower. There was relatively unchanged As proportions in the nonspecifically sorbed and specifically sorbed fractions, but the higher tendency was apparent for Sb in co-amended soils. Compared to As and Sb fractionation in singly amended soils, the proportions of As and Sb in the non-specifically sorbed fraction were higher (e.g. the percentage of As partitioned to the non-specifically sorbed fraction was double that in Asonly soils at two highest treatments) while those in the amorphous Fe and Al oxide fraction tended to be lower. This may be because the simultaneous addition of As and Sb into soils increased amounts of dissolved co-contaminants added, which decreased soil adsorption binding sites. As a result, the sorbed and bound As and Sb decreased and more As and Sb were found in non-specifically sorbed fraction. The As distribution pattern in this study is

supported by Niazi et al. (2011) who studied As fractionation in As spiked soils with various soil properties.

The difference in proportion of As and Sb in each fraction between soil treatments was most likely due to the soil surface adsorption sites being initially unsaturated in the native soil. Thus, when soluble As and Sb was added, As and Sb were first partitioned to the soil solution, followed by the rapid adsorption onto active binding sites. The adsorbed As and Sb then transformed into the inner-sphere complexes over time, with the excess As and Sb added in the high treatments being redistributed among other soil binding phases. When large amounts of As and Sb added (>400 mg As/kg, >1500 mg Sb/kg), the soil adsorption sites may be saturated. This explains the significant portions of As and Sb in non-specifically sorbed fraction in soils AsSb6 and AsSb7.

Overall, As in the non-specifically sorbed and specifically sorbed fractions was higher than Sb in all treatments, indicating that As was more labile than Sb. In our soils the competitive interaction of As and Sb for their soil binding sites appeared to affect the distribution of As and Sb in various soil binding phases and make them more available in the As+Sb co-amended soils, compared to singly amended soils. To our knowledge, this is the first study on comparative interactions of As and Sb in soils.

The partitioning of As and Sb in As+Sb soils was different from that in As and Sb historically co-contaminated soils (chapter 5) (Figure D.1) where As and Sb were low in labile fractions and high in Fe and Al oxides and residual fractions. This possibly occurred because of the variation in aging time and equilibration of soils. Arsenic and Sb in non-specifically sorbed and specifically sorbed fractions (the available fraction) accounted for ~30% and ~18% of total soil As and Sb, respectively, and it is not surprising that they are significantly higher than that of available As and Sb in historically contaminated soils in chapter 5 (Figure D.1). This may be attributed to the soluble form of added As and Sb, and their shorter equilibration time in amended soils compared to the long aging in historically contaminated soils (Tang et al. 2007).

Soil		As con	ncentrations in each frac	ction (mg/kg)	
	Total As	Non-specifically	Specifically sorbed	Amorphous Fe and	Crystalline Fe and
		sorbed		Al oxide	Al oxide
As only					
As1	$46 \pm 3$	$0.457 \pm 0.006 \ (1.0)$	$10.37 \pm 0.05$ (22.7)	$25.9 \pm 0.4 \ (56.5)$	$9.2 \pm 0.5$ (20.1)
As2	$84.2\pm0.7$	$1.12 \pm 0.007 \ (1.3)$	$22.2 \pm 0.1 \ (26.4)$	$49.7 \pm 0.2 (59.0)$	$17.9 \pm 0.3 \ (21.2)$
As3	$165 \pm 3$	3.11 ± 0.08 (1.9)	$49.2 \pm 0.7$ (29.8)	$92.4 \pm 0.7 (56.0)$	$30 \pm 2$ (18.0)
As4	$196 \pm 2$	$4.5 \pm 0.2$ (2.3)	64.1 ± 0.3 (32.6)	110 ± 1 (56.3)	39 ± 3 (19.9)
As5	$300 \pm 5$	$10.4 \pm 0.2 \ (3.5)$	$103.8 \pm 0.8 \ (34.8)$	$160 \pm 4 \ (53.4)$	44 ± 7 (14.7)
As6	$410 \pm 7$	$16.7 \pm 0.7 \ (4.0)$	142 ± 0.1 (34.3)	210 ± 3 (50.0)	58 ± 6 (14.0)
As7	$510 \pm 7$	$27 \pm 1$ (5.3)	180 ± 3 (35.8)	$260 \pm 6 (50.5)$	77 ± 8 (15.2)
As8	$600 \pm 1$	36.8 ± 0.5 (6.1)	220 ± 3 (37.1)	300 ± 3 (49.9)	86 ± 10 (14.3)
As + Sb					
AsSb1	$60\pm 6$	0.520 ± 0.005 (0.9)	$11.62 \pm 0.09$ (19.4)	$27.2 \pm 0.5$ (45.3)	$10.4 \pm 0.8 \ (17.3)$
AsSb2	$87 \pm 2$	$1.19 \pm 0.03 (1.4)$	$23.0 \pm 0.6$ (26.3)	47.1 ± 0.9 (53.8)	$15.9 \pm 0.5 (18.2)$
AsSb3	$175.3\pm0.9$	$3.9 \pm 0.1$ (2.2)	54 ± 1 (30.7)	94 ± 2 (53.9)	35 ± 2 (19.9)
AsSb4	$220 \pm 4$	$5.24 \pm 0.08$ (2.4)	65.1 ± 0.3 (29.3)	$110.9 \pm 0.6 (50.0)$	39 ± 1 (17.6)
AsSb5	$310 \pm 3$	$10.5 \pm 0.2 (3.3)$	100 ± 1 (31.9)	$160 \pm 2 (52.2)$	49 ± 2 (15.7)
AsSb6	$405 \pm 2$	$43.7 \pm 0.4 (10.8)$	$120 \pm 6 (29.8)$	$200 \pm 1 \ (50.2)$	72 ± 3 (17.6)
AsSb7	$510 \pm 10$	$63.9 \pm 0.9 (12.5)$	$150 \pm 4$ (29.3)	250 ± 2 (48.9)	90 ± 2 (17.5)
AsSb8	600 ± 3	-	-	-	-

**Table 6.1** As concentrations in each fraction of soils amended with (i) As only and (ii) with a mixture of As and Sb (mean  $\pm$  SE, n = 3). Numbers in brackets are % of total soil concentrations.

-: not measured due to analysis problems

**Table 6.2** Sb concentrations in each fraction of soils amended with (i) Sb only and (ii) with a mixture of As and Sb (mean  $\pm$  SE, n = 3). Numbers in brackets are % of total soil concentrations.

Soil		Sb conc	centrations in each fi	raction (mg/kg)	
	Total Sb	Non-specifically	Specifically	Amorphous Fe and	Crystalline Fe and
		sorbed	sorbed	Al oxide	Al oxide
Sb only					
Sb1	$40 \pm 1$	$2.54 \pm 0.04$ (6.4)	$3.61 \pm 0.06 (9.1)$	$25.0 \pm 0.2$ (62.7)	$8.0 \pm 0.1$ (20.2)
Sb2	$78 \pm 2$	$5.58 \pm 0.04$ (7.2)	$7.4 \pm 0.1$ (9.6)	$54.7 \pm 0.5 \ (70.5)$	$17.1 \pm 0.8 (22.0)$
Sb3	$160 \pm 9$	$10.69 \pm 0.04$ (6.8)	$14.9 \pm 0.3 \ (9.5)$	$107.7 \pm 0.5 \ (68.9)$	$32.6 \pm 0.3 \ (20.8)$
Sb4	$300 \pm 10$	$21.2 \pm 0.4$ (7.1)	$28.5 \pm 0.2 \ (9.6)$	$200 \pm 2 \ (66.8)$	58 ± 2 (19.6)
Sb5	$650 \pm 20$	$48.5 \pm 0.3 (7.5)$	$59.6 \pm 0.4 \ (9.2)$	$380 \pm 6 (58.9)$	$120 \pm 4 (19.0)$
Sb6	$1560\pm60$	370 ± 4 (23.5)	200 ± 1 (13.0)	860 ± 80 (55.2)	$330 \pm 10 (21.2)$
Sb7	$3000\pm60$	-	-	-	-
Sb8	$5300\pm40$	-	-	-	-
As + Sb					
AsSb1	$45 \pm 4$	$2.73 \pm 0.02$ (6.0)	3.71 ± 0.04 (8.2)	$26.6 \pm 0.6 (58.8)$	$9.0 \pm 0.3 (20.0)$
AsSb2	$73 \pm 7$	$5.58 \pm 0.09 \ (7.7)$	$7.1 \pm 0.1$ (9.8)	51 ± 1 (70.8)	16.1 ± 0.4 (22.2)
AsSb3	$170 \pm 5$	$14.5 \pm 0.3 \ (8.6)$	$16.6 \pm 0.6$ (9.8)	$103 \pm 1 \ (61.1)$	33.6 ± 0.9 (19.9)
AsSb4	$310 \pm 4$	$28.2 \pm 0.2 (9.1)$	$29.8 \pm 0.2 \ (9.6)$	168 ± 1 (54.2)	$55.7 \pm 0.5 (17.9)$
AsSb5	$320 \pm 8$	37.1 ± 0.4 (11.4)	36.6 ± 0.9 (11.3)	195.8 ± 1 (60.4)	62 ± 1 (19.2)
AsSb6	$1620 \pm 30$	570 ± 5 (35.4)	230 ± 9 (14.2)	$800 \pm 10$ (49.0)	$240 \pm 9$ (14.9)
AsSb7	$3400\pm10$	$1310 \pm 20 (38.3)$	$490 \pm 10 (14.4)$	$1650 \pm 20 (48.4)$	460 ± 9 (13.6)
AsSb8	$5800\pm80$	-	-	-	-

-: not measured due to analysis problems

#### 6.3.3. Lability of As and Sb in soils

DGT-labile concentrations ( $C_{DGT}$ ), soil solution concentrations ( $C_{sol}$ ),  $K_d$ , and R values for As and Sb are presented in Tables 6.3 and 6.4.  $C_{DGT-As}$  and  $C_{sol-As}$  increased with As soil treatment concentrations (Table 6.3).  $C_{DGT-As}$  and  $C_{sol-As}$  in As-only soils ranged from 0.02 – 0.14% (9.7 – 853 µg/ As/L) and 0.04 – 0.64% (17.36 – 3860 µg/ As/L) of total soil As concentrations, respectively. Similarly,  $C_{DGT-Sb}$  and  $C_{sol-Sb}$  in Sb-only soils increased with soil Sb concentrations (Table 6.4), with 0.08 – 0.17% (31.6 – 5260 µg Sb/L) and 0.38 – 1.12% (150 – 33700 µg Sb/L) of total soil Sb concentrations, respectively.

The  $C_{DGT}$  of As and Sb were less than their  $C_{sol}$  in all cases, especially for Sb. This is because  $C_{DGT}$  represents labile As and Sb, while  $C_{sol}$  includes both free and complexed As and Sb, and colloids (Zhang & Davison 2015). Some complexes found in soil solutions may be too large to diffuse through the polyacrylamide diffusive gel in DGT, making them non-DGT labile. Decreases in the distribution coefficients (K<sub>d</sub>) of soil solid phase to soil solution concentration of As and Sb occurred with the increasing additions of As and Sb due to an increase in As and Sb in soil solutions, especially at their high treatments. This is consistent with the trend of non-specifically sorbed As and Sb obtained from SEP, the increasing As and Sb load, and potential increasing saturation of soil binding sites.

The R value (R=C<sub>DGT</sub>/C<sub>sol</sub>, 0 < R < 1) obtained from the ratio of DGT-labile and soil solution concentration reflects the capacity of soil to resupply solutes from the solid phase to the soil solution and the kinetics of this process (Harper et al. 1998; Zhang et al. 2006; Zhang et al. 2004). The R values for As and Sb are presented in Tables 6.3 and 6.4. The average R value for As in As-only soils was 0.38, indicating that As was partially resupplied from the solid phase, but insufficient to sustain pore water concentrations. The average R value for Sb in Sb-only soils was 0.2, suggesting that the resupply of Sb from the solid phase was limited and slow. This demonstrates that the sorption of As and Sb onto soils and their resupply to soil solution were controlled by different factors.

Changes in soils that were simultaneously co-contaminated with As and Sb were examined. Soil solution As (0.03 - 0.31% of total As) in As+Sb soils were slightly lower than the Asonly soils (0.04 - 0.64%), whereas soil solution Sb (0.52 - 2.01% of total Sb) in As+Sb soils were about two times higher than the Sb-only soils (0.38 - 1.12%). This phenomenon also occurred for  $C_{DGT-As}$  and  $C_{DGT-Sb}$ , but to a lesser extent. The average percentage of  $C_{DGT-As}$ ,  $C_{Sol-As}$ ,  $C_{DGT-Sb}$ , and  $C_{sol-Sb}$  of their total soil concentrations is illustrated in Figure D.2. This implies that some As and Sb may not be sorbed to the same soil binding sites, possibly because As has a higher affinity for Fe oxides compared to Sb (Qi & Pichler 2017; Wilson et al. 2010), while Sb has stronger binding strength with organic matter (OM) than As. Liang et al. (2014) showed that soluble Fe and dissolved organic carbon (DOC) concentrations increased significantly, especially in high As-amended soils. Based on the Langmuir adsorption model, the binding stability of As/Sb in soil types was also presented in order of As(V) – Fe(III) >> Sb(V) – OM > As(V) – OM > Sb(V) – Fe(III) (Dousova et al. 2015). Thus, when As and Sb were added together, the energetically favourable binding sites such as Fe oxides gave preference to As, leading to more added As in solid phases and more Sb in soil solutions. This is clear from the C<sub>sol-Sb</sub> and reflected in the decrease of K<sub>d</sub> for Sb to low values in the As+Sb soils.

Consequently, the resupply of As in As+Sb soils was slightly lower than the As-only soils, evident by the lower average R value  $(0.33 \pm 0.02 \text{ compared to } 0.38 \pm 0.04, \text{ respectively})$ . As the proportion of C<sub>sol-Sb</sub> increased and K<sub>d</sub> decreased, it was expected that Sb in As+Sb soils would have the greater supply for counteracting the Sb depletion at the vicinity of DGT devices and a proportionally increased C<sub>DGT</sub> compared to that in Sb-only soils. This expectation was based on maintaining the concentration of Sb needed to be resupplied from a non-DGT labile source as a result of the depletion. However, the resupply of Sb in As+Sb soils remained limited or was even lower than that in Sb-only soils, demonstrated by the average R values (0.115  $\pm$  0.009 compared to 0.20  $\pm$  0.01, respectively). This may be because (1) most Sb in soil solutions were inert or in large complexes, making the diffusional transport through polyacrylamide DGT gels restricted, which were unavailable for DGT measurements over short periods and (2) the release of Sb from the solid phase was minimal. Quantification of DOC in soil solutions would be needed for further explaining the reason for low Sb lability. It is possible that sorbed As had higher rates of desorption from the solid phase and little interaction with DOC compared with Sb, which determined the kinetic processes and differences between As and Sb behaviour in soils.

		As in As-only amended soils						As in As+Sb amended soils					
Soil	Total <sub>As</sub> (mg/kg)	C <sub>SEP-labile As</sub> (mg/kg)	C <sub>DGT-As</sub> (µg/L)	C <sub>sol-As</sub> (µg/L)	Ratio R <sub>As</sub> (R=C <sub>DGT</sub> /C <sub>sol</sub> )	K <sub>d</sub> (L/kg)	Soil	Total <sub>As</sub> (mg/kg)	C <sub>SEP-labile As</sub> (mg/kg)	C <sub>DGT-As</sub> (µg/L)	C <sub>sol-As</sub> (µg/L)	Ratio R <sub>As</sub> (R=C <sub>DGT</sub> /C <sub>sol</sub> )	K <sub>d</sub> (L/kg)
As1	$46 \pm 3$	$10.83\pm0.05$	$9.7\pm0.5$	$17.36\pm0.06$	$0.56\pm0.03$	624	AsSb1	$60\pm 6$	$12.14\pm0.09$	$8.2\pm0.5$	$18.6\pm0.1$	$0.44\pm0.03$	652
As2	$84.2\pm0.7$	$23.4\pm0.1$	$22.2\pm0.3$	$47.6\pm0.5$	$0.47\pm0.01$	491	AsSb2	$87\pm2$	$24.2\pm0.6$	$15.7\pm0.2$	$50\pm1$	$0.31\pm0.01$	485
As3	$165\pm3$	$52.3\pm0.7$	$63 \pm 1$	$157\pm5$	$0.40\pm0.02$	333	AsSb3	$175.3\pm0.9$	$58\pm1$	$50\pm3$	$150\pm4$	$0.34\pm0.02$	384
As4	$196\pm2$	$68.5\pm0.4$	$88\pm4$	$233\pm4$	$0.38\pm0.02$	294	AsSb4	$220\pm4$	$70.3\pm0.3$	$90\pm 5$	$260\pm20$	$0.34\pm0.01$	268
As5	$300\pm 5$	$114.2\pm0.8$	$232\pm5$	$650\pm40$	$0.36\pm0.03$	175	AsSb5	$310\pm3$	$110\pm2$	$143\pm8$	$600\pm10$	$0.24\pm0.01$	185
As6	$410\pm7$	$158.5\pm0.7$	$330\pm20$	$1231\pm3$	$0.27\pm0.02$	129	AsSb6	$405\pm2$	$165\pm 6$	$367\pm5$	$850\pm20$	$0.31\pm0.01$	138
As7	$510\pm7$	$210\pm3$	$582\pm 6$	$2560\pm80$	$0.23\pm0.01$	82	AsSb7	$510\pm10$	$214\pm4$	$503\pm9$	$1580\pm40$	$0.32\pm0.01$	136
As8	$600 \pm 1$	$260 \pm 3$	$850\pm20$	$3860\pm200$	$0.22\pm0.01$	67	AsSb8	$600 \pm 3$	-	-	-	-	-

Table 6.3 Total and labile As concentrations measured by SEP, DGT, soil solution in contaminated soils, and R value of As (mean  $\pm$  SE, n = 3).

-: not measured due to analysis problems

Table 6.4 Total and labile Sb concentrations measured by SEP, DGT, soil solution in contaminated soils, and R value of Sb (mean  $\pm$  SE, n = 3).

~		Sb in Sb-only amended soils						Sb in As+Sb amended soils					
Soil	Total <sub>Sb</sub> (mg/kg)	C <sub>SEP-labile Sb</sub> (mg/kg)	C <sub>DGT-Sb</sub> (µg/L)	C <sub>sol-Sb</sub> (µg/L)	Ratio $R_{Sb}$ ( $R = C_{DGT}/C_{sol}$ )	K <sub>d</sub> (L/kg)	Soil	Total <sub>Sb</sub> (mg/kg)	C <sub>SEP-labile Sb</sub> (mg/kg)	C <sub>DGT-Sb</sub> (µg/L)	C <sub>sol-Sb</sub> (µg/L)	Ratio $R_{Sb}$ ( $R = C_{DGT}/C_{sol}$ )	K <sub>d</sub> (L/kg)
Sb1	$40 \pm 1$	$6.15\pm0.07$	$31.6\pm0.3$	$150\pm3$	$0.210\pm0.005$	41	AsSb1	$45\pm4$	$6.44\pm0.04$	$38.2\pm0.7$	$236\pm1$	$0.161\pm0.002$	27
Sb2	$78\pm2$	$13.0\pm0.1$	$74.8\pm0.9$	$340\pm3$	$0.222\pm0.001$	39	AsSb2	$73\pm7$	$12.7\pm0.1$	$67\pm2$	$670\pm5$	$0.095\pm0.002$	18
Sb3	$160\pm9$	$25.6\pm0.3$	$170 \pm 1$	$740 \pm 10$	$0.226 \pm 0.002$	34	AsSb3	$170\pm5$	$31.1\pm0.7$	$167\pm5$	$1767\pm2$	$0.095\pm0.003$	18
Sb4	$300\pm10$	$49.7\pm0.5$	$360\pm10$	$1650\pm20$	$0.219\pm0.005$	30	AsSb4	$310\pm4$	$57.9\pm0.6$	$406\pm 6$	$3130\pm60$	$0.130\pm0.001$	19
Sb5	$650\pm20$	$108.1\pm0.5$	$850\pm10$	$4250\pm20$	$0.201\pm0.002$	25	AsSb5	$320\pm8$	$74 \pm 1$	$450\pm20$	$4550\pm20$	$0.098 \pm 0.004$	16
Sb6	$1560\pm60$	$570\pm4$	$2690\pm20$	$16850\pm100$	$0.159 \pm 0.001$	34	AsSb6	$1620\pm30$	$806\pm10$	$3620\pm70$	$32650\pm400$	$0.111\pm0.002$	25
Sb7	$3000\pm60$	-	$5260\pm100$	$33700\pm200$	$0.156 \pm 0.004$	-	AsSb7	$3400\pm10$	$1800\pm20$	$6410\pm200$	$57080\pm200$	$0.112\pm0.003$	32
Sb8	$5300 \pm 40$	-	-	-	-	-	AsSb8	$5800\pm80$	-	-	-	-	-

-: not measured due to analysis problems

Arsenic and Sb in As+Sb soils were much more labile than As and Sb in historically cocontaminated soils (chapter 5) on account of the greater proportion of their  $C_{DGT}$  and  $C_{sol}$ (Figure D.3), despite their lower soil concentrations. This could de due to the shorter equilibration time and soluble As and Sb added.

# 6.3.4. Uptake of As and Sb by water spinach (Ipomoea aquatica)

#### 6.3.4.1. Plant growth yield

Water spinach cultivated in As-only soils showed a decrease in root and shoot biomass with increasing As concentrations (Figure 6.1A). Although the plants exhibited growth inhibition, toxicity symptoms such as wilting or chlorosis were not visible. The yield decreases were significant (p<0.05) when total soil As concentrations reached 300 mg/kg, corresponding to 653  $\mu$ g As/L in soil pore water. At soil concentrations of 510 mg As/kg, the root and shoot yield were 0.09 and 1.63 g/pot, respectively, which were 15.2% and 7.7% of the corresponding yield in control soils. Plant death occurred at the highest treatment of 600 mg As/kg soil.

For Sb, compared to controls, there was a slight decrease in the yield of *I. aquatica* grown in Sb-only soils of <300 mg/kg, followed by a gradual increase in plant biomass cultivated in soils  $\geq$ 300 mg Sb/kg, and a significant increase in plant tissues grown in soils >3000 mg Sb/kg (Figure 6.1B). This trend possibly resulted from the additional amounts of potassium accompanied with the use of KSb(OH)<sub>6</sub> as the Sb source, which stimulated the plant growth. For *I. aquatica* cultivated in As+Sb soils, the plant yield significantly decreased compared with the control yield, even when the plant was exposed to the lowest As+Sb treatment (Figure 6.1C). This indicates that the plants were sensitive to being exposed to a high amount of available As and Sb, leading to plant stress.



**Figure 6.1** Biomass of *I. aquatica* cultivated in soils amended with different concentrations of As and Sb. Each value is mean  $\pm$  SE (n = 3). The asterisks in the figure indicate significant differences (p<0.05) compared with the control plant tissues.

#### 6.3.4.2. Accumulation of As and Sb in *I. aquatica* tissues

Arsenic and Sb concentrations in roots and shoots of *I. aquatica* grown in contaminated soils are presented in Figure 6.2. In As-only soils, As concentrations in roots linearly increased with increasing soil As concentrations, reaching at 1260 mg As/kg roots when *I. aquatica* was exposed to 410 mg As/kg. Above this concentration, there was a decrease of root As. The similar trend was observed for shoot As, but the bioaccumulation being highest at 82 mg As/kg shoots in a lower soil concentration (Figure 6.2A). The decrease in the translocation of As to shoots corresponded with the significant decrease in plant biomass. On average, As accumulation in *I. aquatica* roots (100 – 1260 mg As/kg) was 13 times higher than shoots (14.3 – 82 mg As/kg). The average bioaccumulation factor (BAF) based on the ratio of total root concentration and total soil concentration (mg/kg, DW) of As was 3.62. The average translocation factor (TF) determined by the ratio of total shoot concentration (mg/kg, DW) of As was 0.09.

For Sb-only soils, Sb concentrations in roots linearly increased with increasing soil Sb until 1560 mg Sb/kg, after which it decreased. The same pattern for Sb concentrations in shoots was observed, but it was highest at the higher soil concentration (3000 mg Sb/kg) (Figure 6.2C). This occurred at the same time as a significant increase in root biomass (Figure 6.1B) which may have stimulated the translocation of Sb to shoots. The decrease in Sb bioaccumulation occurred at much higher soil concentration compared to the As scenario, suggesting that Sb was accumulated via a different uptake mechanism to As, or Sb induced a lower degree of toxicity. Like As, Sb bioaccumulation in roots was 12.6 times higher than shoots. The average BAF and TF of Sb was 0.16 and 0.1, respectively.

Root and shoot As uptake by *I. aquatica* cultivated in As+Sb soils was similar to that in Asonly soils. The root As concentration was highest in 310 mg As/kg soil and then declined as the soil As concentration increased, while the shoot As reached the accumulation saturation at 220 mg As/kg soil (Figure 6.2B). This may have occurred because the plant biomass significantly decreased. Roots also accumulated much more As than shoots, but to a lesser extent (a factor of 11) than that in As-only soils. The BAF of As (2.8) in As+Sb soils was lower than that in As-only soils, which can be explained by the slightly lower labile As discussed in section 6.3.3. The competitive uptake of the co-occurring Sb might also contribute to the lower BAF.

In regards to the response of *I. aquatica* exposed to As+Sb soils, the saturation of Sb bioaccumulation in roots and shoots followed the same pattern as Sb-only soils, but to a greater extent (Figure 6.2D). The average BAF of Sb in As+Sb soils was 0.39, 2.4 times higher than in Sb-only soils, which might be due to the higher labile Sb discussed in sections 6.3.2 and 6.3.3. However, based on the DGT measurements the resupply of Sb in As+Sb soils was known to be lower than that in Sb-only soils. This indicates that the sustained Sb concentration in soil solution was likely related to *I. aquatica* roots activities. Plants produce root exudates such as organic compounds which may displace adsorbed Sb from the solid phase to increase Sb solution concentration, replenishing Sb in the rhizosphere soil. Another possibility for the increase of Sb bioaccumulation is that As enhanced the Sb uptake by plant. Although *I. aquatica* accumulated more Sb in both roots and shoots in As+Sb treatments, the average translocation factor of Sb was slightly lower in co-contaminated soils than that in Sb-only soils. This might be due to the reduction of plant biomass which inhibited the translocation of Sb.

The decrease in As bioaccumulation corresponded with the significant decrease in plant yield. The occurrence of the lowest root and shoot As concentrations in the highest As treatments is most likely because of the severe growth retardation, which may have limited the transfer of As from soil to roots. In addition, plant detoxification mechanisms may have restricted the transport of As to roots and subsequent translocation to shoots. This is supported by previous studies in spinach shoot, amaranth and silverbeet (Helgesen & Larsen 1998; Rahman & Naidu 2009). According to Rahman and Naidu (2009), the likely mechanism utilised by vegetable crops to tolerate As is avoidance via limiting the transport of As from roots to shoots, possibly due to diverting energy from biomass production and increasing As accumulated in roots. This mechanism seems to be appropriate to *I. aquatica* in this study, which explains the accumulation saturation of shoots occurring earlier than that of roots.



Figure 6.2 As and Sb concentrations in *I. aquatica* tissues (mean  $\pm$  SE, n = 3) following cultivation in contaminated soils for 35 days.

The bioaccumulation factors of As and Sb varied among soils, depending upon their phytoavailabilities in soils. Compared to the historically As+Sb contaminated soils (chapter 5), the recently As+Sb amended soils exhibited substantial plant yield reduction and much greater bioaccumulation of As and Sb in *I. aquatica* cultivated in these soils, although the total soil As and Sb in these soils were much lower than historically contaminated soils (Figure 6.3). This is in agreement with the SEP and DGT data showing greater proportion of C<sub>SEP</sub>, C<sub>DGT</sub>, and C<sub>sol</sub> of As and Sb. The results reconfirm that total soil concentration is not appropriate for risk assessments of contaminated soils.



**Figure 6.3** Comparison of bioaccumulation factors (BAF) of As and Sb in water spinach (*I. aquatica*) cultivated in recently As+Sb amended soils ( $\Box$ ) and historically As+Sb contaminated soils ( $\blacksquare$ ) (chapter 5).

Overall, there is strong evidence of interaction between As and Sb with respect to *I. aquatica*. One hypothesis is competitive accumulation between As(III) and Sb(III) through aquaporins due to the similarities of glycerol,  $As(OH)_3$ , and  $Sb(OH)_3$  in terms of structure, neutral charge, and molecular size (Bhattacharjee et al. 2008). As(III) is transported via Si transport pathways which belong to the aquaporin family (Ma et al. 2008). The aquaporins allow the uncharged molecules with similar size relative to Sb to pass through (Mathews et al. 2011). Meharg and Jardine (2003) observed an inhibition of As accumulation by the Sb for rice. However, the interaction between As(III) and Sb(III) was not expected to be dominant in the current study since the As(V) and Sb(V) are the dominant species in aerobic soils (Okkenhaug et al. 2011).

It has been established that As(V) is transported through phosphate transporters (Meharg & Hartley-Whitaker 2002), but such interactions were not observed for Sb(V) and phosphate in Zea mays Tschan et al. (2008). Competition between As(V) and Sb(V) for the same membrane channels has not been established (Tschan et al. 2009b). Thus, since the mechanism of Sb(V) uptake is still unknown, it can only be speculated that As(V) altered membrane integrity and hence the permeability of Sb(V). For the present study, it is possible that the addition of a more toxic element (As) into soils damaged the cell membrane and allowed Sb adsorption and transportation. The stimulation of Sb uptake due to As presence and the inhibition of As uptake due to Sb presence observed here are supported by hydroponic and quartz substrate studies by Feng et al. (2011) and Müller et al. (2013) reporting that the addition of As resulted in an increase of Sb uptake by Ashyperaccumulator Pteris cretica and Pteris vitta. Further analysis of subcellular distribution by Feng et al. (2011) showed that the increase of Sb concentrations associated with more transfer of Sb into the cytosol fractions, and less in the cell wall and cytoplasmic organelles. In contrast, the presence of Sb decreased As uptake by Pteris cretica when exposed to high concentrations of As and Sb, leading to the reduction of As in cytosol fractions in plant tissues, suggesting that cytosol may be an important reservoir. In the current study, water spinach (I. aquatica) can be classified as an As accumulator plant due to BAF of As being > 1 in both historically and recently contaminated soils. Thus it is not unreasonable to extrapolate these mechanisms of uptake and distribution in "accumulators" of As and Sb by *I. aquatica*. The determination of As and Sb in subcellular fractions of *I. aquatica* tissues needs to be done to confirm those similarities.

#### 6.3.5. The relationship between As and Sb bioaccumulation and soil concentrations

The relationship between As and Sb bioaccumulation and soil concentrations was only linear until a certain concentration after which bioaccumulation decreased corresponded with growth inhibition (Figure 6.2). Correlation analyses were performed on data where dose responses of plant roots gave linear relationships (expressed as correlation coefficients) between tissue As and Sb concentrations and their labile fractions measured by SEP, DGT, and soil solution concentrations (Table 6.5). Arsenic concentrations in roots were significantly correlated with all measures of labile fractions compared to shoot As concentrations in shoots may be strongly controlled by plant physiology. This is evident by the fact that at the 300 mg As/kg soil treatment, the roots continued to accumulate As, while shoot As decreased. The shoot As concentrations were weakly correlated with soil As concentrations (low correlation coefficients, Table 6.5), which can be explained by the contrasting relationships between increased As concentrations in soils and decreased shoot As concentrations and plant yield.

As discussed in section 6.3.4, shoot As concentrations increased with increasing soil As concentrations up to a certain exposure concentration, after which As accumulation decreased corresponding to the reduction in plant biomass. When the saturated bioaccumulation data point was excluded (from soil treatment >300 mgAs/kg soil), shoot As concentrations were significantly correlated with all measures,  $C_{SEP}$ ,  $C_{DGT}$ , and  $C_{sol}$  (correlation coefficients for shoot As concentrations > 0.9 for all cases). This illustrated that all measures worked well in predicting As uptake by *I. aquatica* grown in the low range of labile As in soils ( $C_{SEP-As} <70$  mg/kg,  $C_{DGT-As} <90$  µg/L,  $C_{sol-As} <260$  µg/L). For *I. aquatica* grown in As-only soils,  $C_{SEP-As}$  best predicted As concentrations in tissues. This implies that the readily available and exchangeable As phases in the soils were important and responsible for plant uptake. For *I. aquatica* cultivated in As+Sb soils, the performance of DGT in predicting As in plants was improved and  $C_{SEP-As}$  and  $C_{DGT-As}$  better predicted As in plant tissues than  $C_{sol-As}$ . These results suggest that there was a labile pool of As from the

solid phase and dissolved complexes that were consistent with the R-value results (section 6.3.3), which is important from the bioavailability perspective. It is interesting that the labile extraction ( $C_{SEP}$ ) was more highly correlated than DGT, perhaps there was direct uptake by roots in contact with soils.

In contrast, all lability measures were similar in predicting Sb accumulation by *I. aquatica*.  $C_{SEP}$ ,  $C_{DGT}$ , and  $C_{sol}$  and tissue concentrations correlated more strongly for Sb than for As, which suggests that the reduction of plant yield resulted more from As than from Sb exposure. The Sb concentrations in shoots obtained from As+Sb soils were slightly less correlated with  $C_{SEP-Sb}$  and  $C_{DGT-Sb}$  in the soils than from Sb-only soils. This might be the higher Sb bioaccumulation in As+Sb soils due to the presence of As in soils which contributed to the slightly weaker relationships between shoot Sb concentrations and these soil labile fractions. Similar to As, Sb concentrations in shoots of *I. aquatica* grown in As+Sb soils were not as well predicted by  $C_{SEP-Sb}$  and  $C_{DGT-Sb}$  in the soils as the root Sb concentrations. The explanation for As can be applied for the Sb scenario here, which is because root accumulation was affected by soil processes, while shoot uptake was regulated by plant physiology to control the translocation of Sb from roots to shoots.

**Table 6.5** Correlation coefficients ( $R^2$ ) between As and Sb concentration in tissues of *I. aquatica* cultivated in different soils and  $C_{SEP}$ ,  $C_{DGT}$ , and  $C_{sol}$  of As and Sb in soils.

Treatments	As		Treatments	Sb	
	C <sub>Root-As</sub>	C <sub>Shoot-As</sub>		C <sub>Root-Sb</sub>	$C_{\text{Shoot-Sb}}$
As-only			Sb-only		
$C_{SEP-As}$	$0.96^{**}$	0.72	$C_{SEP-Sb}$	$0.99^{**}$	$0.99^{**}$
C <sub>DGT-As</sub>	$0.82^{*}$	0.49	C <sub>DGT-Sb</sub>	$0.99^{**}$	$0.99^{**}$
$C_{sol-As}$	$0.80^{*}$	0.47	$C_{sol-Sb}$	$0.99^{**}$	$0.99^{**}$
As+Sb			As+Sb		
$C_{SEP-As}$	$0.97^{**}$	0.69	$C_{SEP-Sb}$	$0.98^{**}$	$0.91^{*}$
C <sub>DGT-As</sub>	$0.96^{**}$	0.63	C <sub>DGT-Sb</sub>	$0.99^{**}$	$0.85^{*}$
$C_{sol-As}$	$0.86^{*}$	0.46	$C_{sol-Sb}$	$0.96^{**}$	$0.95^{**}$

\*\*: statistically significant (p<0.01)

\*: statistically significant (p<0.05)

#### 6.3.6. Risk assessment of As and Sb in *I. aquatica* to human health

Similar to water spinach (*I. aquatica*) grown in historically contaminated soils (chapter 5), the As concentrations in edible shoots of *I. aquatica* cultivated in amended soils (mg/kg WW, Table 6.6) were higher than the maximum permissible value in food, specific to cereals (1 mg As/kg WW) set by the Food Standard Agency of Australia and New Zealand (FSANZ 2013) and permissible value for food (0.1 mg As/kg, WW) recommended by joint Food and Agricultural Organisaion (FAO)/World Health Organisation (WHO) (JECFA 2011). Thus, there is high risk of consuming this leafy vegetable cultivated in the studied As-contaminated soils. The joint FAO/WHO Expert Committee on Food Additives (JECFA) has not reported guidelines relating to Sb safety. Nevertheless, the USEPA has recommended an oral reference dose (RfD) for Sb being 0.0004 mg/kg of body weight (BW)/day. The exposure parameters used for the calculation of estimated daily intake (EDI), hazard quotient (HQ), and cancer risk (CR) in sections 5.2.7 and 5.3.6 of chapter 5 were applied in this chapter.

The values of EDI, HQ, and CR of As and Sb through the consumption of *I. aquatica* shoots are presented in Tables 6.6 and 6.7, respectively. The EDI values of As for both adults and children were higher than the acceptable limit or the oral reference dose of As (RfD,  $3 \times 10^{-4}$  mg/kg/day) set by USEPA (2011). The HQ values were calculated to assess the potential risk of non-cancer adverse health effect to human through As and Sb ingestion. The HQ values of As were > 1 for all cases, ranging from 3.91 - 22.3 for adults and 6.43 – 36.7 for children, indicating the substantial risk of non-carcinogenic effects for both adults and children consuming *I. aquatica* cultivated in contaminated soils with As  $\geq 46$  mg/kg corresponding to C<sub>DGT-As</sub>  $\geq 9.7$  µg/L and C<sub>sol-As</sub>  $\geq 17.36$  µg/L. The EDI values for Sb for both adults and children consuming *I. aquatica* from Sb-only soil ( $\geq 650$  mg Sb/kg corresponding to C<sub>DGT-Sb</sub>  $\geq 406$  µg/L and C<sub>sol-Sb</sub>  $\geq 3130$  µg/L) were higher than the oral reference dose of Sb (RfD,  $4 \times 10^{-4}$  mg/kg/day) set by USEPA (1991). The HQ values of Sb were > 1 for Sb-only soil ( $\geq 650$  mg Sb/kg) and As+Sb soil ( $\geq 310$  mg Sb/kg), ranging from 1.13 – 30.5 for adults and 1.87 – 50.2 for children, indicating the significant risk of

non-carcinogenic effects for both adults and children consuming *I. aquatica* cultivated in Sb-contaminated soils in those concentration ranges.

The higher potential risk of *I. aquatica* from As+Sb soil compared to Sb-only soil is due to its higher lability of Sb in these As+Sb soils resulting in higher Sb bioaccumulation. It is noticeable that As presented higher non-carcinogenic risks than Sb, demonstrated by the much higher HQ values of As derived from the consumption of *I. aquatica* grown in soils with comparable concentrations of As and Sb. In addition, HQ values of As were much higher than 1 when *I. aquatica* grown in the lowest soil treatments was consumed, while HQ values of Sb were only higher than 1 when *I. aquatica* exposed to Sb-only soil ( $\geq$ 650 mg Sb/kg) and As+Sb soil ( $\geq$ 310 mg Sb/kg) was ingested.

The carcinogenic risk (CR) was calculated to assess the probability of the exposed population developing cancer risk through ingestion of As-contaminated vegetables during their lifetime. Due to the lack of the cancer slope for Sb, only CR derived from the consumption of As-enriched *I. aquatica* was calculated. The CR values increased with the increasing soil As and far exceeded the probability of 1 in 10,000, indicating an increase in probability of cancer development via consumption of *I. aquatica* shoots cultivated in soils contaminated with As  $\geq$ 46 mg/kg corresponding to C<sub>DGT-As</sub>  $\geq$ 9.7 µg/L and C<sub>sol-As</sub>  $\geq$ 17.36 µg/L.

In general, the potential health risks and cancer risks were greater for children than adults. The consumption of *I. aquatica* cultivated in amended soils with such concentrations poses much higher risks (both non-carcinogenic and carcinogenic effects) to humans than that in historically contaminated soils (section 5.3.6, chapter 5). This is due to the higher edible shoot bioaccumulation resulting from the higher lability of As and Sb in freshly amended soils compared to historically contaminated soils.

Soil	Treatment	Soil As (mg/kg DW)	Shoot As*	Estimated daily	v intake (EDI)	Hazard quo	otient (HQ)	Cancer r	isk (CR)
5011			(mg/kg WW)	Adults	Children	Adults	Children	Adults	Children
As-only	As1	46	1.43	$11.7 \times 10^{-4}$	$19.3 \times 10^{-4}$	3.91	6.43	$17.6 \times 10^{-4}$	$86.9  imes 10^{-4}$
	As2	84	3.49	$28.6\times10^{4}$	$47.0  imes 10^{-4}$	9.52	15.7	$42.8\times10^{4}$	$212\times10^{4}$
	As3	165	6.63	$54.3  imes 10^{-4}$	$89.5  imes 10^{-4}$	18.1	29.8	$81.5 imes10^{-4}$	$403  imes 10^{-4}$
	As4	196	8.16	$66.8  imes 10^{-4}$	$110  imes 10^{-4}$	22.3	36.7	$100.2\times10^{4}$	$495\times 10^{\text{-4}}$
As+Sb	AsSb1	60	1.65	$13.5 \times 10^{-4}$	$22.3 \times 10^{-4}$	4.51	7.42	$20.3  imes 10^{-4}$	$100  imes 10^{-4}$
	AsSb2	87	3.75	$30.7\times10^{4}$	$50.5  imes 10^{-4}$	10.2	16.8	$46.0\times10^{4}$	$227  imes 10^{-4}$
	AsSb3	175	6.45	$52.9  imes 10^{-4}$	$87.1  imes 10^{-4}$	17.6	29.0	$79.3  imes 10^{-4}$	$392\times 10^{\text{-}4}$
	AsSb4	220	8.08	$66.2  imes 10^{-4}$	$109  imes 10^{-4}$	22.1	36.3	$99.3\times10^{\text{-}4}$	$490\times 10^{\text{-}4}$

**Table 6.6** Values of shoot As (mg/kg, WW), EDI (mg/kg/day), non-carcinogenic risk (HQ), and cancer risk (CR) for the ingestion of water spinach (*I. aquatica*) shoots grown in As-only and As+Sb soils.

\* Concentration of As in shoots of *I. aquatica*; parameters are presented where a dose response was observed.

Soil	Treatment	Soil Sb	Shoot Sb <sup>*</sup>	Estimated daily	y intake (EDI)	Hazard qu	otient (HQ)
5011	Traincin	(mg/kg DW)	(mg/kg WW)	Adults	Children	Adults	Children
Sb-only	Sb1	40	0.08	$0.68  imes 10^{-4}$	$1.11  imes 10^{-4}$	0.17	0.28
	Sb2	78	0.11	$0.91\times 10^{\text{-}4}$	$1.50 imes10^{-4}$	0.23	0.37
	Sb3	160	0.15	$1.26\times 10^{\text{-}4}$	$2.07  imes 10^{-4}$	0.31	0.52
	Sb4	300	0.24	$2.01  imes 10^{-4}$	$3.30\times10^{4}$	0.50	0.83
	Sb5	650	0.55	$4.54\times10^{4}$	$7.47 imes10^{-4}$	1.13	1.87
	Sb6	1560	1.41	$11.6\times10^{4}$	$19.0\times10^{4}$	2.89	4.76
	Sb7	3000	3.63	$29.7\times10^{\text{-}4}$	$48.9\times10^{4}$	7.43	12.2
As+Sb	AsSb1	45	0.06	$0.51  imes 10^{-4}$	$0.84 imes10^{-4}$	0.13	0.21
	AsSb2	73	0.10	$0.78  imes 10^{-4}$	$1.29  imes 10^{-4}$	0.20	0.32
	AsSb3	170	0.19	$1.59\times10^{4}$	$2.63\times 10^{4}$	0.40	0.66
	AsSb4	310	0.46	$3.75  imes 10^{-4}$	$6.18  imes 10^{-4}$	0.94	1.54
	AsSb5	320	0.89	$7.27  imes 10^{-4}$	$12.0\times10^{4}$	1.82	2.99
	AsSb6	1620	9.96	$81.6\times10^{4}$	$134  imes 10^{-4}$	20.4	33.6
	AsSb7	3400	14.89	$122  imes 10^{-4}$	$201  imes 10^{-4}$	30.5	50.2

**Table 6.7** Values of shoot Sb (mg/kg, WW), EDI (mg/kg/day), non-carcinogenic risk (HQ), and cancer risk (CR) for the ingestion of water spinach (*I. aquatica*) shoots grown in Sb-only and As+Sb soils.

\* Concentration of Sb in shoots of *I. aquatica*; parameters are presented where a dose response was observed.

It is important to note that the consumption of *I. aquatica* cultivated in Sb-only soils ( $\geq 650 \text{ mg Sb/kg}$ ) poses unacceptable non-carcinogenic risk to humans, but *I. aquatica* grew healthily, had no physical toxicity symptoms, and greater biomass of edible shoots. As discussed in section 6.3.4.1, the tissue biomass of *I. aquatica* grown in Sb-only soils displayed the increasing trend with the addition of soluble Sb into soils. Consequently, the Sb-enriched edible shoots of *I. aquatica* cultivated in Sb-only soils cannot be recognised and distinguished compared to As-enriched shoots of *I. aquatica* in Sb-only soils displayed to high soil concentrations. This implies that growing *I. aquatica* in Sb-contaminated soils should be in caution because the plant presents hidden risks and may pose non-carcinogenic risks to human health via the food chain.

## 6.4. Conclusions

The SEP data showed that amorphous and crystalline iron oxides were the most two dominant fractions for Sb, whereas As was mainly found in amorphous iron oxide and specifically sorbed fractions. High loads of both As and Sb into soils increased the proportions of As and Sb in non-specifically sorbed fraction. The presence of As in As+Sb soils significantly increased Sb concentrations in soil solutions. However, the resupply of Sb in As+Sb soils was lower than that in Sb-only soils and the resupply of Sb was slower than that of As.

The *I. aquatica* response varied according to concentration ranges of As and Sb in soils, demonstrated by bioaccumulation of As and Sb saturated at a certain soil concentration, then decreasing with the significant decline in plant yield. Root As and Sb concentrations were >10 times higher than their shoot concentrations. The interesting finding was the clear enhancement of Sb uptake by *I. aquatica* cultivated in As+Sb soils compared with Sb-only soils. On average, the increase of Sb accumulation was not accompanied by an increased translocation of Sb from roots to shoots.

The strong positive correlations between labile fractions of As and Sb in soils and their concentrations in *I. aquatica* tissues were only observed in non-toxic ranges, confirming that labile measures do not work well in predicting the bioaccumulation of elements when plants suffer from toxic effects.  $C_{SEP}$ ,  $C_{DGT}$ , and  $C_{sol}$  more strongly correlated with tissue concentrations of Sb than As and root concentrations than shoots. SEP, DGT, and soil solution concentrations were relatively similar in predicting As and Sb

accumulation by *I. aquatica*. DGT performed comparatively well with other measures, but DGT measurements are probably simpler than some of the other measurements which required single or multiple extractions of soil samples and soil characterisation. The combination of these techniques was useful in understanding the biogeochemical behaviour of As and Sb in individual and co-occurring soils as well as interactive effects of As and Sb on accumulation by *I. aquatica*.

The As concentrations in shoots of *I. aquatica* far exceeded the maximum permissible values set by the Food Standard Agency of Australia and New Zealand (1 mg As/kg, WW) and FAO/WHO (0.1 mg As/kg, WW) and As posed higher non-carcinogenic and carcinogenic risks than Sb. This is the first study illustrating the competitive interactions of As and Sb in aerobic soils and in an important edible vegetable, thus is of value for risk assessments of As and Sb contaminated soils and managements of their concentrations in edible vegetables for food safety.

# Chapter 7. Mobilisation of As and Sb in soils induced by redox changes identified by DGT and their uptake by water spinach (*Ipomoea aquatica*)

# 7.1. Introduction

Arsenic (As) and antimony (Sb) frequently co-occur in extremely high concentrations on mining and smelting sites and surrounding soils (Anawar et al. 2011; Ashley et al. 2007; Okkenhaug et al. 2012; Pratas et al. 2005). The retention and mobilisation of As and Sb in soils depends on the soil properties (e.g. Fe oxides, organic matter), environmental conditions (e.g. redox potential, pH), and their speciation (Wilson et al. 2010). Iron (hydr)oxides are important sorbents for the immobilisation of As and Sb in soils via adsorption, precipitation, and co-precipitation (Mitsunobu et al. 2006; Wilson et al. 2010), thus the lability and solubility of As and Sb in oxic soils were low (Álvarez-Ayuso et al. 2012; Ettler et al. 2010; Kim et al. 2014; Liu et al. 2015; Okkenhaug et al. 2012).

When As contaminated soils were subjected to anoxic conditions (e.g. waterlogging), the As was rapidly mobilised, leading to increased soluble As in soil pore water, dominated by As(III) (Garnier et al. 2010; Takahashi et al. 2004; Xu et al. 2008). Xu et al. (2008) reported that arsenic concentrations in soil pore water were 7-16 and 4-13 times higher under the waterlogged than under the non-waterlogged conditions in soil without As addition and in soils amended with 10 mg As/kg, respectively. In contrast to As, studies on Sb behaviour in anoxic soils are inconsistent and mechanisms controlling Sb speciation and concentrations have been unclear. Mitsunobu et al. (2006) reported only 6% of Sb present as Sb(III) in the solid phase and soil solution under reducing conditions. Later studies reported the effect of redox condition on Sb speciation with Sb(III) constituting 5 - 19% of total Sb in mining-impacted soil pore water (Okkenhaug et al. 2012) and 60% of total Sb in anaerobic shooting range soil pore water (Wan et al. 2013a). It has been reported that reducing conditions generally decreased Sb mobility (Mitsunobu et al. 2006; Wan et al. 2013b), whereas an incubation study by (Hockmann et al. 2014a) stated that reducing conditions actually increased Sb mobility. The varied results of Sb speciation in reduced soil pore water are likely due to changes in Sb

153

species during sampling and analysis. According to Oorts et al. (2008), Sb(III) is very unstable and is oxidized to Sb(V) within 2 days under oxic conditions. It is difficult to preserve and analyse Sb species in environmental samples (Hansen & Pergantis 2008). Thus, collection and analysis of Sb species in soil pore water for accurate assessment of Sb behaviour in contaminated soils and its risks still remains a challenge.

As discussed in previous chapters, diffusive gradients in thin films (DGT) and diffusive equilibration in thin films (DET) techniques are in situ samplers allowing the in-situ study of distribution and concentration profiles of solute in sediment pore water at higher spatial resolution than conventional techniques (Davison et al. 2000; Zhang et al. 1995; Zhang et al. 2002). The Metsorb DGT coupled with DET for Fe was applied to investigate the sediment biogeochemistry of As and Fe in spatial resolution and demonstrated the mechanism of As mobilisation in sediments (Bennett et al. 2012b). The As and Sb species in sediments have not been investigated using DGT because the mercapto-silica DGT for As(III) and Sb(III) measurements has recently been developed and initially applied to measure As(III) in water (Bennett et al. 2011) and Sb(III) in water (Bennett et al. 2016b). With the newly developed techniques, we are able to measure in-situ total inorganic As and Sb, together with individual As and Sb species in water, sediment, and soil pore water by simultaneous use of the combination of Metsorb DGT and mercapto-silica DGT, avoiding artifacts of species changes during sample collection and analysis. The use of Metsorb DGT and mercapto-silica DGT, coupled with Fe-DET, will allow the mechanisms of As and Sb mobilisation in soils to be investigated.

In terms of the effect of waterlogging on As and Sb uptake by plants, waterlogging Ascontaminated soils increased As accumulation in plants cultivated in those soils (Punshon et al. 2017; Wu et al. 2017; Xu et al. 2008). Xu et al. (2016) stated that growing rice in As-contaminated soils subjected to flooding dramatically enhanced As concentrations in rice grains by a factor of 10-15 fold. Arsenic concentrations in rice roots were markedly higher in anoxic treatments (147 - 243 mg/kg) compared to oxic treatments (88.8 - 218 mg/kg) (Wu et al. 2017). A preliminary study on Sb uptake by plants grown in waterlogged contaminated soils showed a contrasting result. Waterlogging Sb-contaminated soils increased Sb concentrations in shoots of *Lolium perenne* by 10 times, but decreased uptake by *Holcus lanatus* by 80% (Wan et al. 2013a). Most studies on the effect of waterlogging on As and Sb uptake by plants used singly contaminated soils as test soils and rice and grass as test plants. There has been no published study on the effects of flooding on simultaneous uptake of As and Sb by edible crops cultivated in co-contaminated soils and their competitive uptake by vegetables exposed to flooded contaminated soils.

Previous studies (chapters 3, 4, 5) showed that As and Sb in historically contaminated soils were mainly present in the Fe oxides and residual phases and their lability in these aerobic soils were low. It is expected that when the historically contaminated soils are subject to waterlogging, As and Sb will be more labile and mobilised, which leads to an increase in their accumulation in the cultivated edible plants. In addition, the results of chapter 5 showed that water spinach (Ipomoea aquatica) grown in historically contaminated soils accumulated As in edible shoots at high concentrations exceeding the maximum permissible value for food (1 mg As/kg, WW) set by the Food Standard Agency of Australia and New Zealand (FSANZ 2013), although the plant did not exhibit toxic symptoms such as wilting and chlorosis. Furthermore, I. aquatica is a common edible vegetable in Asia and can grow in both terrestrial and semiaquatic soil systems. All these matters become alarming concerns. Thus, it is essential to examine the behaviour of As and Sb in contaminated soils and I. aquatica exposed under different scenarios and determine As and Sb bioaccumulation and speciation in I. aquatica cultivated under various soil conditions to further accurately assess risks to humans and ensure food safety.

This study aims to investigate (i) changes in solubility of As and Sb and their mobilisation mechanisms in historically co-contaminated soils, freshly co-contaminated soils, and soils freshly contaminated with As only and Sb only in response to waterlogging using several DGT/DET techniques; (ii) the effect of using semiaquatic cultivation on the solubility and speciation of As and Sb in soils; (iii) the uptake of As and Sb by water spinach (*I. aquatica*) grown in terrestrial and flooded contaminated soils; and (iv) the speciation transformation of As in the *I. aquatica* by HPLC-ICP-MS. To our best knowledge, this is the first study to measure the dynamics and chemistry of As and Sb species in flooded contaminated soils using the DGT and DET and to determine for the first time As speciation in flooded *I. aquatica* compared to that in non-flooded *I. aquatica*. In addition, the study on As-only amended soil, Sb-only amended soil, and As+Sb amended soil allows us to understand if there is the competitive effect of As on the solubility of Sb in anoxic soils and vice versa.
## 7.2. Methods

### 7.2.1. General method

General washing methods were done as per section 2.1.

# Materials, reagents, and standards for HPLC-ICP-MS

Sub-boiled double distilled nitric acid (Aristar, BDH) and deionised water were used in the extraction of As species from plant samples. Ammonium dihydrogen orthophosphate (Suprapur, Merck) and pyridine (Extra Pure, Merck) were used for the mobile phases. Ammonia solution (>99.9%, Aldrich) and formic acid (Extra Pure, Fluka) were used for pH adjustment of mobile phases for HPLC. Arsenous acid (As(III)), arsenic acid (As(V)), methylarsonic acid (MA), and dimethylarsinic acid (DMA) were prepared by dissolving sodium arsenite, sodium arsenate heptahydrate (AJAX Laboratory Chemicals), disodium methylarsenate (Alltech-Specialists), and sodium dimethylarsinic (Alltech-Specialists), respectively, in deionised water immediately before use.

## 7.2.2. Experimental design

# 7.2.2.1. Soil preparation

A historically As and Sb-contaminated soil was collected from a decommissioned antimony processing facility in Urunga NSW, Australia. Details about soil physicochemical properties were described previously (section 3.2.2.1). An uncontaminated soil (control soil) was collected in Wollongong, NSW. The soils were air-dried and crushed to pass a 2-mm sieve. The historically contaminated soil was thoroughly mixed with the uncontaminated soil to establish one low treatment (Aged-L) and one high treatment (Aged-H) of As and Sb. In addition, separate portions of uncontaminated soil was amended with either solutions of sodium arsenate, potassium antimonate, or their mixture to establish one treatment of each spiked soil, namely Asonly soil (As50), Sb-only soil (Sb300), and As+Sb soil (As50Sb300). Following the addition of As/Sb solutions into soils, the spiked soils were homogenised by mixing and turning over vigorously. Moisture content of soils was retained at 60% of maximum water holding capacity (MWHC) after spiking. The artificially contaminated soils were air-dried, crushed using a mortar and pestle, and mixed homogeneously for the bioassays.

All test soils were applied nutrients and allowed to equilibrate as described in section 2.4.2. Soils As50, Sb300, and As50Sb300 are considered recently contaminated soils.

### 7.2.2.2. Experimental treatment and management

Pots were filled with 4 kg of each processed soil. Soil pots without water spinach (*Ipomoea aquatica*) and pots with *I. aquatica* were prepared in triplicate. Two water regimes were applied: (i) non-flooded, in which soils were maintained at 60% of MWHC and (ii) flooded, in which soils were thoroughly mixed with water and flooded to about 4-5 cm above the surface which is typical of semiaquatic agronomy practices. All soil pots were transferred to controlled growth chambers, as described in section 2.4.2. After one week of flooding, a pair of Metsorb-DGT and mercapto-silica-DGT probes was gently inserted into the centre of each flooded soil treatment pot for 24 h to measure labile As and Sb (see DGT procedures below for details). *I. aquatica* were then transplanted into all soil pots (both flooded and non-flooded treatments) the following day. Water was added daily to maintain flooding of soils and pots were randomly rearranged every week.

After 5 weeks of growth in soils, the second DGT deployment was conducted in the same manner as the first deployment. *I. aquatica* were harvested the following day and washed vigorously with deionised water to remove adhering soil before being separated into shoots and roots with a stainless steel blade. After removal of the iron layer from the roots (see procedure below) and washing with deionised water, roots and shoots were freeze dried for 48 h at -50 °C. All dried samples were ground to a fine powder under liquid nitrogen with a ceramic mortar and pestle and stored in clean plastic bags for further analysis.

# **Extraction of iron plaque**

Iron plaque on fresh roots was extracted using dithionite-citrate-bicarbonate (DCB) method as described previously (Liu et al. 2004). The whole fresh root system was soaked in 50 mL of 0.03 M sodium citrate (Na<sub>3</sub>C<sub>6</sub>H<sub>5</sub>O<sub>7</sub>.2H<sub>2</sub>O) and 0.125 M sodium bicarbonate (NaHCO<sub>3</sub>) for 10 min at room temperature (20 - 25 °C). Then 0.6 g of sodium dithionite (Na<sub>2</sub>S<sub>2</sub>O<sub>4</sub>) was added and each root was incubated for another 60 min. Subsequently, roots were taken out and rinsed three times with deionised water. All rinsed water was collected and combined with its corresponding DCB extract. The volume of final solutions was made up to 100 mL with deionised water.

## 7.2.3. Soil and plant analyses

### 7.2.3.1. Total As and Sb analysis

The digestion and analysis of soil and plant samples were done as per sections 5.2.4 and 2.9. For quality control, reagent blanks, duplicate analyses, certified reference materials, and matrix spike recovery were used. A soil certified reference material (Montana II soil, SRM 2711a; NIST) returned recoveries of 86-118% for the studied analytes. A plant certified reference material (Tomato leaves, SRM 1573a; NIST) had typical recoveries of 88-116% of the studied analytes.

### 7.2.3.2. HPLC-ICP-MS inorganic As analysis

The extraction of inorganic As species was performed as per Foster et al. (2007) and Maher et al. (2013) with minor modifications. Briefly, an accurately known mass (approximately 0.2 g) of water spinach sample (*I. aquatica*) was weighed into 55 mL polytetrafluoroacetate (PTFE) digestion vessels (CEM) and combined with 5 mL of 2% v/v HNO<sub>3</sub>. Samples were heated in a Mars Xpress microwave oven at 110 °C for 10 min, cooled to room temperature and combined with 5 mL of deionised water. The supernatants were removed by centrifuging at 5000 rpm for 10 min and filtering. The extract was further diluted 1:1 (v/v) with deionised water prior to chromatography to get adequate peak separation between arsenite and DMA.

The HPLC system consisted of a Perkin-Elmer series 200 mobile phase delivery and autosampler system (Perkin-Elmer). Column conditions used for the separation of anionic As species are described by Kirby et al. (2004). A PEEK PRP-X100 anion exchange column (250 mm × 4.6 mm, 10  $\mu$ m) (Phenomenex) using 20 mM ammonium phosphate buffer mobile phase at pH 4.5, flow rate 1.5 mL min<sup>-1</sup>, column temperature of 40 °C, and injection volume of 40  $\mu$ L was used for the identification and quantification of As(III), As(V), DMA, and MA. The eluant from HPLC columns was directed by PEEK (polyether-ether-ketone) (i.d. 0.02 mm) (Supelco) capillary tubing into a Ryton cross-flow nebulizer of a Perkin-Elmer Elan-6000 ICP-MS, which was used to monitor the signal intensity of As at m/z 75. The potential interference to As (m/z 75) from chloride (<sup>40</sup>Ar<sup>35</sup>Cl) was determined by monitoring chloride at m/z 35, <sup>40</sup>Ar<sup>37</sup>Cl at m/z 77. Selenium was monitored at m/z 82 as a cross check on <sup>40</sup>Ar<sup>37</sup>Cl interference.

### 7.2.4. DGT procedures

### 7.2.4.1. DGT probe preparation

Metsorb DGT probes for measuring total labile inorganic As and Sb and mercapto-silica DGT probes for measuring labile As(III) and Sb(III) were prepared as per section 2.7.

# 7.2.4.2. DGT deployment and retrieval

All DGT probes were deoxygenated by immersion in a clean 0.01 M NaCl solution and bubbled with nitrogen for 24 h prior to deployment. A pair of deoxygenated DGT probes (one Metsorb-DGT and one mercapto-silica-DGT) arranged back to back, was gently inserted into the flooded soil pots. DGT deployment was performed one day before transplanting plants and one day before harvesting plants at  $25 \pm 1$  °C for 24 h. On retrieval, the soil water interface (SWI) was marked, the probes were rinsed well with deionised water to remove sediment particles, immediately put in cleaned plastic zip lock bags, and kept in a refrigerator until disassembly. The diffusive gels from mercapto-silica DGT probes were used as diffusive equilibration in thin films (DET) for measuring Fe, thus mercapto-silica DGT probes were disassembled immediately. On disassembly, binding and diffusive gels for Fe were sliced at 10 mm intervals with two slices above the SWI and eight slices below the SWI using a Teflon coated razor blade. Each DET slice was eluted in 3 mL of 2% HNO<sub>3</sub> for 24 h (Gao et al. 2007; Garnier et al. 2015; Gregusova & Docekal 2013). Each DGT resin slice was eluted in 1 mL of solution with a final concentration of 1 M NaOH : 1 M H<sub>2</sub>O<sub>2</sub> for 24 h (Bennett et al. 2016b; Panther et al. 2013). The DGT extracts were diluted 20 fold prior to analyis by ICP-MS.

Blank DGT probes were treated in the same way as the sample probes. DGT-measured concentrations,  $C_{DGT}$ , of As and Sb were calculated using Equation 7.1 (Zhang & Davison 1995).

$$C_{DGT} = M\Delta g / (DAt)$$
 Equation 7.1

Where M is the mass of elements on the binding gel,  $\Delta g$  is the thickness of the diffusive gel (0.8 mm) plus the thickness of filter membrane (0.1 mm), D is the diffusion coefficient of elements in the diffusive gel (5.69 × 10<sup>-6</sup> cm<sup>2</sup> s<sup>-1</sup> for As(V), 9.04 × 10<sup>-6</sup> cm<sup>2</sup> s<sup>-1</sup> for As(III) and 6.21 × 10<sup>-6</sup> cm<sup>2</sup> s<sup>-1</sup> for Sb(V), 9.42 × 10<sup>-6</sup> cm<sup>2</sup> s<sup>-1</sup> for Sb(III) at 25 °C), A is the exposed area of the gel slice (cm<sup>2</sup>), and t is the deployment time (s).

### 7.2.5. Data analysis

Statistical analysis and calculations of bioaccumulation factor (BAF) and translocation factor (TF) were performed as per section 3.2.7.

### 7.3. Results and discussion

#### 7.3.1. DGT measurements in flooded historically contaminated soils

# 7.3.1.1. Dynamics of lability and speciation of As and Sb in flooded historically contaminated soils

The C<sub>DGT</sub> profiles of As, As(III), As(V), Sb, Sb(III), and Sb(V) in soil Aged-L were determined one and five weeks after flooding with and without plants, and are presented in Figure 7.1. After 1 week of flooding (before transplanting), C<sub>DGT-Sb</sub> was higher than C<sub>DGT-As</sub> in both the overlying water (OLW) and soil layer. C<sub>DGT-As</sub> and C<sub>DGT-Sb</sub> slightly increased at mid-profiles and then decreased (Figure 7.1A). The higher C<sub>DGT-As</sub> and C<sub>DGT-Sb</sub> in the OLW indicates that there were fluxes of As and Sb into the OLW. For speciation, both C<sub>DGT-As(III)</sub> and C<sub>DGT-Sb(III)</sub> were constantly low until a depth of 10-mm, after which C<sub>DGT-As(III)</sub> exhibited a sharp increase from 20 to 40-mm depth, followed by a slight increase, while C<sub>DGT-Sb(III)</sub> displayed a sharp increase to from 20 to 50-mm depth, followed by a decrease. The average proportion of C<sub>DGT-As(III)</sub> and C<sub>DGT-Sb(III)</sub> in deep soil layer (from 20-mm depth downwards) accounted for 62% and 11% of total labile As and Sb species, respectively (Figure 7.1B). The presence of trivalent species indicates that the anaerobic condition developed and the mobilisation of As and Sb occurred at the initial stage of plant bioassay. In contrast, both CDGT-As(V) and CDGT-Sb(V) were high in the OLW and displayed a decreasing trend from the OLW until the bottom of the probe (Figure 7.1C).  $C_{DGT-Sb(V)}$  showed virtually the same patterns to  $C_{DGT-Sb}$ because Sb(V) was the dominant inorganic Sb species.

After 5-week flooding (at the end of the plant bioassay), the profiles of  $C_{DGT-As}$ ,  $C_{DGT-Sb}$ , and their species in flooded soils varied greatly compared to the initial stage. In unplanted soils, interestingly,  $C_{DGT-As}$  increased considerably from 10 to 40-mm depth, followed by a gradual increase and was higher than that in 1-week flooded soil. In contrast,  $C_{DGT-Sb}$  was only higher in the OLW, but significantly lower in deep soil (below 10-mm depth) compared to that in 1-week flooded soil (Figure 7.1D).  $C_{DGT-Sb}$ profile also had a peak at 40-mm soil depth.  $C_{DGT-Sb}$  was higher than  $C_{DGT-As}$  in the OLW and 10-mm soil depth, but was much lower than  $C_{DGT-As}$  below 10-mm depth. For speciation, the profile of C<sub>DGT-As(III)</sub> exhibited the same trend as C<sub>DGT-As</sub>, with the average proportion of C<sub>DGT-As(III)</sub> in deep soil (below 10-mm depth) constituting 91% of C<sub>DGT-As</sub> (Figure 7.1E). This suggests that As(III) played an important role in the changes in As profile and mobilisation. C<sub>DGT-Sb(III)</sub> significantly increased from 10 to 30-mm depth, followed by fluctuation with sub-peaks at 40-mm and 60-mm depth, which is similar to the trend of C<sub>DGT-As(III)</sub>. The average proportion of C<sub>DGT-Sb(III)</sub> in deep soil (below 10-mm depth) constituted 43% of C<sub>DGT-Sb</sub> (Figure 7.1E). The increase in proportion of As(III) and Sb(III) indicates that the anaerobic condition increasingly developed with depth and over time in the soil system. In contrast to C<sub>DGT-As(III)</sub> and C<sub>DGT-Sb(III)</sub>, C<sub>DGT-As(V)</sub> and C<sub>DGT-Sb(V)</sub> were high in the OLW and to 10-mm soil depth and tended to be depleted in the deep layers with sub-peaks at 40-mm depth for both C<sub>DGT</sub>.  $A_{S(V)}$  and  $C_{DGT-Sb(V)}$  (Figure 7.1F). Overall, in deep soils (below 10-mm depth) flooded for 5 weeks, the average values of C<sub>DGT-As(III)</sub> and C<sub>DGT-As(V)</sub> significantly increased, whereas C<sub>DGT-Sb(III)</sub> and C<sub>DGT-Sb(V)</sub> considerably decreased. This is consistent with the substantial increase in C<sub>DGT-As</sub> and significant decrease in C<sub>DGT-Sb</sub> and also led to the much lower C<sub>DGT-Sb</sub> compared to C<sub>DGT-As</sub>.

The increase in As(III) with flooding time and soil depth is supported by numerous studies which showed that As is mobilised and released into the soil solution in response to flooding (Stroud et al. 2011; Takahashi et al. 2004; Xu et al. 2008; Yamaguchi et al. 2011). According to Stroud et al. (2011), As(III) was dominant in pore waters of soils contaminated with As to different degrees and from different sources under flooded conditions (66 - 100%, average 93%, on day 21). The flooded incubation of soils also led to a decrease in redox potential and a rise in As(III) fraction in the solid phase up to 80% of the total soil As (Yamaguchi et al. 2011). A study on As speciation in soil water collected under various soil depths showed that As was dissolved as As(V) (94%) at 10-mm depth, while As(III) was the primary species (99%) at 90-mm depth.

In contrast to As, previous studies on Sb speciation in soil solution under different redox conditions showed inconsistent results. The current study shows that the average Sb(III) proportion in anoxic zone (below 20 mm depth) was 43% for soils, while Mitsunobu et al. (2006) reported that Sb(V) was a stable form and Sb(III) accounted for only 6% of total Sb in soil solution collected at a depth of 90 mm. According to Okkenhaug et al. (2012), Sb(III) concentration in flooded soil pore water accounted for 5 – 19% of total Sb. The lower proportions of Sb(III) for Mitsunobu et al. (2006) and Okkenhaug et al.

(2012) may be due to oxidation of Sb(III) to Sb(V) during the sampling soil at different depths, transportation, and soil solution extraction, and storage or the less reducing conditions. Another study on the release of Sb from contaminated soil under various redox conditions by Hockmann et al. (2014a) mentioned that 85% of soluble Sb was present as Sb(III) on day 64 of incubation in lactate-amended microcosms.

The As concentration profiles closely match with its own Fe concentration profiles, illustrated by the fact that both As and Fe concentrations significantly increase from 10mm to 40-mm depth, but the contrasting trend was observed for the Sb and Fe concentration relationship (Figure E.2). This indicates that the increased mobilisation of As in flooded soils was strongly associated with the increased reductive dissolution of Fe oxyhydroxides, releasing Fe(II) and previously adsorbed or co-precipitated As into pore water. This mechanism of As mobilisation has been observed previously and known as the primary mechanism of flooded soil and sediment As mobilisation (Bennett et al. 2012b; Masscheleyn et al. 1991; Takahashi et al. 2004). Furthermore, the reduction of As(V) to As(III) led to the As mobilisation. It has been debated if As(V) was released into the soil solution along with the dissolution of Fe oxyhydroxides and subsequently reduced to As(III) in the soil solution or if adsorbed As(V) was reduced to As(III) on the soil solid phase and subsequently partitioned into the soil solution. Yamaguchi et al. (2011) and Takahashi et al. (2004) reported that the As desorption followed by As(V) reduction on the solid phase contributed half of the increase in As concentration in soil pore water under reducing conditions.

The current results are supported by previous studies on the mobilisation of As in sediment by DGT reporting that the mobility of As increased with sediment depth and closely related with the dissolution of Fe oxyhydroxides (Bennett et al. 2012a; Gao et al. 2017). Hossain et al. (2012) studied As dynamics in mangrove sediments and reported that As concentrations down depth profiles were more associated with elevated Fe and Mn concentrations than with organic matter. Conversely, the negative or no relationships for Sb indicates that the mobilisation of Sb was not directly associated with the chemistry of Fe in this study, which is supported by a recent study on the remobilisation of Sb in sediment cores collected in the field (Gao et al. 2016).



**Figure 7.1** Depth profiles of As and Sb species in soil Aged-L after 1-week flooding without plants (A, B, C) and 5-week flooding without plants (D, E, F) and with plants (G, H, I). The dotted line represents the soil water interface.

The concentrations of Sb, Sb(III), and Sb(V) in deep soil after 5 weeks of flooding was lower than that after 1 week of flooding (reflected by lower C<sub>DGT-Sb</sub>, C<sub>DGT-Sb</sub>(III), C<sub>DGT-</sub>  $_{Sb(V)}$ ), which is in contrast with a study by Okkenhaug et al. (2012) who observed the increased dissolved Sb concentrations in flooded soil solution over time. However, the current result is in agreement with Wan et al. (2013b) who studied on the solubility and speciation of Sb in waterlogged soil in a tank experiment. The decrease in Sb lability after 5-weeks of flooding in the current study may be due to (1) Sb(V) being adsorbed to re-precipitated less crystalline Fe oxyhydroxides to a larger extent, a process suggested by Mitsunobu et al. (2006), (2) Sb(V) being reduced to Sb(III) which was then immobilised by adsorption to Fe oxyhydroxides due to its higher affinity to Fe oxyhydroxides (Hockmann et al. 2014a; Leuz et al. 2006; Mitsunobu et al. 2010; Wan et al. 2013a), (3) Sb(III) precipitated with sulphide resulted from the microbial reduction of sulphate (Hockmann et al. 2014b), and (4) more Sb was in inert forms or in complexes having large molecules, which are non-DGT labile. The third and fourth explanations seem to be more reasonable since (1) the water table was maintained during the flooding period and the re-precipitation of Fe oxyhydroxides was not observed and (2) the reduction of Fe oxyhydroxides occurred, which would release readsorbed Sb into soil pore water together with As, however this was not the driving mechanism as a decrease in labile Sb was observed. Thus, the coupled Sb release and Fe mineral reduction was not the case for our results. The DGT probe for sulphide measurement and Sb K-edge XANES analysis of solid phase speciation in flooded soil should be applied for further explanation of the Sb mobilisation mechanism.

In comparison of  $C_{DGT-As}$  and  $C_{DGT-Sb}$  in the flooded soil system, As and Sb show contrasting trends with concentrations increasing for As and decreasing for Sb with depth.  $C_{DGT-Sb}$  was much lower than  $C_{DGT-As}$  below 10-mm depth, even though total soil Sb was much higher than total soil As. These results might be due to (1) more Sb being in particulate colloids, (2) Sb being released and resupplied to soil solution very slowly in response to DGT depletion over the 24-h deployment, leading to less Sb measured, and (3) more Sb being present in inert forms or in large organic complexes which were not measured by DGT. However, the first possibility can be ruled out since  $C_{DGT-Sb}$  was much higher than  $C_{DGT-As}$  in the OLW and surface soil.

# 7.3.1.2. Effect of plants on the speciation and solubility of As and Sb in flooded historically contaminated soils

For planted soil, the C<sub>DGT</sub> profiles of As and its species generally displayed the same patterns as those in unplanted soil. C<sub>DGT-Sb</sub> and C<sub>DGT-Sb(V)</sub> shared the same pattern which was a gradually decrease from the OLW to the bottom of probe, but was less pronounced than their decreasing trend in unplanted soil. Both As and Sb were present at lower concentrations of trivalent species and higher concentrations of pentavalent species. Compared to unplanted soils, the proportion of C<sub>DGT-As(III)</sub> and C<sub>DGT-Sb(III)</sub> decreased to ~59 and ~12% of  $C_{DGT-As}$  and  $C_{DGT-Sb}$ , respectively. The presence of these lower trivalent species is likely due to diffusion of oxygen from plant root aerenchyma into the rhizosphere creating a narrow zone of oxidising environments around plant roots, enhancing the oxidation of trivalent species to pentavalent species (María-Cervantes et al. 2010; Nishiuchi et al. 2012). The average  $C_{DGT-As(V)}$  and  $C_{DGT-Sb(V)}$ (below 10-mm depth) in the planted soil was 9.8 and 6.7 times higher than that in the unplanted soil, respectively. In addition, due to the likely higher oxidising environments around the root zones, the reduction of Fe oxyhydroxides releasing adsorbed As and Fe(II) and the reduction of As(V) to As(III) occurred to a lesser extent in soils with plants, leading to the decrease in trivalent species and increase in pentavalent species.

Overall,  $C_{DGT-As}$  decreased, but  $C_{DGT-Sb}$  increased in the planted soil system (below 10mm depth). The decrease in  $C_{DGT-As}$  is most likely due to As accumulated by plants, and Fe plaque coating plant roots adsorbing and co-precipitating As, counteracting the increase of As mobility. The overall increase in  $C_{DGT-Sb}$  in the flooded soil system might be because Sb(V), the dominant Sb species, was adsorbed less effectively than As(V) on the Fe plaque (Qi & Pichler 2017), enhancing more soluble Sb in the solution phase. Another possibility is that Sb was accumulated less by plants than As, leading to an increase in Sb in the soil solution. The current results reconfirm the effect of plant roots on the mobility of surrounding contaminants reported by previous studies (María-Cervantes et al. 2010; Wan et al. 2013b). It is noticeable that  $C_{DGT}$  of As, Sb, and their species in the OLW in planted soil system were lower than those in unplanted soil system. As observed, plant roots developed from the plant node below the water table. This suggests that the absorption of As and Sb in the OLW in planted soil system by newly developed roots possibly occur, which may contribute to their overall uptake by plants. The  $C_{DGT}$  profiles of As, As(III), As(V), Sb, Sb(III), and Sb(V) in soil Aged-H after one-week flooding and five-week flooding with and without plants are presented in Figure E.3. In general, all profile shapes closely resemble those in soil Aged-L, illustrated by the increasing trends, decreasing trends, minima, and maxima occurring at the same locations, but to a greater extent due to soils contaminated with higher concentrations of As and Sb. This indicates that the performance of DGT in assessing labile As and Sb in flooded soils is reproducible and these profiles accurately reflect the dynamic of labile As and Sb species. This is an important observation as no replicate measurements were performed due to the volume of data collected in this study. Thus, the interpretation of As and Sb behaviour in soil Aged-L can be applied for soil Aged-H.

In summary, historically contaminated soils subjected to waterlogging resulted in the significant increase of total labile As in soils with As(III) being the dominant species. Despite labile Sb decreased after 5 weeks of flooding, Sb(III) was present >10% of total labile Sb in the flooded soil system. As(III) and Sb(III) are more toxic than As(V) and Sb(V); hence, waterlogging contaminated soils poses higher risks of As and Sb toxicity to the environment and human health.

### 7.3.2. DGT measurements in flooded recently contaminated soils

# **7.3.2.1.** Dynamics of lability and speciation of As in flooded recently contaminated soils and the effect of Sb on As lability

The C<sub>DGT</sub> profiles of As, As(III), and As(V) in As-only and As+Sb soils flooded for 1 and 5 weeks, with and without plants, are presented in Figure 7.2. After 1 week of flooding, C<sub>DGT</sub> in As-only soil generally fluctuated with small peaks along the depth profile, whereas C<sub>DGT</sub> in As+Sb soil gradually decreased with depth. For speciation, C<sub>DGT-As(III)</sub> in As-only significantly increased in deep soil with a maxima from 40 to 70mm depth, while C<sub>DGT-As(III)</sub> in As+Sb soil was relatively unchanged with depth. The average proportion of C<sub>DGT-As(III)</sub> in deep soil (below 10-mm depth) was <20%. As(V) was a dominant species, thus the profiles of C<sub>DGT-As(V)</sub> displayed the same patterns as C<sub>DGT-As</sub> in both soils. Overall, after 1 week of flooding C<sub>DGT</sub> of As and its species were higher in deep As-only soil than As+Sb soil.



**Figure 7.2** Depth profiles of As species concentrations in soil recently contaminated with As-only and with As+Sb soils after 1-week flooding without plants (A, B, C), 5-week flooding without plants (D, E, F) and with plants (G, H, I), measured by DGT. The dotted line represents the soil water interface.

After 5 weeks of flooding, C<sub>DGT-As</sub> in As-only soil exhibited the same pattern in 1-week flooded soil, with a peak at mid-profile (30-mm depth). In As+Sb soil, C<sub>DGT-As</sub> followed the same pattern as in As-only soil, but the visible peak at mid-profile and sub-peak at the bottom were larger (Figure 7.2D). For speciation, C<sub>DGT-As(III)</sub> in both soils had an excellent comparability, with maxima and constantly unchanged values occurring at the same positions. The noticeable result is that the peak of C<sub>DGT-As(III)</sub> at mid-profile was larger in As+Sb soil than in As-only soil (Figure 7.2E). As expected, C<sub>DGT-As(III)</sub> in Asonly and As+Sb soils increased (a factor of 3.3 and 6.6, respectively) compared to that in 1-week flooded soils, with the average proportion of C<sub>DGT-As(III)</sub> in deep soil (below – 10 mm) being ~75% of C<sub>DGT-As</sub>. In contrast, C<sub>DGT-As(V)</sub> in both soils decreased in a fluctuating manner with depth, with minima and maxima at mid-profile corresponding with maxima and minima of As(III), Fe(II), and total labile As. Thus, it can be generalized that the zone around 20 to 30-mm depth (mid-profile) was the active zone for As mobilisation and Fe reduction. The relatively unchanged trends from -40 mm downwards indicate the possibly completely anoxic conditions in very deep soils and most of the newly formed trivalent species were immobilised by the precipitation or readsorption on the secondary Fe minerals. It seems that the presence of Sb led to the increase in As mobilisation in the flooded soil.

Overall, compared to  $C_{DGT-As}$  in 1-week flooded soils,  $C_{DGT-As}$  in As-only soil decreased by about half, while that in As+Sb soil increased. Thus,  $C_{DGT-As}$  in As+Sb soil was higher than that in As-only soil, meaning that the presence of co-contaminant (Sb) enhanced the As lability in As+Sb soil. This can be explained by the adsorption strength of Sb(III), Sb(V), As(III), and As(V) on Fe minerals and organic matter and the precipitation capacity with sulphide resulting from the reduction of sulphate. Sb(III) binds to the soil much more strongly than As(III) because Sb(III) binds to Fe (hydr)oxides through stable bidentate inner-sphere complexes (Scheinost et al. 2006), whereas As(III) sorbs to Fe (hydr)oxides via multiple inner-sphere and outer-sphere complexes (Fendorf & Kocar 2009). In addition, Sb(V) has higher affinity to organic matter (OM) than As(V) (Dousova et al. 2015). Therefore, Sb could outcompete As in the adsorption on Fe (hydr)oxides and OM, resulting in lesser binding sites for As. Furthermore, given that sulphide was produced, Sb(III) would preferentially precipitate with sulphide before iron and arsenic due to the increasing solubility constant of Sb<sub>2</sub>S<sub>3</sub> (K<sub>sp</sub> =  $1.6 \times 10^{-93}$ ), FeS (K<sub>sp</sub> =  $6.3 \times 10^{-18}$ ), and As<sub>2</sub>S<sub>3</sub> (K<sub>sp</sub> =  $8.4 \times 10^{-16}$ ). These suggest that Sb should be involved in the adsorption and precipitation processes in priority in As+Sb soil, reserving As species in the solution phase. As a result, with the same given binding sites in two soils, As was more labile in As+Sb soil than in As-only soil. In As-only soil, due to Sb absence a proportion of As(III) and Fe(II) may have precipitated with sulphides, leading to the decrease in  $C_{DGT-As}$  (Figure 7.2D) and  $C_{DET-Fe}$  (Figure E.1) after 5-week flooding. This is supported by a previous study on the effect of sulphate reduction on the As mobility during waterlogging of As-contaminated soil by Burton et al. (2014). The authors stated that the microbial sulphate reduction lowered concentrations of pore water Fe(II) resulted from Fe(III) reduction as a result of the formation of FeS. The newly formed FeS subsequently sequestrated considerable amounts of As. The high-resolution transmission electron microscopy showed that the solid phase As was retained as an As<sub>2</sub>S<sub>3</sub>-like complex associated with FeS (mackinawite).

The dynamics of As and Sb in soils under flooding may be influenced by processes such as dissolution of Fe (hydr)oxides, precipitation of sulfide or by precipitation of authigenic minerals (María-Cervantes et al. 2010). The released As during the dissolution of Fe (hydr)oxides may be subsequently re-adsorbed to the secondary Fe minerals formed during the development of anoxic conditions (e.g. sederite, magnetite), which was observed by previous studies (Tufano et al. 2008; Yamaguchi et al. 2011). The formation of meta-stable metal sulfides is believed to be responsible for the sequestration of As, Sb, and metals (Du Laing et al. 2009; Otero et al. 2009). Moreover, green rust minerals with a structure of brucite-type layers of Fe<sup>3+</sup> and Fe<sup>2+</sup> sandwiched by anion layers and water molecules have been indicated for reducing contaminants in soil solutions (Génin et al. 2001; María-Cervantes et al. 2010). These processes seem to be reasonable in our study given that Fe(II), As, and Sb concentrations remained relatively unchanged or lower in deep soil (below 40-mm depth). It is possible that the formation of secondary Fe minerals or insoluble substance and complexes prevented the dissolution of Fe, As, and Sb, thereby resulting in the lower partitioning of As and Sb into soil solution phase.

With respect to mechanism of As mobilisation, it seems that the mobilisation of As in recently contaminated soils is different from that in historically contaminated soils. The results show that As(III) increased as anoxic environments increased over time (Figure 7.2); however, As concentrations displayed a relatively constant trend in deep soil (from

-40 mm downwards), while Fe concentrations exhibited fluctuating decreasing trends (Figure E.1), meaning that the reduction of Fe (hydr)oxides was unlikely to occur in this zone. The further relationship analysis on As and Fe concentration data shows that there was no or minor negative relationships between soluble As and Fe concentrations, indicating that the mobilisation of As in the recently contaminated soils was not associated with the reduction of Fe (hydr)oxides in deep soils. In addition,  $C_{DGT-As(V)}$  decreased and  $C_{DGT-As(III)}$  increased during flooding time, indicating that conversion of As(V) to As(III) did occur. Based on the As fractionation in As-amended soils (section 6.3.2, chapter 6), As in these soils was highly labile and present in the non-specifically sorbed and specifically sorbed (SO<sub>4</sub><sup>2-</sup> and PO<sub>4</sub><sup>3-</sup> extractable) fractions up to 35%. Thus, it can be speculated that the desorption of As(V) from the soil solid phase via the exchangeable process due to the presence of phosphate and sulphate as nutrients and the subsequent reduction of As(V) to As(III) into pore water were strongly involved in the mobilisation of As in recently contaminated soils.

# 7.3.2.2. Effect of plants on the speciation and solubility of As in flooded recently contaminated soils

In planted soils after 5-weeks of flooding, all  $C_{DGT}$  of As and its species in As-only soil exhibited an increasing trend with depth, which is different from those in unplanted soils (Figure 7.2.G). The  $C_{DGT}$  profiles of As and its species in As+Sb soil also displayed the increasing trend and shared the same feature which was a broad maxima at -20 to -40 mm. This can be considered as the active zone of chemistry of As and Fe as well as the plant root activity. Interestingly, these peaks increased with Sb added and the plants extended this process over a greater depth and caused As(V) mobilisation. Similar to unplanted soils after 5-week of flooding, the values of  $C_{DGT-As}$ ,  $C_{DGT-As(III)}$ , and  $C_{DGT-As(V)}$  in As+Sb soil were higher than those in As-only soil.

 $C_{DGT-As(III)}$  and  $C_{DGT-As(V)}$  in two soils generally increased with depth (Figures 7.2H, 7.2I); however, the proportions and DGT concentrations of As(III) generally decreased especially from the surface soil to 60-mm depth while As(V) increased compared to those in no plant treatments. The average  $C_{DGT-As(V)}$  (from 20-mm depth downwards) in As-only and As+Sb planted soils was over 6 times higher than that in the unplanted soils, respectively. This may be due to possible narrow oxidative zones created around the roots resulting in the oxidation of As(III) to As(V) as discussed earlier in section 7.3.1.2, leading to the decrease in As(III) and increase in As(V). In addition, the oxygen

released from plant roots might have led to the oxidation of precipitates of sulphide (e.g. FeS, FeS<sub>2</sub>, As<sub>2</sub>S<sub>3</sub>, Sb<sub>2</sub>S<sub>3</sub>), releasing As, Fe, Sb into the solution phase. Additionally, the root exudates (e.g. amino acids, organic acids) and protons might have contributed to the release of As and Sb bound to organic Fe complexes. These can be illustrated by the increase in Fe (Figure E.1) and As over depth (Figures 7.2G, H, I) in planted soils. The variation in As speciation in the planted recently contaminated soils were similar to that in planted historic soils. However, the net of these processes led to the higher  $C_{DGT-As}$  in planted recently contaminated soils than in unplanted recently contaminated soils. The likely reason for this is unclear. This may be because plants were exposed to much greater labile As in recently contaminated soils, thus, the plants may use internal modifications to limit the As transport through root cell or there was more Fe plaque formed. This resulted in the restriction of As uptake, contributing to the increase in As concentrations in planted recently contaminated soils.

# **7.3.2.3.** Dynamics of speciation and lability of Sb in flooded recently contaminated soils and the effect of As on Sb lability

The C<sub>DGT</sub> profiles of Sb, Sb(III), and Sb(V) in Sb-only and As+Sb soils flooded for 1 and 5 weeks with and without plants are presented in Figure 7.3. In 1-week flooded recently contaminated soils, the profiles of C<sub>DGT</sub> of Sb and its species displayed the same trend as those in flooded historic soils. C<sub>DGT-Sb</sub> in two soils gradually decreased with depth and had relatively identical profiles. For speciation, C<sub>DGT-Sb(III)</sub> in the two soils gradually increased with sub-peaks till the bottom of the probe, being <20% of C<sub>DGT-Sb</sub> in deep soils (below 10-mm depth) (Figure 7.3B), while C<sub>DGT-Sb(V)</sub> in two soils gradually decreased with depth, being the dominant species in soils (the average of Sb(V) below 20 mm >80%). The profiles of C<sub>DGT-Sb(V)</sub> were consistent with the profiles of C<sub>DGT-Sb</sub> (Figure 7.3A, 7.3C).

After 5-week flooding, the profiles of  $C_{DGT-Sb}$  and  $C_{DGT-Sb(V)}$  consistently displayed the same patterns, which sharply decreased from the OLW to 20-mm depth, after which they generally remained stable.  $C_{DGT-Sb(III)}$  in two soils showed completely different trends with those in 1-week flooded soils.  $C_{DGT-Sb(III)}$  in the two soils exhibited relatively unchanged trends, except a peak at 20-mm depth in As+Sb soil, corresponding with the maxima of As and Fe (Figures 7.2E and E.1).



**Figure 7.3** Depth profiles of Sb species concentrations in soil recently contaminated with Sb-only and with As+Sb soils after 1-week flooding without plants (A, B, C), 5-week flooding without plants (D, E, F) and with plants (G, H, I), measured by DGT. The dotted line represents the soil water interface.

Overall,  $C_{DGT-Sb}$  in Sb-only soil was slightly higher than that in As+Sb soil, especially in OLW and 10-mm depth soil. It seems that there was little effect of As on the lability and solubility of Sb in the flooded soils studied. The values of  $C_{DGT-Sb(III)}$ ,  $C_{DGT-Sb(V)}$ , and  $C_{DGT-Sb}$  (below 20-mm depth) in 5-week flooded soils were much lower than those in soils flooded for 1 week. The lower  $C_{DGT-Sb}$  in 5-week flooded soils was possibly due to a combination of the increased adsorption of Sb(V) on dissolved organic matter (DOM) via metal (e.g. Fe) bridges, the immobilisation of the newly reduced Sb(III) on Fe hydr(oxides) via adsorption, and the precipitation of the newly reduced Sb(III) with sulphide (as discussed in section 7.3.1.1). The possible large complexes and inner precipitates formed may also contribute to the decrease in  $C_{DGT-Sb}$  in soil pore water because they were not measured by DGT.

The current results show that Sb and its species did not vary with soil depth (from 40mm downwards), possibly indicating the completely anoxic conditions in deep soils. It seems that everything in the deep soil system was at equilibrium. Alternatively, the relatively unchanged trends of C<sub>DGT</sub> of Sb and Sb species with depth and the decrease in C<sub>DGT-Sb</sub> under more reducing conditions may have resulted from the variation of Fe hydroxide properties known as the host phase of As and Sb. Under anaerobic conditions, the reductive dissolution of Fe(III) to Fe(II) and the reprecipitation due to the oxidation to Fe(III) hydroxide might be occurring in the soil system. Furthermore, the structural order of the host phase might decrease. According to Richmond et al. (2004), ferrihydrite with a decreased structural order had greater surface areas and stronger adsorptive capacity. It can be speculated that the greater surface areas of disordered host phase, Fe hydroxides, resulted in the higher partitioning of Sb to soil under anoxic conditions (Mitsunobu et al. 2006). Moreover, desorbed Sb at the initial stage may readsorb to the secondary Fe mineral (e.g. siderite, magnetite) and precipitate with sulphide resulted from sulphate reduction during the development of anaerobic conditions. In order to further understand the immobilisation of Sb during the anoxic environment, the analysis of solid phase speciation by XAFS is needed.

Regarding the mechanism of Sb mobilisation in flooded recently contaminated soils, the results show contrasting trends between Sb(III) and Fe(II), which is similar to that in historic soils, indicating that the mobilisation of Sb was not directly associated with the reduction of Fe (hydr)oxides. This is in agreement with the remobilisation of Sb in sediments collected by a core sampler and assessed by DGT (Gao et al. 2016). In Gao's

study the release of Sb was not associated with Fe and Mn in sediments. Like As mobilisation in recently contaminated soils, the mechanism of Sb mobilisation may include the desorption of Sb(V) from soil, the reduction of Sb(V) to Sb(III), and the subsequent immobilisation of the newly formed Sb(III).

# 7.3.2.4. Effect of plants on the speciation and solubility of Sb in flooded recently contaminated soils

In flooded planted soils, the profiles of  $C_{DGT-Sb}$  in two soils consistently displayed the same patterns, with little variation in deep soils (Figure 7.3G). In contrast,  $C_{DGT-Sb(III)}$  exhibited different manner with the presence of plants in soils.  $C_{DGT-Sb(III)}$  in Sb-only and As+Sb soils displayed the same pattern which was a considerable increase reaching peaks at 40-mm depth, followed by a gradual decrease down the soil depth, but the profile of  $C_{DGT-Sb(III)}$  in Sb-only soil had a much broader maxima than that in As+Sb soil (Figure 7.3H). Interestingly, the plants facilitated the release of Sb(III), especially in Sb-only soils. As a result,  $C_{DGT-Sb(III)}$  in deep soils (below – 10 mm) increased by a factor of ~1.8 compared to unplanted soils, while  $C_{DGT-Sb(V)}$  slightly decreased. The patterns of  $C_{DGT-Sb(V)}$  in two soils were relatively identical and similar to that of  $C_{DGT-Sb}$ .

The increase in C<sub>DGT-Sb(III)</sub> in planted soils may be due to the oxygen supplied from the plant roots may have caused the sulphide oxidation, mobilising Sb, Fe, and As precipitated with sulphides, a process reported by Otero et al. (2009), María-Cervantes et al. (2010), and Williams et al. (2014). This is demonstrated by the increase in Sb(III), As(III), and Fe(II) with soil depth, especially large maxima at the zone -20 to -40 mm depth which may be the active zone of plant roots. In addition, organic exudates exerted in the rhizosphere may have facilitated the mobilisation of bound-Sb on the host phase. According to Ptak and McBride (2015), organic acid exudation can mobilise Sb bound by organic Fe complexes and the increase in maize rhizosphere Sb bioavailability and in plant tissues were observed. The decrease in C<sub>DGT-Sb(V)</sub> may be due to the uptake of Sb(V) by plant roots. Overall, the net of these processes led to the lower  $C_{DGT-Sb}$  in planted soils than in unplanted soils. This may be because of plant uptake of Sb, especially Sb(V) and adsorption of Sb on the Fe plaque coated on plant roots, lowering the entire labile Sb. Fe plaque around plant roots has been recognised to be a scavenger of Sb and As by previous studies (Cui et al. 2015; Okkenhaug et al. 2012; Tripathi et al. 2014).

In summary, the presence of plants promoted the release of trivalent species from the solid phase and the oxidation of trivalent species to pentavalent species in flooded soils. The overall  $C_{DGT-As}$  decreased, while  $C_{DGT-Sb}$  increased in flooded historically contaminated soils with plants compared to unplanted soils. The opposite was observed for flooded recently contaminated soils with plants. It seems that the labile As in flooded recently contaminated soils was higher than the element acquisition of plants. As discussed in section 7.3.2.2, the plants were exposed to much greater labile As in recently contaminated soils, thus, the plants may use internal modifications to restrict the As transport through root cell. This led to the restriction of As uptake, contributing to the higher  $C_{DGT-As}$  in planted recently contaminated soils. In contrast, Sb uptake by plants in recently contaminated soils may be determined by the mass flow due to increased plant biomass and plant stress, facilitating Sb transport to roots. This led to the decrease in  $C_{DGT-As}$  in planted recently contaminated soils.

# 7.3.3. Accumulation of As and Sb in water spinach (*Ipomoea aquatica*)

*I. aquatica* is a very popular edible vegetable in Asia and can be grown in land and semiaquatic systems. The biomass of *I. aquatica* cultivated in flooded soils was higher than that in non-flooded soils. Of flooded plants, the biomass of *I. aquatica* exposed to flooded soil Aged-H was lowest. The plants looked healthy and exhibited no indicators of phytotoxicity (e.g. discoloration, stunting, wilting).

Interestingly, both roots and shoots of *I. aquatica* accumulated more As and Sb under flooded conditions than non-flooding (Figure 7.4). Arsenic concentrations in roots and shoots of *I. aquatica* were 2.7 - 3.1 and 2.2 - 3 times higher under flooded conditions than under non-flooded conditions, respectively, which is supported by the higher soluble As in flooded soils as discussed in sections 7.3.1 and 7.3.2. The translocation factors (TF) for As in flooded treatments ranged from 0.25 - 0.5 being similar to TF for As (0.24 - 0.53) in non-flooded soils (Table 7.1). It is clear that an increase in As bioavailability under flooded conditions resulted in the enhanced As accumulation. Xu et al. (2008) reported that growing rice in flooded paddy soils for 117 days markedly increased As in rice shoots and grains, which was >10 times higher than As in rice grown in aerobic soils.



**Figure 7.4** Concentrations of As in roots (A) and shoots (B), Sb in roots (C) and shoots (D) of *I. aquatica* cultivated in non-flooded and flooded soils. Data are presented as mean  $\pm$  SE (n = 3).

Soil	As								
		Non-Flooded				Flooded			
	Total soil As	BAF	TF		Total soil As	BAF	TF		
	(mg/kg)				(mg/kg)				
Aged-L	$105 \pm 10$	$0.44 \pm 0.08$	$0.36\pm0.1$		$105 \pm 10$	$1.38\pm0.2$	$0.30\pm0.04$		
Aged-H	$420 \pm 10$	$0.82\pm0.09$	$0.24\pm0.03$		$420 \pm 10$	$2.3 \pm 0.2$	$0.25\pm0.01$		
As50	$52 \pm 1$	$1.3 \pm 0.2$	$0.5\pm0.1$		$52 \pm 1$	$3.5 \pm 0.1$	$0.50\pm0.04$		
As50Sb300	$57 \pm 1$	$1.7 \pm 0.1$	$0.528 \pm 0.004$		$57 \pm 1$	$4.6\pm0.1$	$0.44\pm0.02$		
Soil	Sb								
		Non-Flooded		_	Flooded				
	Total soil Sb	BAF	TF	_	Total soil Sb	BAF	TF		
	(mg/kg)				(mg/kg)				
Aged-L	$120 \pm 10$	$0.004\pm0.001$	$0.29\pm0.05$		$120 \pm 10$	$0.029 \pm 0.001$	$0.13\pm0.02$		
Aged-H	$620 \pm 4$	$0.0045 \pm 0.0005$	$0.16\pm0.03$		$620 \pm 4$	$0.017\pm0.002$	$0.13\pm0.02$		
Sb300	$280 \pm 10$	$0.0522 \pm 0.0007$	$0.19\pm0.04$		$280 \pm 10$	$0.119 \pm 0.007$	$0.18\pm0.01$		
As50Sb300	$280 \pm 10$	$0.057\pm0.006$	$0.21\pm0.03$		$280 \pm 10$	$0.12\pm0.01$	$0.23\pm0.02$		

**Table 7.1** Total soil concentrations, bioaccumulation factor (BAF), and translocation factor (TF) of As and Sb in *I. aquatica* cultivated in soils under non-flooding and flooding. Data are presented as mean  $\pm$  SE (n = 3).

Similar to As concentrations in shoots, *I. aquatica* shoots took up 2.3 - 3.3 times more Sb under flooded conditions than under non-flooded conditions. Sb accumulation in anaerobically grown roots was also enhanced compared to aerobic treatments, but to a greater extent (2.0 - 5.7 times). However, a close inspection on increase differences among soil treatments between two water regimes shows that *I. aquatica* cultivated in historically contaminated soils exhibited a higher increase of As and Sb concentrations in plant tissues compared to the plants exposed to recently contaminated soils. For example, As and Sb concentrations in flooded roots grown in soil Aged-L was 3.1 and 5.7 times higher than that in non-flooded roots, respectively, while the factor of 2.6 for As and 2.0 for Sb in roots grown in soil As50Sb300. This suggests that the release and resupply of As and Sb from dissolved complexes in soil solution and from the solid phase of historically contaminated soils contributed to the uptake by plants at a greater extent.

Sb bioaccumulation was higher under flooded conditions than non-flooded conditions, in spite of total DGT-labile Sb decreased at the fifth week of flooding. This may be because plants subjected to anoxic stress increased the apoplastic bypass (Wan et al. 2013b), which enabled more Sb to be accumulated and transported from roots to xylem. The mass flow was likely to be important for Sb accumulation by *I. aquatica*. In addition, the presence of plants enhanced the lability of Sb, which may contribute to the increase in Sb uptake by *I. aquatica*. Another possibility is that Sb(V), the dominant species in soil systems was preferential for Sb uptake by *I. aquatica*. Wan et al. (2013a, 2013b) reported that waterlogging enhanced Sb concentrations in shoots of *L. perenne* by about 10 times. In the current study, the translocation factors for Sb in the flooded and non-flooded soils were in the range of 0.13 - 0.23 and 0.16 - 0.29, respectively (Table 7.1).

*I. aquatica* cultivated in As+Sb soils (As50Sb300, 57 mg As/kg) accumulated more As than in historically contaminated soils (Aged-L, 105 mg As/kg), by a factor of 1.8 - 2.1 for roots and 2.7 - 3.5 for shoots. Similarly, although total Sb concentration in As+Sb soils (As50Sb300, 280 mg Sb/kg) was approximately two times lower than that in the historically contaminated soil (Aged-H, 620 mg Sb/kg), Sb uptake by roots and shoots of *I. aquatica* cultivated in the As50Sb300 soil was 3.1 - 5.7 and 5.1 - 7.5 times higher than in the Aged-H soil. This indicates that total soil concentrations are not a good indicator for the prediction of plant uptake of metalloids and soil risk assessment. It also confirms that biogeochemical processes and bioavailability of metalloids plays an important role in predicting their accumulation by plants and accurate assessment of contaminated soils.

Arsenic concentrations in *I. aquatica* grown in As+Sb soil (As50Sb300) were approximately 1.5 times greater than that in As-only soil (As50), although total As concentrations in two soils were relatively similar. This indicates that the greater As bioaccumulation is likely due to the presence of Sb in the former soil, which is consistent with the higher DGT-labile As in soil pore water. In contrast, Sb concentrations in tissues of *I. aquatica* exposed to As+Sb soil (As50Sb300) and Sb-only soil (Sb300) with similar total soil Sb of 280 mg/kg were not significantly different, indicating that the presence of As was unlikely to affect Sb solubility and uptake by *I. aquatica* in this study. This is supported by the DGT-labile Sb being relatively similar in both soils.

In regards to the trend of As and Sb uptake by *I. aquatica*, As and Sb concentrations were significantly higher in roots than in shoots (Figure 7.4), which is in agreement with previous studies (chapters 5 and 6) on *I. aquatica* bioaccumulation in As and Sb contaminated soils under aerobic conditions, where As and Sb concentrations in roots were about 14 and 10 times higher than shoots, respectively. Since the Fe plaque coating on roots in this study was removed, the root As and Sb concentrations reflect the true accumulation by roots, suggesting the firm conclusion that As and Sb were

sequestrated in *I. aquatica* roots. This uptake tendency was also found for other plants (e.g. rice). Higher As and Sb concentrations in rice roots than shoots were reported by Okkenhaug et al. (2012) and Smith et al. (2008a). The average bioaccumulation factors (BAF) of As in non-flooded (0.63) and flooded historically contaminated soils (1.84) were 147 and 90 times higher than those of Sb (0.004 under non-flooding and 0.02 under flooding), respectively. This is consistent with the trend in chapter 5.

## 7.3.4. Arsenic and antimony speciation in water spinach (*Ipomoea aquatica*)

# 7.3.4.1. As speciation in non-flooded I. aquatica

The extraction efficiencies of As species in roots and shoots of *I. aquatica* grown in non-flooded soil were in the range of 86 - 102%, with higher proportions of As extracted from roots (Table 7.2). The high extraction efficiency is in agreement with published studies using the same extractant (dilute HNO<sub>3</sub>) to extract As species from edible vegetables (Sadee et al. 2016), rice grains (Maher et al. 2013), and marine plants (Foster et al. 2007). These studies demonstrated that dilute HNO<sub>3</sub> returned quantitative extraction of As species in samples and no appreciable oxidation of As(III) and As(V) was observed.

Speciation analysis revealed that only As(III) and As(V) were identified in roots and shoots of non-flooded *I. aquatica*, with variation of As species between plant tissues. As(III) was the dominant species in roots (62 - 84%), followed by As(V) (14 - 31%). In contrast to As speciation in roots of *I. aquatica*, the proportion of As(III) and As(V) in shoots was not significantly different, ranging from 39 to 47% of total As in shoots, with the exception for Aged-L soil. It seems that there was an equilibrium between As(III) and As(V) in shoots.

The speciation in non-flooded *I. aquatica* roots is supported by (Yao et al. 2009) who reported that As(III) accounted for 71.3 - 81.1% of total As in roots of water spinach grown in soil amended with As-contaminated chicken manure, followed by As(V) (15.3 – 22%), and DMA (3.7 - 6.8%) which is derived from chicken manure. In the current study, MMA and DMA were not identified in *I. aquatica* tissues, suggesting that the methylation process was unlikely to occur in the studied soil and plant.

A study on As speciation in rice by Smith et al. (2008a) also showed the predominance of inorganic As species in rice roots, with As(III) comprising 57 - 78% of total As, whereas As(V) constituting 16 - 27%. Numerous studies on As speciation have reported

that inorganic As species are dominant species in terrestrial plants including edible vegetables (e.g. lettuce, chard, radish) (Muñoz et al. 2002; Smith et al. 2009; Tlustoš et al. 2002). Some As speciation studies identified the predominance of As(III) in tissues of *Pteris vittata* (Lombi et al. 2002; Webb et al. 2003), *Pityro-gramma calomelanos* (Francesconi et al. 2002), and mung bean roots accumulating up to 90% of As(III) (Smith et al. 2009).

High proportions of As(III) in roots may be due to the direct uptake of As(III) from the soil solution or the uptake of As(V) from soil solution, followed by the reduction of As(V) to As(III) in roots as a tolerant mechanism of plant to avoid As stress (Meharg 1994). In the studied aerobic soils, *I. aquatica* was likely to be exposed to As(V), the dominant species in oxic conditions (Sadiq 1997; Smith 1998; Wilson et al. 2010), thus the second pathway seems to be more reasonable for the non-flooded *I. aquatica*. The contribution of As(V) from soils to root uptake may be significant compared to As(III).

Several studies reported that once As(V) accumulated in roots, As(V) is quickly reduced to As(III) in root cells (Lombi et al. 2002; Pickering et al. 2000; Webb et al. 2003). Pickering et al. (2000) reported the main metabolic pathway of As in Indian mustard (*Brassica juncea*) involving processes of roots significantly accumulating As(V) which was rapidly reduced to As(III), and that the majority of As was restricted in roots, only a small amount of As(III) and As(V) was transferred to shoots of Indian mustard. This metabolic pathway of As seems to also occurred in *I. aquatica*, demonstrated by the much higher As concentrations in roots than shoots.

The reduction of As(V) to As(III) has been found to be a strategy for the detoxification of As in plants, which involves As(V) reduced to As(III) via enzyme arsenate reductase, then As(III) complexed with thiols and sequestrated in root vacuoles (Álvarez-Ayuso et al. 2016; Tripathi et al. 2007; Zhao et al. 2009). Álvarez-Ayuso et al. (2016) found a negative correlation between non-protein acid soluble thiol concentrations in rye roots and As concentrations in rye shoots, indicating that As complexation with thiols plays an important role to restrict As transport from rye roots to shoots.

		Non-flooded										
Treatment	Root						Shoot					
	Total As	As <sup>III</sup>	As <sup>v</sup>	Total As in	Extracted		Total As	As <sup>III</sup>	$As^{V}$	Total As in	Extracted	
	(mg/kg)	(mg/kg)	(mg/kg)	extracts (mg/kg)	(%)		(mg/kg)	(mg/kg)	(mg/kg)	extracts (mg/kg)	(%)	
Aged-L	$47 \pm 8$	33 ± 2 (74)	$14 \pm 7 (27)$	$48 \pm 10$	$102 \pm 2$		$15 \pm 3$	$10 \pm 4$ (60)	$5.2 \pm 0.4$ (38)	15 ±5	$99\pm9$	
Aged-H	$348 \pm 40$	$216 \pm 30$ (62)	$108 \pm 30 (31)$	$324 \pm 40$	$93 \pm 5$		$81 \pm 6$	33 ± 5 (41)	$40 \pm 3$ (40)	$73 \pm 4$	$91\pm2$	
As50	$67 \pm 8$	$57 \pm 6 (84)$	$10 \pm 2(14)$	$66 \pm 10$	$99 \pm 1$		$33 \pm 4$	13 ± 1 (39)	$16 \pm 2$ (47)	$29 \pm 3$	$86.1\pm0.6$	
As50Sb300	$99 \pm 6$	$64 \pm 20$ (64)	$29 \pm 10$ (30)	$93 \pm 10$	$94 \pm 4$		$52 \pm 3$	$22 \pm 2$ (42)	24.1 ± 0.7 (46)	$46 \pm 2$	$88.9\pm0.7$	
	Flooded											
Treatment		Root					Shoot					
	Total As	As <sup>III</sup>	As <sup>v</sup>	Total As in	Extracted		Total As	As <sup>III</sup>	$As^{V}$	Total As in	Extracted	
	(mg/kg)	(mg/kg)	(mg/kg)	extracts (mg/kg)	(%)		(mg/kg)	(mg/kg)	(mg/kg)	extracts (mg/kg)	(%)	
Aged-L	$144 \pm 20$	$0.20 \pm 0.07 \ (0.1)$	$140 \pm 20 \ (99.9)$	$140\pm20$	$96 \pm 2$		$42 \pm 1$	-	41 ± 1 (99.9)	$41 \pm 1$	$98 \pm 1$	
Aged-H	$973 \pm 90$	$0.73 \pm 0.06 \ (0.08)$	$937 \pm 80 \ (99.9)$	$938\pm80$	$96.4\pm0.9$		$245 \pm 10$	$0.03 \pm 0.03 \ (0.01)$	$248 \pm 10 \ (99.99)$	$248 \pm 10$	$101.1\pm0.4$	
As50	$183 \pm 5$	$0.7 \pm 0.5 \; (0.4)$	183 ± 7 (99.6)	$184 \pm 6$	$100 \pm 1$		$91 \pm 7$	$0.01 \pm 0.01 \; (0.01)$	$99 \pm 6  (99.99)$	$99 \pm 6$	$109 \pm 2$	
As50Sb300	$262 \pm 7$	$0.6 \pm 0.5 \; (0.2)$	$267 \pm 10 \ (99.8)$	$267 \pm 10$	$102 \pm 2$		$114 \pm 4$	-	117 ± 3 (99.99)	$117 \pm 3$	$104 \pm 2$	

**Table 7.2** Total As and As species concentrations in roots and shoots of *I. aquatica* cultivated in non-flooded and flooded soils. Data are presented as mean  $\pm$  SE (n = 3). Values in parentheses are the percentage of As species. There was no detectable MMA and DMA.

-: not detected

Regarding how As species transported from roots to shoots, both As(III) and As(V) were transported to shoots or only As(V) was transported to shoots, followed by the reduction of As(V) to As(III) in shoots. The translocation pattern of As species from roots to shoots of *I. aquatica* is still unclear. For rice, the reduction of As(V) to As(III) and the complexation of As(III) with thiols occurred in rice shoots (Tripathi et al. 2007; Zhao et al. 2009).

#### 7.3.4.2. As speciation in flooded *I. aquatica*

The extraction efficiencies of As species in roots and shoots of *I. aquatica* grown in flooded soil were in the range of 96 - 109%. For flooded *I. aquatica*, the As uptake and speciation was expected to be similar to non-flooded I. aquatica, but with higher proportion of total As and As(III) in plant tissues. This is because As was mobilised and soluble As increased in flooded soils as illustrated in sections 7.3.1 and 7.3.2, with As(III) as the dominant species in soil pore water. Thus, As(III) was expected to be present at higher proportions in flooded I. aquatica. However, the results of As speciation showed that As(V) was the main As species in flooded plant tissues, while As(III) was present in trace amounts (Table 7.2), which is opposite to both the expected results and the As speciation in non-flooded *I. aquatica*, and also opposite to the common species found in flooded paddy rice. These speciation results indicate differences between non-flooded and flooded I. aquatica in regards to the internal modifications of arsenic, perhaps due to different regulations of arsenic in the cell wall or different binding of arsenic to phytochelatins (Bergqvist & Greger 2012; Tripathi et al. 2007). The relative amount of As species in flooded roots and shoots were similar (Table 7.2), indicating that the same processes throughout the *I. aquatica* body may occur in response to arsenic or internal alteration of arsenic species did not occur within the I. aquatica.

The accumulation and speciation of As in tissues of flooded *I. aquatica* may involve (1) the uptake of As(V) by roots, followed by the translocation of As(V) from roots to shoots or (2) the uptake of As(III) or both As(III) and As(V), followed by the oxidation of As(III) to As(V) in root cells and the subsequent translocation of As(V) to shoots. A previous study on the accumulation and speciation of As in plants collected from different habitats (submerged, emergent, and terrestrial plants) reported that As(V) was the predominant species in submerged plants (Bergqvist & Greger 2012). The authors only investigated the patterns of As uptake by plants for As phytoremediation purposes.

The soil/sediment chemistry and plant characteristics were not presented and the mechanism of As accumulation and speciation in the plants were not explored.

As discussed in sections 7.3.1.2 and 7.3.2.2, the  $C_{DGT-As(III)}$  decreased and  $C_{DGT-As(V)}$ increased by a factor of 10 for flooded historically contaminated soils and 32 for flooded recently contaminated soils with I. aquatica compared to flooded soils without I. aquatica, indicating the diffusion of oxygen from above-ground parts to roots and the presence of oxygen in the rhizosphere. This supports the oxidation of As(III) to As(V) in rhizosphere and roots. In addition, I. aquatica is known as a semiaquatic plant (Shaibur et al. 2009) and grew healthily in the studied flooded soils. The roots of I. aquatica in flooded soils were better developed than those in non-flooded soils and no toxicity symptoms were observed. This indicates that I. aquatica oxygenated the rhizosphere to stimulate its growth under flooded conditions where oxygen was supposed to be microbially depleted and the oxidation of reduced species may be accompanied, the process supported by María-Cervantes et al. (2010). Taking all information into account, the oxidation of As(III) to As(V) in flooded planted soils and I. aquatica seems to be a reasonable hypothesis in this case. The current research is the first study on the As speciation in flooded water spinach (I. aquatica) in comparison with that in non-flooded soils. The mechanism of As uptake and speciation in I. aquatica, especially in flooded scenario, is still unknown. Further studies on this matter are warranted to explain this unusual trend.

### 7.3.4.3. Sb speciation in plants

In the current study, speciation of Sb in *I. aquatica* tissues was not determined because the extraction of Sb species in plant materials is yet to be successfully developed. The efficiency of published extraction procedures was low and varied greatly among extractants and materials. For example, citric acid extraction of rice roots gave a yield of  $92 \pm 11\%$ , while only  $55 \pm 14\%$  of Sb was extracted from total Sb in rice shoots (Okkenhaug et al. 2012). In addition, the change of Sb species during extraction was also problematic, counteracting the successful development of extraction of Sb species in plants. For instance, oxalic and ascorbic acid extraction was claimed to be suitable extractant for trimethylantimony (TMSb), but this new method did not completely discriminate Sb(III) and Sb(V) due to the reduction of Sb(V) to Sb(III) during extraction (Mestrot et al. 2016). The development of a reliable extraction method for Sb species in plants is warranted. It is known that Sb(V) is the dominant species in oxic soils (Wilson et al. 2010; Wilson et al. 2014). Given that the proportion of Sb(III) increased in the anoxic soils illustrated by DGT-labile Sb(III) slightly increasing with flooded soil depth, especially at the initial stage of plant bioassay, Sb(V) was still dominant in soil pore water. Previous studies showed that Sb(III) was more strongly adsorbed on Fe (hydr)oxides than Sb(V) (Johnson et al. 2005; Leuz et al. 2006), thus the presence of iron plaque on plant roots might favour the influx of Sb(V) into roots. This is because Sb(V) less get trapped by the iron plaque. It can be speculated that uptake of Sb(V) by roots of *I. aquatica* cultivated in the studied soils would be a preferential pathway. Previous research by Okkenhaug et al. (2012) showed that Sb was mainly present as Sb(V) in roots and shoots of flooded paddy rice, partially supporting our speculation.

### 7.3.4.4. Practical implications

Arsenic concentrations in edible shoots of *I. aquatica* grown in non-flooded and flooded contaminated soils significantly exceeded the Australian maximum permissible value for food (1 mg As/kg, wet weight, WW) (FSANZ 2013); with an average water content of shoots of 90%, the value is equivalent to 10 mg As/kg, dry weight (DW). This indicates that the consumption of *I. aquatica* grown in the studied soils, even in soils with the lowest labile As, poses high risks to humans. Since flooding resulted in an increase of As accumulation in *I. aquatica* by >2 times, the intake of *I. aquatica* exposed to flooded contaminated soils is at higher risks.

In addition to exceeding the permissible value, the more alarming concerns are that *I. aquatica* grew healthily with no toxic signs observed and As was only present as inorganic As species. It is well known that the toxicity of inorganic As species is higher than organic species, in which As(III) is more toxic than As(V) (Mandal & Suzuki 2002). Taking these results into consideration, the consumption of the *I. aquatica* exposed to As-contaminated soils, especially under flooding poses severe risks to human health via the food chain. Based on the results of As risk assessment in section 5.3.6 (chapter 5) and 6.3.6 (chapter 6), it can be inferred that the ingestion of As-contaminated shoots of *I. aquatica* at concentrations in this study would greatly cause unacceptable non-carcinogenic and carcinogenic risks to both adults and children. Therefore, suitable strategies for the management of historically contaminated soils and suitable vegetable-growing practices to minimise cancer-related risks are required.

# 7.3.5. Evaluating the performance of DGT as a predictor of bioaccumulation and speciation in *I. aquatica*

As discussed in section 7.3.3, total soil concentration is not a good indicator for predicting metalloid uptake by plants and soil risk assessment. In addition, changes in the lability and speciation of As and Sb in planted soils occurred compared to unplanted soils. To better evaluate the performance of DGT in predicting As and Sb uptake by *I. aquatica* under different redox conditions, relationships between As and Sb bioaccumulation and their total soil concentrations and  $C_{DGT}$  measured at the initial stage and the end of bioassay were considered. As observed, the fibrous roots of *I. aquatica* developed healthily and occupied the entire deep soil layer; thus, relationships between bioaccumulation and  $C_{DGT}$  were examined using the average  $C_{DGT}$  below 10-mm soil depth (Figure 7.5).

The results also showed that the DGT measure in 5-week flooded soils with or without plants was a better predictor of As accumulation in *I. aquatica* ( $R^2 = 0.975 - 0.996$  for roots,  $R^2 = 0.847 - 0.959$  for shoots) than total soil As ( $R^2 = 0.944$  for roots,  $R^2 = 0.785$  for shoots), illustrated by higher correlation coefficients for both roots and shoots. The As concentrations in roots was more strongly correlated with  $C_{DGT-As}$  in 5-week flooded soils without plants ( $R^2 = 0.975$ ) than that in 1-week flooded soils without plants ( $R^2 = 0.975$ ) than that in 1-week flooded soils without plants ( $R^2 = 0.975$ ) than that in 1-week flooded soils without plants ( $R^2 = 0.746$ ). This suggests that the strong increase in As mobility after 5 weeks contributed to As accumulation by *I. aquatica*, whereas after 1 week the system may have been in transition still. The correlation between As shoot concentration and  $C_{DGT-As}$  in 5-week flooded soils without plants ( $R^2 = 0.847$ ) was relatively similar to that in 1-week flooded soils with *I. aquatica* was considered, greater correlations between  $C_{DGT-As}$  and As concentrations in both roots ( $R^2 = 0.996$ ) and shoots ( $R^2 = 0.959$ ) were observed. These suggests that the DGT measurements in planted soils captured changes in As mobility induced by *I. aquatica*, leading to the better reflection of As uptake by tissues of *I. aquatica*.

As with As, DGT outperformed total soil Sb in predicting Sb accumulation in tissues of *I. aquatica* grown in flooded soils ( $R^2 > 0.92$  for  $C_{DGT-Sb}$  and Sb bioaccumulation,  $R^2 < 0.01$  for total soil Sb and Sb bioaccumulation). This is demonstrated by the very strong relationships between Sb concentrations in *I. aquatica* tissues and  $C_{DGT-Sb}$  compared to the very weak relationships observed for total soil Sb (Figure 7.5). In contrast,  $C_{DGT-Sb}$  measured in 5-week flooded soils with or without plants was as well correlated with Sb

bioaccumulation as with the 1-week flooded soils, which suggests that the Sb biogeochemistry responded to the waterlogged conditions more quickly than As. It seems that the Sb uptake by roots and the Sb translocation from roots to shoots were proportional to available concentrations. Sb in the soil systems was at equilibrium. As discussed in sections 7.3.1 and 7.3.2, the mobility of Sb decreased as flooding time increased, but Sb bioaccumulation in flooded plants increased compared to that in non-flooded plants. This suggests that the mass flow pathway and apoplastic bypass from roots to shoots may be important for Sb uptake and translocation due to *I. aquatica* subjected to physical damage and anaerobic stress.

The correlation analysis was also performed on  $C_{DGT-As(III)}$ ,  $C_{DGT-As(V)}$  and As(III) and As(V) in *I. aquatica* tissues to evaluate the possible preferential As species for uptake by flooded *I. aquatica* (Figure 7.6). The results showed that  $C_{DGT-As(III)}$  measured in 1-week flooded soils, 5-week flooded unplanted soils, and 5-week flooded planted soils (from 20 mm below the soil surface downwards) were not correlated with As(III) concentrations in roots. In contrast,  $C_{DGT-As(V)}$  measured in 5-week flooded soils with or without plants was strongly correlated with As(V) concentrations in tissues of *I. aquatica* (R<sup>2</sup> >0.88), and was better correlated with As(V) in *I. aquatica* tissues than that in 1-week flooded soils (R<sup>2</sup> <0.01). Similar to  $C_{DGT-As}$ ,  $C_{DGT-As(V)}$  measured in flooded soils with plants was the best predictor of As(V) concentrations in both roots (R<sup>2</sup> = 0.975) and shoots (R<sup>2</sup> = 0.985) of flooded *I. aquatica* in this study.

In addition, although As(III) concentrations in flooded soils were increased with the studied flooding time, As(III) concentrations were low in roots, and present at trace amounts or not-detected in shoots, while As(V) was the main species in tissues. Thus, the relationships between  $C_{DGT-As(III)}$  in flooded soils and As(V) in *I. aquatica* tissues were also performed to identify if As(III) was taken up by roots and if there was redox reaction in flooded *I. aquatica*. Interestingly, the results showed that  $C_{DGT-As(III)}$  in flooded soils were strongly correlated with As(V) in *I. aquatica* tissues ( $R^2 = 0.948 - 0.985$  for roots and  $R^2 = 0.779 - 0.864$  for shoots), even with  $C_{DGT-As(III)}$  in 1-week of flooding (Figure 7.7). This suggests that As(III) may have been accumulated at the initial stage of plant growth. Similar to  $C_{DGT-As(V)}$ ,  $C_{DGT-As(III)}$  in 5-week flooded with plants was best correlated with As(V) in both roots and shoots of *I. aquatica*. This confirms that DGT measures included changes in As speciation induced by plants, resulting in the better reflection of plant uptake of As species.

In summary, after 1 week of flooding As(III) in flooded soil was not correlated with As(III) in plants, but strongly correlated with As(V) in plants. There was very weak relationship between As(V) in plants and in soils. These suggest that As(III) was preferentially accumulated by plant roots and subsequently oxidised to As(V). After 5 weeks of flooding, there was also a very week correlation between As(III) in soils and As(III) in plants and strong correlations between both As(III) and As(V) in flooded soils and As(V) in plants. These suggest that both As(III) and As(V) in flooded soils and As(V) in plants. These suggest that both As(III) and As(V) were accumulated by plant roots, followed by the oxidation of As(III) to As(V) in plants. Taking all into consideration, these correlations support the discussion in section 7.3.4 that roots accumulated As(III) or both As(III) and As(V), followed by the oxidation of As(V) and the subsequent translocation of As(V) to shoots. This also explains the results of speciation analysis showing that As(V) was the main species in flooded *I. aquatica*.

Overall, the *in situ* sampling capability, which can be readily utilized in different environmental conditions, and the selective measurement capability for As(III) made DGT an effective technique for this study. Therefore, DGT is a very promising tool for measuring the bioavailability and speciation of As and Sb in soils, identifying differences in their biogeochemical behaviour, and predicting their uptake by plants under various conditions.



**Figure 7.5** Relationships between As and Sb bioaccumulation and their average  $C_{DGT}$  measured in 1-week flooded soils without plants, 5-week flooded soils with plants (from -20 mm downwards). Roots ( $\diamond$ ). All data presented as mean  $\pm$  SE (n = 3 for tissue concentrations, n = 7 for  $C_{DGT}$ ) have error bars; many are smaller than symbols.



**Figure 7.6** Relationships between As(III) and As(V) concentrations in plant tissues and  $C_{DGT-As(III)}$ ,  $C_{DGT-As(V)}$  measured in 1-week flooded soils without plants, 5-week flooded soils with plants (from -20 mm downwards). Roots ( $\circ$ ), Shoots ( $\blacktriangle$ ). All data presented as mean  $\pm$  SE (n = 3 for tissue concentrations, n = 7 for  $C_{DGT}$ ) have error bars; many are smaller than symbols.



**Figure 7.7** Relationships between As(V) concentrations in plant tissues and  $C_{DGT-As(III)}$  measured in 1-week flooded soils without plants, 5-week flooded soils without plants, and 5-week flooded soils with plants (from -20 mm downwards). Roots ( $\circ$ ), Shoots ( $\blacktriangle$ ). All data presented as mean  $\pm$  SE (n = 3 for tissue concentrations, n = 7 for  $C_{DGT}$ ) have error bars; many are smaller than symbols.

### 7.4. Conclusions

Our first use of the combined mercapto-silica-DGT, Metsorb-DGT, and DET for Fe successfully examined the dynamics of lability and speciation of As and Sb in flooded soils and differentiated the mechanisms of the mobilisation of As and Sb in soils under changes in redox conditions. The reductive dissolution of Fe (hydr)oxides was the main mechanism for As mobilisation in historically flooded soils, while the Sb mobilisation was not directly associated with Fe (hydr)oxide reduction and may involve the desorption of Sb(V) from soil, followed by the reduction of Sb(V) to Sb(III) and the immobilisation of Sb(III). During the flooding time of contaminated soils the lability of As increased, while that of Sb decreased, which may be due to the precipitation with sulphides. Thus, DGT probes for sulphide measurements need to be applied in future studies to fully explain the mechanism of Sb mobilisation.

The application of multi-DGT probes also demonstrated the differences in As and Sb behaviour in contaminated soils subjected to anoxic conditions in spatial resolution. The effect of *I. aquatica* on the changes in As and Sb speciation in flooded planted soils were also identified by DGTs. Due to the presence of Sb the lability of As in As+Sb soil was higher than that in As-only soil, leading to higher As bioaccumulation in *I. aquatica*. The relationships between As(III)/As(V) in soils measured by DGTs and As(III)/As(V) in plants determined by HPLC-ICP-MS allowed predicting the preferential As species uptake by flooded *I. aquatica* and explaining the proposed As speciation in flooded *I. aquatica*. This study has demonstrated that DGT is most useful under changing conditions and was a superior method in this study.

The waterlogging of contaminated soils resulted in much elevated concentrations of As and Sb in flooded *I. aquatica*, compared to those in non-flooded *I. aquatica*. The concentrations of As in edible shoots far exceeded the maximum permissible value for food (1 mg As/kg, WW). In addition, only inorganic As species were identified in *I. aquatica* tissues, with As(III) being the dominant species in non-flooded *I. aquatica*, while As(V) being the main species in flooded *I. aquatica*. As a result, the ingestion of edible shoots of *I. aquatica* exposed to soils contaminated with As poses dangerous risks for humans via the food chain. Therefore, the appropriate management of As and Sb-contaminated soils and agronomic practices are crucial to ensure environmental health and food safety.

191
### **Chapter 8. Concluding remarks and future directions**

The aim of this study was to investigate the biogeochemical behaviour of Sb compared to that of As in contaminated soils in relation to their uptake by importantly agricultural plants identified by the diffusive gradients in thin films (DGT) technique in comparison to selective extractions. This was successfully obtained by achieving the following research objectives:

- Investigating the partitioning of As and Sb in various soil binding phases and their lability in different types of contaminated soils;
- Evaluating the performance of the DGT technique in measuring labile As and Sb in historically co-contaminated soils to two cultivars of radish (*Raphanus sativus*) compared to other techniques;
- Assessing the performance of the DGT technique in measuring the lability of As and Sb in historically co-contaminated soils with varying physicochemical soil properties and recently contaminated soils and predicting their accumulation by water spinach (*Ipomoea aquatica*) compared to other techniques;
- Investigating the fractionation and lability of As and Sb in single and multicontaminant amended soils and their competitive interactions when they coexist in soils in relation to their uptake by water spinach (*Ipomoea aquatica*);
- Investigating the mobilisation and speciation of As and Sb in soils induced by redox changes using multiple binding resins in DGT;
- Evaluating changes in As and Sb bioaccumulation and speciation in water spinach (*Ipomoea aquatica*) cultivated in contaminated soils subjected to different redox environmental conditions, which was linked to changes in As and Sb mobility and speciation in soils under such changing conditions predicted by DGTs.

This chapter draws together the research outcomes from this thesis. It will consider (1) the biogeochemical behaviour of As and Sb in various soil types under different redox conditions using DGT, SEP, and soil solution analysis; (2) the competitive interactions of As and Sb in soils under different redox conditions using DGTs; (3) uptake of As and Sb by a variety of edible vegetables under various growing conditions, focusing uptake

patterns, speciation transformation, and risk assessment of As and Sb-enriched vegetables to human health; (4) evaluation of the combination of multiple DGT to differentiate behavior of As and Sb in contaminated soils ; (5) evaluation of the performance of DGT in predicting As and Sb uptake by vegetables compared to other soil extraction methods.

### 8.1. The geochemical behaviour of As and Sb in contaminated soils

The partitioning and lability of As and Sb in contaminated soils determined by sequential extraction and DGT were deeply discussed in sections 3.3.2, 3.3.3, 4.3.2, 4.3.3, 5.3.2, 5.3.3, 7.3.1, and 7.3.2. The test soils were many series of historically co-contaminated soils (mining soils) with various soil physicochemical properties (e.g. Fe, organic matter, phosphorus), series of historically As-contaminated soil (cattle dip soil), and series of recently contaminated soils (As-only soils, Sb-only soils, and As+Sb soils) having comparable soil mineralogy.

# **8.1.1.** The partitioning and lability of As and Sb in various soils determined by sequential extraction procedure (SEP)

In historically co-contaminated soils, irrespective of historic soil types with different soil physicochemical properties, As and Sb associated with non-specifically sorbed and specifically sorbed fractions were low (<10 and 1% of total soil As and Sb, respectively). The sum of As and Sb in these two fractions were defined as the bioavailable fraction which are readily mobilised and available for plant uptake (Wenzel et al. 2001). As was mainly associated with amorphous and crystalline Fe and Al oxide fractions (>60% of total soil As), while Sb predominantly bound to the residual phase (>60% of total soil Sb), followed by the crystalline phase (up to 29%). The lower proportions of Sb in the metal oxide phases and the higher percentage of Sb in the residual phase compared to As could be because of Sb association with organic matter, sulphides, or resistant minerals. Fe oxides and organic matter are phases that are somewhat (moderately) labile especially under certain biogeochemical conditions. Minerals associates with silicates and sulphides are non-labile. Thus, in historically co-contaminated soils Sb was biogeochemically much less mobile than As despite total soil Sb being higher.

In the recently contaminated soils studied, Sb was mainly found in amorphous and crystalline iron oxides fractions, whereas As was primarily partitioned in amorphous iron oxide and specifically sorbed fractions. Regardless of singly or co-contaminated soils, the lability of Sb was less than that of As. The non-specifically sorbed and specifically sorbed Sb and As (the bioavailable fraction) in recently contaminated soils constituted 18% and 30% of total Sb and As, respectively, which were significantly higher than that in historically contaminated soils. This data suggests that As and Sb in recently contaminated soils have high lability, which is attributed to the soluble form of added As and Sb, and their shorter equilibration time in spiked soils compared to the long aging in historically contaminated soils. Irrespective of contamination sources and soil types, the Fe and Al oxides were the main sinks for these metalloids, especially As.

# **8.1.2.** Labile As and Sb in various soils measured by the diffusive gradients in thin films (DGT) technique

The DGT technique successfully identified differences in the biogeochemical behaviour of As and Sb in contaminated soils with various soil physicochemical properties, illustrated by the variation in the R-value ( $C_{DGT}/C_{sol}$ ) and distribution coefficients ( $K_d$ ). The As and Sb soil solution and DGT-labile concentrations accounted for only a small proportion of total soil concentrations (<0.1% for historically contaminated soils,  $\leq 2\%$ for recently contaminated soils). Arsenic was partially resupplied from the solid phase to soil solution (R-value ranging from 0.21 - 0.65), while the resupply of Sb from the solid phase was consistently low (R-value being 0.12 - 0.19) indicating recharge of Sb in adjacent porewaters by diffusion only. The kinetic resupply of As varied with soil physicochemical properties, which was highest in the soil having the lowest iron oxides. Unlike As, the resupply rate of Sb was relatively unchanged across soil types with various soil physicochemical properties, demonstrated by the relatively constant Rvalue and K<sub>d</sub> among soils, which confirms that Sb was biogeochemically immobilised in contaminated soils. DGT was sensitive to differentiate the lability, capacity, and kinetics of As and Sb release from the solid phase of various soils. DGT results were consistent with and complementary to the selective extraction data.

# 8.1.3. The dynamics of lability and speciation of As and Sb in contaminated soils induced by redox changes identified by DGT

Historically and recently contaminated soils were subjected to two water regimes: nonflooded and flooded. Mercapto-silica-DGT probes for sampling As(III) and Sb(III) and Metsorb-DGT probes for sampling total labile As and Sb were deployed into flooded soils after 1 week (without plants) and 5 weeks of flooding (with and without plants). The diffusive gels of mercapto-silica DGT probes were used as diffusive equilibration in thin films (DET) for measuring Fe, giving pore water Fe concentrations in flooded soils.

The combined mercapto-silica-DGT, Metsorb-DGT, and DET for Fe allowed successful examination of the speciation and lability of As and Sb in flooded soils and differentiated the mechanisms of As and Sb mobilisation in soils under changing redox conditions. During extended flooding of historically contaminated soils the lability of As increased with soil depth and over time, while lability of Sb decreased, which may be due to Sb precipitation with sulphides. The reductive dissolution of Fe (hydr)oxides was the main mechanism for As mobilisation in flooded soils, while the Sb mobilisation was not directly associated with Fe (hydr)oxide reduction. In flooded soils As(III) was the dominant As species, whereas Sb(V) was the dominant Sb species.

The dynamics of As and Sb in recently contaminated soils exhibited the same patterns as in historically contaminated soils. However, their mobilisation mechanism may be involved the desorption of As and Sb from soil solid phases due to exchangeable processes, followed by the reduction of As(V)/Sb(V) to As(III)/Sb(III). The presence of *I. aquatica* in flooded soils lowered trivalent species and enhanced pentavalent species of As and Sb in flooded historically contaminated soils compared to flooded soils without plants, which was likely due to the oxygen released from the roots creating the narrow oxidising zone around roots. This study highlights a strong advantage of *in situ* DGT measurements, which able to be done in different redox conditions and combine with selective analysis.

### 8.2. Competitive interactions of As and Sb in soils under different redox conditions

The chemical fractionation, lability, and kinetic resupply of As and Sb in As+Sb soils were compared to those in singly contaminated (As-only, Sb-only) soils under aerobic conditions, assessed by SEP and DGT. The DGT profiles of As, Sb, and their species in these flooded soils were also examined by multiple *in situ* DGT samplers. These results provided a better understanding of the competitive interactions between As and Sb in soils under changes in redox conditions.

In comparison with As and Sb fractionation in singly contaminated (As-only and Sbonly) soils, the proportions of As and Sb in the non-specifically sorbed fraction of As+Sb soils were higher, while those in the amorphous Fe oxide fraction tended to be lower. This is likely due to the simultaneous addition of As and Sb into soils, resulting in the decrease in binding sites and the redistribution of other substances. Thus, the sorbed and bound As and Sb decreased and more As and Sb were found in the available fractions. The competitive interaction of As and Sb for their preferred soil binding sites affected the distribution of As and Sb in various soil binding phases, which made them more labile and available in the As+Sb co-contaminated soils, compared to singly contaminated soils.

The average proportion of  $C_{sol-As}$  was lower in As+Sb soils than in As-only soils under aerobic conditions, whereas that of  $C_{sol-Sb}$  in As+Sb soils was ~2 times higher than in Sb-only soils. This phenomenon also occurred for  $C_{DGT-As}$  and  $C_{DGT-Sb}$ , but to a lesser extent (as discussed in section 6.3.3). In contrast, when singly contaminated (As-only and Sb-only) and As+Sb contaminated soils were subject to flooding, the presence of As did not cause differences in the dynamics of Sb speciation and lability in Sb-only and As+Sb soils. In contrast, the presence of Sb enhanced the higher lability of As in As+Sb soil than As-only soil (as discussed in section 7.3.2).

The variation in competitive interactions of As and Sb is likely a result of the differences in their adsorption and precipitation capacity in the soil systems, which is dependent on their species present. The pentavalent and trivalent species were the dominant species in aerobic and anaerobic soils, respectively. As(V) has a higher affinity for Fe oxides compared to Sb(V) (Qi & Pichler 2017; Wilson et al. 2010), while Sb(V) has stronger binding strength with organic matter (OM) than As(V). The binding stability of As/Sb in soil types was also presented in order of As(V) – Fe(III) >> Sb(V) -OM > As(V) - OM > Sb(V) - Fe(III) (Dousova et al. 2015). Thus, when As and Sb were added together in soils under aerobic condition, Fe oxides gave preference to As, leading to more added As in solid phases and more added Sb in soil solutions. In anaerobic soils, Sb(III) has higher affinity for Fe oxides (Qi & Pichler 2017) and greater precipitation rate with sulphides than As(III) due to its lower solubility constant (as discussed in section 7.3.2). Hence, Sb would preferentially undergo adsorption and precipitation in the anoxic soil system, leading to higher amounts of As in As+Sb soil than in As-only soil where labile As decreased due to As involved in adsorption and precipitation processes.

It is clear that the speciation of As and Sb in the soil environment determines their lability and solubility, which in turn influences their accumulation by plants.

# **8.3.** Uptake of As and Sb by a variety of vegetables cultivated in various soils under different redox conditions

#### 8.3.1. Uptake trend

The bioaccumulation of As and Sb in the cherry bell radish (*Raphanus sativus*) cultivated in historically contaminated soils was very low due to the low lability of As and Sb in soils. There was more arsenic accumulated in roots and shoots than Sb; however, Sb had a higher ability to translocate from roots to shoots than As (a factor of 2.5), to the extent that much higher Sb concentrations were observed in shoots, suggesting a quite different pattern of transport within the plant. The white icicle radish displayed the same uptake trend as the cherry belle radish, but was less able to translocate As and Sb from roots to shoots than the cherry belle cultivar. These results suggest that the *R. sativus* uptake was clearly element and cultivar-specific.

In contrast, the leafy vegetable, water spinach (Ipomoea aquatica) cultivated in historically contaminated soils had a high capacity to accumulate As and Sb with concentrations in roots being 10 and 14 times higher than shoots, respectively. The bioaccumulation factor for As was >1, thus *I. aquatica* was considered as an As accumulator plant. The average bioaccumulation factor for As in *I. aquatica* grown in historically contaminated soils was 55.7-fold higher than that of Sb, while the translocation factor of Sb was 2.2-fold higher than that of As, indicating differences in uptake patterns. The same uptake patterns were also observed for As and Sb in I. aquatica grown in singly and co-amended soils. Compared with single-element soils under anaerobic conditions, the average bioaccumulation factor of Sb in As+Sb soils significantly increased, while that of As slightly decreased. The results demonstrated that the presence of a more toxic element (As) stimulated Sb uptake by I. aquatica cultivated in aerobic soils. The opposite phenomenon occurred for their interactive effects in *I. aquatica* grown in anaerobic soils. The results showed that the presence of As in flooded As+Sb soils was less likely to affect Sb uptake by *I. aquatica*, but the presence of Sb in flooded As+Sb soils led to the greater As bioaccumulation in I. aquatica exposure to As+Sb soil than As-only soil. This is consistent with the variation in DGT-labile As and Sb in soils induced by redox changes.

Concentrations of As and Sb in tissues of flooded *I. aquatica* were 2.2 - 3.1 and 2.0 - 5.7 times higher than those in non-flooded *I. aquatica*, respectively. Thus, the consumption of edible shoots of *I. aquatica* cultivated in flooded contaminated soils poses >2 times higher risks to humans. Irrespective of growing conditions and soil types, concentrations of As and Sb in roots of *I. aquatica* were significantly higher than in shoots.

#### 8.3.2. Speciation transformation of As in *I. aquatica*

Inorganic As species were found in both roots and shoots of *I. aquatica* with both As(III) and As(V) present in non-flooded *I. aquatica*, which is supported by previous studies on As speciation in terrestrial plants. By contrast, mainly As(V) was present in flooded *I. aquatica*, which is opposite to the As speciation in flooded paddy rice. The predominance of As(V) in flooded *I. aquatica* was also opposite to the expected results that As(III) would be present in flooded *I. aquatica* at a greater extent than that in non-flooded *I. aquatica* because As(III) was the dominant species in flooded contaminated soils. This can be explained by the DGT data obtained in flooded planted soils showing that  $C_{DGT-As(III)}$  decreased, but  $C_{DGT-AS(V)}$  considerably increased compared to flooded soils without plants. This indicates the diffusion of oxygen from above-ground parts to roots, the presence of oxygen in the rhizosphere, and the oxidation of As(III) to As(V). The analyses of relationships between As(III)/As(V) in soils and As(III)/As(V) in plant tissues demonstrate that roots accumulated As(III) or both As(III) and As(V), followed by the oxidation of As(III) to As(V) and the subsequent translocation of As(V) to shoots (as discussed in section 7.3.5).

Sb species in *I. aquatica* tissues were not determined, but it was speculated that Sb(V) uptake by roots of *I. aquatica* grown in the studied soils would be favoured relative to Sb(III) (as discussed in section 7.3.4).

#### 8.3.3. Risk assessment of As and Sb in *I. aquatica* to human health

Even for *I. aquatica* grown in the lowest treatment of historically contaminated soils under aerobic conditions, the shoots of *I. aquatica* accumulated As at very high concentrations which far exceeded the permissible limit of As in food set by the Food Standard Agency of China, Australia and New Zealand, and FAO/WHO, but no visible toxicity symptoms was observed in the plant, which is an alarming observation. *I. aquatica* presented direct risks to human health where  $C_{sol-As} \ge 5.27 \mu g/L$  and  $C_{DGT-As}$   $\geq$ 3.42 µg/L. The USEPA hazard quotient above 1, and cancer risk value above 10<sup>-4</sup> for both adults and children, confirm high cancer risks to humans from consumption of As-contaminated *I. aquatica*.

Interestingly, the *I. aquatica* exposed to flooded soils grew healthily, had greater biomass compared to non-flooded *I. aquatica*, and did not exhibited visible toxicity symptoms. Inorganic As species were mainly found in *I. aquatica* tissues and known to be more toxic than organic As species. Consequently, the intake of *I. aquatica* cultivated in contaminated soils poses severe risks to humans via the food chain, especially when the semiaquatic growing practice was applied.

The ingestion of *I. aquatica* cultivated in Sb-only soils ( $\geq 650 \text{ mg Sb/kg}$ ) poses unacceptable non-carcinogenic risk to humans, but it looked healthy and there is no sign of toxicity symptoms. The Sb-enriched edible shoots of *I. aquatica* cultivated in Sbonly soils displayed the increasing trend of biomass with the increasing amounts of soluble Sb added into soils. As a result, Sb-contaminated shoots cannot be recognised and distinguished from uncontaminated shoots, compared to As-enriched shoots of *I. aquatica* cultivated in As-contaminated soils where shoot biomass decreased when the plant exposed to high soil concentrations. This implies that growing *I. aquatica* in Sbcontaminated soils should be avoided because the plant presents hidden risks and may pose non-carcinogenic risks to human health via the food chain.

The results suggest that the highly As-contaminated soil and underground water in some Asian regions may be dangerous and not suitable for cultivating water spinach (*I. aquatica*). Thus, cultivation of *I. aquatica* in As-contaminated soils or water or irrigation of *I. aquatica* with As-enriched water should be restricted in terms of food safety and human health. Appropriate strategies for the management of contaminated soils and suitable vegetable-growing practices to minimise As and Sb enrichment in soil and vegetables are crucial for the development of sustainable agriculture.

Overall, regardless of plant species, both As and Sb were restricted in roots and the translocation of Sb from roots to shoots was more efficient than As. The accumulation of As and Sb was higher in the leafy vegetable than in root vegetables. The uptake of As and Sb was clearly element, plant, and cultivar specific. In addition, the presented results indicate that total soil concentrations are not a good indicator for the prediction of plant uptake of metalloids. It also confirms that the biogeochemical processes and the

bioavailability of metalloids plays an important role in predicting their accumulation by plants.

# 8.4. Evaluation of the performance of DGT in predicting As and Sb uptake by vegetables compared with other soil extraction methods

The As and Sb concentrations in tissues of the tested vegetables were compared with labile As and Sb fractions in soils measured by DGT (as  $C_{DGT}$ ), soil solution analysis ( $C_{sol}$ ), and sequential extraction ( $C_{SEP-labile}$ ). The concentrations of As and Sb in tissues of radishes were highly positively correlated with all labile As and Sb fractions ( $R^2 = 0.96 - 0.99$  for cherry bell radish and  $R^2 = 0.92 - 0.99$  for white icicle radish in historically co-contaminated soils,  $R^2 = 0.87 - 0.97$  for As in white icicle radish grown in cattle dip soils). This indicates that tissue As and Sb concentrations in two cultivars of radish were well predicted by these labile measures.

The  $C_{SEP-labile}$ ,  $C_{DGT}$ , and  $C_{sol}$  were better predictors for As and Sb in radish roots (greater correlation coefficients) than in shoots, with the exception of  $C_{sol-As}$  and  $C_{SEP-labile Sb}$ . Arsenic and Sb accumulation by the white icicle radish from the test soils had a stronger relationship with  $C_{DGT}$  of As and Sb than their  $C_{sol}$ , especially from historically As-only-contaminated soils, which is illustrated by the higher correlation coefficients for  $C_{DGT}$ . From the predictive perspective, DGT exhibited better performance in predicting the phytoavailability of As and Sb in the studied soils, compared to soil solution concentration method.

Arsenic concentrations in tissues of *I. aquatica* cultivated in historically contaminated soils were better correlated with  $C_{SEP-labile}$ ,  $C_{DGT}$ , and  $C_{sol}$  than Sb, illustrated by greater correlation coefficients for As. When the soil Sb associated with amorphous Fe and Al oxides were considered, a greater correlation between the extractable Sb in soil and Sb in roots was observed. This indicates that Sb bound to amorphous Fe and Al oxides may contribute to the Sb uptake by *I. aquatica* or that the plant induced the release and resupply of Sb from the soil solid phase, which was not included in labile measures for Sb in soils. The coupling of DGT and sequential extractions was useful in predicting potentially phytoavailable As and Sb in contaminated soils to *I. aquatica*, evident from the As and Sb high-uptake-capacity plant, and providing understanding about potential soil processes in the root zone.

In contrast, Sb concentrations in tissues of *I. aquatica* cultivated in recently contaminated soils were more strongly correlated with  $C_{SEP-labile}$ ,  $C_{DGT}$ , and  $C_{sol}$  than for As. This may suggest that the reduction of *I. aquatica* yield grown in the freshly amended soils resulted from As than from Sb exposure. Arsenic concentrations in roots were significantly correlated with  $C_{SEP-labile}$ ,  $C_{DGT}$ , and  $C_{sol}$  compared to shoot As concentrations. The weaker correlations for As concentrations in shoots and soils were likely due to the contrasting relationships between increased As concentrations in soils and decreased shoot As concentrations and plant yield. When the saturated bioaccumulation data point was excluded, shoot As concentrations were significantly correlated with all measures,  $C_{SEP}$ ,  $C_{DGT}$ , and  $C_{sol}$ . The results demonstrated that all measures worked well in predicting As uptake by *I. aquatica* grown in the low range of labile As in soils. For *I. aquatica* cultivated in As+Sb soils,  $C_{DGT-As}$  and  $C_{SEP-As}$  predicted As concentrations in *I. aquatica* tissues better than  $C_{sol-As}$ .

Similar to radish (*R. sativus*), As and Sb concentrations in root of water spinach (*I. aquatica*) were better predicted than shoot As and Sb concentrations, which can be explained by the fact that roots directly contact soils, whereas bioaccumulation in shoots depends on the translocation from roots to shoots, usually governed by plant physiology. Overall, the prediction ability of these techniques varied with element, plant parts, plant species, and cultivars. DGT performed comparatively well with other measures, but DGT measurements are probably simpler than some of the other measurements which required single or multiple extractions of soil samples and soil characterisation. The combination of DGT, soil solution analysis, and SEP provided a better understanding of the biogeochemical behaviour of As and Sb in soils and useful mechanistic data (e.g. R,  $K_d$ ). Compared to labile extractions, the dynamic DGT technique was sensitive to differentiate the lability, capacity, and kinetics of As and Sb release from the solid phase of various soils.

Interestingly, DGT outperformed total soil As and Sb in predicting As and Sb bioaccumulation in flooded *I. aquatica*. DGT was able to capture changes in As mobility and speciation induced by flooded *I. aquatica*, leading to the better reflection of As uptake by tissues of *I. aquatica*. Importantly, the relationships between As(III)/As(V) in soils measured by DGTs and As(III)/As(V) in plants measured by HPLC-ICP-MS allowed predicting the preferential As species uptake by flooded *I. aquatica*. On the long-

term basis of risk assessment, DGT could be a promising tool in assessing and predicting bioavailable As and Sb in contaminated soils to plants.

# **8.5.** Evaluation of the combination of multiple DGT and DET to investigate behaviour of As and Sb in contaminated soils

Arsenic in soil solution was found to be partially resupplied from the solid phase. The resupply of Sb from the solid phase of historically contaminated soil is known to be slower than that of As. However, the biogeochemistry of As is strongly linked to that of Fe, thus the resupply of As from the solid phase is likely to be controlled by the redox reactions of Fe in sediment and soil. The strong adsorption of As on Fe oxides in aerobic conditions would limit the As resupply, while the reductive dissolution of Fe oxides in anaerobic conditions would partially resupply soluble As to DGT devices. DGT truly provides a useful technique for the interpretation of processes and mechanistic interactions occurring in sediment and soil and avoids the interferences and drawbacks of the traditional sampling techniques, which is supported by (Bennett et al. 2012a).

In this research the 3-mercapto-silica DGT was also used to selectively measure As(III) and Sb(III) in flooded soil pore water at high spatial resolution. The Metsorb DGT was simultaneously used to measure total labile inorganic As and Sb in flooded soil pore water at high spatial resolution. The difference between these two DGT measurements allows us to determine As(V) and Sb(V). The *in situ* characteristic of DGT sampling techniques coupled with the selectivity for redox states eliminates the potential changes in speciation that may occur during the collection of a sediment or soil core and subsequent processing steps such as extracting pore water samples, storing, and analysing them. This aspect is very important for the investigation of biogeochemistry of As and Sb which are very sensitive to environmental redox changes. The oxidation of As(III) to As(V), Sb(III) to Sb(V), and Fe(II) to Fe(III) during pore water processing steps confounds the redox chemistry and the relationships with their mobility.

The coupling of multiple DGT and DET techniques for Fe(II) allows us to measure codistributions of As species, Sb species, and Fe(II) at the same spatial location within flooded soil, eliminating artifacts related to the heterogeneous distribution of solutes within flooded soils and their pore water and providing accurate comparison of As and Sb biogeochemical behaviour. The combined application of *in situ* DGT and DET sampling techniques avoids changes in speciation and provides chemical profiles at high spatial resolution. Compared to the combined use of core-slicing, pore water sampler techniques and subsequent speciation analysis by HPLC-ICP-MS, DGT is likely the more reliable, economical and ease-for-use technique.

In addition, the dynamic DGT technique integrates soil properties and environmental conditions affecting soil processes, which determines the biogeochemical behaviour of As and Sb. DGT was non-destructive and far less laborious than some of the other measurements requiring single or multiple extractions of soil samples and soil characterisation. On the long-term basis of risk assessment, DGT provides a promise for assessing bioavailable As and Sb in contaminated soils routinely.

### **8.6. Future directions**

#### 8.6.1. Geochemical behaviour of As and Sb in soils

This research indicates the contrasting geochemical behaviour of As and Sb in contaminated soils. Irrespective of soil types, although both oxyanions are bound to the labile exchangeable fraction sufficiently to interact with changes in concentration of the other element and also to be taken up by plants, Sb was predominantly bound to the residual phase, whereas As mainly associated with Fe and Al oxides. The explanation was made that Sb was possibly associated with organic matter and sulphide fractions or other resistant minerals in soils, which was not included in the SEP used in this research. Thus, other extraction procedures targeting the organic matter and sulphide fractions of the distribution of Sb in various soil binding phases.

This research also demonstrates that regardless of soil types and contamination source, As was more labile and sustained than Sb. Arsenic was partially resupplied from the solid phase and its resupply varied with soil physicochemical properties. By contrast, the resupply of Sb was virtually minimal and relatively unchanged across soil types with various physicochemical properties. In addition, when contaminated soils were subject to flooding, labile As increased with soil depth and over time, while labile Sb decreased. The results showed that the mobilisation of As was associated with the reductive dissolution of Fe (hydr)oxides, but it was not the case for Sb. The explanation of the lower lability of Sb under various soil redox conditions was based on more Sb complexing with organic matter and in the inert forms (e.g. precipitation with sulphides), which were non-DGT labile. The precipitation with sulphides would be also a reasonable explanation for the higher lability of As in As-only soil compared to As+Sb soil (as discussed in section 7.3.2.1). Hence, DGT probes for sulphide measurements and the analysis of solid phase speciation by XAFS need to be applied in future studies to fully interpret the mechanism of Sb immobilisation.

To understand their competitive adsorption by soil compositions, soils should have identical soil properties, mineralogy, and course of aging. Our first study on the competitive interactions between As and Sb in soils used a series of soils amended with As only, Sb only, and mixture of As and Sb which simultaneously added into soils in which amounts of Sb were equivalent and higher than As in soil treatments. The results presented in chapters 6 and 7 show that there were interactive effects of As and Sb in soils for their binding sites, which determined the changes in their lability in co-amended soils under anaerobic and aerobic conditions. The order and amounts of As and Sb added into soils would also be important in controlling their interactive effects on soil binding sites and in turn determining their lability in soils. This warrants further studies to fully understand the competitive interactions of As and Sb in soils and manage contaminated soils. To do this, a study could increase the concentration of one element while keeping the other constant and vice versa, which would be more intensive than anything attempted in this study.

### 8.6.2. Uptake of As and Sb by plants

The results in this thesis show that As bioaccumulation in all test vegetables grown in various soils under different conditions was much higher than Sb bioaccumulation, but the translocation of Sb from roots to shoots was consistently higher than that of As. In addition, the presence of As in soils enhanced the bioaccumulation factor of Sb in As+Sb soils compared to Sb-only soils. The possibility was made that the more toxic element (As) damaged the cell membrane and allowed the absorption and transportation of the less toxic Sb in plants. The mechanism of Sb uptake by plants remains relatively unknown. Thus, the determination of As and especially Sb in subcellular fractions of plant tissues needs to be done to enable a better understanding of their mechanisms.

The analysis of As species in flooded water spinach (*I. aquatica*) shows that the dominant As species in flooded *I. aquatica* was As(V). This is opposite to the expected results that As(III) would be present in flooded *I. aquatica* at higher concentrations than

that in non-flooded *I. aquatica* because As(III) was the dominant species in flooded contaminated soils. The dominance of As(V) in tissues of flooded *I. aquatica* was an unusual trend. This also warrants further studies to fully understand the mechanism behind this trend. Future studies on the speciation of As and Sb in fresh tissues of flooded *I. aquatica* using X-ray absorption spectroscopy would be expected to fully explain their uptake and metabolism in flooded *I. aquatica*.

A reliable procedure for the extraction and analysis of Sb species in plants has not been successfully developed in the literature. *I. aquatica* accumulated Sb at high concentrations, but the plant looked healthy and had great biomass, which poses chronic risks to humans consuming the contaminated *I. aquatica*. The Sb toxicity depends on its species. Thus, the development of a reliable protocol for the extraction and analysis of Sb species in plants is essential for the interpretation of Sb uptake and metabolism in plants and the development of sustainable agriculture.

The results in chapter 7 (sections 7.3.1.2, 7.3.2.2, and 7.3.2.4) show that the presence of *I. aquatica* in flooded contaminated soils altered the speciation of As and Sb in the soils identified by multiple profiles of  $C_{DGT-As(III)}$ ,  $C_{DGT-As(V)}$ ,  $C_{DGT-Sb(III)}$ , and  $C_{DGT-Sb(V)}$ , which influences the uptake and speciation of As and Sb in flooded *I. aquatica*. Hence, the application of DGT-planar optode sandwich sensor coupled with LA-ICP-MS for chemical imaging of labile As and Sb and *in situ* measurement of pH and oxygen in the rhizophere of flooded *I. aquatica* is warranted. This is supported by Williams et al. (2014) who applied DGT-planar optode sandwich sensor to evaluate the localised mobilisation of As, Fe(II), and Pb at the boundary of the aerobic rhizophere region of flooded rice roots, accompanying with the decrease in pH. These studies would enable a better understanding of the behaviour of As and Sb in such environments and contribute to the explanation of As and Sb speciation in flooded *I. aquatica*.

### References

Abedin, M. J., Feldmann, J. & Meharg, A. A. 2002, 'Uptake kinetics of arsenic species in rice plants', *Plant Physiology*, vol. 128, no. 3, pp. 1120-1128.

Abernathy, C. O., Liu, Y. P., Longfellow, D., Aposhian, H. V., Beck, B., Fowler, B., Goyer, R., Menzer, R., Rossman, T., Thompson, C. & Waalkes, M. 1999, 'Meeting on Arsenic: Health Effects, Mechanisms of Actions, and Research Issues, Hunt Valley, Maryland, 22-24 September 1997', *Environmental Health Perspectives*, vol. 107, no. 7, pp. 593-597.

Adriano, D. C. 2001, *Trace Elements in Terrestrial Environments*, 2 edn, Springer-Verlag, New York.

Agbenin, J. O. & Welp, G. 2012, 'Bioavailability of copper, cadmium, zinc, and lead in tropical savanna soils assessed by diffusive gradient in thin films (DGT) and ion exchange resin membranes', *Environmental Monitoring and Assessment*, vol. 184, no. 4, pp. 2275-2284.

Ahmed, Z. U., Panaullah, G. M., Gauch Jr, H., McCouch, S. R., Tyagi, W., Kabir, M. S. & Duxbury, J. M. 2011, 'Genotype and environment effects on rice (Oryza sativa L.) grain arsenic concentration in Bangladesh', *Plant and Soil*, vol. 338, no. 1, pp. 367-382.

Alloway, B. J. 1995, Heavy metals in soils, Blackie Academic & Professional, London.

Álvarez-Ayuso, E., Abad-Valle, P., Murciego, A. & Villar-Alonso, P. 2016, 'Arsenic distribution in soils and rye plants of a cropland located in an abandoned mining area', *Science of the Total Environment*, vol. 542, pp. 238-246.

Álvarez-Ayuso, E., Otones, V., Murciego, A., García-Sánchez, A. & Regina, I. S. 2012, 'Antimony, arsenic and lead distribution in soils and plants of an agricultural area impacted by former mining activities', *Science of the Total Environment*, vol. 439, pp. 35-43. Álvarez-Ayuso, E., Otones, V., Murciego, A., García-Sánchez, A. & Regina, I. S. 2013, 'Mobility and phytoavailability of antimony in an area impacted by a former stibnite mine exploitation', *Science of the Total Environment*, vol. 449, pp. 260-268.

Amato, E. D., Simpson, S. L., Jarolimek, C. V. & Jolley, D. F. 2014, 'Diffusive gradients in thin films technique provide robust prediction of metal bioavailability and toxicity in estuarine sediments', *Environmental Science and Technology*, vol. 48, no. 8, pp. 4485-4494.

Anawar, H. M., Freitas, M. C., Canha, N. & Regina, I. S. 2011, 'Arsenic, antimony, and other trace element contamination in a mine tailings affected area and uptake by tolerant plant species', *Environmental Geochemistry and Health*, vol. 33, no. 4, pp. 353-362.

Anawar, H. M., Garcia-Sanchez, A. & Santa Regina, I. 2008, 'Evaluation of various chemical extraction methods to estimate plant-available arsenic in mine soils', *Chemosphere*, vol. 70, no. 8, pp. 1459-1467.

Antunes, P. M. C., Berkelaar, E. J., Boyle, D., Hale, B. A., Hendershot, W. & Voigt, A. 2006, 'The biotic ligand model for plants and metals: Technical challenges for field application', *Environmental Toxicology and Chemistry*, vol. 25, no. 3, pp. 875-882.

Arai, Y. 2010, 'Arsenic and Antimony', in *Trace Elements in Soils*, John Wiley & Sons, Ltd, pp. 381-407.

Ashley, P. M., Graham, B. P., Tighe, M. K. & Wolfenden, B. J. 2007, 'Antimony and arsenic dispersion in the Macleay River catchment, New South Wales: A study of the environmental geochemical consequences', *Australian Journal of Earth Sciences*, vol. 54, no. 1, pp. 83-103.

Ashley, P. M. & Lottermoser, B. G. 1999, 'Arsenic contamination at the Mole River mine, northern New South Wales', *Australian Journal of Earth Sciences*, vol. 46, no. 6, pp. 861-874.

Bacon, J. R. & Davidson, C. M. 2008, 'Is there a future for sequential chemical extraction?', *Analyst*, vol. 133, no. 1, pp. 25-46.

Bagherifam, S., Lakzian, A., Fotovat, A., Khorasani, R. & Komarneni, S. 2014, 'In situ stabilization of As and Sb with naturally occurring Mn, Al and Fe oxides in a calcareous soil: Bioaccessibility, bioavailability and speciation studies', *Journal of Hazardous Materials*, vol. 273, pp. 247-252.

Baig, J. A., Kazi, T. G., Shah, A. Q., Kandhro, G. A., Afridi, H. I., Khan, S. & Kolachi,
N. F. 2010, 'Biosorption studies on powder of stem of Acacia nilotica: Removal of arsenic from surface water', *Journal of Hazardous Materials*, vol. 178, no. 1-3, pp. 941-948.

Baker, A. J. M. 1981, 'Accumulators and Excluders - Strategies in the Response of Plants to Heavy Metals', *Journal of Plant Nutrition*, vol. 3, no. 1-4, pp. 643-654.

Barber, S. A. 1995, *Soil nutrient bioavailability: a mechanistic approach*, John Wiley and Sons, New York.

Baroni, F., Boscagli, A., Di Lella, L. A., Protano, G. & Riccobono, F. 2004, 'Arsenic in soil and vegetation of contaminated areas in southern Tuscany (Italy)', *Journal of Geochemical Exploration*, vol. 81, no. 1-3, pp. 1-14.

Baroni, F., Boscagli, A., Protano, G. & Riccobono, F. 2000, 'Antimony accumulation in Achillea ageratum, Plantago lanceolata and Silene vulgaris growing in an old Sb-mining area', *Environmental Pollution*, vol. 109, no. 2, pp. 347-352.

Bedabati Chanu, L. & Gupta, A. 2016, 'Phytoremediation of lead using Ipomoea aquatica Forsk. in hydroponic solution', *Chemosphere*, vol. 156, pp. 407-411.

Bennett, W. W., Arsic, M., Panther, J. G., Welsh, D. T. & Teasdale, P. R. 2016a, 'Binding Layer Properties', in W. Davison (ed.), *Diffusive Gradients in Thin-Films for Environmental Measurements*, Cambridge University Press, Cambridge, pp. 32-65.

Bennett, W. W., Arsic, M., Welsh, D. T. & Teasdale, P. R. 2016b, 'In situ speciation of dissolved inorganic antimony in surface waters and sediment porewaters: Development of a thiol-based diffusive gradients in thin films technique for SbIII', *Environmental Science: Processes and Impacts*, vol. 18, no. 8, pp. 992-998.

Bennett, W. W., Teasdale, P. R., Panther, J. G., Welsh, D. T. & Jolley, D. F. 2010, 'New diffusive gradients in a thin film technique for measuring inorganic arsenic and selenium(IV) using a titanium dioxide based adsorbent', *Analytical Chemistry*, vol. 82, no. 17, pp. 7401-7407.

Bennett, W. W., Teasdale, P. R., Panther, J. G., Welsh, D. T. & Jolley, D. F. 2011, 'Speciation of dissolved inorganic arsenic by diffusive gradients in thin films: Selective binding of As III by 3-mercaptopropyl-functionalized silica gel', *Analytical Chemistry*, vol. 83, no. 21, pp. 8293-8299.

Bennett, W. W., Teasdale, P. R., Panther, J. G., Welsh, D. T., Zhao, H. & Jolley, D. F. 2012a, 'Investigating arsenic speciation and mobilization in sediments with DGT and DET: A mesocosm evaluation of oxic-anoxic transitions', *Environmental Science and Technology*, vol. 46, no. 7, pp. 3981-3989.

Bennett, W. W., Teasdale, P. R., Welsh, D. T., Panther, J. G., Stewart, R. R., Price, H. L. & Jolley, D. F. 2012b, 'Inorganic arsenic and iron(II) distributions in sediment porewaters investigated by a combined DGTcolourimetric DET technique', *Environmental Chemistry*, vol. 9, no. 1, pp. 31-40.

Bergqvist, C. & Greger, M. 2012, 'Arsenic accumulation and speciation in plants from different habitats', *Applied Geochemistry*, vol. 27, no. 3, pp. 615-622.

Bhattacharjee, H., Mukhopadhyay, R., Thiyagarajan, S. & Rosen, B. P. 2008, 'Aquaglyceroporins: Ancient channels for metalloids', *Journal of Biology*, vol. 7, no. 9.

Bhattacharya, P., Samal, A. C., Majumdar, J. & Santra, S. C. 2010a, 'Accumulation of arsenic and its distribution in rice plant (Oryza sativa L.) in Gangetic West Bengal, India', *Paddy and Water Environment*, vol. 8, no. 1, pp. 63-70.

Bhattacharya, P., Samal, A. C., Majumdar, J. & Santra, S. C. 2010b, 'Arsenic contamination in rice, wheat, pulses, and vegetables: A study in an arsenic affected area of West Bengal, India', *Water, Air, and Soil Pollution*, vol. 213, no. 1-4, pp. 3-13.

Bhattacharya, P., Samal, A. C., Majumdar, J. & Santra, S. C. 2010c, 'Uptake of arsenic in rice plant varieties cultivated with arsenic rich groundwater', *EnvironmentAsia*, vol. 3, no. 2, pp. 34-37.

Bhattacharya, P., Welch, A. H., Stollenwerk, K. G., McLaughlin, M. J., Bundschuh, J. & Panaullah, G. 2007, 'Arsenic in the environment: Biology and Chemistry', *Science of the Total Environment*, vol. 379, no. 2-3, pp. 109-120.

Bhatti, S. M., Anderson, C. W. N., Stewart, R. B. & Robinson, B. H. 2013, 'Risk assessment of vegetables irrigated with arsenic-contaminated water', *Environmental Sciences: Processes and Impacts*, vol. 15, no. 10, pp. 1866-1875.

Bissen, M. & Frimmel, F. H. 2003, 'Arsenic — a Review. Part I: Occurrence, Toxicity, Speciation, Mobility', *Acta hydrochimica et hydrobiologica*, vol. 31, no. 1, pp. 9-18.

Bogdan, K. & Schenk, M. K. 2009, 'Evaluation of soil characteristics potentially affecting arsenic concentration in paddy rice (Oryza sativa L.)', *Environmental Pollution*, vol. 157, no. 10, pp. 2617-2621.

Brandstetter, A., Lombi, E., Wenzel, W. W. & Adriano, D. C. 2000, 'Arseniccontaminated soils: I. Risk assessment', *Remediation Engineering of Contaminated Soils*, pp. 715-737.

Burló, F., Guijarro, I., Carbonell-Barrachina, A. A., Valero, D. & Martínez-Sánchez, F. 1999, 'Arsenic species: Effects on and accumulation by tomato plants', *Journal of Agricultural and Food Chemistry*, vol. 47, no. 3, pp. 1247-1253.

Burton, E. D., Johnston, S. G. & Kocar, B. D. 2014, 'Arsenic mobility during flooding of contaminated soil: The effect of microbial sulfate reduction', *Environmental Science and Technology*, vol. 48, no. 23, pp. 13660-13667.

Buschmann, J., Kappeler, A., Lindauer, U., Kistler, D., Berg, M. & Sigg, L. 2006, 'Arsenite and arsenate binding to dissolved humic acids: Influence of pH, type of humic acid, and aluminum', *Environmental Science and Technology*, vol. 40, no. 19, pp. 6015-6020.

Buschmann, J. & Sigg, L. 2004, 'Antimony(III) binding to humic substances: Influence of pH and type of humic acid', *Environmental Science and Technology*, vol. 38, no. 17, pp. 4535-4541.

Buurman, P., Lagen, B. v. & Velthorst, E. J. (eds) 1996, *Manual for soil and water analysis*, Backhuys, Leiden, The Netherlands.

Cai, F., Ren, J., Tao, S. & Wang, X. 2016, 'Uptake, translocation and transformation of antimony in rice (Oryza sativa L.) seedlings', *Environmental Pollution*, vol. 209, pp. 169-176.

Campbell, P. G. C. 1995, 'Interaction between trace metals and aquatic organisms: A critique of the free-ion activity model', in A. Tessier & D. R. Turner (eds), *Metal speciation and bioavailability in aquatic systems*, John Wiley & Son, New York, NY, pp. 45-97.

Cao, X. & Ma, L. Q. 2004, 'Effects of compost and phosphate on plant arsenic accumulation from soils near pressure-treated wood', *Environmental Pollution*, vol. 132, no. 3, pp. 435-442.

Cao, X., Ma, L. Q. & Shiralipour, A. 2003, 'Effects of compost and phosphate amendments on arsenic mobility in soils and arsenic uptake by the hyperaccumulator, Pteris vittata L', *Environmental Pollution*, vol. 126, no. 2, pp. 157-167.

Carbonell-Barrachina, A. A., Burló, F., Burgos-Hernández, A., López, E. & Mataix, J. 1997, 'The influence of arsenite concentration on arsenic accumulation in tomato and bean plants', *Scientia Horticulturae*, vol. 71, no. 3-4, pp. 167-176.

Carbonell-Barrachina, A. A., Burló, F., López, E. & Martínez-Sánchez, F. 1999a, 'Arsenic toxicity and accumulation in radish as affected by arsenic chemical speciation', *Journal of Environmental Science and Health - Part B Pesticides, Food Contaminants, and Agricultural Wastes*, vol. 34, no. 4, pp. 661-679.

Carbonell-Barrachina, A. A., Burló, F., Valero, D., López, E., Martínez-Romero, D. & Martínez-Sánchez, F. 1999b, 'Arsenic toxicity and accumulation in turnip as affected by arsenic chemical speciation', *Journal of Agricultural and Food Chemistry*, vol. 47, no. 6, pp. 2288-2294.

Casado, M., Anawar, H. M., Garcia-Sanchez, A. & Regina, I. S. 2007, 'Antimony and arsenic uptake by plants in an abandoned mining area', *Communications in Soil Science and Plant Analysis*, vol. 38, no. 9-10, pp. 1255-1275.

Casiot, C., Ujevic, M., Munoz, M., Seidel, J. L. & Elbaz-Poulichet, F. 2007, 'Antimony and arsenic mobility in a creek draining an antimony mine abandoned 85 years ago (upper Orb basin, France)', *Applied Geochemistry*, vol. 22, no. 4, pp. 788-798.

Cattani, I., Capri, E., Boccelli, R. & Del Re, A. A. M. 2009, 'Assessment of arsenic availability to roots in contaminated Tuscany soils by a diffusion gradient in thin films (DGT) method and uptake by Pteris vittata and Agrostis capillaris', *European Journal of Soil Science*, vol. 60, no. 4, pp. 539-548.

Ceriotti, G. & Amarasiriwardena, D. 2009, 'A study of antimony complexed to soilderived humic acids and inorganic antimony species along a Massachusetts highway', *Microchemical Journal*, vol. 91, no. 1, pp. 85-93.

Chaignon, V., Sanchez-Neira, I., Herrmann, P., Jaillard, B. & Hinsinger, P. 2003, 'Copper bioavailability and extractability as related to chemical properties of contaminated soils from a vine-growing area', *Environmental Pollution*, vol. 123, no. 2, pp. 229-238.

Chaturvedi, I. 2006, 'Effects of arsenic concentrations on growth and arsenic uptake and accumulation by rice (Oryza sativa) genotypes', *Electronic Journal of Environmental, Agricultural and Food Chemistry*, vol. 5, no. 5, pp. 1546-1552.

Chen, W., Li, L., Chang, A. C., Wu, L., Chaney, R. L., Smith, R. & Ajwa, H. 2009, 'Characterizing the solid-solution partitioning coefficient and plant uptake factor of As, Cd, and Pb in California croplands', *Agriculture, Ecosystems and Environment*, vol. 129, no. 1-3, pp. 212-220.

Chowdhury, U. K., Biswas, B. K., Chowdhury, T. R., Samanta, G., Mandal, B. K., Basu, G. C., Chanda, C. R., Lodh, D., Saha, K. C., Mukherjee, S. K., Roy, S., Kabir, S., Quamruzzaman, Q. & Chakraborti, D. 2000, 'Groundwater arsenic contamination in Bangladesh and West Bengal, India', *Environmental Health Perspectives*, vol. 108, no. 5, pp. 393-397.

Clarisse, O. & Hintelmann, H. 2006, 'Measurements of dissolved methylmercury in natural waters using diffusive gradients in thin film (DGT)', *Journal of Environmental Monitoring*, vol. 8, no. 12, pp. 1242-1247.

Clemente, R., Dickinson, N. M. & Lepp, N. W. 2008, 'Mobility of metals and metalloids in a multi-element contaminated soil 20 years after cessation of the pollution source activity', *Environmental Pollution*, vol. 155, no. 2, pp. 254-261.

Cobb, G. P., Sands, K., Waters, M., Wixson, B. G. & Dorward-King, E. 2000, 'Accumulation of heavy metals by vegetables grown in mine wastes', *Environmental Toxicology and Chemistry*, vol. 19, no. 3, pp. 600-607.

Codling, E. E., Chaney, R. L. & Green, C. E. 2015, 'Accumulation of Lead and Arsenic by Carrots Grown on Lead-Arsenate Contaminated Orchard Soils', *Journal of Plant Nutrition*, vol. 38, no. 4, pp. 509-525.

Codling, E. E., Chaney, R. L. & Green, C. E. 2016, 'Accumulation of Lead and Arsenic by Potato Grown on Lead–Arsenate-Contaminated Orchard Soils', *Communications in Soil Science and Plant Analysis*, vol. 47, no. 6, pp. 799-807.

Conesa, H. M., Wieser, M., Studer, B. & Schulin, R. 2011, 'Effects of vegetation and fertilizer on metal and Sb plant uptake in a calcareous shooting range soil', *Ecological Engineering*, vol. 37, no. 4, pp. 654-658.

Corrales, I., Barceló, J., Bech, J. & Poschenrieder, C. 2014, 'Antimony accumulation and toxicity tolerance mechanisms in Trifolium species', *Journal of Geochemical Exploration*, vol. 147, no. PB, pp. 167-172.

Creger, T. L. & Peryea, F. J. 1994, 'Phosphate fertilizer enhances arsenic uptake by apricot liners grown in lead-arsenate-enriched soil', *HortScience*, vol. 29, no. 2, pp. 88-92.

Cui, X. D., Wang, Y. J., Hockmann, K. & Zhou, D. M. 2015, 'Effect of iron plaque on antimony uptake by rice (Oryza sativa L.)', *Environmental Pollution*, vol. 204, pp. 133-140.

Cusnir, R., Steinmann, P., Bochud, F. & Froidevaux, P. 2014, 'A DGT Technique for Plutonium Bioavailability Measurements', *Environmental science & technology*, vol. 48, no. 18, pp. 10829-10834.

Dahlqvist, R., Zhang, H., Ingri, J. & Davison, W. 2002, 'Performance of the diffusive gradients in thin films technique for measuring Ca and Mg in freshwater', *Analytica Chimica Acta*, vol. 460, no. 2, pp. 247-256.

Das, D. K., Sur, P. & Das, K. 2008, 'Mobilisation of arsenic in soils and in rice (Oryza sativa L.) plants affected by organic matter and zinc application in irrigation water contaminated with arsenic', *Plant, Soil and Environment*, vol. 54, no. 1, pp. 30-37.

Dasgupta, T., Hossain, S. A., Meharg, A. A. & Price, A. H. 2004, 'An arsenate tolerance gene on chromosome 6 of rice', *New Phytologist*, vol. 163, no. 1, pp. 45-49.

Davison, W., Fones, G., Harper, M., Teasdale, P. & Zhang, H. 2000, 'Dialysis, DET and DGT: In situ diffusional techniques for studying water, sediments and soils', *In Situ Monitoring of Aquatic Systems: Chemical Analysis and Speciation*, pp. 495-569.

Davison, W. & Zhang, H. 1994, 'In situ speciation measurements of trace components in natural waters using thin-film gels', *Nature*, vol. 367, no. 6463, pp. 546-548.

Davison, W. & Zhang, H. 2012, 'Progress in understanding the use of diffusive gradients in thin films (DGT) back to basics', *Environmental Chemistry*, vol. 9, no. 1, pp. 1-13.

De Oliveira, L. K., Melo, C. A., Goveia, D., Lobo, F. A., Armienta Hernández, M. A., Fraceto, L. F. & Rosa, A. H. 2015, 'Adsorption/desorption of arsenic by tropical peat: Influence of organic matter, iron and aluminium', *Environmental Technology (United Kingdom)*, vol. 36, no. 2, pp. 149-159.

Degryse, F. & Smolders, E. 2016, 'DGT and Bioavailability', in W. Davison (ed.), *Diffusive Gradients in Thin-Films for Environmental Measurements*, Cambridge University Press, Cambridge, pp. 174-215.

Degryse, F., Smolders, E., Zhang, H. & Davison, W. 2009, 'Predicting availability of mineral elements to plants with the DGT technique: a review of experimental data and interpretation by modelling', *Environmental Chemistry*, vol. 6, no. 3, pp. 198-218.

Denys, S., Tack, K., Caboche, J. & Delalain, P. 2009, 'Bioaccessibility, solid phase distribution, and speciation of Sb in soils and in digestive fluids', *Chemosphere*, vol. 74, no. 5, pp. 711-716.

Deschamps, E., Ciminelli, V. S. T., Weidler, P. G. & Ramos, A. Y. 2003, 'Arsenic sorption onto soils enriched in Mn and Fe minerals', *Clays and Clay Minerals*, vol. 51, no. 2, pp. 197-204.

Ding, S., Xu, D., Sun, Q., Yin, H. & Zhang, C. 2010, 'Measurement of Dissolved Reactive Phosphorus Using the Diffusive Gradients in Thin Films Technique with a High-Capacity Binding Phase', *Environmental science & technology*, vol. 44, no. 21, pp. 8169-8174.

Dixit, S. & Hering, J. G. 2003, 'Comparison of arsenic(V) and arsenic(III) sorption onto iron oxide minerals: Implications for arsenic mobility', *Environmental Science and Technology*, vol. 37, no. 18, pp. 4182-4189.

Dobran, S. & Zagury, G. J. 2006, 'Arsenic speciation and mobilization in CCAcontaminated soils: Influence of organic matter content', *Science of the Total Environment*, vol. 364, no. 1-3, pp. 239-250.

Dočekalová, H., Škarpa, P. & Dočekal, B. 2015, 'Diffusive gradient in thin films technique for assessment of cadmium and copper bioaccessibility to radish (Raphanus sativus)', *Talanta*, vol. 134, pp. 153-157.

Doherty, S. J., Tighe, M. K. & Wilson, S. C. 2017, 'Evaluation of amendments to reduce arsenic and antimony leaching from co-contaminated soils', *Chemosphere*, vol. 174, pp. 208-217.

Douay, F., Pelfrêne, A., Planque, J., Fourrier, H., Richard, A., Roussel, H. & Girondelot, B. 2013, 'Assessment of potential health risk for inhabitants living near a former lead smelter. Part 1: Metal concentrations in soils, agricultural crops, and homegrown vegetables', *Environmental Monitoring and Assessment*, vol. 185, no. 5, pp. 3665-3680.

Douay, F., Pruvot, C., Roussel, H., Ciesielski, H., Fourrier, H., Proix, N. & Waterlot, C. 2008, 'Contamination of urban soils in an area of Northern France polluted by dust

emissions of two smelters', Water, Air, and Soil Pollution, vol. 188, no. 1-4, pp. 247-260.

Dousova, B., Buzek, F., Herzogova, L., Machovic, V. & Lhotka, M. 2015, 'Effect of organic matter on arsenic(V) and antimony(V) adsorption in soils', *European Journal of Soil Science*, vol. 66, no. 1, pp. 74-82.

Drahota, P., Grösslová, Z. & Kindlová, H. 2014, 'Selectivity assessment of an arsenic sequential extraction procedure for evaluating mobility in mine wastes', *Analytica Chimica Acta*, vol. 839, pp. 34-43.

Du Laing, G., Rinklebe, J., Vandecasteele, B., Meers, E. & Tack, F. M. G. 2009, 'Trace metal behaviour in estuarine and riverine floodplain soils and sediments: A review', *Science of the Total Environment*, vol. 407, no. 13, pp. 3972-3985.

Dwivedi, S., Tripathi, R. D., Srivastava, S., Singh, R., Kumar, A., Tripathi, P., Dave, R., Rai, U. N., Chakrabarty, D., Trivedi, P. K., Tuli, R., Adhikari, B. & Bag, M. K. 2010, 'Arsenic affects mineral nutrients in grains of various Indian rice (Oryza sativa L.) genotypes grown on arsenic-contaminated soils of West Bengal', *Protoplasma*, vol. 245, no. 1, pp. 113-124.

Environmental Health Australia - enHealth 2012, *Environmental health risk assessment: Guidelines for assessing human health risks from environmental hazards*, Canberra.

Ernstberger, H., Davison, W., Zhang, H., Andrew, T. Y. E. & Young, S. 2002, 'Measurement and dynamic modeling of trace metal mobilization in soils using DGT and DIFS', *Environmental Science and Technology*, vol. 36, no. 3, pp. 349-354.

Ernstberger, H., Zhang, H., Tye, A., Young, S. & Davison, W. 2005, 'Desorption kinetics of Cd, Zn, and Ni measured in soils by DGT', *Environmental Science and Technology*, vol. 39, no. 6, pp. 1591-1597.

Esteban, E., Carpena, R. O. & Meharg, A. A. 2003, 'High-affinity phosphate/arsenate transport in white lupin (Lupinus albus) is relatively insensitive to phosphate status', *New Phytologist*, vol. 158, no. 1, pp. 165-173.

Ettler, V., Mihaljevič, M., Šebek, O. & Nechutný, Z. 2007, 'Antimony availability in highly polluted soils and sediments - A comparison of single extractions', *Chemosphere*, vol. 68, no. 3, pp. 455-463.

Ettler, V., Tejnecký, V., Mihaljevič, M., Šebek, O., Zuna, M. & Vaněk, A. 2010, 'Antimony mobility in lead smelter-polluted soils', *Geoderma*, vol. 155, no. 3-4, pp. 409-418.

Farooq, M. A., Islam, F., Ali, B., Najeeb, U., Mao, B., Gill, R. A., Yan, G., Siddique, K.
H. M. & Zhou, W. 2016, 'Arsenic toxicity in plants: Cellular and molecular mechanisms of its transport and metabolism', *Environmental and Experimental Botany*, vol. 132, pp. 42-52.

Fendorf, S., Eick, M. J., Grossl, P. & Sparks, D. L. 1997, 'Arsenate and chromate retention mechanisms on goethite. 1. Surface structure', *Environmental Science and Technology*, vol. 31, no. 2, pp. 315-320.

Fendorf, S., Herbel, M. J., Tufano, K. J. & Kocar, B. D. 2007, 'Biogeochemical Processes Controlling the Cycling of Arsenic in Soils and Sediments', in *Biophysico-Chemical Processes of Heavy Metals and Metalloids in Soil Environments*, pp. 313-338.

Fendorf, S. & Kocar, B. D. 2009, 'Chapter 3 Biogeochemical Processes Controlling the Fate and Transport of Arsenic. Implications for South and Southeast Asia', *Advances in agronomy*, vol. 104, pp. 137-164.

Feng, R., Wei, C., Tu, S., Ding, Y., Wang, R. & Guo, J. 2013, 'The uptake and detoxification of antimony by plants: A review', *Environmental and Experimental Botany*, vol. 96, pp. 28-34.

Feng, R., Wei, C., Tu, S., Tang, S. & Wu, F. 2011, 'Simultaneous hyperaccumulation of arsenic and antimony in Cretan brake fern: Evidence of plant uptake and subcellular distributions', *Microchemical Journal*, vol. 97, no. 1, pp. 38-43.

Fergusson, J. E. 1990, *The heavy elements: chemistry, environmental impact and health effects*, Pergamon Press, Oxford, England.

Filella, M., Belzile, N. & Chen, Y. W. 2002a, 'Antimony in the environment: A review focused on natural waters I. Occurence', *Earth-Science Reviews*, vol. 57, no. 1-2, pp. 125-176.

Filella, M., Belzile, N. & Chen, Y. W. 2002b, 'Antimony in the environment: A review focused on natural waters II. Relevant solution chemistry', *Earth-Science Reviews*, vol. 59, no. 1-4, pp. 265-285.

Filella, M., Belzile, N. & Lett, M. C. 2007, 'Antimony in the environment: A review focused on natural waters. III. Microbiota relevant interactions', *Earth-Science Reviews*, vol. 80, no. 3-4, pp. 195-217.

Filella, M., Williams, P. A. & Belzile, N. 2009, 'Antimony in the environment: Knowns and unknowns', *Environmental Chemistry*, vol. 6, no. 2, pp. 95-105.

Fitz, W. J. & Wenzel, W. W. 2002, 'Arsenic transformations in the soil-rhizosphereplant system: Fundamentals and potential application to phytoremediation', *Journal of Biotechnology*, vol. 99, no. 3, pp. 259-278.

Fitz, W. J., Wenzel, W. W., Zhang, H., Nurmi, J., Štipek, K., Fischerova, Z., Schweiger, P., Köllensperger, G., Ma, L. Q. & Stingeder, G. 2003, 'Rhizosphere Characteristics of the Arsenic Hyperaccumulator Pteris vittata L. and Monitoring of Phytoremoval Efficiency', *Environmental Science and Technology*, vol. 37, no. 21, pp. 5008-5014.

Flynn, H. C., Meharg, A. A., Bowyer, P. K. & Paton, G. I. 2003, 'Antimony bioavailability in mine soils', *Environmental Pollution*, vol. 124, no. 1, pp. 93-100.

Foster, S., Maher, W., Krikowa, F. & Apte, S. 2007, 'A microwave-assisted sequential extraction of water and dilute acid soluble arsenic species from marine plant and animal tissues', *Talanta*, vol. 71, no. 2, pp. 537-549.

Francesconi, K., Visoottiviseth, P., Sridokchan, W. & Goessler, W. 2002, 'Arsenic species in an arsenic hyperaccumulating fern, Pityrogramma calomelanos: a potential phytoremediator of arsenic-contaminated soils', *Science of the Total Environment*, vol. 284, no. 1, pp. 27-35.

Frost, R. R. & Griffin, R. A. 1977, 'Effect of pH on adsorption of arsenic and selenium from landfill leachate by clay minerals', *Soil Science Society of America Journal*, vol. 41, no. 1, pp. 53-57.

FSANZ (Food Standard Agency of Australia and New Zealand) 2013, Australia New Zealand Food Standards Code - Standard 1.4.1 - Contaminants and Natural Toxicants F2013C00140, Canberra.

Fu, Y., Chen, M., Bi, X., He, Y., Ren, L., Xiang, W., Qiao, S., Yan, S., Li, Z. & Ma, Z. 2011, 'Occurrence of arsenic in brown rice and its relationship to soil properties from Hainan Island, China', *Environmental Pollution*, vol. 159, no. 7, pp. 1757-1762.

Fu, Z. 2016, 'Influence of reducing conditions on the release of antimony and arsenic from a tailings sediment', *Journal of Soils and Sediments*, vol. 16, no. 10, pp. 2471-2481.

Fu, Z., Wu, F., Mo, C., Deng, Q., Meng, W. & Giesy, J. P. 2016, 'Comparison of arsenic and antimony biogeochemical behavior in water, soil and tailings from Xikuangshan, China', *Science of the Total Environment*, vol. 539, pp. 97-104.

Gál, J., Hursthouse, A. & Cuthbert, S. 2007, 'Bioavailability of arsenic and antimony in soils from an abandoned mining area, Glendinning (SW Scotland)', *Journal of Environmental Science and Health - Part A Toxic/Hazardous Substances and Environmental Engineering*, vol. 42, no. 9, pp. 1263-1274.

Gál, J., Hursthouse, A. S. & Cuthbert, S. J. 2006, 'Chemical availability of arsenic and antimony in industrial soils', *Environmental Chemistry Letters*, vol. 3, no. 4, pp. 149-153.

Gao, L., Gao, B., Xu, D., Peng, W., Lu, J. & Gao, J. 2017, 'Assessing remobilization characteristics of arsenic (As) in tributary sediment cores in the largest reservoir, China', *Ecotoxicology and Environmental Safety*, vol. 140, pp. 48-54.

Gao, L., Gao, B., Zhou, H., Xu, D., Wang, Q. & Yin, S. 2016, 'Assessing the remobilization of Antimony in sediments by DGT: A case study in a tributary of the Three Gorges Reservoir', *Environmental Pollution*, vol. 214, pp. 600-607.

Gao, Y., De Craemer, S. & Baeyens, W. 2014, 'A novel method for the determination of dissolved methylmercury concentrations using diffusive gradients in thin films technique', *Talanta*, vol. 120, pp. 470-474.

Gao, Y., Leermakers, M., Elskens, M., Billon, G., Ouddane, B., Fischer, J. C. & Baeyens, W. 2007, 'High resolution profiles of thallium, manganese and iron assessed by DET and DGT techniques in riverine sediment pore waters', *Science of the Total Environment*, vol. 373, no. 2-3, pp. 526-533.

Gao, Y., Lesven, L., Gillan, D., Sabbe, K., Billon, G., De Galan, S., Elskens, M., Baeyens, W. & Leermakers, M. 2009, 'Geochemical behavior of trace elements in sub-tidal marine sediments of the Belgian coast', *Marine Chemistry*, vol. 117, no. 1-4, pp. 88-96.

García-Sanchez, A., Alvarez-Ayuso, E. & Rodriguez-Martin, F. 2002, 'Sorption of As(V) by some oxyhydroxides and clay minerals. Application to its immobilization in two polluted mining soils', *Clay minerals*, vol. 37, no. 1, pp. 187-194.

Garmo, Ø. A., Røyset, O., Steinnes, E. & Flaten, T. P. 2003, 'Performance study of diffusive gradients in thin films for 55 elements', *Analytical Chemistry*, vol. 75, no. 14, pp. 3573-3580.

Garnier, J. M., Garnier, J., Jézéquel, D. & Angeletti, B. 2015, 'Using DET and DGT probes (ferrihydrite and titanium dioxide) to investigate arsenic concentrations in soil porewater of an arsenic-contaminated paddy field in Bangladesh', *Science of the Total Environment*, vol. 536, pp. 306-315.

Garnier, J. M., Travassac, F., Lenoble, V., Rose, J., Zheng, Y., Hossain, M. S., Chowdhury, S. H., Biswas, A. K., Ahmed, K. M., Cheng, Z. & van Geen, A. 2010, 'Temporal variations in arsenic uptake by rice plants in Bangladesh: The role of iron plaque in paddy fields irrigated with groundwater', *Science of the Total Environment*, vol. 408, no. 19, pp. 4185-4193.

Gebel, T. 1997, 'Aresnic and antimony: Comparative approach on mechanistic toxicology', *Chemico-Biological Interactions*, vol. 107, no. 3, pp. 131-144.

Gebel, T. 1999, 'Environmental medicine and toxicology of the metalloid antimony', *Umweltmedizin in Forschung und Praxis*, vol. 4, no. 5, pp. 259-267.

Geng, C. N., Zhu, Y. G., Liu, W. J. & Smith, S. E. 2005, 'Arsenate uptake and translocation in seedlings of two genotypes of rice is affected by external phosphate concentrations', *Aquatic Botany*, vol. 83, no. 4, pp. 321-331.

Génin, J.-M. R., Refait, P., Bourrié, G., Abdelmoula, M. & Trolard, F. 2001, 'Structure and stability of the Fe(II)–Fe(III) green rust "fougerite" mineral and its potential for reducing pollutants in soil solutions', *Applied Geochemistry*, vol. 16, no. 5, pp. 559-570.

Gobran, G. R., Wenzel, W. W. & Lombi, E. 2001, *Trace elements in the rhizosphere*, CRC Press, Boca Raton, Fla.

Goldberg, S. & Glaubig, R. A. 1988, 'Anion sorption on a calcareous, montmorillonitic soil - arsenic', *Soil Science Society of America Journal*, vol. 52, no. 5, pp. 1297-1300.

Gonzaga, M. I. S., Santos, J. A. G. & Ma, L. Q. 2008, 'Phytoextraction by arsenic hyperaccumulator Pteris vittata L. from six arsenic-contaminated soils: Repeated harvests and arsenic redistribution', *Environmental Pollution*, vol. 154, no. 2, pp. 212-218.

Gorny, J., Billon, G., Lesven, L., Dumoulin, D., Madé, B. & Noiriel, C. 2015, 'Arsenic behavior in river sediments under redox gradient: A review', *Science of the Total Environment*, vol. 505, pp. 423-434.

Göthberg, A., Greger, M. & Bengtsson, B. E. 2002, 'Accumulation of heavy metals in water spinach (Ipomoea aquatica) cultivated in the Bangkok region, Thailand', *Environmental Toxicology and Chemistry*, vol. 21, no. 9, pp. 1934-1939.

Grafe, M., Eick, M. J. & Grossl, P. R. 2001, 'Adsorption of arsenate (V) and arsenite (III) on goethite in the presence and absence of dissolved organic carbon', *Soil Science Society of America Journal*, vol. 65, no. 6, pp. 1680-1687.

Gregusova, M. & Docekal, B. 2013, 'High resolution characterization of uranium in sediments by DGT and DET techniques ACA-S-12-2197', *Analytica Chimica Acta*, vol. 763, pp. 50-56.

Guan, D. X., Williams, P. N., Luo, J., Zheng, J. L., Xu, H. C., Cai, C. & Ma, L. Q. 2015, 'Novel precipitated zirconia-based DGT technique for high-resolution imaging of oxyanions in waters and sediments', *Environmental Science and Technology*, vol. 49, no. 6, pp. 3653-3661.

Gutierrez, J., Hong, C. O., Lee, B. H. & Kim, P. J. 2010, 'Effect of steel-making slag as a soil amendment on arsenic uptake by radish (Raphanus sativa L.) in an upland soil', *Biology and Fertility of Soils*, vol. 46, no. 6, pp. 617-623.

Hajiani, N. J., Ghaderian, S. M., Karimi, N. & Schat, H. 2017, 'A comparison of antimony accumulation and tolerance among Achillea wilhelmsii, Silene vulgaris and Thlaspi arvense', *Plant and Soil*, vol. 412, no. 1-2, pp. 267-281.

Hammel, W., Debus, R. & Steubing, L. 2000, 'Mobility of antimony in soil and its availability to plants', *Chemosphere*, vol. 41, no. 11, pp. 1791-1798.

Hamon, R. E., Lombi, E., Fortunati, P., Nolan, A. L. & McLaughlin, M. J. 2004, 'Coupling Speciation and Isotope Dilution Techniques to Study Arsenic Mobilization in the Environment', *Environmental Science and Technology*, vol. 38, no. 6, pp. 1794-1798.

Hansen, H. R. & Pergantis, S. A. 2008, 'Analytical techniques and methods used for antimony speciation analysis in biological matrices', *Journal of Analytical Atomic Spectrometry*, vol. 23, no. 10, pp. 1328-1340.

Harper, M. P., Davison, W. & Tych, W. 1999, 'Estimation of pore water concentrations from DGT profiles: A modelling approach', *Aquatic Geochemistry*, vol. 5, no. 4, pp. 337-355.

Harper, M. P., Davison, W. & Tych, W. 2000, 'DIFS—a modelling and simulation tool for DGT induced trace metal remobilisation in sediments and soils', *Environmental Modelling & Software*, vol. 15, no. 1, pp. 55-66.

Harper, M. P., Davison, W., Zhang, H. & Tych, W. 1998, 'Kinetics of metal exchange between solids and solutions in sediments and soils interpreted from DGT measured fluxes', *Geochimica et Cosmochimica Acta*, vol. 62, no. 16, pp. 2757-2770.

Hartley, W., Dickinson, N. M., Clemente, R., French, C., Piearce, T. G., Sparke, S. & Lepp, N. W. 2009, 'Arsenic stability and mobilization in soil at an amenity grassland overlying chemical waste (St. Helens, UK)', *Environmental Pollution*, vol. 157, no. 3, pp. 847-856.

Hartley, W., Wu, C., Dickinson, N. M., Riby, P., Lepp, N. W. & Wong, M. H. 2010, 'Arsenic mobility and bioavailability in flooded industrially polluted UK soils', *Land Contamination and Reclamation*, vol. 18, no. 3, pp. 267-278.

Hartmann, H. T., Flocker, W. J. & Kofranek, A. M. 1981, *Plant science: growth, development and utilization of cultivated plants*, Prentice-Hall, Inc, Englewood Cliffs, New Jersey.

Harvey, C. F., Swartz, C. H., Badruzzaman, A. B. M., Keon-Blute, N., Yu, W., Ali, M.
A., Jay, J., Beckie, R., Niedan, V., Brabander, D., Oates, P. M., Ashfaque, K. N., Islam,
S., Hemond, H. F. & Ahmed, M. F. 2002, 'Arsenic mobility and groundwater extraction in Bangladesh', *Science*, vol. 298, no. 5598, pp. 1602-1606.

He, M. 2007, 'Distribution and phytoavailability of antimony at an antimony mining and smelting area, Hunan, China', *Environmental Geochemistry and Health*, vol. 29, no. 3, pp. 209-219.

He, M., Wang, X., Wu, F. & Fu, Z. 2012, 'Antimony pollution in China', *Science of the Total Environment*, vol. 421-422, pp. 41-50.

He, M. & Yang, J. 1999, 'Effects of different forms of antimony on rice during the period of germination and growth and antimony concentration in rice tissue', *Science of the Total Environment*, vol. 243-244, pp. 149-155.

Heikens, A., Panaullah, G. M. & Meharg, A. A. 2007, 'Arsenic behaviour from groundwater and soil to crops: Impacts on agriculture and food safety', *Reviews of Environmental Contamination and Toxicology*, vol. 189, pp. 43-87.

Heiri, O., Lotter, A. F. & Lemcke, G. 2001, 'Loss on ignition as a method for estimating organic and carbonate content in sediments: Reproducibility and comparability of results', *Journal of Paleolimnology*, vol. 25, no. 1, pp. 101-110.

Helgesen, H. & Larsen, E. H. 1998, 'Bioavailability and speciation of arsenic in carrots grown in contaminated soil', *Analyst*, vol. 123, no. 5, pp. 791-796.

Hinsinger, P. 2000, 'Bioavailability of Trace Elements as Related to Root-Induced Chemical Changes in the Rhizosphere', in *Trace Elements in the Rhizosphere*, CRC Press.

Hockmann, K., Lenz, M., Tandy, S., Nachtegaal, M., Janousch, M. & Schulin, R. 2014a, 'Release of antimony from contaminated soil induced by redox changes', *Journal of Hazardous Materials*, vol. 275, pp. 215-221.

Hockmann, K., Tandy, S., Lenz, M., Reiser, R., Conesa, H. M., Keller, M., Studer, B. & Schulin, R. 2015, 'Antimony retention and release from drained and waterlogged shooting range soil under field conditions', *Chemosphere*, vol. 134, pp. 536-543.

Hockmann, K., Tandy, S., Lenz, M. & Schulin, R. 2014b, 'Antimony leaching from contaminated soil under manganese- and iron-reducing conditions: Column experiments', *Environmental Chemistry*, vol. 11, no. 6, pp. 624-631.

Hooda, P. S. 2010, 'Assessing Bioavailability of Soil Trace Elements', in P. S. Hooda (ed.), *Trace Elements in Soils*, John Wiley & Sons, Ltd, pp. 227-265.

Horswell, J. & Speir, T. 2006, 'Arsenic phytotoxicity: Effect on crop yield and crop quality', in R. Naidu, et al. (eds), *Managing Arsenic in the Environment: From Soil to Human Health*, CSIRO Publishing, Collingwood, VIC, Australia, pp. 183-207.

Hossain, M., Williams, P. N., Mestrot, A., Norton, G. J., Deacon, C. M. & Meharg, A.
A. 2012, 'Spatial heterogeneity and kinetic regulation of arsenic dynamics in mangrove sediments: The Sundarbans, Bangladesh', *Environmental Science and Technology*, vol. 46, no. 16, pp. 8645-8652.

Hossain, M. B., Jahiruddin, M., Loeppert, R. H., Panaullah, G. M., Islam, M. R. & Duxbury, J. M. 2009, 'The effects of iron plaque and phosphorus on yield and arsenic accumulation in rice', *Plant and Soil*, vol. 317, no. 1-2, pp. 167-176.

Huang, G., Chen, Z., Zhang, Y., Liu, F., Wang, J. & Hou, Q. 2016, 'Changes of arsenic fractionation and bioaccessibility in wastewater-irrigated soils as a function of aging: Influence of redox condition and arsenic load', *Geoderma*, vol. 280, pp. 1-7.

Huang, R. Q., Gao, S. F., Wang, W. L., Staunton, S. & Wang, G. 2006, 'Soil arsenic availability and the transfer of soil arsenic to crops in suburban areas in Fujian Province, southeast China', *Science of the Total Environment*, vol. 368, no. 2-3, pp. 531-541.

Huang, Y., Chen, Z. & Liu, W. 2012, 'Influence of iron plaque and cultivars on antimony uptake by and translocation in rice (Oryza sativa L.) seedlings exposed to Sb(III) or Sb(V)', *Plant and Soil*, vol. 352, no. 1-2, pp. 41-49.

Huang, Z. E. C., Chen, T. B., Lei, M., Liu, Y. R. & Hu, T. D. 2008, 'Difference of toxicity and accumulation of methylated and inorganic arsenic in arsenic-hyperaccumulating and -hypertolerant plants', *Environmental Science and Technology*, vol. 42, no. 14, pp. 5106-5111.

Husson, O. 2013, 'Redox potential (Eh) and pH as drivers of soil/plant/microorganism systems: A transdisciplinary overview pointing to integrative opportunities for agronomy', *Plant and Soil*, vol. 362, no. 1-2, pp. 389-417.

Huynh, T., Zhang, H. & Noller, B. 2012, 'Evaluation and application of the diffusive gradients in thin films technique using a mixed-binding gel layer for measuring inorganic arsenic and metals in mining impacted water and soil', *Analytical Chemistry*, vol. 84, no. 22, pp. 9988-9995.

Huynh, T. T., Zhang, H., Laidlaw, W. S., Singh, B. & Baker, A. J. M. 2010, 'Plantinduced changes in the bioavailability of heavy metals in soil and biosolids assessed by DGT measurements', *Journal of Soils and Sediments*, vol. 10, no. 6, pp. 1131-1141.

Isayenkov, S. V. & Maathuis, F. J. M. 2008, 'The Arabidopsis thaliana aquaglyceroporin AtNIP7;1 is a pathway for arsenite uptake', *FEBS Letters*, vol. 582, no. 11, pp. 1625-1628.

Islam, F. S., Gault, A. G., Boothman, C., Polya, D. A., Chamok, J. M., Chatterjee, D. & Lloyd, J. R. 2004, 'Role of metal-reducing bacteria in arsenic release from Bengal delta sediments', *Nature*, vol. 430, no. 6995, pp. 68-71.

JECFA 2011, Report of the fifth session of the Codex Committee on Contaminants in Food, Joint FAO/WHO Food Standards Programme, Netherlands.

Jedynak, L., Kowalska, J., Kossykowska, M. & Golimowski, J. 2010, 'Studies on the uptake of different arsenic forms and the influence of sample pretreatment on arsenic speciation in White mustard (Sinapis alba)', *Microchemical Journal*, vol. 94, no. 2, pp. 125-129.

Jiang, Y., Zeng, X., Fan, X., Chao, S., Zhu, M. & Cao, H. 2015, 'Levels of arsenic pollution in daily foodstuffs and soils and its associated human health risk in a town in Jiangsu Province, China', *Ecotoxicology and Environmental Safety*, vol. 122, pp. 198-204.

Johnson, C. A., Moench, H., Wersin, P., Kugler, P. & Wenger, C. 2005, 'Solubility of antimony and other elements in samples taken from shooting ranges', *Journal of Environmental Quality*, vol. 34, no. 1, pp. 248-254.

Jones, C. A., Inskeep, W. P. & Neuman, D. R. 1997, 'Arsenic transport in contaminated mine tailings following liming', *Journal of Environmental Quality*, vol. 26, no. 2, pp. 433-439.

Juhasz, A. L., Smith, E., Weber, J., Naidu, R., Rees, M., Rofe, A., Kuchel, T. & Sansom, L. 2008, 'Effect of soil ageing on in vivo arsenic bioavailability in two dissimilar soils', *Chemosphere*, vol. 71, no. 11, pp. 2180-2186.

Kabata-Pendias, A. 2011, *Trace elements in soils and plants*, 4th edn, CRC Press, Boca Raton, Fl.

Kabata-Pendias, A. & Mukherjee, A. B. 2007, *Trace elements from soil to human*, Springer-Verlag, Berlin Heidelberg.

Keon, N. E., Swartz, C. H., Brabander, D. J., Harvey, C. & Hemond, H. F. 2001, 'Validation of an arsenic sequential extraction method for evaluating mobility in sediments', *Environmental Science and Technology*, vol. 35, no. 13, pp. 2778-2784.

Khan, S., Cao, Q., Zheng, Y. M., Huang, Y. Z. & Zhu, Y. G. 2008, 'Health risks of heavy metals in contaminated soils and food crops irrigated with wastewater in Beijing, China', *Environmental Pollution*, vol. 152, no. 3, pp. 686-692.

Khan, S., Rehman, S., Zeb Khan, A., Amjad Khan, M. & Tahir Shah, M. 2010, 'Soil and vegetables enrichment with heavy metals from geological sources in Gilgit, northern Pakistan', *Ecotoxicology and Environmental Safety*, vol. 73, no. 7, pp. 1820-1827.

Khan, S., Reid, B. J., Li, G. & Zhu, Y. G. 2014, 'Application of biochar to soil reduces cancer risk via rice consumption: A case study in Miaoqian village, Longyan, China', *Environment International*, vol. 68, pp. 154-161.

Kilgour, D. W., Moseley, R. B., Barnett, M. O., Savage, K. S. & Jardine, P. M. 2008, 'Potential negative consequences of adding phosphorus-based fertilizers to immobilize lead in soil', *Journal of Environmental Quality*, vol. 37, no. 5, pp. 1733-1740.

Kim Anh, T., Kim, D. D., Kuschk, P., Tua, T. V., Hue, N. T. & Minh, N. N. 2013, 'Effect of soil pH on As hyperaccumutation capacity in fern species, Pityrogramma calomelanos', *Journal of Environmental Biology*, vol. 34, no. 2, pp. 237-242.

Kim, E. J., Yoo, J. C. & Baek, K. 2014, 'Arsenic speciation and bioaccessibility in arsenic-contaminated soils: Sequential extraction and mineralogical investigation', *Environmental Pollution*, vol. 186, pp. 29-35.

Kirby, J., Maher, W., Ellwood, M. & Krikowa, F. 2004, 'Arsenic Species Determination in Biological Tissues by HPLCICPMS and HPLCHGICPMS', *Australian Journal of Chemistry*, vol. 57, no. 10, pp. 957-966.

Larios, R., Fernández-Martínez, R., LeHecho, I. & Rucandio, I. 2012a, 'A methodological approach to evaluate arsenic speciation and bioaccumulation in different plant species from two highly polluted mining areas', *Science of the Total Environment*, vol. 414, pp. 600-607.

Larios, R., Fernández-Martínez, R. & Rucandio, I. 2012b, 'Comparison of three sequential extraction procedures for fractionation of arsenic from highly polluted

227
mining sediments', Analytical and Bioanalytical Chemistry, vol. 402, no. 9, pp. 2909-2921.

Larios, R., Fernández-Martínez, R. & Rucandio, I. 2013, 'Assessment of a sequential extraction procedure for arsenic partitioning and application to samples from different pollution sources', *Analytical Methods*, vol. 5, no. 16, pp. 4096-4104.

Lehto, N. J. 2016, 'Principles and Application in Soils and Sediments', in W. Davison (ed.), *Diffusive Gradients in Thin-Films for Environmental Measurements*, Cambridge University Press, Cambridge, pp. 123-145.

Lehto, N. J., Davison, W., Zhang, H. & Tych, W. 2006a, 'An evaluation of DGT performance using a dynamic numerical model', *Environmental Science and Technology*, vol. 40, no. 20, pp. 6368-6376.

Lehto, N. J., Davison, W., Zhang, H. & Tych, W. 2006b, 'Theoretical comparison of how soil processes affect uptake of metals by diffusive gradients in thinfilms and plants', *Journal of Environmental Quality*, vol. 35, no. 5, pp. 1903-1913.

Leuz, A. K., Mönch, H. & Johnson, C. A. 2006, 'Sorption of Sb(III) and Sb(V) to goethite: Influence on Sb(III) oxidation and mobilization', *Environmental Science and Technology*, vol. 40, no. 23, pp. 7277-7282.

Li, J., Wei, Y., Zhao, L., Zhang, J., Shangguan, Y., Li, F. & Hou, H. 2014, 'Bioaccessibility of antimony and arsenic in highly polluted soils of the mine area and health risk assessment associated with oral ingestion exposure', *Ecotoxicology and Environmental Safety*, vol. 110, pp. 308-315.

Liang, S., Guan, D. X., Ren, J. H., Zhang, M., Luo, J. & Ma, L. Q. 2014, 'Effect of aging on arsenic and lead fractionation and availability in soils: Coupling sequential extractions with diffusive gradients in thin-films technique', *Journal of Hazardous Materials*, vol. 273, pp. 272-279.

Lin, C. F., Wu, C. H. & Lai, H. T. 2008, 'Dissolved organic matter and arsenic removal with coupled chitosan/UF operation', *Separation and Purification Technology*, vol. 60, no. 3, pp. 292-298.

Lin, Z. & Puls, R. W. 2000, 'Adsorption, desorption and oxidation of arsenic affected by clay minerals and aging process', *Environmental Geology*, vol. 39, no. 7, pp. 753-759.

Lintschinger, J., Michalke, B., Schulte-Hostede, S. & Schramel, P. 1998, 'Studies on speciation of antimony in soil contaminated by industrial activity', *International Journal of Environmental Analytical Chemistry*, vol. 72, no. 1, pp. 11-25.

Liu, C., Yu, H. Y., Li, F., Xu, X. & Wang, Q. 2015, 'Arsenic availability in rice from a mining area: Is amorphous iron oxide-bound arsenic a source or sink?', *Environmental Pollution*, vol. 199, pp. 95-101.

Liu, F., De Cristofaro, A. & Violante, A. 2001, 'Effect of pH, phosphate and oxalate on the adsorption/desorption of arsenate on/from goethite', *Soil Science*, vol. 166, no. 3, pp. 197-208.

Liu, W. J., Zhu, Y. G., Smith, F. A. & Smith, S. E. 2004, 'Do iron plaque and genotypes affect arsenate uptake and translocation by rice seedlings (Oryza sativa L.) grown in solution culture?', *Journal of Experimental Botany*, vol. 55, no. 403, pp. 1707-1713.

Livesey, N. T. & Huang, P. M. 1981, 'Adsorption of arsenate by soils and its relation to selected chemical properties and anions', *Soil Science*, vol. 131, no. 2, pp. 88-94.

Lock, A., Wallschläger, D., McMurdo, C., Tyler, L., Belzile, N. & Spiers, G. 2016, 'Validation of an updated fractionation and indirect speciation procedure for inorganic arsenic in oxic and suboxic soils and sediments', *Environmental Pollution*, vol. 219, pp. 1102-1108.

Lomaglio, T., Hattab-Hambli, N., Bret, A., Miard, F., Trupiano, D., Scippa, G. S., Motelica-Heino, M., Bourgerie, S. & Morabito, D. 2016, 'Effect of biochar amendments on the mobility and (bio) availability of As, Sb and Pb in a contaminated mine technosol', *Journal of Geochemical Exploration*.

Lombi, E. & Holm, P. E. 2010, 'Metalloids, Soil Chemistry and the Environment', in T. P. Jahn & G. P. Bienert (eds), *MIPs and Their Role in the Exchange of Metalloids*, Springer New York, New York, NY, pp. 33-44.

Lombi, E., Zhao, F.-J., Fuhrmann, M., Ma, L. Q. & McGrath, S. P. 2002, 'Arsenic distribution and speciation in the fronds of the hyperaccumulator Pteris vittata', *New Phytologist*, vol. 156, no. 2, pp. 195-203.

Lorenz, S. E., Hamon, R. E. & McGrath, S. P. 1994, 'Differences between soil solutions obtained from rhizosphere and non-rhizosphere soils by water displacement and soil centrifugation', *European Journal of Soil Science*, vol. 45, no. 4, pp. 431-438.

Lund, U. & Fobian, A. 1991, 'Pollution of two soils by arsenic, chromium and copper, Denmark', *Geoderma*, vol. 49, no. 1-2, pp. 83-103.

Luo, J. 2008, 'Effect of soil dynamics on metal uptake by plants and diffusive gradients in thin-films', PhD thesis, Lancaster University.

Luo, J., Cheng, H., Ren, J., Davison, W. & Zhang, H. 2014, 'Mechanistic insights from DGT and soil solution measurements on the uptake of Ni and Cd by radish', *Environmental Science and Technology*, vol. 48, no. 13, pp. 7305-7313.

Luo, J., Zhang, H., Santner, J. & Davison, W. 2010, 'Performance characteristics of diffusive gradients in thin films equipped with a binding gel layer containing precipitated ferrihydrite for measuring arsenic(V), Selenium(VI), Vanadium(V), and Antimony(V)', *Analytical Chemistry*, vol. 82, no. 21, pp. 8903-8909.

Ma, J. F., Yamaji, N., Mitani, N., Xu, X. Y., Su, Y. H., McGrath, S. P. & Zhao, F. J. 2008, 'Transporters of arsenite in rice and their role in arsenic accumulation in rice grain', *Proceedings of the National Academy of Sciences of the United States of America*, vol. 105, no. 29, pp. 9931-9935.

Macgregor, K., MacKinnon, G., Farmer, J. G. & Graham, M. C. 2015, 'Mobility of antimony, arsenic and lead at a former antimony mine, Glendinning, Scotland', *Science of the Total Environment*, vol. 529, pp. 213-222.

Maher, W., Foster, S., Krikowa, F., Donner, E. & Lombi, E. 2013, 'Measurement of Inorganic Arsenic Species in Rice after Nitric Acid Extraction by HPLC-ICPMS: Verification Using XANES', *Environmental science & technology*, vol. 47, no. 11, pp. 5821-5827.

Mahimairaja, S., Bolan, N. S., Adriano, D. C. & Robinson, B. 2005, 'Arsenic contamination and its risk management in complex environmental settings', *Advances in agronomy*, vol. 86, pp. 1-82.

Manaka, M. 2006, 'Amount of amorphous materials in relationship to arsenic, antimony, and bismuth concentrations in a brown forest soil', *Geoderma*, vol. 136, no. 1-2, pp. 75-86.

Mandal, B. K. & Suzuki, K. T. 2002, 'Arsenic round the world: A review', *Talanta*, vol. 58, no. 1, pp. 201-235.

Manful, G. A., Verloo, M. & De Spiegeleer, F. 1989, 'Arsenate sorption by soils in relation to pH and selected anions', *Pedologie*, vol. 39, no. 1, pp. 55-68.

Manning, B. A. & Goldberg, S. 1996a, 'Modeling arsenate competitive adsorption on kaolinite, montmorillonite and illite', *Clays and Clay Minerals*, vol. 44, no. 5, pp. 609-623.

Manning, B. A. & Goldberg, S. 1996b, 'Modeling competitive adsorption of arsenate with phosphate and molybdate on oxide minerals', *Soil Science Society of America Journal*, vol. 60, no. 1, pp. 121-131.

Manning, B. A. & Goldberg, S. 1997, 'Arsenic(III) and arsenic(V) adsorption on three California soils', *Soil Science*, vol. 162, no. 12, pp. 886-895.

María-Cervantes, A., Conesa, H. M., González-Alcaraz, M. N. & Álvarez-Rogel, J. 2010, 'Rhizosphere and flooding regime as key factors for the mobilisation of arsenic and potentially harmful metals in basic, mining-polluted salt marsh soils', *Applied Geochemistry*, vol. 25, no. 11, pp. 1722-1733.

Marin, A. R., Masscheleyn, P. H. & Patrick Jr, W. H. 1992, 'The influence of chemical form and concentration of arsenic on rice growth and tissue arsenic concentration', *Plant and Soil*, vol. 139, no. 2, pp. 175-183.

Marschner, H. 1995, *Mineral Nutrition of Higher Plants*, 2 edn, Academic Press, London.

Marschner, P. & Rengel, Z. 2012, 'Chapter 12 - Nutrient Availability in Soils', in *Marschner's Mineral Nutrition of Higher Plants (Third Edition)*, Academic Press, San Diego, pp. 315-330.

Mason, S., Hamon, R., Nolan, A., Zhang, H. & Davison, W. 2005, 'Performance of a mixed binding layer for measuring anions and cations in a single assay using the diffusive gradients in thin films technique', *Analytical Chemistry*, vol. 77, no. 19, pp. 6339-6346.

Mason, S., McNeill, A., McLaughlin, M. J. & Zhang, H. 2010, 'Prediction of wheat response to an application of phosphorus under field conditions using diffusive gradients in thin-films (DGT) and extraction methods', *Plant and Soil*, vol. 337, no. 1, pp. 243-258.

Masscheleyn, P. H., Delaune, R. D. & Patrick Jr, W. H. 1991, 'Effect of redox potential and pH on arsenic speciation and solubility in a contaminated soil', *Environmental Science and Technology*, vol. 25, no. 8, pp. 1414-1419.

Mathews, S., Rathinasabapathi, B. & Ma, L. Q. 2011, 'Uptake and translocation of arsenite by Pteris vittata L.: Effects of glycerol, antimonite and silver', *Environmental Pollution*, vol. 159, no. 12, pp. 3490-3495.

Matschullat, J. 2000, 'Arsenic in the geosphere - A review', *Science of the Total Environment*, vol. 249, no. 1-3, pp. 297-312.

McBride, M. B. 2013, 'Arsenic and lead uptake by vegetable crops grown on historically contaminated orchard soils', *Applied and Environmental Soil Science*, vol. 2013.

McBride, M. B., Simon, T., Tam, G. & Wharton, S. 2013, 'Lead and arsenic uptake by leafy vegetables grown on contaminated soils: Effects of mineral and organic amendments', *Water, Air, and Soil Pollution*, vol. 224, no. 1.

McLaren, R. G., Naidu, R., Smith, J. & Tiller, K. G. 1998, 'Fractionation and distribution of arsenic in soils contaminated by cattle dip', *Journal of Environmental Quality*, vol. 27, no. 2, pp. 348-354.

McLaughlin, M. J. 2001, 'Bioavailability of metals to terrestrial plants', in H. E. Allen (ed.), *Bioavailability of metals in terrestrial ecosystems: Importance of partitioning for bioavailability to invertebrates, microbes, and plants*, SETAC Press, Pensacola, FL, pp. 39-68.

McLaughlin, M. J., Smolders, E., Degryse, F. & Rietra, R. 2011, 'Uptake of metals from soil into vegetables', in F. A. Swartjes (ed.), *Dealing with contaminated sites: from theory towards practical application*, Springer, Dordrecht, pp. 325-367.

Meharg, A. A. 1994, 'Integrated tolerance mechanisms: constitutive and adaptive plant responses to elevated metal concentrations in the environment', *Plant, Cell & Environment*, vol. 17, no. 9, pp. 989-993.

Meharg, A. A. & Hartley-Whitaker, J. 2002, 'Arsenic uptake and metabolism in arsenic resistant and nonresistant plant species', *New Phytologist*, vol. 154, no. 1, pp. 29-43.

Meharg, A. A. & Jardine, L. 2003, 'Arsenite transport into paddy rice (Oryza sativa) roots', *New Phytologist*, vol. 157, no. 1, pp. 39-44.

Meharg, A. A. & Macnair, M. R. 1990, 'An altered phosphate uptake system in arsenate-tolerant Holcus lanatus L', *New Phytologist*, vol. 116, no. 1, pp. 29-35.

Meharg, A. A. & Macnair, M. R. 1992, 'Suppression of the high affinity phosphate uptake system: A mechanism of arsenate tolerance in Holcus lanatus L', *Journal of Experimental Botany*, vol. 43, no. 4, pp. 519-524.

Meharg, A. A. & Rahman, M. 2003, 'Arsenic contamination of Bangladesh paddy field soils: Implications for rice contribution to arsenic consumption', *Environmental Science and Technology*, vol. 37, no. 2, pp. 229-234.

Mendoza-Cózatl, D. G., Jobe, T. O., Hauser, F. & Schroeder, J. I. 2011, 'Long-distance transport, vacuolar sequestration, tolerance, and transcriptional responses induced by cadmium and arsenic', *Current Opinion in Plant Biology*, vol. 14, no. 5, pp. 554-562.

Meng, X. Y., Qin, J., Wang, L. H., Duan, G. L., Sun, G. X., Wu, H. L., Chu, C. C., Ling, H. Q., Rosen, B. P. & Zhu, Y. G. 2011, 'Arsenic biotransformation and volatilization in transgenic rice', *New Phytologist*, vol. 191, no. 1, pp. 49-56.

Mengel, K. & Kirkby, E. A. 2001, *Principles of plant nutrition*, 5 edn, Springer, Netherlands.

Menzie, N. & Guppy, C. 2000, 'In-situ soil solution extraction with polyacrylonitrile hollow-fibers', *Communications in Soil Science and Plant Analysis*, vol. 31, no. 11-14, pp. 1875-1886.

Mestrot, A., Ji, Y., Tandy, S. & Wilcke, W. 2016, 'A novel method to determine trimethylantimony concentrations in plant tissue', *Environmental Chemistry*, vol. 13, no. 6, pp. 919-926.

Meunier, L., Walker, S. R., Wragg, J., Parsons, M. B., Koch, I., Jamieson, H. E. & Reimer, K. J. 2010, 'Effects of soil composition and mineralogy on the bioaccessibility of arsenic from tailings and soil in gold mine districts of nova scotia', *Environmental Science and Technology*, vol. 44, no. 7, pp. 2667-2674.

Mitsunobu, S., Harada, T. & Takahashi, Y. 2006, 'Comparison of antimony behavior with that of arsenic under various soil redox conditions', *Environmental Science and Technology*, vol. 40, no. 23, pp. 7270-7276.

Mitsunobu, S., Takahashi, Y. & Terada, Y. 2010, 'µ-XANES evidence for the reduction of Sb(V) to Sb(III) in soil from Sb mine tailing', *Environmental Science and Technology*, vol. 44, no. 4, pp. 1281-1287.

Mojsilovic, O., McLaren, R. G. & Condron, L. M. 2011, 'Modelling arsenic toxicity in wheat: Simultaneous application of diffusive gradients in thin films to arsenic and phosphorus in soil', *Environmental Pollution*, vol. 159, no. 10, pp. 2996-3002.

Montes-Bayón, M., Meija, J., LcDuc, D. L., Terry, N., Caruso, J. A. & Sanz-Medel, A. 2004, 'HPLC-ICP-MS and ESI-Q-TOF analysis of biomolecules induced in Brassica juncea during arsenic accumulation', *Journal of Analytical Atomic Spectrometry*, vol. 19, no. 1, pp. 153-158.

Moreno-Jiménez, E., Esteban, E. & Peñalosa, J. M. 2012, 'The Fate of Arsenic in Soil-Plant Systems', in D. M. Whitacre (ed.), *Reviews of Environmental Contamination and Toxicology*, Springer New York, New York, NY, pp. 1-37. Müller, K., Daus, B., Mattusch, J., Vetterlein, D., Merbach, I. & Wennrich, R. 2013, 'Impact of arsenic on uptake and bio-accumulation of antimony by arsenic hyperaccumulator Pteris vittata', *Environmental Pollution*, vol. 174, pp. 128-133.

Müller, K., Daus, B., Morgenstern, P. & Wennrich, R. 2007, 'Mobilization of antimony and arsenic in soil and sediment samples - Evaluation of different leaching procedures', *Water, Air, and Soil Pollution*, vol. 183, no. 1-4, pp. 427-436.

Muñoz, O., Diaz, O. P., Leyton, I., Nuñez, N., Devesa, V., Súñer, M. A., Vélez, D. & Montoro, R. 2002, 'Vegetables collected in the cultivated Andean area of Northern Chile: Total and inorganic arsenic contents in raw vegetables', *Journal of Agricultural and Food Chemistry*, vol. 50, no. 3, pp. 642-647.

Murciego, A. M., Sánchez, A. G., González, M. A. R., Gil, E. P., Gordillo, C. T., Fernández, J. C. & Triguero, T. B. 2007, 'Antimony distribution and mobility in topsoils and plants (Cytisus striatus, Cistus ladanifer and Dittrichia viscosa) from polluted Sb-mining areas in Extremadura (Spain)', *Environmental Pollution*, vol. 145, no. 1, pp. 15-21.

Murray, H., Thompson, K. & Macfie, S. M. 2009, 'Site-and species-specific patterns of metal bioavailability in edible plants', *Botany*, vol. 87, no. 7, pp. 702-711.

Nakamaru, Y., Tagami, K. & Uchida, S. 2006, 'Antimony mobility in Japanese agricultural soils and the factors affecting antimony sorption behavior', *Environmental Pollution*, vol. 141, no. 2, pp. 321-326.

Nakamaru, Y. M. & Altansuvd, J. 2014, 'Speciation and bioavailability of selenium and antimony in non-flooded and wetland soils: A review', *Chemosphere*, vol. 111, pp. 366-371.

Nakamaru, Y. M. & Sekine, K. 2008, 'Sorption behavior of selenium and antimony in soils as a function of phosphate ion concentration', *Soil Science and Plant Nutrition*, vol. 54, no. 3, pp. 332-341.

Ngo, L. K., Pinch, B. M., Bennett, W. W., Teasdale, P. R. & Jolley, D. F. 2016, 'Assessing the uptake of arsenic and antimony from contaminated soil by radish (Raphanus sativus) using DGT and selective extractions', *Environmental Pollution*, vol. 216, pp. 104-114.

Niazi, N. K., Singh, B. & Shah, P. 2011, 'Arsenic speciation and phytoavailability in contaminated soils using a sequential extraction procedure and XANES spectroscopy', *Environmental Science and Technology*, vol. 45, no. 17, pp. 7135-7142.

Nishiuchi, S., Yamauchi, T., Takahashi, H., Kotula, L. & Nakazono, M. 2012, 'Mechanisms for coping with submergence and waterlogging in rice', *Rice*, vol. 5, no. 1, p. 2.

Nolan, A. L., Lombi, E. & McLaughlin, M. J. 2003, 'Metal Bioaccumulation and Toxicity in Soils—Why Bother with Speciation?', *Australian Journal of Chemistry*, vol. 56, no. 3, pp. 77-91.

Nolan, A. L., Zhang, H. & McLaughlin, M. J. 2005, 'Prediction of zinc, cadmium, lead, and copper availability to wheat in contaminated soils using chemical speciation, diffusive gradients in thin films, extraction, and isotopic dilution techniques', *Journal of Environmental Quality*, vol. 34, no. 2, pp. 496-507.

Nowack, B., Koehler, S. & Schulin, R. 2004, 'Use of Diffusive Gradients in Thin Films (DGT) in Undisturbed Field Soils', *Environmental Science and Technology*, vol. 38, no. 4, pp. 1133-1138.

O'Neill, P. 1995, 'Arsenic', in *Heavy Metals in Soils*, Blackie Academic & Professional, London, pp. 105-121.

Oelkers, E. H., Sherman, D. M., Ragnarsdottir, K. V. & Collins, C. 1998, 'An EXAFS spectroscopic study of aqueous antimony(III)-chloride complexation at temperatures from 25 to 250°C', *Chemical Geology*, vol. 151, no. 1-4, pp. 21-27.

Okkenhaug, G., Grasshorn Gebhardt, K. A., Amstaetter, K., Lassen Bue, H., Herzel, H., Mariussen, E., Rossebø Almås, Å., Cornelissen, G., Breedveld, G. D., Rasmussen, G. & Mulder, J. 2016, 'Antimony (Sb) and lead (Pb) in contaminated shooting range soils: Sb and Pb mobility and immobilization by iron based sorbents, a field study', *Journal of Hazardous Materials*, vol. 307, pp. 336-343.

Okkenhaug, G., Zhu, Y. G., He, J., Li, X., Luo, L. & Mulder, J. 2012, 'Antimony (Sb) and Arsenic (As) in Sb mining impacted paddy soil from Xikuangshan, China: Differences in mechanisms controlling soil sequestration and uptake in Rice', *Environmental Science and Technology*, vol. 46, no. 6, pp. 3155-3162.

Okkenhaug, G., Zhu, Y. G., Luo, L., Lei, M., Li, X. & Mulder, J. 2011, 'Distribution, speciation and availability of antimony (Sb) in soils and terrestrial plants from an active Sb mining area', *Environmental Pollution*, vol. 159, no. 10, pp. 2427-2434.

Onken, B. M. & Adriano, D. C. 1997, 'Arsenic availability in soil with time under saturated and subsaturated conditions', *Soil Science Society of America Journal*, vol. 61, no. 3, pp. 746-752.

Onken, B. M. & Hossner, L. R. 1995, 'Plant uptake and determination of arsenic species in soil solution under flooded conditions', *Journal of Environmental Quality*, vol. 24, no. 2, pp. 373-381.

Oorts, K., Smolders, E., Degryse, F., Buekers, J., Gascó, G., Cornelis, G. & Mertens, J. 2008, 'Solubility and toxicity of antimony trioxide (Sb2O3) in soil', *Environmental Science and Technology*, vol. 42, no. 12, pp. 4378-4383.

Österlund, H., Chlot, S., Faarinen, M., Widerlund, A., Rodushkin, I., Ingri, J. & Baxter, D. C. 2010, 'Simultaneous measurements of As, Mo, Sb, V and W using a ferrihydrite diffusive gradients in thin films (DGT) device', *Analytica Chimica Acta*, vol. 682, no. 1-2, pp. 59-65.

Otero, X. L., Ferreira, T. O., Huerta-Díaz, M. A., Partiti, C. S. M., Souza, V., Vidal-Torrado, P. & Macías, F. 2009, 'Geochemistry of iron and manganese in soils and sediments of a mangrove system, Island of Pai Matos (Cananeia — SP, Brazil)', *Geoderma*, vol. 148, no. 3, pp. 318-335.

Panther, J. G., Bennett, W. W., Welsh, D. T. & Teasdale, P. R. 2014, 'Simultaneous measurement of trace metal and oxyanion concentrations in water using diffusive gradients in thin films with a chelex-metsorb mixed binding layer', *Analytical Chemistry*, vol. 86, no. 1, pp. 427-434.

237

Panther, J. G., Stewart, R. R., Teasdale, P. R., Bennett, W. W., Welsh, D. T. & Zhao, H. 2013, 'Titanium dioxide-based DGT for measuring dissolved As(V), V(V), Sb(V), Mo(VI) and W(VI) in water', *Talanta*, vol. 105, pp. 80-86.

Panther, J. G., Stillwell, K. P., Powell, K. J. & Downard, A. J. 2008, 'Development and application of the diffusive gradients in thin films technique for the measurement of total dissolved inorganic arsenic in waters', *Analytica Chimica Acta*, vol. 622, no. 1, pp. 133-142.

Panther, J. G., Teasdale, P. R., Bennett, W. W., Welsh, D. T. & Zhao, H. 2010, 'Titanium dioxide-based DGT technique for in situ measurement of dissolved reactive phosphorus in fresh and marine waters', *Environmental Science and Technology*, vol. 44, no. 24, pp. 9419-9424.

Pérez-Sirvent, C., Martínez-Sánchez, M. J., Martínez-López, S., Bech, J. & Bolan, N. 2012, 'Distribution and bioaccumulation of arsenic and antimony in Dittrichia viscosa growing in mining-affected semiarid soils in southeast Spain', *Journal of Geochemical Exploration*, vol. 123, pp. 128-135.

Pérez-Sirvent, C., Martínez-Sánchez, M. J., Martínez-López, S. & Hernández-Córdoba,
M. 2011, 'Antimony distribution in soils and plants near an abandoned mining site', *Microchemical Journal*, vol. 97, no. 1, pp. 52-56.

Pickering, I. J., Prince, R. C., George, M. J., Smith, R. D., George, G. N. & Salt, D. E. 2000, 'Reduction and coordination of arsenic in Indian mustard', *Plant Physiology*, vol. 122, no. 4, pp. 1171-1177.

Pierart, A., Shahid, M., Séjalon-Delmas, N. & Dumat, C. 2015, 'Antimony bioavailability: Knowledge and research perspectives for sustainable agricultures', *Journal of Hazardous Materials*, vol. 289, pp. 219-234.

Pierce, M. L. & Moore, C. B. 1982, 'Adsorption of arsenite and arsenate on amorphous iron hydroxide', *Water Research*, vol. 16, no. 7, pp. 1247-1253.

Pigna, M., Cozzolino, V., Giandonato Caporale, A., Mora, M. L., Di Meo, V., Jara, A. A. & Violante, A. 2010, 'Effects of phosphorus fertilization on arsenic uptake by wheat

grown in polluted soils', *Journal of Soil Science and Plant Nutrition*, vol. 10, no. 4, pp. 428-442.

Pratas, J., Prasad, M. N. V., Freitas, H. & Conde, L. 2005, 'Plants growing in abandoned mines of Portugal are useful for biogeochemical exploration of arsenic, antimony, tungsten and mine reclamation', *Journal of Geochemical Exploration*, vol. 85, no. 3, pp. 99-107.

Price, H. L., Teasdale, P. R. & Jolley, D. F. 2013, 'An evaluation of ferrihydrite- and Metsorb<sup>TM</sup>-DGT techniques for measuring oxyanion species (As, Se, V, P): Effective capacity, competition and diffusion coefficients', *Analytica Chimica Acta*, vol. 803, pp. 56-65.

Ptak, C. & McBride, M. 2015, 'Organically complexed iron enhances bioavailability of antimony to maize (Zea mays) seedlings in organic soils', *Environmental Toxicology and Chemistry*, vol. 34, no. 12, pp. 2732-2738.

Punshon, T., Jackson, B. P., Meharg, A. A., Warczack, T., Scheckel, K. & Guerinot, M.L. 2017, 'Understanding arsenic dynamics in agronomic systems to predict and prevent uptake by crop plants', *Science of the Total Environment*, vol. 581-582, pp. 209-220.

Qafoku, N. P., Kukier, U., Sumner, M. E., Miller, W. P. & Radcliffe, D. E. 1999, 'Arsenate displacement from fly ash in amended soils', *Water, Air, and Soil Pollution*, vol. 114, no. 1-2, pp. 185-198.

Qi, C., Wu, F., Deng, Q., Liu, G., Mo, C., Liu, B. & Zhu, J. 2011, 'Distribution and accumulation of antimony in plants in the super-large Sb deposit areas, China', *Microchemical Journal*, vol. 97, no. 1, pp. 44-51.

Qi, P. & Pichler, T. 2017, 'Competitive adsorption of As(III), As(V), Sb(III) and Sb(V) onto ferrihydrite in multi-component systems: Implications for mobility and distribution', *Journal of Hazardous Materials*, vol. 330, pp. 142-148.

Raab, A., Feldmann, J. & Meharg, A. A. 2004, 'The nature of arsenic-phytochelatin complexes in Holcus lanatus and Pteris cretica', *Plant Physiology*, vol. 134, no. 3, pp. 1113-1122.

Raab, A., Schat, H., Meharg, A. A. & Feldmann, J. 2005, 'Uptake, translocation and transformation of arsenate and arsenite in sunflower (Helianthus annuus): Formation of arsenic-phytochelatin complexes during exposure to high arsenic concentrations', *New Phytologist*, vol. 168, no. 3, pp. 551-558.

Rahaman, S., Sinha, A. C. & Mukhopadhyay, D. 2011, 'Effect of water regimes and organic matters on transport of arsenic in summer rice (Oryza sativa L.)', *Journal of Environmental Sciences*, vol. 23, no. 4, pp. 633-639.

Rahman, F. & Naidu, R. 2009, 'The influence of arsenic speciation (AsIII & amp; AsV) and concentration on the growth, uptake and translocation of arsenic in vegetable crops (silverbeet and amaranth): Greenhouse study', *Environmental Geochemistry and Health*, vol. 31, no. SUPPL. 1, pp. 115-124.

Rahman, M. A. & Hasegawa, H. 2011, 'High levels of inorganic arsenic in rice in areas where arsenic-contaminated water is used for irrigation and cooking', *Science of the Total Environment*, vol. 409, no. 22, pp. 4645-4655.

Rahman, M. S., Clark, M. W., Yee, L. H., Comarmond, M. J., Payne, T. E., Kappen, P.
& Mokhber-Shahin, L. 2017, 'Arsenic solid-phase speciation and reversible binding in long-term contaminated soils', *Chemosphere*, vol. 168, pp. 1324-1336.

Rapant, S., Dietzová, Z. & Cicmanová, S. 2006, 'Environmental and health risk assessment in abandoned mining area, Zlata Idka, Slovakia', *Environmental Geology*, vol. 51, no. 3, pp. 387-397.

Rayment, G. E. & Lyons, D. J. 2011, *Soil chemical methods - Australasia*, CSIRO Publishing, Collingwood, VIC, Australia.

Redman, A. D., Macalady, D. L. & Ahmann, D. 2002, 'Natural organic matter affects Arsenic speciation and sorption onto hematite', *Environmental Science and Technology*, vol. 36, no. 13, pp. 2889-2896.

Rehman, Z. U., Khan, S., Qin, K., Brusseau, M. L., Shah, M. T. & Din, I. 2016, 'Quantification of inorganic arsenic exposure and cancer risk via consumption of vegetables in southern selected districts of Pakistan', *Science of the Total Environment*, vol. 550, pp. 321-329.

Ren, J., Ma, L. Q., Sun, H., Cai, F. & Luo, J. 2014, 'Antimony uptake, translocation and speciation in rice plants exposed to antimonite and antimonate', *Science of the Total Environment*, vol. 475, pp. 83-89.

Renella, G., Landi, L., Valori, F. & Nannipieri, P. 2007, 'Microbial and hydrolase activity after release of low molecular weight organic compounds by a model root surface in a clayey and a sandy soil', *Applied Soil Ecology*, vol. 36, no. 2-3, pp. 124-129.

Richmond, W. R., Loan, M., Morton, J. & Parkinson, G. M. 2004, 'Arsenic Removal from Aqueous Solution via Ferrihydrite Crystallization Control', *Environmental science & technology*, vol. 38, no. 8, pp. 2368-2372.

Robinson, B. H., Bischofberger, S., Stoll, A., Schroer, D., Furrer, G., Roulier, S., Gruenwald, A., Attinger, W. & Schulin, R. 2008, 'Plant uptake of trace elements on a Swiss military shooting range: Uptake pathways and land management implications', *Environmental Pollution*, vol. 153, no. 3, pp. 668-676.

Rosas-Castor, J. M., Guzmán-Mar, J. L., Hernández-Ramírez, A., Garza-González, M. T. & Hinojosa-Reyes, L. 2014, 'Arsenic accumulation in maize crop (Zea mays): A review', *Science of the Total Environment*, vol. 488-489, no. 1, pp. 176-187.

Rowland, H. A. L., Polya, D. A., Lloyd, J. R. & Pancost, R. D. 2006, 'Characterisation of organic matter in a shallow, reducing, arsenic-rich aquifer, West Bengal', *Organic Geochemistry*, vol. 37, no. 9, pp. 1101-1114.

Roy, W. R., Hassett, J. J. & Griffin, R. A. 1986, 'Competitive interactions of phosphate and molybdate on arsenate adsorption', *Soil Science*, vol. 142, no. 4, pp. 203-210.

Saada, A., Breeze, D., Crouzet, C., Cornu, S. & Baranger, P. 2003, 'Adsorption of arsenic (V) on kaolinite and on kaolinite-humic acid complexes role of humic acid nitrogen groups', *Chemosphere*, vol. 51, no. 8, pp. 757-763.

Sadee, B. A., Foulkes, M. E. & Hill, S. J. 2016, 'A study of arsenic speciation in soil, irrigation water and plant tissue: A case study of the broad bean plant, Vicia faba', *Food Chemistry*, vol. 210, pp. 362-370.

Sadiq, M. 1997, 'Arsenic chemistry in soils: An overview of thermodynamic predictions and field observations', *Water, Air, and Soil Pollution*, vol. 93, no. 1-4, pp. 117-136.

Sahoo, P. K. & Kim, K. 2013, 'A review of the arsenic concentration in paddy rice from the perspective of geoscience', *Geosciences Journal*, vol. 17, no. 1, pp. 107-122.

Sanderson, P., Naidu, R. & Bolan, N. 2014, 'Ecotoxicity of chemically stabilised metal(loid)s in shooting range soils', *Ecotoxicology and Environmental Safety*, vol. 100, no. 1, pp. 201-208.

Santner, J., Larsen, M., Kreuzeder, A. & Glud, R. N. 2015, 'Two decades of chemical imaging of solutes in sediments and soils - a review', *Analytica Chimica Acta*, vol. 878, pp. 9-42.

Santner, J., Zhang, H., Leitner, D., Schnepf, A., Prohaska, T., Puschenreiter, M. & Wenzel, W. W. 2012, 'High-resolution chemical imaging of labile phosphorus in the rhizosphere of Brassica napus L. cultivars', *Environmental and Experimental Botany*, vol. 77, pp. 219-226.

Scheinost, A. C., Rossberg, A., Vantelon, D., Xifra, I., Kretzschmar, R., Leuz, A. K., Funke, H. & Johnson, C. A. 2006, 'Quantitative antimony speciation in shooting-range soils by EXAFS spectroscopy', *Geochimica et Cosmochimica Acta*, vol. 70, no. 13, pp. 3299-3312.

Schmoger, M. E. V., Oven, M. & Grill, E. 2000, 'Detoxification of arsenic by phytochelatins in plants', *Plant Physiology*, vol. 122, no. 3, pp. 793-801.

Selim Reza, A. H. M., Jean, J. S., Yang, H. J., Lee, M. K., Woodall, B., Liu, C. C., Lee,
J. F. & Luo, S. D. 2010, 'Occurrence of arsenic in core sediments and groundwater in the Chapai-Nawabganj District, northwestern Bangladesh', *Water Research*, vol. 44, no. 6, pp. 2021-2037.

Senila, M., Tanaselia, C. & Rimba, E. 2013, 'Investigations on arsenic mobility changes in rizosphere of two ferns species using DGT technique', *Carpathian Journal of Earth and Environmental Sciences*, vol. 8, no. 3, pp. 145-154. Serafimovska, J. M., Arpadjan, S., Stafilov, T. & Tsekova, K. 2013, 'Study of the antimony species distribution in industrially contaminated soils', *Journal of Soils and Sediments*, vol. 13, no. 2, pp. 294-303.

Seyfferth, A. L., Webb, S. M., Andrews, J. C. & Fendorf, S. 2010, 'Arsenic localization, speciation, and co-occurrence with iron on rice (Oryza sativa L) roots having variable Fe coatings', *Environmental Science and Technology*, vol. 44, no. 21, pp. 8108-8113.

Shaibur, M. R., Islam, T. & Kawai, S. 2009, 'Response of leafy vegetable Kalmi (water spinach; Ipomoea aquatica L.) at elevated concentrations of arsenic in hydroponic culture', *Water, Air, and Soil Pollution*, vol. 202, no. 1-4, pp. 289-300.

Sharifi, R., Moore, F. & Keshavarzi, B. 2014, 'Potential health risks of arsenic, antimony and mercury in the Takab geothermal field, NW Iran', *International Journal of Environmental Studies*, vol. 71, no. 3, pp. 372-390.

Shtangeeva, I., Bali, R. & Harris, A. 2011, 'Bioavailability and toxicity of antimony', *Journal of Geochemical Exploration*, vol. 110, no. 1, pp. 40-45.

Shtangeeva, I., Niemelä, M. & Perämäki, P. 2014, 'Effects of soil amendments on antimony uptake by wheat', *Journal of Soils and Sediments*, vol. 14, no. 4, pp. 679-686.

Shtangeeva, I., Steinnes, E. & Lierhagen, S. 2012, 'Uptake of different forms of antimony by wheat and rye seedlings', *Environmental Science and Pollution Research*, vol. 19, no. 2, pp. 502-509.

Signes-Pastor, A., Burló, F., Mitra, K. & Carbonell-Barrachina, A. A. 2007, 'Arsenic biogeochemistry as affected by phosphorus fertilizer addition, redox potential and pH in a west Bengal (India) soil', *Geoderma*, vol. 137, no. 3-4, pp. 504-510.

Simpson, S. L., Yverneau, H., Cremazy, A., Jarolimek, C. V., Price, H. L. & Jolley, D.
F. 2012, 'DGT-induced copper flux predicts bioaccumulation and toxicity to bivalves in sediments with varying properties', *Environmental Science and Technology*, vol. 46, no. 16, pp. 9038-9046.

Smith, E. 1998, 'Arsenic in the Soil Environment: A Review', *Advances in agronomy*, vol. 64, no. c, pp. 149-195.

Smith, E., Juhasz, A. L. & Weber, J. 2009, 'Arsenic uptake and speciation in vegetables grown under greenhouse conditions', *Environmental Geochemistry and Health*, vol. 31, no. SUPPL. 1, pp. 125-132.

Smith, E., Juhasz, A. L., Weber, J. & Naidu, R. 2008a, 'Arsenic uptake and speciation in rice plants grown under greenhouse conditions with arsenic contaminated irrigation water', *Science of the Total Environment*, vol. 392, no. 2-3, pp. 277-283.

Smith, E., Naidu, R. & Alston, A. M. 1999, 'Chemistry of arsenic in soils: I. Sorption of arsenate and arsenite by four Australian soils', *Journal of Environmental Quality*, vol. 28, no. 6, pp. 1719-1726.

Smith, E., Naidu, R. & Alston, A. M. 2002, 'Chemistry of inorganic arsenic in soils: II. Effect of phosphorus, sodium, and calcium on arsenic sorption', *Journal of Environmental Quality*, vol. 31, no. 2, pp. 557-563.

Smith, E., Smith, J., Smith, L., Biswas, T., Correll, R. & Naidu, R. 2003, 'Arsenic in Australian environment: An overview', *Journal of Environmental Science and Health - Part A Toxic/Hazardous Substances and Environmental Engineering*, vol. 38, no. 1, pp. 223-239.

Smith, K. S. & Huyck, H. L. O. 1999, 'An overview of the abundance, relative mobility, bioavailability, and human toxicity of metals', *Reviews in Economic Geology*, vol. 6 A, pp. 29-70.

Smith, P. G., Koch, I. & Reimer, K. J. 2008b, 'Uptake, transport and transformation of arsenate in radishes (Raphanus sativus)', *Science of the Total Environment*, vol. 390, no. 1, pp. 188-197.

Sochaczewski, Ł., Tych, W., Davison, B. & Zhang, H. 2007, '2D DGT induced fluxes in sediments and soils (2D DIFS)', *Environmental Modelling & Software*, vol. 22, no. 1, pp. 14-23.

Song, J., Zhao, F. J., McGrath, S. P. & Luo, Y. M. 2006, 'Influence of soil properties and aging on arsenic phytotoxicity', *Environmental Toxicology and Chemistry*, vol. 25, no. 6, pp. 1663-1670.

Soriano-Disla, J. M., Gómez, I., Navarro-Pedreño, J. & Lag-Brotons, A. 2010, 'Evaluation of single chemical extractants for the prediction of heavy metal uptake by barley in soils amended with polluted sewage sludge', *Plant and Soil*, vol. 327, no. 1, pp. 303-314.

Steely, S., Amarasiriwardena, D. & Xing, B. 2007, 'An investigation of inorganic antimony species and antimony associated with soil humic acid molar mass fractions in contaminated soils', *Environmental Pollution*, vol. 148, no. 2, pp. 590-598.

Stockdale, A., Davison, W. & Zhang, H. 2009, 'Micro-scale biogeochemical heterogeneity in sediments: A review of available technology and observed evidence', *Earth-Science Reviews*, vol. 92, no. 1-2, pp. 81-97.

Stroud, J. L., Khan, M. A., Norton, G. J., Islam, M. R., Dasgupta, T., Zhu, Y. G., Price, A. H., Meharg, A. A., McGrath, S. P. & Zhao, F. J. 2011, 'Assessing the labile arsenic pool in contaminated paddy soils by isotopic dilution techniques and simple extractions', *Environmental Science and Technology*, vol. 45, no. 10, pp. 4262-4269.

Sun, Q., Chen, J., Zhang, H., Ding, S., Li, Z., Williams, P. N., Cheng, H., Han, C., Wu, L. & Zhang, C. 2014, 'Improved diffusive gradients in thin films (DGT) measurement of total dissolved inorganic arsenic in waters and soils using a hydrous zirconium oxide binding layer', *Analytical Chemistry*, vol. 86, no. 6, pp. 3060-3067.

Sun, Q., Zhang, L., Ding, S., Li, C., Yang, J., Chen, J. & Wang, P. 2015, 'Evaluation of the diffusive gradients in thin films technique using a mixed binding gel for measuring iron, phosphorus and arsenic in the environment', *Environmental Sciences: Processes and Impacts*, vol. 17, no. 3, pp. 570-577.

Takahashi, T., Shozugawa, K. & Matsuo, M. 2010, 'Contribution of amorphous iron compounds to adsorptions of pentavalent antimony by soils', *Water, Air, and Soil Pollution*, vol. 208, no. 1-4, pp. 165-172.

Takahashi, Y., Minamikawa, R., Hattori, K. H., Kurishima, K., Kihou, N. & Yuita, K. 2004, 'Arsenic Behavior in Paddy Fields during the Cycle of Flooded and Non-flooded Periods', *Environmental Science and Technology*, vol. 38, no. 4, pp. 1038-1044.

Takamatsu, T., Aoki, H. & Yoshida, T. 1982, 'Determination of arsenate, arsenite, monomethylarsonate, and dimethylarsinate in soil polluted with arsenic', *Soil Science*, vol. 133, no. 4, pp. 239-246.

Tamás, M. J. 2016, 'Cellular and molecular mechanisms of antimony transport, toxicity and resistance', *Environmental Chemistry*, vol. 13, no. 6, pp. 955-962.

Tang, X. Y., Zhu, Y. G., Shan, X. Q., McLaren, R. & Duan, J. 2007, 'The ageing effect on the bioaccessibility and fractionation of arsenic in soils from China', *Chemosphere*, vol. 66, no. 7, pp. 1183-1190.

Teasdale, P. R., Hayward, S. & Davison, W. 1999, 'In situ, high-resolution measurement of dissolved sulfide using diffusive gradients in thin films with computerimaging densitometry peter', *Analytical Chemistry*, vol. 71, no. 11, pp. 2186-2191.

Telford, K., Maher, W., Krikowa, F., Foster, S., Ellwood, M. J., Ashley, P. M., Lockwood, P. V. & Wilson, S. C. 2009, 'Bioaccumulation of antimony and arsenic in a highly contaminated stream adjacent to the Hillgrove Mine, NSW, Australia', *Environmental Chemistry*, vol. 6, no. 2, pp. 133-143.

Thanabalasingam, P. & Pickering, W. F. 1986, 'Arsenic sorption by humic acids', *Environmental Pollution. Series B, Chemical and Physical*, vol. 12, no. 3, pp. 233-246.

Thanabalasingam, P. & Pickering, W. F. 1990, 'Specific sorption of antimony (III) by the hydrous oxides of Mn, Fe, and Al', *Water, Air, and Soil Pollution*, vol. 49, no. 1-2, pp. 175-185.

Thibault, D. H. & Sheppard, M. I. 1992, 'A disposable system for soil pore-water extraction by centrifugation', *Communications in Soil Science and Plant Analysis*, vol. 23, no. 13-14, pp. 1629-1641.

Tiensing, T., Preston, S., Strachan, N. & Paton, G. I. 2001, 'Soil solution extraction techniques for microbial ecotoxicity testing: A comparative evaluation', *Journal of Environmental Monitoring*, vol. 3, no. 1, pp. 91-96.

Tighe, M., Ashley, P., Lockwood, P. & Wilson, S. 2005a, 'Soil, water, and pasture enrichment of antimony and arsenic within a coastal floodplain system', *Science of the Total Environment*, vol. 347, no. 1-3, pp. 175-186.

Tighe, M. & Lockwood, P. 2007, 'Importance of noncrystalline hydroxide phases in sequential extractions to fractionate antimony in acid soils', *Communications in Soil Science and Plant Analysis*, vol. 38, no. 11-12, pp. 1487-1501.

Tighe, M., Lockwood, P. & Wilson, S. 2005b, 'Adsorption of antimony(v) by floodplain soils, amorphous iron(III) hydroxide and humic acid', *Journal of Environmental Monitoring*, vol. 7, no. 12, pp. 1177-1185.

Tisarum, R., Chen, Y., Dong, X., Lessl, J. T. & Ma, L. Q. 2015, 'Uptake of antimonite and antimonate by arsenic hyperaccumulator Pteris vittata: Effects of chemical analogs and transporter inhibitor', *Environmental Pollution*, vol. 206, pp. 49-55.

Tisarum, R., Lessl, J. T., Dong, X., De Oliveira, L. M., Rathinasabapathi, B. & Ma, L. Q. 2014, 'Antimony uptake, efflux and speciation in arsenic hyperaccumulator Pteris vittata', *Environmental Pollution*, vol. 186, pp. 110-114.

Tlustoš, P., Balík, J., Száková, J. & Pavlíková, D. 1998, 'The accumulation of arsenic in radish biomass when different forms of As were applied in the soil', *Rostlinna Vyroba*, vol. 44, no. 1, pp. 7-13.

Tlustoš, P., Goessler, W., Száková, J. & Balík, J. 2002, 'Arsenic compounds in leaves and roots of radish grown in soil treated by arsenite, arsenate and dimethylarsinic acid', *Applied Organometallic Chemistry*, vol. 16, no. 4, pp. 216-220.

Tripathi, R. D., Srivastava, S., Mishra, S., Singh, N., Tuli, R., Gupta, D. K. & Maathuis,F. J. M. 2007, 'Arsenic hazards: strategies for tolerance and remediation by plants',*Trends in Biotechnology*, vol. 25, no. 4, pp. 158-165.

Tripathi, R. D., Tripathi, P., Dwivedi, S., Kumar, A., Mishra, A., Chauhan, P. S., Norton, G. J. & Nautiyal, C. S. 2014, 'Roles for root iron plaque in sequestration and uptake of heavy metals and metalloids in aquatic and wetland plants', *Metallomics*, vol. 6, no. 10, pp. 1789-1800.

Tschan, M., Robinson, B., Johnson, C. A., Bürgi, A. & Schulin, R. 2010, 'Antimony uptake and toxicity in sunflower and maize growing in SbIII and SbV contaminated soil', *Plant and Soil*, vol. 334, no. 1, pp. 235-245.

Tschan, M., Robinson, B. & Schulin, R. 2008, 'Antimony uptake by Zea mays (L.) and Helianthus annuus (L.) from nutrient solution', *Environmental Geochemistry and Health*, vol. 30, no. 2, pp. 187-191.

Tschan, M., Robinson, B. H., Nodari, M. & Schulin, R. 2009a, 'Antimony uptake by different plant species from nutrient solution, agar and soil', *Environmental Chemistry*, vol. 6, no. 2, pp. 144-152.

Tschan, M., Robinson, B. H. & Schulin, R. 2009b, 'Antimony in the soil - Plant system - A review', *Environmental Chemistry*, vol. 6, no. 2, pp. 106-115.

Tu, S. & Ma, L. Q. 2003, 'Interactive effects of pH, arsenic and phosphorus on uptake of As and P and growth of the arsenic hyperaccumulator Pteris vittata L. under hydroponic conditions', *Environmental and Experimental Botany*, vol. 50, no. 3, pp. 243-251.

Tufano, K. J., Reyes, C., Saltikov, C. W. & Fendorf, S. 2008, 'Reductive processes controlling arsenic retention: Revealing the relative importance of iron and arsenic reduction', *Environmental Science and Technology*, vol. 42, no. 22, pp. 8283-8289.

Turpeinen, R., Pantsar-Kallio, M., Häggblom, M. & Kairesalo, T. 1999, 'Influence of microbes on the mobilization, toxicity and biomethylation of arsenic in soil', *Science of the Total Environment*, vol. 236, no. 1-3, pp. 173-180.

Uddh-Söderberg, T. E., Gunnarsson, S. J., Hogmalm, K. J., Lindegård, M. I. B. G. & Augustsson, A. L. M. 2015, 'An assessment of health risks associated with arsenic exposure via consumption of homegrown vegetables near contaminated glassworks sites', *Science of the Total Environment*, vol. 536, pp. 189-197.

Uroic, M. K., Salaün, P., Raab, A. & Feldmann, J. 2012, 'Arsenate impact on the metabolite profile, production, and arsenic loading of xylem sap in cucumbers (Cucumis sativus L.)', *Frontiers in Physiology*, vol. 3 APR.

USEPA 1979, Water Related Fate of the 129 Priority Pollutants, Washington, DC, USA.

USEPA 1991, Integrated risk information system, Antimony, (CASRN 7440-36-0).

USEPA 1998, Integrated Risk Information System (IRIS), Arsenic, inorganic (CASRN 7440-38-2).

USEPA 2011, *Exposure Factors Handbook 2011 Edition (Final Report)*, Washington, DC, EPA/600/R-09/052F, 2011.

Vaculík, M., Jurkovič, A., Matejkovič, P., Molnárová, M. & Lux, A. 2013, 'Potential risk of arsenic and antimony accumulation by medicinal plants naturally growing on old mining sites', *Water, Air, and Soil Pollution*, vol. 224, no. 5.

Vaculík, M., Mrázová, A. & Lux, A. 2015, 'Antimony (SbIII) reduces growth, declines photosynthesis, and modifies leaf tissue anatomy in sunflower (Helianthus annuus L.)', *Environmental Science and Pollution Research*, vol. 22, no. 23, pp. 18699-18706.

Vamerali, T., Bandiera, M. & Mosca, G. 2010, 'Field crops for phytoremediation of metal-contaminated land. A review', *Environmental Chemistry Letters*, vol. 8, no. 1, pp. 1-17.

Van Herreweghe, S., Swennen, R., Vandecasteele, C. & Cappuyns, V. 2003, 'Solid phase speciation of arsenic by sequential extraction in standard reference materials and industrially contaminated soil samples', *Environmental Pollution*, vol. 122, no. 3, pp. 323-342.

Vazquez Reina, S., Esteban, E. & Goldsbrough, P. 2005, 'Arsenate-induced phytochelatins in white lupin: influence of phosphate status', *Physiologia Plantarum*, vol. 124, no. 1, pp. 41-49.

Verbruggen, N., Hermans, C. & Schat, H. 2009, 'Mechanisms to cope with arsenic or cadmium excess in plants', *Current Opinion in Plant Biology*, vol. 12, no. 3, pp. 364-372.

Visoottiviseth, P., Francesconi, K. & Sridokchan, W. 2002, 'The potential of Thai indigenous plant species for the phytoremediation of arsenic contaminated land', *Environmental Pollution*, vol. 118, no. 3, pp. 453-461.

Vithanage, M., Dabrowska, B. B., Mukherjee, A. B., Sandhi, A. & Bhattacharya, P. 2012, 'Arsenic uptake by plants and possible phytoremediation applications: A brief overview', *Environmental Chemistry Letters*, vol. 10, no. 3, pp. 217-224.

Wan, X. M., Tandy, S., Hockmann, K. & Schulin, R. 2013a, 'Changes in Sb speciation with waterlogging of shooting range soils and impacts on plant uptake', *Environmental Pollution*, vol. 172, pp. 53-60.

Wan, X. M., Tandy, S., Hockmann, K. & Schulin, R. 2013b, 'Effects of waterlogging on the solubility and redox state of Sb in a shooting range soil and its uptake by grasses: A tank experiment', *Plant and Soil*, vol. 371, no. 1-2, pp. 155-166.

Wanat, N., Joussein, E., Soubrand, M. & Lenain, J. F. 2014, 'Arsenic (As), antimony (Sb), and lead (Pb) availability from Au-mine Technosols: A case study of transfer to natural vegetation cover in temperate climates', *Environmental Geochemistry and Health*, vol. 36, no. 4, pp. 783-795.

Wang, J., Bai, L., Zeng, X., Su, S., Wang, Y. & Wu, C. 2014, 'Assessment of arsenic availability in soils using the diffusive gradients in thin films (DGT) technique - A comparison study of DGT and classic extraction methods', *Environmental Sciences: Processes and Impacts*, vol. 16, no. 10, pp. 2355-2361.

Wang, J., Zhao, F. J., Meharg, A. A., Raab, A., Feldmann, J. & McGrath, S. P. 2002, 'Mechanisms of arsenic hyperaccumulation in Pteris vittata. Uptake kinetics, interactions with phosphate, and arsenic speciation', *Plant Physiology*, vol. 130, no. 3, pp. 1552-1561.

Wang, L. & Duan, G. 2009, 'Effect of external and internal phosphate status on arsenic toxicity and accumulation in rice seedlings', *Journal of Environmental Sciences*, vol. 21, no. 3, pp. 346-351.

Wang, S. & Mulligan, C. N. 2006, 'Occurrence of arsenic contamination in Canada: Sources, behavior and distribution', *Science of the Total Environment*, vol. 366, no. 2-3, pp. 701-721.

Wang, X., Sato, T., Xing, B. & Tao, S. 2005, 'Health risks of heavy metals to the general public in Tianjin, China via consumption of vegetables and fish', *Science of the Total Environment*, vol. 350, no. 1, pp. 28-37.

Wang, Y., Ding, S., Shi, L., Gong, M., Xu, S. & Zhang, C. 2017, 'Simultaneous measurements of cations and anions using diffusive gradients in thin films with a ZrO-Chelex mixed binding layer', *Analytica Chimica Acta*, vol. 972, pp. 1-11.

Wang, Y., Zeng, X., Lu, Y., Su, S., Bai, L., Li, L. & Wu, C. 2015, 'Effect of aging on the bioavailability and fractionation of arsenic in soils derived from five parent materials in a red soil region of Southern China', *Environmental Pollution*, vol. 207, pp. 79-87.

Warnken, J., Ohlsson, R., Welsh, D. T., Teasdale, P. R., Chelsky, A. & Bennett, W. W. 2017, 'Antimony and arsenic exhibit contrasting spatial distributions in the sediment and vegetation of a contaminated wetland', *Chemosphere*, vol. 180, pp. 388-395.

Warren, G. P. & Alloway, B. J. 2003, 'Reduction of arsenic uptake by lettuce with ferrous sulfate applied to contaminated soil', *Journal of Environmental Quality*, vol. 32, no. 3, pp. 767-772.

Warren, G. P., Alloway, B. J., Lepp, N. W., Singh, B., Bochereau, F. J. M. & Penny, C. 2003, 'Field trials to assess the uptake of arsenic by vegetables from contaminated soils and soil remediation with iron oxides', *Science of the Total Environment*, vol. 311, no. 1-3, pp. 19-33.

Webb, S. M., Gaillard, J.-F., Ma, L. Q. & Tu, C. 2003, 'XAS Speciation of Arsenic in a Hyper-Accumulating Fern', *Environmental science & technology*, vol. 37, no. 4, pp. 754-760.

Wei, C., Deng, Q., Wu, F., Fu, Z. & Xu, L. 2011, 'Arsenic, antimony, and bismuth uptake and accumulation by plants in an old antimony mine, China', *Biological Trace Element Research*, vol. 144, no. 1-3, pp. 1150-1158.

Wenzel, W. W. 2009, 'Rhizosphere processes and management in plant-assisted bioremediation (phytoremediation) of soils', *Plant and Soil*, vol. 321, no. 1-2, pp. 385-408.

Wenzel, W. W., Kirchbaumer, N., Prohaska, T., Stingeder, G., Lombi, E. & Adriano, D.C. 2001, 'Arsenic fractionation in soils using an improved sequential extraction procedure', *Analytica Chimica Acta*, vol. 436, no. 2, pp. 309-323.

Williams, P. N., Santner, J., Larsen, M., Lehto, N. J., Oburger, E., Wenzel, W., Glud, R. N., Davison, W. & Zhang, H. 2014, 'Localized flux maxima of arsenic, lead, and iron around root apices in flooded lowland rice', *Environmental Science and Technology*, vol. 48, no. 15, pp. 8498-8506.

Williams, P. N., Villada, A., Deacon, C., Raab, A., Figuerola, J., Green, A. J., Feldmann, J. & Meharg, A. A. 2007, 'Greatly enhanced arsenic shoot assimilation in rice leads to elevated grain levels compared to wheat and barley', *Environmental Science and Technology*, vol. 41, no. 19, pp. 6854-6859.

Williams, P. N., Zhang, H., Davison, W., Meharg, A. A., Hossain, M., Norton, G. J., Brammer, H. & Islam, M. R. 2011, 'Organic matter-solid phase interactions are critical for predicting arsenic release and plant uptake in Bangladesh paddy soils', *Environmental Science and Technology*, vol. 45, no. 14, pp. 6080-6087.

Wilson, N. J., Craw, D. & Hunter, K. 2004, 'Antimony distribution and environmental mobility at an historic antimony smelter site, New Zealand', *Environmental Pollution*, vol. 129, no. 2, pp. 257-266.

Wilson, S. C., Leech, C. D., Butler, L., Lisle, L., Ashley, P. M. & Lockwood, P. V. 2013, 'Effects of nutrient and lime additions in mine site rehabilitation strategies on the accumulation of antimony and arsenic by native Australian plants', *Journal of Hazardous Materials*, vol. 261, pp. 801-807.

Wilson, S. C., Lockwood, P. V., Ashley, P. M. & Tighe, M. 2010, 'The chemistry and behaviour of antimony in the soil environment with comparisons to arsenic: A critical review', *Environmental Pollution*, vol. 158, no. 5, pp. 1169-1181.

Wilson, S. C., Tighe, M., Paterson, E. & Ashley, P. M. 2014, 'Food crop accumulation and bioavailability assessment for antimony (Sb) compared with arsenic (As) in contaminated soils', *Environmental science and pollution research international*, vol. 21, no. 20, pp. 11671-11681.

Winship, K. A. 1987, 'Toxicity of antimony and its compounds', *Adverse Drug Reactions and Acute Poisoning Review*, vol. 6, no. 2, pp. 67-90.

Wu, C., Huang, L., Xue, S. G., Pan, W. S., Zou, Q., Hartley, W. & Wong, M. H. 2017, 'Oxic and anoxic conditions affect arsenic (As) accumulation and arsenite transporter expression in rice', *Chemosphere*, vol. 168, pp. 969-975.

Wu, C., Ye, Z., Shu, W., Zhu, Y. & Wong, M. 2011a, 'Arsenic accumulation and speciation in rice are affected by root aeration and variation of genotypes', *Journal of Experimental Botany*, vol. 62, no. 8, pp. 2889-2898.

Wu, Z., He, M. & Lin, C. 2011b, 'In situ measurements of concentrations of Cd, Co, Fe and Mn in estuarine porewater using DGT', *Environmental Pollution*, vol. 159, no. 5, pp. 1123-1128.

Xi, J. & He, M. 2013, 'Removal of Sb(III) and Sb(V) from aqueous media by goethite', *Water Quality Research Journal of Canada*, vol. 48, no. 3, pp. 223-231.

Xi, J., He, M. & Kong, L. 2016, 'Adsorption of antimony on kaolinite as a function of time, pH, HA and competitive anions', *Environmental earth sciences*, vol. 75, no. 2, pp. 1-7.

Xu, H., Allard, B. & Grimvall, A. 1988, 'Influence of pH and organic substance on the adsorption of As(V) on geologic materials', *Water, Air, and Soil Pollution*, vol. 40, no. 3, pp. 293-305.

Xu, H., Allard, B. & Grimvall, A. 1991, 'Effects of acidification and natural organic materials on the mobility of arsenic in the environment', *Water, Air, and Soil Pollution*, vol. 57, no. 1, pp. 269-278.

Xu, X. Y., McGrath, S. P., Meharg, A. A. & Zhao, F. J. 2008, 'Growing rice aerobically markedly decreases arsenic accumulation', *Environmental Science and Technology*, vol. 42, no. 15, pp. 5574-5579.

Yamaguchi, N., Nakamura, T., Dong, D., Takahashi, Y., Amachi, S. & Makino, T. 2011, 'Arsenic release from flooded paddy soils is influenced by speciation, Eh, pH, and iron dissolution', *Chemosphere*, vol. 83, no. 7, pp. 925-932.

Yao, L., Li, G., Dang, Z., Yang, B., He, Z. & Zhou, C. 2010, 'Uptake and transport of Roxarsone and its metabolites in water spinach as affected by phosphate supply', *Environmental Toxicology and Chemistry*, vol. 29, no. 4, pp. 947-951.

Yao, L. X., Li, G. L., Dang, Z., He, Z. H., Zhou, C. M. & Yang, B. M. 2009, 'Arsenic uptake by two vegetables grown in two soils amended with As-bearing animal manures', *Journal of Hazardous Materials*, vol. 164, no. 2-3, pp. 904-910.

Young, S. D., Zhang, H., Tye, A. M., Maxted, A., Thums, C. & Thornton, I. 2005, 'Characterizing the availability of metals in contaminated soils. I. The solid phase: Sequential extraction and isotopic dilution', *Soil Use and Management*, vol. 21, no. SUPPL., pp. 450-458.

Yu, Y., Zhang, S., Huang, H., Luo, L. & Wen, B. 2009, 'Arsenic accumulation and speciation in maize as affected by inoculation with arbuscular mycorrhizal fungus glomus mosseae', *Journal of Agricultural and Food Chemistry*, vol. 57, no. 9, pp. 3695-3701.

Zhang, H. & Davison, W. 1995, 'Performance characteristics of diffusion gradients in thin films for the in situ measurement of trace metals in aqueous solution', *Analytical Chemistry*, vol. 67, no. 19, pp. 3391-3400.

Zhang, H. & Davison, W. 2000, 'Direct in situ measurements of labile inorganic and organically bound metal species in synthetic solutions and natural waters using diffusive gradients in thin films', *Analytical Chemistry*, vol. 72, no. 18, pp. 4447-4457.

Zhang, H. & Davison, W. 2006, 'Predicting metal uptake by plants using the DGT technique', in I. Twardowska, et al. (eds), *Soil and Water Pollution Monitoring, Protection and Remediation*, Springer Netherlands, Dordrecht, pp. 187-197.

Zhang, H. & Davison, W. 2015, 'Use of diffusive gradients in thin-films for studies of chemical speciation and bioavailability', *Environmental Chemistry*, vol. 12, no. 2, pp. 85-101.

Zhang, H., Davison, W., Gadi, R. & Kobayashi, T. 1998a, 'In situ measurement of dissolved phosphorus in natural waters using DGT', *Analytica Chimica Acta*, vol. 370, no. 1, pp. 29-38.

Zhang, H., Davison, W., Knight, B. & McGrath, S. 1998b, 'In situ measurements of solution concentrations and fluxes of trace metals in soils using DGT', *Environmental Science and Technology*, vol. 32, no. 5, pp. 704-710.

Zhang, H., Davison, W., Miller, S. & Tych, W. 1995, 'In situ high resolution measurements of fluxes of Ni, Cu, Fe, and Mn and concentrations of Zn and Cd in porewaters by DGT', *Geochimica et Cosmochimica Acta*, vol. 59, no. 20, pp. 4181-4192.

Zhang, H., Davison, W., Mortimer, R. J. G., Krom, M. D., Hayes, P. J. & Davies, I. M. 2002, 'Localised remobilization of metals in a marine sediment', *Science of the Total Environment*, vol. 296, no. 1-3, pp. 175-187.

Zhang, H., Davison, W., Tye, A. M., Crout, N. M. J. & Young, S. D. 2006, 'Kinetics of zinc and cadmium release in freshly contaminated soils', *Environmental Toxicology and Chemistry*, vol. 25, no. 3, pp. 664-670.

Zhang, H., Lombi, E., Smolders, E. & McGrath, S. 2004, 'Kinetics of Zn release in soils and prediction of Zn concentration in plants using diffusive gradients in thin films', *Environmental Science and Technology*, vol. 38, no. 13, pp. 3608-3613.

Zhang, H. & Selim, H. M. 2008, 'Reaction and transport of arsenic in soils: Equilibrium and kinetic modeling', *Advances in agronomy*, vol. 98, pp. 45-115.

Zhang, H., Zhao, F. J., Sun, B., Davison, W. & McGrath, S. P. 2001, 'A new method to measure effective soil solution concentration predicts copper availability to plants', *Environmental Science and Technology*, vol. 35, no. 12, pp. 2602-2607.

Zhang, J. & Duan, G. L. 2008, 'Genotypic difference in arsenic and cadmium accumulation by rice seedlings grown in hydroponics', *Journal of Plant Nutrition*, vol. 31, no. 12, pp. 2168-2182.

Zhao, F. J., Ma, J. F., Meharg, A. A. & McGrath, S. P. 2009, 'Arsenic uptake and metabolism in plants', *New Phytologist*, vol. 181, no. 4, pp. 777-794.

Zhao, F. J., McGrath, S. P. & Meharg, A. A. 2010, 'Arsenic as a food chain contaminant: Mechanisms of plant uptake and metabolism and mitigation strategies', *Annual Review of Plant Biology*, vol. 61, pp. 535-559.

## Appendix A

## **Supplementary Information for Chapter 3**

	As concentrations in different fractions (mg/kg)					Sb concentrations in different fractions (mg/kg)				
Soil	Non- specifically sorbed (A)	Specifically sorbed (B)	Amorphous Fe and Al oxides (C)	Crystalline Fe and Al oxides (D)	Residual (E)	Non- specifically sorbed (A)	Specifically sorbed (B)	Amorphous Fe and Al oxides (C)	Crystalline Fe and Al oxides (D)	Residual (E)
<b>S</b> 1	$0.048 \pm 0.004$	$0.87\pm0.01$	$4.4 \pm 0.8$	$4.7 \pm 0.3$	$3.3\pm0.9$	$0.056\pm0.001$	$0.084 \pm 0.001$	$1.4 \pm 0.1$	$2.3\pm0.2$	$7.9\pm0.7$
S2	$0.076\pm0.003$	$1.43\pm0.04$	$8.4\pm0.7$	$7 \pm 1$	8 ± 3	$0.102\pm0.003$	$0.193 \pm 0.004$	$4.14\pm0.06$	$4.5\pm0.4$	$17\pm2$
<b>S</b> 3	$0.096 \pm 0.004$	$1.76\pm0.09$	$90 \pm 1$	$14 \pm 1$	$5\pm 2$	$0.152\pm0.003$	$0.25\pm0.02$	$4.1\pm0.6$	$9\pm1$	$25\pm2$
<b>S</b> 4	$0.127\pm0.003$	$2.39\pm0.06$	$14.6\pm0.6$	$10 \pm 1$	$12 \pm 3$	$0.212\pm0.001$	$0.379 \pm 0.008$	$7.7\pm0.4$	$8.0\pm0.3$	$39\pm5$
S5	$0.150\pm0.003$	$2.81\pm0.06$	$19\pm1$	$13.1\pm0.4$	$14 \pm 2$	$0.261\pm0.002$	$0.466\pm0.009$	$9.75\pm0.09$	$10.6\pm0.9$	$50\pm2$
<b>S</b> 6	$0.167\pm0.007$	$3.14\pm0.06$	$160 \pm 2$	$19.8\pm0.8$	$12 \pm 2$	$0.293\pm0.008$	$0.539 \pm 0.009$	$7.8\pm0.8$	$13.7\pm0.5$	$58\pm4$
<b>S</b> 7	$0.207\pm0.005$	$3.76\pm0.01$	$27.9\pm0.8$	$20.3\pm0.9$	$25 \pm 3$	$0.212\pm0.001$	$0.641 \pm 0.008$	$14.4\pm0.8$	$15\pm2$	$82\pm10$
<b>S</b> 8	$0.230\pm0.008$	$4.00\pm0.06$	$30.6\pm0.6$	$19.70\pm0.08$	$25 \pm 3$	$0.261\pm0.002$	$0.70\pm0.03$	$18\pm2$	$16.5\pm0.4$	$92\pm 6$
<b>S</b> 9	$0.25\pm0.01$	$4.96 \pm 0.08$	$25\pm5$	$40 \pm 3$	$23 \pm 6$	$0.482\pm0.007$	$0.86 \pm 0.01$	$12 \pm 1$	$25 \pm 2$	$110\pm4$
S10	$0.49\pm0.02$	$7.58\pm0.06$	$53.4\pm0.8$	$36.2\pm0.8$	$87 \pm 10$	$0.78\pm0.03$	$1.36\pm0.03$	$28 \pm 1$	$28.8\pm0.9$	$220\pm10$
S11	$0.55\pm0.03$	$9.2\pm0.1$	$57.3\pm0.8$	$63 \pm 9$	$77\pm20$	$0.99\pm0.04$	$1.61\pm0.03$	$28\pm2$	$43\pm 6$	$290\pm30$
S12	$1.49\pm0.05$	$17.0\pm0.4$	$115 \pm 2$	$110 \pm 5$	$160 \pm 20$	$1.75\pm0.01$	$3.0 \pm 0.1$	$45\pm7$	$79\pm3$	$590\pm30$

**Table A.1** As and Sb concentrations in different fractions in soils (mg/kg, dry mass, mean  $\pm$  SE, n = 3).



**Figure A.1** Arsenic (a) and antimony (b) concentrations in *R. sativus* roots and shoots (mg/kg dry mass, mean  $\pm$  SE, n = 4).

## Appendix B

**Supplementary Information for Chapter 4** 

	As concentrations in different fractions (mg/kg)									
Soil	Non-specifically sorbed	Specifically sorbed	Amorphous Fe and Al oxides	Crystalline Fe and Al oxides	Residual					
Q1	$0.228 \pm 0.004$	$1.76\pm0.03$	$9.62\pm0.06$	$4.0 \pm 0.4$	$1.7\pm0.5$					
Q2	$0.34\pm0.02$	$2.8\pm0.1$	$15.8\pm0.5$	$4.5\pm0.2$	$2 \pm 1$					
Q3	$0.68\pm0.03$	$4.2\pm0.2$	$23 \pm 1$	$5.8\pm0.7$	$2\pm 2$					
Q4	$0.88\pm0.02$	$5.1 \pm 0.1$	$34.3\pm0.9$	$7.7\pm0.9$	$3 \pm 1$					
Q5	$1.183\pm0.005$	$6.78\pm0.03$	$41.1\pm0.6$	$8.8\pm0.5$	$4.1\pm0.9$					
Q6	$1.40\pm0.02$	$7.9\pm0.2$	$47 \pm 1$	$8.1\pm0.1$	$3 \pm 1$					
Q7	$1.58\pm0.03$	$9.37\pm0.06$	$55 \pm 1$	$10.6\pm0.8$	$4\pm 2$					
Q8	$1.73\pm0.04$	$10.11\pm0.06$	$61.3\pm0.8$	$10.2\pm0.4$	$8\pm2$					
Q9	$2.03\pm0.05$	$13.0\pm0.6$	$77.3\pm0.8$	$14.2\pm0.6$	$9\pm1$					
Q10	$3.013 \pm 0.004$	$19.4\pm0.2$	$120 \pm 1$	$20.5\pm0.9$	$9\pm4$					
Q11	$3.55\pm0.01$	$27 \pm 1$	$155.6\pm0.7$	$28.0\pm0.3$	$22\pm9$					
Q12	$6.7 \pm 0.1$	$55.2\pm0.6$	$370 \pm 10$	$71 \pm 1$	$35 \pm 10$					

**Table B.1** As concentrations in individual different fractions in cattle dip soils (mg/kg dry mass, mean  $\pm$  SE, n = 3).

	As concentrations in different fractions (mg/kg)					Sb concentrations in different fractions (mg/kg)				
Soil	Non-specifically sorbed	Specifically sorbed	Amorphous Fe and Al oxides	Crystalline Fe and Al oxides	Residual	Non-specifically sorbed	Specifically sorbed	Amorphous Fe and Al oxides	Crystalline Fe and Al oxides	Residual
<b>S</b> 1	$0.041\pm0.002$	$0.822\pm0.005$	$4.6\pm0.3$	$4.9\pm0.2$	$2.7\pm0.4$	$0.051 \pm 0.001$	$0.074\pm0.001$	$1.52\pm0.01$	$2.42\pm0.06$	$6.9\pm0.6$
S2	$0.0476 \pm 0.0008$	$1.01\pm0.07$	$6.7\pm0.5$	$5.7\pm0.3$	$5\pm1$	$0.088 \pm 0.002$	$0.167\pm0.006$	$4.17\pm0.04$	$4.72\pm0.02$	$14.1\pm0.6$
<b>S</b> 3	$0.103\pm0.002$	$2.12\pm0.03$	$12.3\pm0.7$	$19 \pm 1$	$5\pm3$	$0.158 \pm 0.004$	$0.261\pm0.004$	$5.3\pm0.2$	$10.9\pm0.2$	$26\pm7$
<b>S</b> 4	$0.119\pm0.002$	$2.53\pm0.06$	$17.4\pm0.6$	$12.6\pm0.4$	$11.5\pm0.8$	$0.198 \pm 0.009$	$0.350\pm0.009$	$8.5\pm0.3$	$9.26\pm0.04$	$36\pm7$
<b>S</b> 5	$0.135\pm0.002$	$2.80\pm0.05$	$21.5\pm0.7$	$14.8\pm0.4$	$13 \pm 6$	$0.240\pm0.004$	$0.426\pm0.006$	$11 \pm 1$	$12.1\pm0.4$	$46\pm2$
<b>S</b> 6	$0.169\pm0.002$	$3.53\pm0.09$	$20\pm1$	$25 \pm 1$	$11 \pm 10$	$0.261\pm0.004$	$0.477\pm0.007$	$8.7\pm0.3$	$14.8\pm0.7$	$51\pm3$
<b>S</b> 7	$0.175\pm0.002$	$3.67\pm0.04$	$30.9\pm0.7$	$22.7\pm0.8$	$22\pm9$	$0.212\pm0.005$	$0.67\pm0.01$	$18.4\pm0.6$	$20.2\pm0.5$	$86\pm20$
<b>S</b> 8	$0.198 \pm 0.006$	$3.93\pm0.05$	$34 \pm 1$	$22.2\pm0.1$	$22\pm2$	$0.246\pm0.009$	$0.68\pm0.006$	$21 \pm 1$	$20.6\pm0.7$	$91\pm20$
<b>S</b> 9	$0.216\pm0.004$	$4.86\pm0.04$	$28\pm2$	$40.7\pm0.7$	$20\pm5$	$0.57\pm0.03$	$1.00\pm0.03$	$18.24\pm0.09$	$36\pm2$	$130\pm20$
S10	$0.298 \pm 0.008$	$5.41\pm0.07$	$45\pm2$	$31 \pm 2$	$61 \pm 3$	$0.66\pm0.03$	$1.14\pm0.05$	$30.4\pm0.9$	$33 \pm 1$	$190\pm4$
<b>S</b> 11	$0.491 \pm 0.005$	$9.3\pm0.1$	$68 \pm 1$	$72 \pm 2$	$73 \pm 30$	$1.00\pm0.04$	$1.59\pm0.02$	$37 \pm 2$	$57 \pm 2$	$300\pm20$
S12	$1.36\pm0.02$	$16.7\pm0.2$	$130 \pm 4$	$120 \pm 4$	$150\pm30$	$1.51\pm0.05$	$2.6\pm0.1$	$55 \pm 4$	$92 \pm 3$	$520\pm50$

**Table B.2** As and Sb concentrations in individual different fractions in mining soils (mg/kg dry mass, mean  $\pm$  SE, n = 3).

Soil	As - Cattle dip		Soil	As - Mining		Sb - Mi	Sb - Mining	
	BAF	TF		BAF	TF	BAF	TF	
Q1	$0.0085 \pm 0.0003$	$0.80\pm0.03$	S1	$0.010\pm0.001$	$0.77\pm0.06$	$0.0030 \pm 0.0004$	$1.0 \pm 0.2$	
Q2	$0.011\pm0.003$	$0.5 \pm 0.1$	S2	$0.011 \pm 0.002$	$0.5 \pm 0.1$	$0.0020 \pm 0.0002$	$1.0 \pm 0.3$	
Q3	$0.0099 \pm 0.0009$	$0.57\pm0.05$	<b>S</b> 3	$0.0054 \pm 0.0001$	$0.5 \pm 0.1$	$0.0011 \pm 0.0001$	$1.7 \pm 0.6$	
Q4	$0.013\pm0.002$	$0.37\pm0.06$	<b>S</b> 4	$0.0060 \pm 0.0006$	$0.37\pm0.02$	$0.0015 \pm 0.0002$	$0.9 \pm 0.1$	
Q5	$0.0082 \pm 0.0005$	$0.49\pm0.05$	S5	$0.006\pm0.001$	$0.46 \pm 0.1$	$0.0013 \pm 0.0002$	$1.2 \pm 0.3$	
Q6	$0.009\pm0.001$	$0.43\pm0.07$	<b>S</b> 6	$0.0051 \pm 0.0004$	$0.43\pm0.07$	$0.0014 \pm 0.0004$	$1.1 \pm 0.2$	
Q7	$0.0083 \pm 0.0005$	$0.8 \pm 0.3$	<b>S</b> 7	$0.0054 \pm 0.0007$	$0.39\pm0.02$	$0.0010 \pm 0.0003$	$1.2 \pm 0.3$	
Q8	$0.0079 \pm 0.0006$	$0.6 \pm 0.1$	<b>S</b> 8	$0.0047 \pm 0.0003$	$0.35\pm0.01$	$0.0008 \pm 0.0001$	$1.0 \pm 0.03$	
Q9	$0.009\pm0.001$	$0.5 \pm 0.1$	<b>S</b> 9	$0.0046 \pm 0.0003$	$0.37\pm0.01$	$0.0010 \pm 0.0004$	$0.9 \pm 0.2$	
Q10	$0.0068 \pm 0.0004$	$0.53\pm0.02$	S10	$0.0037 \pm 0.0002$	$0.46\pm0.06$	$0.0008 \pm 0.0003$	$1.2 \pm 0.3$	
Q11	$0.008\pm0.002$	$0.62\pm0.07$	S11	$0.0039 \pm 0.0004$	$0.46\pm0.06$	$0.0009 \pm 0.0002$	$0.9 \pm 0.1$	
Q12	$0.009 \pm 0.001$	$0.9 \pm 0.2$	S12	$0.0033 \pm 0.0003$	$0.44\pm0.08$	$0.0007 \pm 0.0001$	$1.4\pm0.2$	

**Table B.3** Bioaccumulation factor (BAF) and translocation factor (TF) for As and Sb in white icicle radish (mean  $\pm$  SE, n = 4).


**Figure B.1** The relationship between DGT-labile ( $C_{DGT}$ ) As and Sb after 24-h deployment and concentrations in extracted soil solutions (mean  $\pm$  SE, n = 3). Dotted line represents a 1:1 gradient, regression set through zero. Note that total Sb soil concentrations were higher than total As concentrations in the same soils. Arsenic in bioassay cattle dip soils (A), As in bioassay mining soils (B), and Sb in bioassay mining soils (C).



**Figure B.2** Arsenic and antimony concentrations in white icicle radish roots and shoots (mg/kg dry mass, mean  $\pm$  SE, n = 4). Arsenic in cattle dip soils (A), As in mining soils (B), and Sb in mining soils (C).

# Appendix C

# **Supplementary Information for Chapter 5**

Soil	Treatment	As concentrations in different fractions (mg/kg)					Sb concentrations in different fractions (mg/kg)				
		Non- specifically sorbed	Specifically sorbed	Amorphous Fe and Al oxides	Crystalline Fe and Al oxides	Residual	Non- specifically sorbed	Specifically sorbed	Amorphous Fe and Al oxides	Crystalline Fe and Al oxides	Residual
L	<b>S</b> 1	$0.059\pm0.007$	$3.41\pm0.05$	$18.38\pm0.2$	$14 \pm 1$	$7 \pm 1$	$0.130\pm0.005$	$0.195\pm0.001$	$5.8\pm0.2$	$14.6\pm0.8$	$31\pm 6$
	S2	$0.38 \pm 0.01$	$12.72\pm0.06$	$67\pm2$	$49\pm8$	33	$0.657\pm0.007$	$1.02\pm0.01$	$20.7\pm0.6$	$55\pm7$	$124\pm30$
	<b>S</b> 3	$0.541\pm0.002$	$25.8\pm0.2$	$139 \pm 3$	$102 \pm 4$	1	$1.30\pm0.02$	$2.47\pm0.01$	$46 \pm 1$	$109 \pm 3$	$169\pm30$
	S4	$2.06\pm0.03$	$48.9\pm0.5$	$294\pm10$	$220\pm2$	110	$3.34\pm0.02$	$5.21\pm0.08$	$94 \pm 2$	$240\pm20$	$530\pm200$
	S5	$8.3\pm0.1$	$113.6\pm0.3$	$627\pm7$	$535\pm10$	$130\pm50$	$7.93 \pm 0.05$	$12.6\pm0.1$	$270 \pm 3$	$620\pm20$	$1360\pm100$
М	S6	$0.169 \pm 0.003$	$3.95\pm0.03$	$32.0\pm0.9$	$27\pm 6$	3	$0.176\pm0.005$	$0.231 \pm 0.007$	$9.8\pm0.4$	$27\pm 6$	$54 \pm 6$
	<b>S</b> 7	$0.54\pm0.02$	$12.2\pm0.2$	$76\pm5$	$83 \pm 6$	5	$0.48 \pm 0.01$	$0.82\pm0.02$	$23.2\pm0.6$	$73\pm8$	$120\pm30$
	S8	$1.31\pm0.05$	$23.8\pm0.9$	$168 \pm 6$	$130\pm10$	8	$1.02\pm0.03$	$1.79\pm0.06$	$50 \pm 3$	$127\pm10$	$350\pm200$
	S9	$4.7\pm0.1$	$58.9\pm0.3$	$413\pm7$	$340\pm10$	$130\pm60$	$2.68\pm0.01$	$4.39\pm0.06$	$122 \pm 2$	$330\pm20$	$730\pm60$
	S10	$20.56\pm0.04$	$126\pm4$	$740\pm300$	$960\pm60$	$780\pm300$	$5.03\pm0.02$	$11 \pm 1$	$300 \pm 100$	$1100\pm5$	$2380\pm200$
Н	S11	$0.28\pm0.01$	$6.3 \pm 0.1$	$44 \pm 2$	$38.6\pm0.2$	12 ± 9	$0.236\pm0.006$	$0.330 \pm 0.005$	$13.5\pm0.5$	$38.6\pm0.2$	91 ± 7
	S12	$0.55\pm0.01$	$11.2\pm0.2$	$73 \pm 2$	$53 \pm 1$	$65 \pm 30$	$0.42\pm0.01$	$0.65\pm0.01$	$22.8\pm0.6$	$52 \pm 1$	$150\pm30$
	<b>S</b> 13	$1.77\pm0.01$	$28.8\pm0.3$	$188\pm2$	$169\pm 6$	$130 \pm 30$	$1.022\pm0.006$	$1.75\pm0.02$	$58 \pm 1$	$152\pm8$	$450\pm70$
	S14	$6.58 \pm 0.09$	$77 \pm 1$	$490\pm20$	$433\pm4$	$270\pm50$	$2.70\pm0.06$	$4.55\pm0.07$	$180 \pm 20$	$480\pm10$	$940\pm60$
	S15	$44.4\pm0.7$	$163 \pm 1$	$1620\pm10$	$1400\pm20$	$950\pm200$	$5.9\pm0.1$	$14.8\pm0.1$	$590\pm10$	$1740\pm30$	$4260 \pm 500$

**Table C.1** As and Sb concentrations in different fractions in soils (mg/kg, dry mass, mean  $\pm$  SE, n = 3).

**Table C.2** Total and labile concentrations measured by SEP, DGT, soil solution in a concentration gradient obtained from historically contaminated soils, and R values of As and Sb (mean  $\pm$  SE, n = 3). C<sub>SEP-labile</sub> represents sum of the non-specifically sorbed and specifically sorbed fractions obtained from SEP.

Soil	Treatment				As		Sb					
		Total <sub>As</sub> (mg/kg)	C <sub>SEP-As</sub> (mg/kg)	C <sub>sol-As</sub> (µg/L)	C <sub>DGT-As</sub> (µg/L)	Ratio R <sub>As</sub> =C <sub>DGT</sub> /C <sub>sol</sub>	Total <sub>Sb</sub> (mg/kg)	C <sub>SEP-Sb</sub> (mg/kg)	C <sub>sol-Sb</sub> (µg/L)	C <sub>DGT-Sb</sub> (µg/L)	Ratio R <sub>Sb</sub> =C <sub>DGT</sub> /C <sub>sol</sub>	
L	<b>S</b> 1	$42.1\pm0.2$	$3.47\pm0.05$	$5.27\pm0.01$	$3.42\pm0.08$	$0.65\pm0.01$	$52\pm 6$	$0.325\pm0.005$	$18.8\pm0.1$	$2.20\pm0.02$	$0.1173 \pm 0.0003$	
	S2	$160 \pm 30$	$13.09\pm0.07$	$18.5\pm0.5$	$10.4\pm0.9$	$0.57\pm0.06$	$200\pm30$	$1.68 \pm 0.02$	$71\pm1$	$14 \pm 2$	$0.20\pm0.02$	
	<b>S</b> 3	$270\pm9$	$26.4\pm0.2$	$26.1\pm0.1$	$15.6\pm0.4$	$0.60\pm0.02$	$330 \pm 30$	$3.77\pm0.02$	$133.8\pm0.8$	$24.7\pm0.2$	$0.189 \pm 0.001$	
	<b>S</b> 4	$670\pm100$	$51.0\pm0.5$	$61 \pm 1$	$41.9\pm0.6$	$0.682\pm0.001$	$870 \pm 170$	$8.56 \pm 0.08$	$268.0\pm0.6$	$56 \pm 4$	$0.21\pm0.01$	
	S5	$1400\pm50$	$121.9\pm0.3$	$290.7\pm0.5$	$176.9\pm0.8$	$0.609\pm0.002$	$2277 \pm 130$	$20.6\pm0.1$	$580\pm2$	$122 \pm 1$	$0.214\pm0.001$	
М	S6	65 ± 3	$4.12\pm0.03$	$8.48\pm0.09$	3.1 ± 0.3	$0.36 \pm 0.04$	$90.7\pm0.6$	$0.41\pm0.01$	$20.4\pm0.3$	$2.74\pm0.08$	$0.137 \pm 0.006$	
	<b>S</b> 7	$176 \pm 2$	$12.8\pm0.2$	$22 \pm 1$	$10.3\pm0.2$	$0.46\pm0.03$	$220\pm30$	$1.30\pm0.03$	$47\pm2$	$7.54\pm0.09$	$0.166\pm0.006$	
	<b>S</b> 8	$330\pm20$	$25.1\pm0.9$	$77 \pm 3$	$24.2\pm0.7$	$0.32\pm0.01$	$530\pm180$	$2.80\pm0.06$	$85\pm1$	$15.6\pm0.2$	$0.187 \pm 0.004$	
	S9	$950\pm50$	$63.6\pm0.4$	$215\pm10$	$85 \pm 3$	$0.40\pm0.03$	$1190\pm60$	$7.07\pm0.07$	$178\pm1$	$33 \pm 1$	$0.189 \pm 0.008$	
	S10	$2630\pm90$	$147 \pm 4$	-	-	-	$3790 \pm 180$	$16 \pm 1$	-	-	-	
Н	S11	$101 \pm 8$	6.6 ± 0.1	$10.2 \pm 0.1$	$4.0 \pm 0.2$	$0.39 \pm 0.03$	$140 \pm 8$	$0.57\pm0.01$	$23.7\pm0.2$	$3.3 \pm 0.2$	$0.143 \pm 0.008$	
	S12	$200\pm30$	$11.8\pm0.2$	$24 \pm 1$	$9.5\pm0.4$	$0.41\pm0.03$	$230\pm30$	$1.06\pm0.02$	$42\pm2$	$7.0 \pm 0.2$	$0.172\pm0.007$	
	S13	$510\pm30$	$30.5\pm0.3$	$63.8\pm0.7$	$24.5\pm0.8$	$0.38\pm0.01$	$670 \pm 70$	$2.77\pm0.02$	$71.1\pm0.7$	$12.8\pm0.5$	$0.183\pm0.005$	
	S14	$1280\pm40$	$83 \pm 1$	$237\pm8$	$114 \pm 4$	$0.48\pm0.01$	$1600 \pm 60$	$7.26\pm0.09$	$177.4\pm0.7$	$38 \pm 2$	$0.22\pm0.01$	
	S15	$4180 \pm 160$	$207\pm2$	-	-	-	$6610\pm460$	$20.7\pm0.2$	-	-	-	

-: not measured due to analysis problems

# Appendix D

**Supplementary Information for Chapter 6** 



**Figure D.1** As and Sb associated with different binding fractions in historically and recently contaminated soils, each fraction as a percentage of the total soil concentration (mean  $\pm$  SE, n = 15 for historically contaminated soils (chapter 5), n = 21 for recently contaminated soils (chapter 6)).



**Figure D.2** The average percentage of  $C_{DGT-As}$  and  $C_{sol-As}$  of total soil As in As-only and As+Sb amended soils (A);  $C_{DGT-Sb}$  and  $C_{sol-Sb}$  of total soil Sb in Sb-only and As+Sb amended soils (B).



Figure D.3 The average percentage of  $C_{DGT-As}$  and  $C_{sol-As}$ ;  $C_{DGT-Sb}$  and  $C_{sol-Sb}$  of total soil As and Sb in historically and recently contaminated soils.

## Appendix E

## **Supplementary Information for Chapter 7**



**Figure E.1** Depth profiles of pore water Fe concentrations in soils after 5 weeks of flooding with and without plants, measured by DET. Unplanted soils (◊), Planted soils (■). Soil Aged-L (A), Soil Aged-H (B), Soil 50As (C), Soil 300Sb (D), Soil 50As300Sb (D). The dotted line represents the soil water interface.



**Figure E.2** Depth profiles of pore water inorganic arsenic and antimony concentrations measured by DGT, and iron(II) concentrations measured by DET in a historically contaminated soil (Aged-L). Unplanted soil (A, C) and Planted soil (B, D). The dotted line represents the soil water interface.



**Figure E.3** Depth profiles of As and Sb species in soil Aged-H after 1-week flooding and 5-week flooding with and without plants. The dotted line represents the soil water interface.